TRANSM	/IITTAL

To: Date: 6/28/2018

THE COUNCIL

From:

THE MAYOR

TRANSMITTED FOR YOUR CONSIDERATION. PLEASE SEE ATTACHED.

(Ana Guerrero) for

ERIC GARCE TI Mayor





Eric Garcetti, Mayor Rushmore D. Cervantes, General Manager

Housing Development Bureau 1200 West 7th Street, Los Angeles, CA 90017 tel 213.808.8638 | fax 213.808.8610 hcidla.lacity.org

June 28, 2018

Council File: Council District: 17-1410, 16-0677

Council District: 14
Contact Persons: Ste

Steven Brady Magdalina Zakaryan (213) 808-8538 (213) 808-8592

Edwin C. Gipson II Sean L. Spear

(213) 808-8597 (213) 808-8901

Honorable Eric Garcetti Mayor, City of Los Angeles Room 303, City Hall 200 N. Spring Street Los Angeles, CA 90012

Attention: Mandy Morales, Legislative Coordinator

COUNCIL TRANSMITTAL: REQUEST FOR AUTHORITY TO ISSUE TAX-EXEMPT MULTIFAMILY CONDUIT REVENUE BONDS IN THE AMOUNT UP TO \$32,007,814 AND AMEND A HOPWA ACQUISITION LOAN FOR THE SP7 PROJECT

## **SUMMARY**

The General Manager of the Los Angeles Housing + Community Investment Department (HCIDLA) respectfully requests that your office review and approve this transmittal and forward it to the City Council for further consideration. Through this transmittal, HCIDLA requests authority to issue tax-exempt multifamily conduit revenue bonds in an amount up to \$32,007,814 for the development of the SP7 scattered site project (Project) and, as part of the acquisition/recapitalization, amend the terms of an approved Housing Opportunities for Persons With HIV/AIDS (HOPWA) acquisition loan (#C-131386, C.F. No. 16-0677) (HOPWA Acquisition Loan) to combine and incorporate the second property and its Promissory Note, Regulatory Agreement, and HOPWA Restrictive Covenant from an existing HOPWA loan (Loan #C-96549). The California Debt Limit Allocation Committee (CDLAC) designated October 1, 2018 as the bond issuance deadline for the Project.

### RECOMMENDATIONS

The General Manager of HCIDLA respectfully requests that the Mayor and City Council:

- I. Adopt the Resolution prepared by the City Attorney provided as Attachment A to this report, authorizing the issuance up to \$32,007,814 in tax-exempt multifamily conduit revenue bonds or notes for the development of the SP7 project;
- II. Authorize the General Manager of HCIDLA, or designee, to negotiate and execute relevant loan documents and amendments to City Contract #C-131386, including but not limited to amending interest rates, loan and affordability terms, and loan amount; executing separate promissory notes

and regulatory agreements for the two properties; subordination to new permanent loans; and, assumption, assignment, and transfer of ownership to a new limited partnership, subject to the approval of the City Attorney as to form; and,

III. Authorize the General Manager of HCIDLA or designee, to negotiate and execute the relevant bond documents for the development, subject to the approval of the City Attorney as to form.

## **BACKGROUND**

## **Project Summary**

The Project is a 100-unit scattered site new construction and rehabilitation development that consists of two sites, located at 519 E. 7<sup>th</sup> Street (Site 1) and 647 S. San Pedro Street (Site 2), in Council District 14. Site 1 is currently improved with a one-story commercial warehouse building built in 1984 that was vacant as of December 2017. The developer plans to demolish the existing building prior to construction of the Project. Site 1 will be a newly constructed, seven-story, 81-unit property comprised of 80 studio units and one two-bedroom manager unit. It will have at-grade parking for staff, include the majority of the Project's amenities, as well as the manager's unit that will serve both properties. Amenities include a community room and kitchen, multiple patios and community gardens, bike parking, laundry room, conference room, elevator service, as well as management and supportive service offices, which will serve both sites. All units will be fully furnished.

Site 2 is currently improved with an affordable residential building built in 1998, and was and continues to be owned and operated by The Skid Row Housing Trust, the Project's sponsor. SP7 Apartments LP (Borrower) will acquire Site 2 at closing. Site 2 currently consists of 18 studio units and one studio manager unit spread across three levels. No reconfiguration of the units will occur; however, the existing manager's unit will be converted for residential use to be consistent with the Site 2 rehabilitation plan. The rehabilitation of Site 2 includes the major building systems, modernization of unit interiors and common area upgrades, addition of washers and dryers in each unit, and accessibility and sustainability upgrades. Site 1 and Site 2 are approximately 50 feet apart, and a single management team will be able to service both properties satisfactorily. As with Site 1, all Site 2 units will be fully furnished. Site 2 has no parking.

The Project will provide a total of 99 studio units for individuals at or below 50% of the area median income (AMI) and one two-bedroom manager's unit. There are several advantages for the development of a scattered site project, including economies of scale with staffing, amenities, and services, and recapitalizing an existing affordable housing development to extend its useful life, continue and extend affordability, and improve the existing building condition and systems.

In June 2017, the Mayor and City Council authorized the sale of HHH Bonds for the Project. Based on HHH funding restrictions, 55 of the 99 units will be restricted as permanent supportive housing (PSH) units, of which 27 units will be set aside for homeless individuals, and 28 units will be set aside for chronically homeless individuals who are either clients of the Los Angeles County Department of Health Services (DHS), veterans or have special needs. Of the 99 units, 31 units will be restricted for residents with special needs (HOPWA), including 22 units at Site 1 and nine units at Site 2. It should be noted that unit restrictions including PSH, chronically homeless and HOPWA may be layered in for a resident that qualifies for multiple restrictions. All 99 affordable units will have rental assistance, either in the form of a Section 8 Project Based Voucher (PBV) (59 units), HUD-VASH Voucher (22 units) for veterans referred by the Veterans Administration (VA), or Shelter Plus Care (18 units). On-site supportive services will be

provided full-time, Monday through Friday, to meet the needs of residents that have serious mental illnesses, substance addiction, or long-term chronic health conditions.

The Project is located just east of the Central Business District in downtown Los Angeles. The developer plans to incorporate sustainable design, and energy efficient features and practices throughout the construction process and following project completion.

## Financing History

On August 25, 2017, the Mayor and City Council provided inducement authority to HCIDLA to issue private activity bonds in the amount up to \$34,731,929 for the SP7 Apartments project, which enabled the Borrower to apply for a tax-exempt bond allocation (C.F. No. 04-2646). On January 18, 2018, HCIDLA submitted two CDLAC applications, for a total allocation of \$32,007,814 of tax-exempt issuance authority on behalf of the Borrower. Due to technical issues with CDLAC's application systems, CDLAC required HCIDLA submit one application for the new construction portion of the Project (Site 1), and a second application for the rehabilitation portion of the Project (Site 2). CDLAC agreed that the scattered site project will be awarded a single amount of bond issuance authority, via a single Bond Resolution, which was awarded on March 21, 2018, with the issuance deadline of October 1, 2018. Per the Tax Equity and Fiscal Responsibility Act of 1982 (TEFRA) requirements, HCIDLA conducted a public hearing on November 20, 2017. Following the TEFRA hearing, on January 22, 2018, the TEFRA Resolution and Minutes were approved by the Mayor and City Council (C.F. No. 17-1410).

## Affordability Restrictions

Pursuant to HCIDLA's Multifamily Bond Policies and Procedures, the Project must provide a public benefit. Therefore, one or more Bond Regulatory Agreements will be executed in connection with the issuance of the tax-exempt bonds. Each Bond Regulatory Agreement will include affordability restrictions throughout a term ending no sooner than the later of: 1) 55 years after the date on which 50% of the dwelling units are first occupied; 2) the date such bonds are paid in full; or, 3) the date on which any Section 8 assistance terminates, if applicable. In addition to the above, the bond award includes a CDLAC Resolution, which restricts the Project's housing units for 55 years.

The Project will also have affordability restrictions applied under a separate agreement with the California Tax Credit Allocation Committee (TCAC), for a term of 55 years, as a result of the allocation of the 4% Low Income Housing Tax Credits (LIHTC).

All of the 99 homeless units will be targeted to households at or below 50% of AMI; including 31 special needs units restricted to qualified HOPWA residents. Tables 1 and 2 below provide a summary of the unit mix for the Project.

	Table 1 Affordability Restrictions – Site 1 (7th Street)					
Unit Type	Units at 30% AMI	Units at 35% AMI	Units at 40% AMI	Units at 50% AMI	Market Rate (Manager's Unit)	Total Number of Units
Studio	25		16	39		80
Two Bedroom					1	1
Total	25		16	39	1	81

	Table 2 Affordability Restrictions – Site 2 (San Pedro Street)					
Unit Type	Units at 30% AMI	Units at 35% AMI	Units at 40% AMI	Units at 50% AMI	Market Rate (Manager's Unit)	Total Number of Units
Studio		12		7		19
Two Bedroom						
Total		12		7		19

### TOTAL UNITS = 100

## Development Team

The Borrower is a California limited partnership, comprised of SP7 Apartments GP LLC (SP7 GP LLC) as the managing general partner with a 0.01% ownership interest, and National Equity Fund, Inc. (NEF) as the investor limited partner with a 99.99% of the limited partnership. The SP7 GP LLC is wholly held by The Skid Row Housing Trust (SRHT). The Borrower entity is currently in compliance with HCIDLA's Business Policies.

Skid Row Housing Trust's key staff members are: Mike Alvidrez, CEO, and Dana Trujillo, Chief Investment and Finance Officer.

The developer, Skid Row Housing Trust, has previously been involved with the development of 31 affordable housing projects consisting of a total of 2,282 housing units.

Borrower:

SP7 Apartments LP

1317 E 7<sup>th</sup> Street

Los Angeles, CA 90021

Contact: Ben Rosen, Director of Real Estate Development

Phone: (213) 683-0522 x 153

Additional Project development team members are:

Architect:

KFA Architecture

1625 Olympic Boulevard Santa Monica, CA 90404

Contact: Wade Killefer, FAIA, Principal

Phone: (310) 399-7975

Attorney:

Bocarsly, Emden, Cowan, Esmail & Arndt

633 West Fifth Street, 64th Floor

Los Angeles, CA 90071 Contact: Nicole Deddens Phone: (213) 239-8029

General Contractor:

**TBD** 

Property Manager:

SRHT Property Management Co.

1317 East 7<sup>th</sup> Street

Los Angeles, CA 90021 Contact: Ken Fewell

Phone: (213) 683-0522 ext. 127

Tax Credit Investor:

National Equity Fund, Inc. 500 South Grand Ave., Floor 23

Los Angeles, CA 90071 Contact: Debbie Burkart, VP Phone: (213) 250-9550

## Financial Structure

The tax-exempt note will be privately placed with MUFG Union Bank, N.A. (Union Bank). Union Bank has proposed a "back-to-back" loan structure. In order to meet Community Reinvestment Act (CRA) requirements, large banking organizations such as Union Bank are required to achieve two types of CRA goals in the markets they serve. In each market, they are required to hold a certain dollar volume of investment activity, and are also required to achieve a certain dollar volume of lending activity in each period. Union Bank's proposed tax-exempt back-to-back loan structure characterizes Union Bank's involvement as a loan of tax-exempt proceeds rather than a purchase of tax-exempt bonds, replacing the Bond with a Note in order to achieve these CRA goals. Union Bank's ability to achieve these goals is fundamental to its ability to continue to allocate significant levels of capital to affordable housing activity in the City.

Instead of purchasing bonds, Union Bank will make a tax-exempt loan to the City of Los Angeles (City) (Funding Loan) in the amount of up to \$32,007,000. The Funding Loan will have interim and permanent phases. The City will loan the proceeds of the Funding Loan to the Borrower (Borrower Loan). The Borrower will use the proceeds from the Borrower Loan to fund the construction phase of the development. The construction loan term will be 20 months (plus one 6 month extension), with a fixed rate, currently estimated at 4.30%. Once the rehabilitation and construction is completed, the construction loan will be paid down with permanent financing sources. The permanent loan, also provided by Union Bank, will be in the amount of up to \$3,216,500 with a 15-year term with an 18-year amortization period. The permanent loan interest rate will be a fixed rate established prior to the initial closing.

The Bond will remain as a single instrument and be governed by the terms of a Construction and Permanent Loan Agreement among the City, Borrower and Union Bank.

The Project's financing structure includes amending the HOPWA Acquisition Loan, which was closed in June 2018, with recapitalizing the existing HOPWA Predevelopment Loan. The Project will include a single loan agreement that will be secured by two promissory notes: the HOPWA Acquisition Loan Promissory Note will be in the amount of \$3,000,000, will bear a 1.0% simple interest rate, and have a 57 year term (two years for construction, then a standard 55 year term); and the HOPWA Predevelopment Loan in the amount of \$950,000 will bear the Applicable Federal Rate ("AFR") interest rate (current estimate is 2.88%), reflecting an increase from the original 0% interest rate, and a 57 year term (two years for construction, then a standard 55 year term). The amended HCIDLA loan and promissory notes will mature at the same time as the separate Bond Regulatory Agreements, a requirement of both Union Bank and NEF.

The financing structure will include HOPWA funds from HCIDLA, the Proposition HHH Permanent Supportive Housing Loan Program (HHH), the California Department of Housing and Community

Development (HCD) Veterans Housing and Homelessness Prevention Program (VHHP), the Community Development Commission of the County of Los Angeles (LACDC), the Federal Home Loan Bank of San Francisco, Affordable Housing Program (FHLB-AHP), the Housing Authority of the City of Los Angeles, Voucher Assisted Supportive Housing with Project Based Vouchers (HACLA-VASH-PBV), and the Los Angeles County Department of Health Services' (DHS) rental subsidy program. In addition to these construction and permanent funding sources, the Project will receive an annual supportive service funding allocation by the DHS in the amount of up to \$437,400 for an initial period of 15 years, to support intensive case management services to 81 homeless DHS patients.

HCIDLA requires that Union Bank meet the City's Responsible Banking Ordinance reporting requirements. At closing, bond counsel will provide the required legal opinions as to the tax-exempt status of the bonds, under federal and state law. The legal and financing documents will include language that establishes the bond structure as a limited obligation, and strictly payable from project revenues. The Project financing complies with both the HCIDLA Bond Policies and City Financial Policies. Additionally, the legal and financing documents will require the Borrower to provide annual statements and information as requested by HCIDLA.

## Sources and Uses

Tables 3, 4 and 5 provide a summary of the sources and uses for the project, including cost per unit and cost category percentages.

Table 3 – Construction Sources					
Construction	<b>Total Sources</b>	Per Unit	% Total		
Union Bank	\$25,198,748	\$251,988	50.7%		
HCIDLA HOPWA Loan (recast)	\$950,000	\$9,500	1.9%		
HCIDLA HOPWA Loan (recast)					
Accrued/Deferred Interest	\$69,825	\$698	0.2%		
HCIDLA – HHH	\$8,500,000	\$85,000	17.1%		
HCIDLA HOPWA Loan (new)	\$3,000,000	\$30,000	6.0%		
CalHFA MHSA	\$2,000,000	\$20,000	4.0%		
CalHFA MHSA Accrued/Deferred					
Interest	\$89,500	\$895	0.2%		
LACDC	\$1,500,000	\$15,000	3.0%		
LACDC Accrued/Deferred Interest	\$67,125	\$671	0.2%		
AHP	\$1,100,000	\$11,000	2.2%		
General Partner Equity	\$4,105,124	\$41,051	8.3%		
Costs Deferred Until Conversion	\$1,359,840	\$13,598	2.7%		
Tax Credit Limited Partner Equity	\$1,723,889	\$17,239	3.5%		
TOTAL	\$49,664,051	\$496,640	100%		

Table 4 – Permanent Sources					
Permanent	<b>Total Sources</b>	Per Unit	% Total		
Permanent Loan	\$2,925,800	\$29,258	5.9%		
HCIDLA HOPWA Loan (recast)	\$950,000	\$9,500	1.9%		
HCIDLA HOPWA Loan (recast)					
Accrued/Deferred Interest	\$69,825	\$698	0.1%		
HCIDLA – HHH Loan	\$12,000,000	\$120,000	24.2%		
HCIDLA HOPWA Loan (new)	\$3,000,000	\$30,000	6.0%		
CalHFA MHSA	\$2,000,000	\$20,000	4.0%		
CalHFA MHSA Accrued/Deferred					
Interest	\$89,500	\$895	0.2%		
HCD VHHP Loan	\$4,657,670	\$46,577	9.4%		
LACDC	\$1,500,000	\$15,000	3.0%		
LACDC Accrued/Deferred Interest	\$67,125	\$671	0.1%		
FHLB AHP Loan	\$1,100,000	\$11,000	2.2%		
General Partner Equity	\$4,105,124	\$41,051	8.3%		
Limited Partner Capital Contribution	\$17,199,007	\$171,990	34.7%		
TOTAL	\$49,664,051	\$496,640	100%		

TO THE RESIDENCE OF THE PARTY O	Table 5 – Uses of Funds				
Uses of Funds	Total Uses	Cost/Unit			
Construction & related Costs	\$28,987,539	\$289,875			
Architect & Engineering	\$1,750,174	\$17,502			
Construction Financing Costs	\$2,807,333	\$28,073			
Permanent Financing Costs	\$25,000	\$250			
Legal fees & related costs	\$174,000	\$1,740			
Reserves & appraisal	\$733,840	\$7,338			
Developer Fee	\$5,505,124	\$55,051			
Total Contingency costs	\$4,395,579	\$43,955			
Other Project Costs	\$5,285,462	\$52,854			
TOTALS	\$49,664,051	\$496,640			

HCIDLA's Bond Team for the financing of the SP7 Apartments project is as follows:

Bond Issuer Municipal Advisor CSG Advisors, Inc. 315 W. 5<sup>th</sup> Street, Suite 302 Los Angeles, CA 90013

Bond Issuer Counsel Los Angeles City Attorney 200 N. Main Street, 9<sup>th</sup> Floor Los Angeles, CA 90012

Bond Counsel Kutak Rock LLP 777 S. Figueroa Street, Suite 4550 Los Angeles, CA 90017 HCIDLA Request for Issuance of Bonds for the SP7 Project Page 8

## **Labor Costs**

Labor costs for the Project are subject to California prevailing wage requirements, and/or federal wage requirements under the Davis-Bacon Act of 1931 and Related Acts, as applicable.

# FISCAL IMPACT STATEMENT

There is no fiscal impact to the General Fund as a result of the issuance of the bonds. The City is a conduit issuer and will not incur liability for repayment of the bonds. The bonds are a limited obligation, payable strictly from revenue derived from the Project. The City will not be obligated to make payments on the bonds.

Prepared By:

Reviewed By:

STEVEN BRADY

Finance Development Officer I

MAGDALINA ZAKARYAN

Finance Development Officer II

Reviewed By:

Reviewed By:

EDWIN C. GIPSON II

Director of Housing and Development

SEAN L. SPEAR

Assistant General Manager

Reviewed By:

Approved By:

LAURA K. GUGLIELMO

**Executive Officer** 

RUSHMORE D. CERVANTES

General Manager

Attachment A – Bond Resolution

Attachment B – Funding Loan Agreement

Attachment C - Borrower Loan Agreement

Attachment D – Bond Regulatory Agreement (519 E. 7th Street)

Attachment E – Bond Regulatory Agreement (647 S. San Pedro Street)

HDB16-034

# Attachment A

Bond Resolution for SP7 on next page.

#### RESOLUTION

### **CITY OF LOS ANGELES**

A RESOLUTION AUTHORIZING THE ISSUANCE, SALE AND DELIVERY OF ONE OR MORE SERIES OF MULTIFAMILY NOTES BY THE CITY OF LOS ANGELES DESIGNATED AS ITS MULTIFAMILY MORTGAGE REVENUE NOTE (SP7 APARTMENTS) SERIES 2018C, IN AN AGGREGATE PRINCIPAL AMOUNT NOT TO EXCEED \$32,007,814 TO PROVIDE FINANCING FOR THE ACQUISITION, CONSTRUCTION, REHABILITATION AND EQUIPPING OF THE SCATTERED SITE MULTIFAMILY HOUSING PROJECT SPECIFIED IN PARAGRAPH 16 HEREOF AND APPROVING AND AUTHORIZING THE EXECUTION AND DELIVERY OF RELATED DOCUMENTS AND AGREEMENTS AND THE TAKING OF RELATED ACTIONS INCLUDING THE EXECUTION OF AMENDATORY DOCUMENTS THERETO.

WHEREAS, the City of Los Angeles (the "City") is authorized, pursuant to the provisions of Section 248, as amended, of the City Charter of the City (the "City Charter") and Article 6.3 of Chapter 1 of Division 11 of the Los Angeles Administrative Code, as amended (the "Law"), to issue its obligations for the purpose of providing financing for the acquisition, construction, rehabilitation, equipping and development of multifamily rental housing for persons and families of low or moderate income (the "Program") which will satisfy the provisions of Chapter 7 of Part 5 of Division 31 of the Health and Safety Code of the State of California (the "Act"); and

WHEREAS, the City desires to issue, pursuant to the Law, and in accordance with the Act, its revenue note to provide financing for the acquisition, rehabilitation, construction and equipping of the scattered site multifamily rental housing project described in paragraph 16 below (the "Project"); and

WHEREAS, the Project is or will be located wholly within the City; and

WHEREAS, it is in the public interest and for the public benefit that the City authorize financing for the Project, and it is within the powers of the City to provide for such a financing and the issuance of such note; and

WHEREAS, the City proposes to issue, pursuant to the Law and in accordance with the Act, its Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C (the "Note") in one or more series in an aggregate principal amount not to exceed \$32,007,814; and

WHEREAS, the City proposes to use the proceeds of the Note to fund a loan to the owner identified in paragraph 16 (the "Owner") to finance a portion of the acquisition, construction,

rehabilitation and equipping of the Project and, if applicable, to pay certain costs of issuance in connection with the issuance of the Note; and

WHEREAS, MUFG Union Bank, N.A., or a subsidiary or affiliate thereof (the "Purchaser"), has expressed its intention to purchase (or to cause a subsidiary or affiliate to purchase) the Note authorized hereby in whole; and

WHEREAS, this City Council (the "City Council") finds that the public interest and necessity require that the City at this time make arrangements for the sale of such Note; and

WHEREAS, the interest on the Note may qualify for a federal tax exemption under Section 142(a)(7) of the Internal Revenue Code of 1986, as amended (the "Code"), only if the Note is approved in accordance with Section 147(f) of the Code; and

WHEREAS, pursuant to the Code, the Note is required to be approved, following a public hearing, by an elected representative of the issuer of the Note and an elected representative of the governmental unit having jurisdiction over the area in which the Project is located; and

WHEREAS, the City Council is the elected legislative body of the City and is the applicable elected representative required to approve the issuance of the Note within the meaning of Section 147(f) of the Code; and

WHEREAS, pursuant to Section 147(f) of the Code, the City caused a notice to appear in the *Los Angeles Times*, which is a newspaper of general circulation in the City, on November 6, 2017 to the effect that a public hearing would be held on November 20, 2017 regarding the issuance of the Note; and

WHEREAS, the Los Angeles Housing and Community Investment Department held said public hearing on such date, at which time an opportunity was provided to present arguments both for and against the issuance of the Note; and

WHEREAS, the minutes of such public hearing, and any written comments received with respect thereto, have been presented to the City Council; and

WHEREAS, the Owner provided to the City the following information as a good faith estimate of the cost of the Note financing, and the City disclosed such information at said public hearing in accordance with Section 5852.1 of the California Government Code: (a) the true interest cost of the Note, (b) the finance charge of the Note, including all third-party expenses, (c) the amount of proceeds received by the City for the sale of the Note less the finance charge of the Note and any reserves or capitalized interest paid or funded with proceeds of the Note and (d) the total payment amount (the "Financing Information"); and

WHEREAS, such Financing Information has been disclosed in connection with the City Council meeting in which this Resolution is approved;

NOW, THEREFORE, BE IT RESOLVED by the City Council of the City of Los Angeles, as follows:

- 1. The recitals hereinabove set forth are true and correct, and the City Council so finds. This Resolution is being adopted pursuant to the Law.
- 2. Pursuant to the Law and in accordance with the Act and the Funding Loan Agreement (as hereinafter defined), a revenue note of the City, to be designated as "City of Los Angeles Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C" in one or more series in an aggregate principal amount not to exceed \$32,007,814 is hereby authorized to be issued. The principal amount of the Note to be issued shall be determined by a Designated Officer (as defined below) in accordance with this Resolution.
- 3. The proposed form of Funding Loan Agreement (the "Funding Loan Agreement") among the City, the Purchaser as Funding Lender and such fiscal agent as appearing in the final form of Funding Loan Agreement executed by a below-defined Designated Officer (the "Fiscal Agent"), in substantially the form attached hereto, is hereby approved along with any additions or supplements which may, in the determination of a Designated Officer, be necessary to document the issuance of the Note authorized hereunder. The Mayor of the City, the General Manager, any Acting General Manager, the Interim General Manager or any Assistant General Manager, Acting Assistant General Manager, Interim Assistant General Manager, Executive Officer, Acting Director or Director—Finance and Development Division of the Housing Development Bureau of the Los Angeles Housing and Community Investment Department (each hereinafter referred to as a "Designated Officer") are hereby authorized and directed to execute and deliver, for and in the name and on behalf of the City, said Funding Loan Agreement, with such additions, changes or corrections (including, without limitation, designation of the Fiscal Agent) as the Designated Officer executing the same may approve upon consultation with the City Attorney and Bond Counsel to the City and approval of the City Attorney, provided that such additions or changes shall not authorize an aggregate principal amount of the Note in excess of the amount stated above, such approval by the City Attorney to be conclusively evidenced by the execution and delivery of the Funding Loan Agreement with such additions, changes or corrections.

Any Designated Officer shall be authorized to approve the appointment of the Fiscal Agent.

- 4. The proposed form of Construction and Permanent Loan Agreement (the "Loan Agreement"), by and among the City, the Purchaser and the below-defined Owner, in substantially the form attached hereto, is hereby approved. Any Designated Officer is hereby authorized and directed, for and in the name and on behalf of the City, to execute the Loan Agreement, with such additions, changes or corrections as the Designated Officer executing the same may approve upon consultation with the City Attorney and Bond Counsel and approval of the City Attorney, such approval to be conclusively evidenced by the execution of said Loan Agreement with such additions, changes or corrections.
- 5. The proposed form of the Note, as set forth in the Funding Loan Agreement, is hereby approved, and the Mayor and City Treasurer, the Interim City

Treasurer or Deputy City Treasurer of the City are hereby authorized and directed to execute, by manual or facsimile signatures of such officers under the seal of the City, and the Fiscal Agent or an authenticating agent, is hereby authorized and directed to authenticate, by manual signatures of one or more authorized officers of the Fiscal Agent or an authenticating agent, the Note in substantially such form and the Fiscal Agent is hereby authorized and directed to sell and deliver the Note to the Purchaser in accordance with the Funding Loan Agreement. The date, maturity dates, interest rate or rates (which may be either fixed or variable), interest payment dates, denomination, form of registration privileges, manner of execution, place of payment, terms of redemption, use of proceeds, series designation and other terms of the Note shall be as provided in the Funding Loan Agreement as finally executed; provided, however, that the aggregate principal amount of the Note shall not exceed \$32,007,814, no interest rate on the Note shall exceed 12% per annum, and the final maturity of the Note shall be no later than May 1, 2058. The initial purchase price of the Note shall be 100% of the principal amount thereof to be paid as advances are made with respect to the Note by the Purchaser. The Note may, if so provided in the Funding Loan Agreement, be issued as a "draw-down" note to be funded over time as provided in the Funding Loan Agreement. Such Note may be delivered in temporary form pursuant to the Funding Loan Agreement if, in the judgment of the City Attorney, delivery in such form is necessary or appropriate until the Note in definitive form can be prepared.

- 6. The proposed form of Regulatory Agreement and Declaration of Restrictive Covenants (the "Regulatory Agreement") to be entered into by and among the City, the Fiscal Agent and the below-defined Owner, substantially in the form attached hereto, is hereby approved. Any Designated Officer is hereby authorized and directed, for and in the name and on behalf of the City, to execute and deliver one or more Regulatory Agreements, with such additions, changes and corrections as the Designated Officer may approve upon consultation with the City Attorney and Bond Counsel and approval of the City Attorney, such approval to be conclusively evidenced by the execution of said Regulatory Agreement with such additions, changes or corrections. Any Designated Officer is hereby authorized and directed for and in the name and on behalf of the City to execute amendments to the Regulatory Agreement in order that interest on the Note remains tax-exempt.
- 7. All actions heretofore taken by the officers and agents of the City with respect to the sale and issuance of the Note are hereby approved, confirmed and ratified, and each Designated Officer of the City, the City Clerk and other properly authorized officers of the City are hereby authorized and directed, for and in the name and on behalf of the City, to do any and all things and take any and all actions and execute and deliver any and all certificates, agreements and other documents, including, but not limited to, those described in the Funding Loan Agreement, the Loan Agreement, the Regulatory Agreement and the other documents herein approved, which they, or any of them, may deem necessary or advisable in order to consummate the lawful issuance and delivery of the Note and the implementation of the Program in accordance with the Act and the Law and this Resolution and resolutions heretofore adopted by the City.

- 8. The City Clerk of the City or any deputy thereof is hereby authorized to countersign or to attest the signature of any Designated Officer and to affix and attest the seal of the City as may be appropriate in connection with the execution and delivery of any of the documents authorized by this Resolution, provided that the due execution and delivery of said documents or any of them shall not depend on such signature of the City Clerk or any deputy thereof or affixing of such seal. Any of such documents may be executed in multiple counterparts.
- 9. In addition to the Designated Officers, any official of the City, including any official of the Los Angeles Housing and Community Investment Department, as shall be authorized in writing by the Mayor of the City, is hereby authorized for and on behalf of the City to execute and deliver any of the agreements, certificates and other documents, except the Note, authorized by this Resolution.
- 10. In accordance with procedures established by the City Charter, the City Council, by adoption and approval of this Resolution and with the concurrence of the Mayor, does hereby direct that the proceeds of the Note be delivered directly to the Fiscal Agent, instead of the City Treasurer, to be deposited into the funds and accounts established under the Funding Loan Agreement.
- 11. Pursuant to Section 147(f) of the Code, the City Council hereby approves the issuance of the Note to finance the Project. It is intended that this Resolution constitute approval of the Note by the applicable elected representative of the issuer of the Note and the applicable elected representative of the governmental unit having jurisdiction over the area in which the Project is located, in accordance with said Section 147(f).
- 12. Pursuant to the City Charter all agreements to which the City is a party shall be subject to approval by the City Attorney as to form.
- 13. Each Designated Officer and other properly authorized officials of the City as specifically authorized under this Resolution are hereby authorized, directed and empowered on behalf of the City and the Council to execute any other additional applications, certificates, agreements, documents or other instruments or any amendments or supplements thereto, subject to approval by the City Attorney as to form, or to do and to cause to be done any and all other acts and things as they may deem necessary or appropriate to carry out for the purpose of the foregoing authorizations and to address any issues arising with respect to the Note or the agreements relating thereto subsequent to their issuance.
- 14. The Note shall contain a recital that it is issued pursuant to the Law and in accordance with the Act.
- 15. This Resolution shall take effect immediately upon its passage and adoption.

16. The "Project" and "Owner" referred to herein are as follows:

Project Name	Number of Units	Address	Owner
SP7 Apartments	100 (including 1 manager unit)	519 E. 7th Street and 647 S. San Pedro Street, Los Angeles, CA 90014	SP7 Apartments LP

[Remainder of page intentionally left blank]

I certify that the foregoing	Resolution was	adopted	by the	e Council	of the	City	of
Los Angeles at its meeting on	, 2018.						
	By						
	Name	:					
	Title						

# Attachment B

Funding Loan Agreement for SP7 on next page.

# FUNDING LOAN AGREEMENT

by and among

MUFG UNION BANK, N.A., as Funding Lender

[FISCAL AGENT], as Fiscal Agent

and

**CITY OF LOS ANGELES**, as Governmental Lender

dated as of [September] 1, 2018

relating to:

\$[32,007,814] City of Los Angeles Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C

# **TABLE OF CONTENTS**

		<b>Page</b>
	ARTICLE I DEFINITIONS AND INTERPRETATION	
1.1.	Definitions	1
1.1.	Interpretation	
1.3.	Recitals, Titles and Headings	
1.0.	ARTICLE II REPRESENTATIONS AND WARRANTIES	
2.1. 2.2.	Representations and Warranties of the Governmental Lender Representations, Warranties and Covenants of the Bank	
	ARTICLE III THE FUNDING LOAN	
3.1.	Closing of the Funding Loan	8
3.2.	Commitment To Execute the Funding Loan Note	9
3.3.	Amount and Source of Funding Loan	9
3.4.	Disbursement of Funding Loan Proceeds	9
	ARTICLE IV	
	LIMITED LIABILITY; NOTE REGISTER	
4.1.	Limited Liability	10
4.2.	Note Register	
4.3.	Transfer of Funding Loan Note	
	ARTICLE V REPAYMENT OF THE FUNDING LOAN	
5.1.	Funding Loan Repayment	12
5.2.	Nature of the Governmental Lender's Obligations	13
	ARTICLE VI FURTHER AGREEMENTS	
6.1.	Successor to the Governmental Lender	14
6.2.	Additional Instruments	14
6.3.	Books and Records	
6.4.	Notice of Certain Events	
6.5.	Compliance With Usury Laws	
6.6.	No Untrue Statements	
6.7.	No Arbitrage	
6.8.	Limitation on Issuance Costs	
6.9.	Federal Guarantee Prohibition	
6.10.	Prohibited Facilities	
6.11.	Use Covenant	
6.12. 6.13	Limitation of Expenditure of Proceeds  Tax-Exempt Status of Funding Loan	
0.13	TAX-EXECON MINUS OF FUNCTION FORD	וח

8.2.       Authority of the Fiscal Agent and Bank; Notices to Governmental Lender       20         8.3.       Successor Fiscal Agent       20         8.4.       Assignment of Rights       21         8.5.       [Reserved]       21         8.6.       Acceptance       21         ARTICLE IX FUNDS AND ACCOUNTS         9.1.       Authorization To Create Funds and Accounts       21         9.2.       Investment of Funds       21         9.3.       Establishment of Funds       21         9.4.       Note Payment Fund       22         9.5.       Expense Fund       22         9.6.       Project Fund       23         9.7.       Rebate Fund       26         9.8.       Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent; Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       28         10.3.       Notice of Defaults       29         10.4.       Certain Rights of Fiscal Agent       29         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31         1	6.14.	Immunities and Limitations of Responsibility of Governmental Lender	17
ARTICLE VIII   AGENCY   20			
ARTICLE VIII AGENCY  8.1. Assignment of Authority 8.2. Authority of the Fiscal Agent and Bank; Notices to Governmental Lender 20 8.3. Successor Fiscal Agent 8.4. Assignment of Rights. 21 8.5. [Reserved] 2.6. Acceptance. 21 8.6. Acceptance. 21 8.7. Conditions.  ARTICLE IX FUNDS AND ACCOUNTS  9.1. Authorization To Create Funds and Accounts. 9.2. Investment of Funds. 9.3. Establishment of Funds. 9.4. Note Payment Fund. 22 9.5. Expense Fund 22 9.6. Project Fund. 23 9.7. Rebate Fund. 24 9.8. Closing Costs Fund. 25 9.8. Closing Costs Fund. 26 9.8. Closing Costs Fund. 27  ARTICLE X THE FISCAL AGENT  10.1. Appointment of Fiscal Agent, Acceptance. 28 10.3. Notice of Defaults. 29 10.4. Certain Duties and Responsibilities of Fiscal Agent. 29 10.5. Not Responsible for Recitals. 30 10.6. May Hold Funding Loan. 31 10.7. Moneys Held in Trust 31 10.8. Compensation and Reimbursement 31 10.9. Fiscal Agent Required; Eligibility 31 10.9. Fiscal Agent Required; Eligibility 31 10.10. Resignation and Removal; Appointment of Successor 32 10.11. Acceptance of Appointment by Successor 33 10.12. Merger, Conversion, Consolidation or Succession to Business 33 10.14. No Recourse Against Officers or Employees of Fiscal Agent 34  ARTICLE XI	7.1.	Security for the Funding Loan	18
AGENCY   20	7.2.	Delivery of Security	19
8.2.       Authority of the Fiscal Ågent and Bank; Notices to Governmental Lender.       20         8.3.       Successor Fiscal Agent       20         8.4.       Assignment of Rights       21         8.5.       [Reserved]       21         8.6.       Acceptance       21         ARTICLE IX FUNDS AND ACCOUNTS         9.1.       Authorization To Create Funds and Accounts       21         9.2.       Investment of Funds       21         9.3.       Establishment of Funds       21         9.4.       Note Payment Fund       22         9.5.       Expense Fund       22         9.6.       Project Fund       23         9.7.       Rebate Fund       26         9.8.       Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent, Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       28         10.3.       Notice of Defaults       29         10.4.       Certain Rights of Fiscal Agent       29         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31			
8.3.       Successor Fiscal Agent       20         8.4.       Assignment of Rights       21         8.5.       [Reserved]       21         8.6.       Acceptance       21         ARTICLE IX FUNDS AND ACCOUNTS         9.1.       Authorization To Create Funds and Accounts       21         9.2.       Investment of Funds       21         9.3.       Establishment of Funds       21         9.4.       Note Payment Fund       22         9.5.       Expense Fund       22         9.6.       Project Fund       23         9.7.       Rebate Fund       26         9.8.       Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent; Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       29         10.4.       Certain Rights of Fiscal Agent       29         10.4.       Certain Rights of Fiscal Agent       29         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31         10.7.       Moneys Held in Trust       31         10.8.	8.1.	Assignment of Authority.	20
8.3.       Successor Fiscal Agent       20         8.4.       Assignment of Rights       21         8.5.       [Reserved]       21         8.6.       Acceptance       21         ARTICLE IX FUNDS AND ACCOUNTS         9.1.       Authorization To Create Funds and Accounts       21         9.2.       Investment of Funds       21         9.3.       Establishment of Funds       21         9.4.       Note Payment Fund       22         9.5.       Expense Fund       22         9.6.       Project Fund       23         9.7.       Rebate Fund       26         9.8.       Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent; Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       29         10.4.       Certain Rights of Fiscal Agent       29         10.4.       Certain Rights of Fiscal Agent       29         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31         10.7.       Moneys Held in Trust       31         10.8.	8.2.	S ,	
8.5.       [Reserved]       21         8.6.       Acceptance       21         8.7.       Conditions       21         ARTICLE IX FUNDS AND ACCOUNTS         9.1.       Authorization To Create Funds and Accounts       21         9.2.       Investment of Funds       21         9.3.       Establishment of Funds       21         9.4.       Note Payment Fund       22         9.5.       Expense Fund       22         9.6.       Project Fund       23         9.7.       Rebate Fund       26         9.8.       Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent; Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       28         10.2.       Certain Rights of Fiscal Agent       29         10.4.       Certain Rights of Forecitals       30         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31         10.7.       Moneys Held in Trust       31         10.8.       Compensation and Reimbursement       31         10.9.       Fiscal A	8.3.		
8.6. Acceptance       21         ARTICLE IX FUNDS AND ACCOUNTS         9.1. Authorization To Create Funds and Accounts         9.2. Investment of Funds       21         9.3. Establishment of Funds       21         9.4. Note Payment Fund       22         9.5. Expense Fund       22         9.6. Project Fund       23         9.7. Rebate Fund       26         9.8. Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1. Appointment of Fiscal Agent; Acceptance       28         10.2. Certain Duties and Responsibilities of Fiscal Agent       28         10.2. Certain Rights of Fiscal Agent       29         10.4. Certain Rights of Fiscal Agent       29         10.5. Not Responsible for Recitals       30         10.6. May Hold Funding Loan       31         10.7. Moneys Held in Trust       31         10.8. Compensation and Reimbursement       31         10.9. Fiscal Agent Required; Eligibility       32         10.10. Resignation and Removal; Appointment of Successor       32         10.11. Acceptance of Appointment by Successor       33         10.12. Merger, Conversion, Consolidation or Succession to Business       33         10.13. Appointment of Co-Fiscal Age	8.4.	Assignment of Rights	21
ARTICLE IX FUNDS AND ACCOUNTS     9.1.   Authorization To Create Funds and Accounts   21     9.2.   Investment of Funds   21     9.3.   Establishment of Funds   21     9.4.   Note Payment Fund   22     9.5.   Expense Fund   22     9.6.   Project Fund   23     9.7.   Rebate Fund   26     9.8.   Closing Costs Fund   27     ARTICLE X THE FISCAL AGENT   28     10.1.   Appointment of Fiscal Agent; Acceptance   28     10.2.   Certain Duties and Responsibilities of Fiscal Agent   28     10.3.   Notice of Defaults   29     10.4.   Certain Rights of Fiscal Agent   29     10.5.   Not Responsible for Recitals   30     10.6.   May Hold Funding Loan   31     10.7.   Moneys Held in Trust   31     10.8.   Compensation and Reimbursement   31     10.9.   Fiscal Agent Required; Eligibility   32     10.10.   Resignation and Removal; Appointment of Successor   33     10.11.   Acceptance of Appointment by Successor   33     10.12.   Merger, Conversion, Consolidation or Succession to Business   33     10.13.   Appointment of Co-Fiscal Agent   34     ARTICLE XI	8.5.	[Reserved]	21
ARTICLE IX FUNDS AND ACCOUNTS  9.1. Authorization To Create Funds and Accounts	8.6.	Acceptance	21
FUNDS AND ACCOUNTS         9.1. Authorization To Create Funds and Accounts       21         9.2. Investment of Funds       21         9.3. Establishment of Funds       21         9.4. Note Payment Fund       22         9.5. Expense Fund       22         9.6. Project Fund       23         9.7. Rebate Fund       26         9.8. Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1. Appointment of Fiscal Agent; Acceptance       28         10.2. Certain Duties and Responsibilities of Fiscal Agent       28         10.2. Certain Rights of Fiscal Agent       29         10.4. Certain Rights of Fiscal Agent       29         10.5. Not Responsible for Recitals       30         10.6. May Hold Funding Loan       31         10.7. Moneys Held in Trust       31         10.9. Fiscal Agent Required; Eligibility       32         10.10. Resignation and Removal; Appointment of Successor       32         10.11. Acceptance of Appointment by Successor       33         10.12. Merger, Conversion, Consolidation or Succession to Business       33         10.13. Appointment of Co-Fiscal Agent       34         ARTICLE XI	8.7.	Conditions	21
9.1. Authorization To Create Funds and Accounts       21         9.2. Investment of Funds       21         9.3. Establishment of Funds       21         9.4. Note Payment Fund       22         9.5. Expense Fund       22         9.6. Project Fund       23         9.7. Rebate Fund       26         9.8. Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1. Appointment of Fiscal Agent; Acceptance       28         10.2. Certain Duties and Responsibilities of Fiscal Agent       28         10.3. Notice of Defaults       29         10.4. Certain Rights of Fiscal Agent       29         10.5. Not Responsible for Recitals       30         10.6. May Hold Funding Loan       31         10.7. Moneys Held in Trust       31         10.8. Compensation and Reimbursement       31         10.9. Fiscal Agent Required; Eligibility       32         10.10. Resignation and Removal; Appointment of Successor       32         10.11. Acceptance of Appointment by Successor       33         10.12 Merger, Conversion, Consolidation or Succession to Business       33         10.13 Appointment of Co-Fiscal Agent       34         ARTICLE XI			
9.2.       Investment of Funds       21         9.3.       Establishment of Funds       21         9.4.       Note Payment Fund       22         9.5.       Expense Fund       22         9.6.       Project Fund       23         9.7.       Rebate Fund       26         9.8.       Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent, Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       28         10.2.       Certain Rights of Fiscal Agent       29         10.4.       Certain Rights of Fiscal Agent       29         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31         10.7.       Moneys Held in Trust       31         10.8.       Compensation and Reimbursement       31         10.9.       Fiscal Agent Required; Eligibility       32         10.10.       Resignation and Removal, Appointment of Successor       32         10.11.       Acceptance of Appointment by Successor       33         10.12.       Merger, Conversion, Consolidation or Succession to Business       33	0.1	Authorization To Crosto Funds and Accounts	21
9.3.       Establishment of Funds       21         9.4.       Note Payment Fund       22         9.5.       Expense Fund       23         9.6.       Project Fund       26         9.7.       Rebate Fund       26         9.8.       Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent; Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       28         10.3.       Notice of Defaults       29         10.4.       Certain Rights of Fiscal Agent       29         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31         10.7.       Moneys Held in Trust       31         10.8.       Compensation and Reimbursement       31         10.9.       Fiscal Agent Required; Eligibility       32         10.10.       Resignation and Removal; Appointment of Successor       32         10.11.       Acceptance of Appointment by Successor       33         10.12.       Merger, Conversion, Consolidation or Succession to Business       33         10.13.       Appointment of Co-Fiscal Agent       34 <td< td=""><td></td><td></td><td></td></td<>			
9.4.       Note Payment Fund.       22         9.5.       Expense Fund.       23         9.6.       Project Fund.       26         9.7.       Rebate Fund.       26         9.8.       Closing Costs Fund.       27         ARTICLE X THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent; Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       28         10.3.       Notice of Defaults       29         10.4.       Certain Rights of Fiscal Agent       29         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31         10.7.       Moneys Held in Trust       31         10.8.       Compensation and Reimbursement       31         10.9.       Fiscal Agent Required; Eligibility       32         10.10.       Resignation and Removal; Appointment of Successor       32         10.11.       Acceptance of Appointment by Successor       33         10.12.       Merger, Conversion, Consolidation or Succession to Business       33         10.13.       Appointment of Co-Fiscal Agent       34         ARTICLE XI			
9.5.       Expense Fund       23         9.6.       Project Fund       26         9.7.       Rebate Fund       26         9.8.       Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent; Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       29         10.4.       Certain Rights of Fiscal Agent       29         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31         10.7.       Moneys Held in Trust       31         10.8.       Compensation and Reimbursement       31         10.9.       Fiscal Agent Required; Eligibility       32         10.10.       Resignation and Removal; Appointment of Successor       32         10.11.       Acceptance of Appointment by Successor       33         10.12.       Merger, Conversion, Consolidation or Succession to Business       33         10.13.       Appointment of Co-Fiscal Agent       34         ARTICLE XI			
9.6.       Project Fund       23         9.7.       Rebate Fund       26         9.8.       Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1.       Appointment of Fiscal Agent; Acceptance       28         10.2.       Certain Duties and Responsibilities of Fiscal Agent       28         10.3.       Notice of Defaults       29         10.4.       Certain Rights of Fiscal Agent       29         10.5.       Not Responsible for Recitals       30         10.6.       May Hold Funding Loan       31         10.7.       Moneys Held in Trust       31         10.8.       Compensation and Reimbursement       31         10.9.       Fiscal Agent Required; Eligibility       32         10.10.       Resignation and Removal; Appointment of Successor       32         10.11.       Acceptance of Appointment by Successor       33         10.12.       Merger, Conversion, Consolidation or Succession to Business       33         10.13.       Appointment of Co-Fiscal Agent       34         ARTICLE XI		•	
9.7. Rebate Fund       26         9.8. Closing Costs Fund       27         ARTICLE X THE FISCAL AGENT         10.1. Appointment of Fiscal Agent; Acceptance       28         10.2. Certain Duties and Responsibilities of Fiscal Agent       28         10.3. Notice of Defaults       29         10.4. Certain Rights of Fiscal Agent       29         10.5. Not Responsible for Recitals       30         10.6. May Hold Funding Loan       31         10.7. Moneys Held in Trust       31         10.8. Compensation and Reimbursement       31         10.9. Fiscal Agent Required; Eligibility       32         10.10. Resignation and Removal; Appointment of Successor       32         10.11. Acceptance of Appointment by Successor       32         10.12. Merger, Conversion, Consolidation or Succession to Business       33         10.13. Appointment of Co-Fiscal Agent       33         10.14. No Recourse Against Officers or Employees of Fiscal Agent       34         ARTICLE XI		•	
ARTICLE X         THE FISCAL AGENT         10.1. Appointment of Fiscal Agent; Acceptance       28         10.2. Certain Duties and Responsibilities of Fiscal Agent       28         10.3. Notice of Defaults       29         10.4. Certain Rights of Fiscal Agent       29         10.5. Not Responsible for Recitals       30         10.6. May Hold Funding Loan       31         10.7. Moneys Held in Trust       31         10.8. Compensation and Reimbursement       31         10.9. Fiscal Agent Required; Eligibility       32         10.10. Resignation and Removal; Appointment of Successor       32         10.11. Acceptance of Appointment by Successor       33         10.12. Merger, Conversion, Consolidation or Succession to Business       33         10.13. Appointment of Co-Fiscal Agent       34         ARTICLE XI			
ARTICLE X THE FISCAL AGENT  10.1. Appointment of Fiscal Agent; Acceptance			
10.1. Appointment of Fiscal Agent; Acceptance			
10.2. Certain Duties and Responsibilities of Fiscal Agent2810.3. Notice of Defaults2910.4. Certain Rights of Fiscal Agent2910.5. Not Responsible for Recitals3010.6. May Hold Funding Loan3110.7. Moneys Held in Trust3110.8. Compensation and Reimbursement3110.9. Fiscal Agent Required; Eligibility3210.10. Resignation and Removal; Appointment of Successor3210.11. Acceptance of Appointment by Successor3310.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI		THE FISCAL AGENT	
10.2. Certain Duties and Responsibilities of Fiscal Agent2810.3. Notice of Defaults2910.4. Certain Rights of Fiscal Agent2910.5. Not Responsible for Recitals3010.6. May Hold Funding Loan3110.7. Moneys Held in Trust3110.8. Compensation and Reimbursement3110.9. Fiscal Agent Required; Eligibility3210.10. Resignation and Removal; Appointment of Successor3210.11. Acceptance of Appointment by Successor3310.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI	10.1.	Appointment of Fiscal Agent: Acceptance	28
10.3. Notice of Defaults2910.4. Certain Rights of Fiscal Agent2910.5. Not Responsible for Recitals3010.6. May Hold Funding Loan3110.7. Moneys Held in Trust3110.8. Compensation and Reimbursement3110.9. Fiscal Agent Required; Eligibility3210.10. Resignation and Removal; Appointment of Successor3210.11. Acceptance of Appointment by Successor3310.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI			
10.4. Certain Rights of Fiscal Agent2910.5. Not Responsible for Recitals3010.6. May Hold Funding Loan3110.7. Moneys Held in Trust3110.8. Compensation and Reimbursement3110.9. Fiscal Agent Required; Eligibility3210.10. Resignation and Removal; Appointment of Successor3210.11. Acceptance of Appointment by Successor3310.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI			
10.5. Not Responsible for Recitals3010.6. May Hold Funding Loan3110.7. Moneys Held in Trust3110.8. Compensation and Reimbursement3110.9. Fiscal Agent Required; Eligibility3210.10. Resignation and Removal; Appointment of Successor3210.11. Acceptance of Appointment by Successor3310.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI			
10.6.May Hold Funding Loan3110.7.Moneys Held in Trust3110.8.Compensation and Reimbursement3110.9.Fiscal Agent Required; Eligibility3210.10.Resignation and Removal; Appointment of Successor3210.11.Acceptance of Appointment by Successor3310.12.Merger, Conversion, Consolidation or Succession to Business3310.13.Appointment of Co-Fiscal Agent3310.14.No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI	10.5.		
10.7. Moneys Held in Trust3110.8. Compensation and Reimbursement3110.9. Fiscal Agent Required; Eligibility3210.10. Resignation and Removal; Appointment of Successor3210.11. Acceptance of Appointment by Successor3310.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI	10.6.		
10.8. Compensation and Reimbursement3110.9. Fiscal Agent Required; Eligibility3210.10. Resignation and Removal; Appointment of Successor3210.11. Acceptance of Appointment by Successor3310.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI	10.7.		
10.10. Resignation and Removal; Appointment of Successor.3210.11. Acceptance of Appointment by Successor.3310.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent.3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI	10.8.		
10.10. Resignation and Removal; Appointment of Successor.3210.11. Acceptance of Appointment by Successor.3310.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent.3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI	10.9.	Fiscal Agent Required; Eligibility	32
10.12. Merger, Conversion, Consolidation or Succession to Business3310.13. Appointment of Co-Fiscal Agent3310.14. No Recourse Against Officers or Employees of Fiscal Agent34ARTICLE XI	10.10.		
10.13. Appointment of Co-Fiscal Agent	10.11.	Acceptance of Appointment by Successor	33
10.14. No Recourse Against Officers or Employees of Fiscal Agent			
ARTICLE XI			
	10.14.	No Recourse Against Officers or Employees of Fiscal Agent	34

11.1.	Default Under Borrower Loan Agreement; Acceleration	34
11.2.	Actions Under Borrower Loan Documents	35
	ARTICLE XII	
	MISCELLANEOUS	
12.1.	Entire Agreement	35
12.2.	Notices	
12.3.	Assignments	37
12.4.	Severability	37
12.5.	Execution of Counterparts	37
12.6.	Amendments, Changes and Modifications	37
12.7.	Governing Law	37
12.8.	Term of Agreement	
12.9.	Survival of Agreement	37
12.10.	Nondiscrimination and Affirmative Action	37
12.11.	Business Tax Registration Certificate	38
12.12.	Child Support Assignment Orders	38
12.13.	Americans with Disabilities Act	39
12.14.	Responsible Banking Ordinance Filing	39
EXHI	BIT A FORM OF FUNDING LOAN NOTE	
EXHI		
EXHI	BIT C FORM OF PROJECT FUND REQUISITION	
EXHI	BIT D FORM OF RESPONSIBLE BANKING ORDINANCE CERTIFICATE	

## **FUNDING LOAN AGREEMENT**

(Back to Back Loan Program)

THIS FUNDING LOAN AGREEMENT, dated as of [September] 1, 2018 (this "Funding Loan Agreement"), is by and among MUFG UNION BANK, N.A., in its capacity as the funding lender hereunder (together with any successor to its rights, duties and obligations hereunder, the "Bank"), the CITY OF LOS ANGELES, a charter city and a municipal corporation of the State of California (together with any successor to its rights, duties and obligations hereunder, the "City" or "Governmental Lender"), and [FISCAL AGENT], in its capacity as the fiscal agent hereunder (together with any successor to its rights, duties and obligations hereunder, the "Fiscal Agent").

For and in consideration of the mutual agreements hereinafter contained, the parties hereto agree as follows:

### **ARTICLE I**

## **DEFINITIONS AND INTERPRETATION**

- 1.1. <u>Definitions</u>. The following words and terms as used in this Agreement shall have the following meanings unless the context or use otherwise requires:
- "Act" means Chapter 7 of Part 5 of Division 31 (commencing with Section 52075) of the Health and Safety Code of the State of California, as now in effect and as it may from time to time hereafter be amended or supplemented to apply to obligations incurred as of the Closing Date.
- "Affiliate" means, as to any Person, any other Person that, directly or indirectly, is in Control of, is Controlled by or is under common Control with such Person.
- "Approved Institutional Buyer" means: (a) a "qualified institutional buyer" as defined in Rule 144A promulgated under the United States Securities Act of 1933, as in effect on the date hereof (the "Securities Act"), or commercial bank having a minimum capital surplus of \$5,000,000,000; (b) an institutional "accredited investor" as defined in Sections 501(a)(1) through (3) of Regulation D promulgated under the Securities Act; (c) an entity that is directly or indirectly wholly owned or controlled by the Bank (being a financial institution described in (a) above); (d) an entity all of the investors in which are described in (a), (b) or (c) above; or (e) a custodian or trustee for a party or parties described in (a), (b) or (c) above.
- "Assignment of Deed of Trust" means that certain [Assignment of Deed of Trust and Related Documents] dated [\_\_\_\_\_] 1, 2018, executed by the Governmental Lender in favor of the Bank.
- "Authorized Representative" shall mean: (i) as to the Governmental Lender, the Mayor, the General Manager, any Interim General Manager, any Acting General Manager, any Assistant General Manager, any Acting Assistant General Manager or Interim Assistant General Manager, Executive Officer or the Director or Acting Director—Finance and Development Division of the Housing Development Bureau of the Los Angeles Housing and Community Investment

Department, and any other, officer or employee of the Governmental Lender designated to perform a specified act, to sign a specified document or to act generally on behalf of the Governmental Lender as evidenced by a written certificate furnished to the Funding Lender, the Fiscal Agent, and the Borrower containing the specimen signature of such person and signed on behalf of the Governmental Lender by the Mayor, the General Manager, any Interim General Manager, any Acting General Manager, any Assistant General Manager, any Acting Assistant General Manager or Interim Assistant General Manager, Executive Officer or the Director or Acting Director— Finance and Development Division of the Housing Development Bureau of the Los Angeles Housing and Community Investment Department. Such certificate may designate an alternate or alternates, each of whom shall be entitled to perform all duties of such Authorized Representative; (ii) as to the Borrower, [NAME], [TITLE] of [ENTITY] or [NAME], [TITLE] of [ENTITY]; (iii) as to the Funding Lender, Perica Bell, Director or Johanna Gullick, Managing Director of MUFG Union Bank, N.A.; and (iv) as to the Fiscal Agent, [ ]. The Funding Lender, the Fiscal Agent, the Bank and the Borrower may from time to time deliver one or more additional certificates designating other Authorized Representative(s).

"Bank" means MUFG Union Bank, N.A., and its successors and assigns in its capacity as the Funding Lender hereunder.

"Borrower" means SP7 Apartments LP, a California limited partnership, and its permitted successors and assigns.

"Borrower Loan" means the mortgage loan originated by the Governmental Lender to the Borrower in the principal amount of \$[32,007,814] pursuant to the terms of the Borrower Loan Agreement.

"Borrower Loan Agreement" means that certain [Construction and Permanent Loan Agreement (Multifamily Housing Back to Back Loan Program)], dated as of [\_\_\_\_\_] 1, 2018, by and among the Borrower, the Governmental Lender and the Bank, as amended and supplemented from time to time, pursuant to which the Borrower Loan is being made.

"Borrower Loan Documents" shall have the meaning ascribed to it in the Borrower Loan Agreement.

"Borrower Note" means that certain [Promissory Note (Multifamily Housing Back to Back Loan Program)] in the original principal amount of \$[32,007,814] evidencing the Borrower Loan, executed by the Borrower in favor of the Governmental Lender, as amended from time to time.

"Borrower Representative" means the [TITLE] of [ENTITY] of the [	]
of the [managing] general partner of the Borrower, or any other officer of the [	] of the
[managing] general partner of the Borrower designated by the [TITLE ABOVE] of [I	ENTITY
ABOVE] of the [] of the [managing] general partner of the Borrowe	r to be a
Borrower Representative for purposes of the Borrower Loan Documents.	

"Business Day" means a day other than a Saturday or Sunday on which the Bank is open for business for the funding of corporate loans.

- "CDLAC" means the California Debt Limit Allocation Committee or any successor thereto.
- "Closing Date" means the date of issuance of the Funding Loan Note for purposes of the Code.
- "Code" or "Internal Revenue Code" means the Internal Revenue Code of 1986, and with respect to a specific section thereof, such reference shall be deemed to include (a) the regulations promulgated by the United States Department of the Treasury under such section, (b) any successor provision of similar import hereafter enacted, (c) any corresponding provision of any subsequent Internal Revenue Code and (d) the regulations promulgated under the provisions described in (b) and (c).
- "Control" shall mean, with respect to any Person, either (i) ownership directly or through other entities of more than 50% of all beneficial equity interest in such Person, or (ii) the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, through the ownership of voting securities, by contract or otherwise.
- "Conversion Date" shall have the meaning given such term in the [Borrower Loan Agreement].
  - "County" means the County of Los Angeles, California.
- "Deed of Trust" means the [Deed of Trust, Assignment of Rents, Security Agreement and Fixture Filing (Multifamily Housing Back to Back Loan Program)], executed by the Borrower and granting a security interest in the Development to the deed of trust trustee identified therein for the benefit of the Governmental Lender and the Bank to secure the Borrower's obligations under the Borrower Note to repay the Borrower Loan, and all obligations related thereto under the Borrower Loan Agreement.
- "**Development**" means the scattered site residential rental facility consisting of 100 units (including 1 manager unit) of multifamily rental housing located at 519 East 7th Street and 647 South San Pedro Street in Los Angeles, California, on the sites described in the Deed of Trust and Exhibit A to the Regulatory Agreement.
- "Event of Default" means any of the events described as an event of default in Section 9.1 hereof.
- "Fiscal Agent" means [FISCAL AGENT] and its successors and assigns in its capacity as the fiscal agent hereunder.
- "Fiscal Agent's Fees" means the ongoing annual compensation and expenses of the Fiscal Agent payable annually in arrears equal to [\_\_\_\_]% of the outstanding principal amount of the Funding Loan Note on each [\_\_\_\_\_] 1, with a minimum annual amount of \$[1,200].
- "Funding Lender" shall mean MUFG Union Bank N.A., a national banking association, and any successor Funding Lender under this Funding Loan Agreement and the Funding Loan Documents.

- "Funding Loan" means the loan originated hereunder by the Bank to the Governmental Lender in an aggregate principal amount of up to \$[32,007,814], evidenced by the Funding Loan Note, for the purpose of enabling the Governmental Lender to make the Borrower Loan to the Borrower pursuant to the terms of the Borrower Loan Agreement.
- "Funding Loan Agreement" means this Funding Loan Agreement, as amended and supplemented from time to time.
- "Funding Loan Documents" means this Funding Loan Agreement, the Funding Loan Note, the Borrower Loan Agreement, the Regulatory Agreement, the Tax Certificate and the Assignment of Deed of Trust.
- "Funding Loan Note" means the City of Los Angeles' Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C which is issued on the Closing Date by the Governmental Lender in favor of the Bank, in the form attached hereto as Exhibit A, as amended and supplemented from time to time.
- "Governmental Lender" means the City of Los Angeles, a charter city and municipal corporation of the State of California, and its successors and assigns.
- "Ineligible Purchaser" shall mean any owner or operator of a multifamily housing facility within the City of Los Angeles which has either: (i) repeatedly violated the building or habitability codes of the City of Los Angeles; or (ii) been convicted of criminal violations relating to the ownership or operation of multifamily housing in the City of Los Angeles.
- **"Law"** means Section 248 of the City Charter of the City and Article 6.3 of Chapter 1 of Division 11 of the Los Angeles Administrative Code, as supplemented and amended to the Closing Date.
- "**Person**" shall mean an individual, a corporation, a partnership, a limited liability company, a limited liability partnership, a limited partnership, a trust, an unincorporated organization or a government or any agency or political subdivision thereof.
- "**Regulations**" means the tax regulations promulgated by the United States Department of the Treasury from time to time pursuant to the Code.
- "Regulatory Agreement" means, collectively, the two Regulatory Agreements and Declarations of Restrictive Covenants, each dated as of [September] 1, 2018, by and among the Governmental Lender, the Fiscal Agent and the Borrower, as amended and supplemented from time to time in accordance with its terms.
- "Representation Letter" shall mean a representation letter in substantially the form attached to this Funding Loan Agreement as Exhibit B.
  - "Requisition" shall have the meaning ascribed to it in Section 9.6(c).
- "Reserved Rights" means the Governmental Lender's rights to enforce and receive payments of money directly and for its own purposes under [Exhibit C and Sections 7.24, 7.27,

7.28, 7.29 and 11.4] (solely as such Sections relate to the Governmental Lender) of the Borrower Loan Agreement, the Governmental Lender's rights to inspect and audit the books, records and premises of the Borrower and of the Development, its right to collect attorneys' fees and related expenses, its right to enforce the Borrower's covenants to comply with applicable federal tax law and State law (including the Law, the Act and the rules and regulations of the Governmental Lender), its right to receive fees and notices and to grant or withhold consents or waivers and to enforce its rights under the Regulatory Agreement and this Funding Loan Agreement, its rights to indemnification by the Borrower under [Section 7.24] of the Borrower Loan Agreement and Section 9 of the Regulatory Agreement, and its rights to amend this Funding Loan Agreement and the Regulatory Agreement in accordance with the provisions hereof and thereof.

"Responsible Banking Ordinance" means Ordinance No. 182138 of the City of Los Angeles amending Chapter 5.1 of the Los Angeles Administrative Code.

"**Security**" shall have the meaning ascribed to it in Section 7.1.

"State" means the State of California.

"**Tax Certificate**" means the Tax Certificate as to Arbitrage and the Provisions of Sections 103 and 141-150 of the Internal Revenue Code of 1986, executed and delivered by the City and the Borrower as of the Closing Date.

"Tax Counsel" means (a) Kutak Rock LLP, or (b) any attorney at law or other firm of attorneys selected by the Borrower and acceptable to the Governmental Lender and the Funding Lender of nationally recognized standing in matters pertaining to the federal tax status of interest on tax exempt obligations issued by states and political subdivisions, and duly admitted to practice law before the highest court of any state of the United States of America, but shall not include counsel for the Borrower.

- 1.2. <u>Interpretation</u>. Unless the context clearly requires otherwise, words of masculine gender shall be construed to include correlative words of the feminine and neuter genders and vice versa, and words of the singular number shall be construed to include correlative words of the plural number and vice versa. This Funding Loan Agreement and all the terms and provisions hereof shall be construed to effectuate the purpose set forth herein and to sustain the validity hereof.
- 1.3. Recitals, Titles and Headings. The terms and phrases used in the recitals of this Funding Loan Agreement have been included for convenience of reference only, and the meaning, construction and interpretation of all such terms and phrases for purposes of this Funding Loan Agreement shall be determined by references to Section 1.1 hereof. The titles and headings of the articles and sections of this Funding Loan Agreement have been inserted for convenience of reference only and are not to be considered a part hereof, and shall not in any way modify or restrict any of the terms or provisions hereof and shall never be considered or given any effect in construing this Funding Loan Agreement or any provision hereof or in ascertaining intent, if any question of intent should arise.

#### **ARTICLE II**

### REPRESENTATIONS AND WARRANTIES

- 2.1. <u>Representations and Warranties of the Governmental Lender</u>. The Governmental Lender makes the following representations and warranties:
  - (a) The Governmental Lender is a charter city and municipal corporation, duly organized and existing under the laws of the State.
  - (b) Under the provisions of the Law and the Act, the Governmental Lender has the power, and has taken all official actions necessary (i) to enter into the Funding Loan Documents to which it is a party, or (ii) to perform its obligations hereunder and thereunder, and (iii) to consummate all other transactions on its part contemplated by this Funding Loan Agreement.
  - (c) The Funding Loan Documents to which the Governmental Lender is a party have been duly executed and delivered by the Governmental Lender and the Governmental Lender has taken such actions as are necessary to cause the Funding Loan Documents to which it is a party, when executed by the other respective parties thereto, to be valid and binding limited obligations of the Governmental Lender, enforceable against the Governmental Lender in accordance with their respective terms, except as limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws affecting the rights of creditors generally.
  - (d) The execution and delivery of Funding Loan Documents to which it is a party, the performance by it of its obligations hereunder and thereunder and the consummation of the transactions on its part contemplated hereby and thereby will not violate any law, rule, regulation or ordinance or any order, judgment or decree of any federal, state or local court, and do not conflict with, or constitute a breach of, or a default under the terms and conditions of, any agreement, instrument or commitment to which the Governmental Lender is a party or by which the Governmental Lender or any of its property is bound.
  - (e) The Governmental Lender has not been served with any action, suit, proceeding, inquiry or investigation and, to the knowledge of the Governmental Lender, no action, suit, proceeding, inquiry or investigation is threatened against the Governmental Lender, by or before any court, governmental agency or public board or body which: (i) affects or questions the existence or the territorial jurisdiction of the Governmental Lender or the title to office of any members of the governing board of the Governmental Lender; (ii) affects or seeks to prohibit, restrain or enjoin the execution and delivery of the Funding Loan Documents to which the Governmental Lender is a party, or the loaning of the Funding Loan as herein set forth; (iii) affects or questions the validity or enforceability of the Funding Loan Documents; or (iv) questions the power or authority of the Governmental Lender to carry out the transactions on its part contemplated by the Funding Loan Documents.

- (f) The revenues and receipts to be derived from the Borrower Loan Agreement and the Borrower Note have not been pledged by the Governmental Lender to secure any of its notes or bonds other than the Funding Loan evidenced by the Funding Loan Note.
- (g) The Governmental Lender will not create, authorize or approve any mortgage, pledge, lien, charge or encumbrance of any kind on the Security or any part thereof prior to or on parity with the lien of this Funding Loan Agreement, except as expressly permitted or contemplated by the Funding Loan Documents.
- (h) CDLAC has provided an allocation of the State of California's 2017 private activity bond volume cap under Section 146 of the Code to the Governmental Lender for the Funding Loan Note. The Governmental Lender has timely made the required carry forward election with respect to such allocation as required by the Code, and the Governmental Lender will comply with the requirements of the Code with respect to such allocation. The Governmental Lender has applied the alternative option under clause (2) of the first paragraph of Section 3.01 of IRS Notice 2011-63 with respect to the issue date of the Funding Loan Note; and, in connection therewith, has included the information on Form 8038 filed for the Funding Loan Note that is required by Section 3.03 of said Notice.

Nothing in this Funding Loan Agreement shall be construed as requiring the Governmental Lender to provide any financing for the Development other than the proceeds of the Funding Loan, or to provide sufficient moneys for all of the costs of the Development.

- 2.2. <u>Representations, Warranties and Covenants of the Bank</u>. The Bank as of the date hereof, represents, warrants and covenants that:
  - (a) The Bank is a national banking association, organized and existing under the laws of the United States and has full legal right, power and authority under the laws of the United States (i) to enter into this Funding Loan Agreement, the Borrower Loan Agreement and the Assignment Agreement, (ii) to perform its obligations hereunder, and (iii) to consummate the transactions on its part contemplated by this Funding Loan Agreement and the Borrower Loan Agreement.
  - (b) This Funding Loan Agreement, the Borrower Loan Agreement and the Assignment Agreement have been duly executed and delivered by the Bank and, when executed by the Governmental Lender and Borrower, as applicable, will constitute valid and binding obligations of the Bank, enforceable against the Bank in accordance with their respective terms, except as limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws or judicial decisions affecting the rights of creditors generally.
  - (c) The execution and delivery of this Funding Loan Agreement, the Borrower Loan Agreement and the Assignment Agreement, the performance by the Bank of its obligations hereunder and thereunder and the consummation of the transactions on its part contemplated hereby and thereby will not violate any law, regulation, rule or ordinance or any order, judgment or decree of any federal, state or local court and do not conflict with, or constitute a breach of, or a default under, any document, instrument or commitment to which the Bank is a party or by which the Bank or any of its property is bound.

- (d) The Bank has not been served with and, to the knowledge of the Bank, there is no action, suit, proceeding, inquiry or investigation by or before any court, governmental agency or public board or body pending or threatened against the Bank which (i) affects or seeks to prohibit, restrain or enjoin the loaning of the amounts set forth herein to the Governmental Lender or the execution and delivery of this Funding Loan Agreement, the Borrower Loan Agreement or the Assignment Agreement, (ii) affects or questions the validity or enforceability of this Funding Loan Agreement or the Borrower Loan Agreement, or (iii) questions the power or authority of the Bank to carry out the transactions on its part contemplated by, or to perform its obligations under, this Funding Loan Agreement and the Borrower Loan Agreement.
- (e) Any certificate for the benefit of the Governmental Lender signed by a representative of the Bank and delivered pursuant to this Funding Loan Agreement, the other Funding Loan Documents or the Borrower Loan Agreement shall be deemed a representation and warranty by the Bank as to the statements made therein.
- (f) The Bank shall advise the Fiscal Agent in writing of any amendment to the amortization schedules attached to the Borrower Loan Agreement.

### **ARTICLE III**

## **THE FUNDING LOAN**

- 3.1. <u>Closing of the Funding Loan</u>. The closing of the Funding Loan shall not occur until the following conditions are met:
  - (a) the Bank shall have received an original executed counterpart of this Funding Loan Agreement, the Funding Loan Note, the Assignment of Deed of Trust, the Regulatory Agreement, the Deed of Trust, the original of the Borrower Note endorsed by the Governmental Lender to the Fiscal Agent, an opinion of Tax Counsel as to the validity of the Funding Loan Note and the exemption of interest thereon from gross income for federal income tax purposes and all of the Borrower Loan Documents;
  - (b) no Event of Default nor any event which with the passage of time and/or the giving of notice would constitute an Event of Default under this Funding Loan Agreement shall have occurred as evidenced by a certificate received from the Governmental Lender;
  - (c) the conditions to the closing of the Borrower Loan, the issuance of the Borrower Note and the initial disbursement of the Borrower Loan as set forth in [Sections 4.1, 4.2 (including, but not limited to, Sections 4.2.2, 4.2.3, 4.2.6(b) and 4.2.9), and 5.1.1] of the Borrower Loan Agreement shall have been satisfied in full;
  - (d) counsel to the Borrower shall have delivered an opinion in form satisfactory to counsel to the Governmental Lender and counsel to the Bank regarding the enforceability against the Borrower of the Borrower Loan Documents to which the Borrower is a party;

- (e) the initial owner of the Funding Loan Note shall have executed and delivered a letter in the form of Exhibit B hereto; and
- (f) all legal matters incident to the transactions contemplated by this Funding Loan Agreement shall be concluded to the reasonable satisfaction of Tax Counsel, counsel to the Governmental Lender, counsel to the Bank and counsel to the Fiscal Agent.
- 3.2. <u>Commitment To Execute the Funding Loan Note</u>. The Governmental Lender agrees to execute and deliver the Funding Loan Note simultaneously with the execution of this Funding Loan Agreement, the Borrower Loan Agreement, the Borrower Note, the Tax Certificate and the Regulatory Agreement.
- 3.3. Amount and Source of Funding Loan. The Bank hereby makes to the Governmental Lender and agrees to fund, and the Governmental Lender hereby accepts from the Bank, upon the terms and conditions set forth herein, the Funding Loan in an aggregate principal amount of up to \$[32,007,814] and agrees to have the proceeds of the Funding Loan applied and disbursed in accordance with the provisions of this Funding Loan Agreement.

## 3.4. Disbursement of Funding Loan Proceeds.

- (a) The Bank and the Governmental Lender hereby authorize and direct the funding and disbursement by the Bank of the initial principal amount of the Funding Loan in the amount of \$[\_\_\_\_\_] on the Closing Date, subject to the satisfaction of all the conditions specified in Section 3.1 above. On the date of execution and delivery of the Funding Loan Note, and the date of execution and delivery of the Borrower Note, the initial proceeds of the Funding Loan in the amount of \$[\_\_\_\_\_] shall be disbursed by the Bank, on behalf of the Governmental Lender, to the escrow agent for the closing of the Borrower Loan to fund the Borrower Loan under and as provided in [Section 5.1.2] of the Borrower Loan Agreement.
- (b) The Bank and the Governmental Lender hereby authorize and direct the funding and disbursement of the remaining principal amount of the Funding Loan (not referenced in Section 3.4(a) above) by the Bank, on behalf of the Governmental Lender, directly to the Fiscal Agent to fund the remaining principal of the Borrower Loan under and as provided in, and subject to the provisions of, [Section 5] of the Borrower Loan Agreement (other than [Section 5.1.1]).
- (c) Prior to the Conversion Date, the Bank shall disburse directly to the Bank on the first Business Day of each month, the accrued interest under the Funding Loan and the Bank will provide the Fiscal Agent with written notice of the amount disbursed pursuant to this Section 3.4(c).
- (d) None of the Bank, the Governmental Lender, or the Fiscal Agent shall be responsible for the application by the Borrower of monies disbursed to or for the account of the Borrower in accordance with this Section 3.4.

(e) From and after the earlier of (i) the Conversion Date (as defined in the Borrower Loan Agreement), or (ii) the date which is three years after the Closing Date), no further advances of the Funding Loan shall occur.

### **ARTICLE IV**

### LIMITED LIABILITY; NOTE REGISTER

4.1. <u>Limited Liability</u>. All obligations and any liability of the Governmental Lender hereunder, under the Funding Loan Note, under the other Funding Loan Documents and under the Borrower Loan Documents shall be a special, limited obligation of the Governmental Lender secured by the funds and other revenues, funds and assets pledged under this Funding Loan Agreement, and not from any other revenues, funds or assets of the Governmental Lender. The Funding Loan Note shall be payable solely from the revenues, funds and income derived from such pledged funds and other security and is and shall always be a valid claim of the owner thereof only against the such pledged funds and other security, which revenues, funds and income shall be used for no other purpose than to pay the principal installments of, redemption premium, if any, and interest on the Funding Loan Note, except as may be expressly authorized otherwise in the Funding Loan Agreement and the Borrower Loan Agreement.

EXCEPT TO THE LIMITED EXTENT PROVIDED ABOVE, NO RECOURSE SHALL BE HAD FOR THE PAYMENT OF THE PRINCIPAL OF, PREMIUM, IF ANY, OR INTEREST ON THE FUNDING LOAN NOTE OR FOR ANY CLAIM BASED THEREON OR UPON ANY OBLIGATION, COVENANT OR AGREEMENT IN THIS FUNDING LOAN AGREEMENT CONTAINED AGAINST THE GOVERNMENTAL LENDER, ANY PAST, PRESENT OR FUTURE MEMBER OF ITS GOVERNING BODY, ITS OFFICERS, ATTORNEYS, ACCOUNTANTS, FINANCIAL ADVISORS, AGENTS OR STAFF OR THE OFFICERS, ATTORNEYS, ACCOUNTANTS, FINANCIAL ADVISORS, AGENTS OR STAFF OF ANY SUCCESSOR PUBLIC ENTITY, AS SUCH, EITHER DIRECTLY OR THROUGH THE GOVERNMENTAL LENDER OR ANY SUCCESSOR PUBLIC ENTITY, UNDER ANY RULE OF LAW OR PENALTY OR OTHERWISE, AND ALL SUCH LIABILITY OF THE GOVERNMENTAL LENDER, ANY MEMBER OF ITS GOVERNING BODY AND ITS OFFICERS, ATTORNEYS, ACCOUNTANTS, FINANCIAL ADVISORS, AGENTS AND STAFF IS HEREBY, AND BY THE ACCEPTANCE OF THE FUNDING LOAN NOTE, EXPRESSLY WAIVED AND RELEASED AS A CONDITION OF, AND IN CONSIDERATION FOR, THE EXECUTION OF THIS FUNDING LOAN AGREEMENT AND THE ISSUANCE OF THE FUNDING LOAN NOTE.

The Governmental Lender shall not be liable for payment of the principal of or interest on the Funding Loan Note or any other costs, expenses, losses, damages, claims or actions, of any conceivable kind on any conceivable theory, under or by reason of or in connection with this Funding Loan Agreement, the Funding Loan Note or any other documents, except only to the extent amounts are received for the payment thereof from the Borrower under the Borrower Loan Agreement.

4.2. <u>Note Register</u>. The Funding Loan Note shall be in fully registered form. The Fiscal Agent shall maintain records (the "**Note Register**") as to the owner of the Funding Loan Note. Any transfer by the Bank of its ownership of the Funding Loan Note (or by any subsequent transferee of the Funding Loan Note) shall be recorded by the Fiscal Agent in the Note Register. The Funding Loan Note shall not be transferred through the services of the Depository Trust Company or any other third-party registrar.

## 4.3. <u>Transfer of Funding Loan Note</u>.

- (a) The Funding Loan Note and the Funding Loan may, in accordance with the terms of this Funding Loan Agreement but in any event subject to the provisions of Section 4.3(b) and (c) hereof, be transferred by the person in whose name it is registered, in person or by his duly authorized attorney, upon surrender of the Funding Loan Note for cancellation at the office of the Fiscal Agent, accompanied by a written instrument of transfer in a form acceptable to the Fiscal Agent, duly executed. Whenever the Funding Loan Note shall be surrendered for transfer, the Fiscal Agent shall authenticate and deliver to the transferee thereof a new Funding Loan Note in the name of the transferee as beneficiary thereof.
- (b) Notwithstanding any other provision hereof, the Funding Loan Note may not be registered in the name of, or transferred to, any person except with the prior written consent of the Governmental Lender, which shall not be unreasonably or arbitrarily withheld, and only to an Approved Institutional Buyer that executes and delivers to the Governmental Lender and the Fiscal Agent a letter substantially in the form attached hereto as Exhibit B; provided, however, that no such letter is required to be executed and no prior written consent of the Governmental Lender shall be required in connection with a transfer of the Funding Loan Note to an Affiliate of the Bank.

Nothing contained in this Section 4.3(b) shall be deemed to limit or otherwise restrict the sale by any owner of the Funding Loan Note of any participation interests in the Funding Loan Note, provided that: (i) such owner shall remain the owner of record in the Note Register of the Fiscal Agent following the sale of any such participation interest; (ii) the purchaser of the participation interest is an Approved Institutional Buyer; (iii) any such participation shall be in a principal amount of at least \$250,000 or, if less, the then outstanding principal amount of the Funding Loan Note; and (iv) the purchaser of such participation interest shall provide a Representation Letter to the Governmental Lender and the Fiscal Agent substantially in the form of Exhibit B hereto.

- (c) The Funding Loan Note may only be transferred in whole.
- (d) In the case of a transfer of the Funding Loan Note and Funding Loan, subject to the foregoing transfer restrictions, the transferor shall provide to the Governmental Lender written notice of such proposed transfer not less than 10 calendar days prior to such proposed transfer, during which time the Governmental Lender shall determine whether the proposed transferee is an Ineligible Purchaser. If the Governmental Lender fails to deliver written notice to the Fiscal Agent of such determination within

10 calendar days of receipt of notice of proposed transfer, the Fiscal Agent shall conclude that such transferee is not an Ineligible Purchaser.

- (e) The Governmental Lender may require the payment by the entity requesting any such transfer of any tax, fee or other governmental charge required to be paid with respect to such transfer. The cost of printing any new Funding Loan Note and any services rendered or any out-of-pocket expenses incurred by the Governmental Lender in connection therewith shall be paid by the transferor of the Funding Loan Note.
- (f) The transferor of the Funding Loan Note shall indemnify and defend the Governmental Lender and the officers, directors, employees, attorneys and agents of the Governmental Lender, past, present and future, against any claim brought by any transferor or transferee of the Funding Loan Note in respect of the Borrower Loan Documents in the event that there occurs a transfer of the Funding Loan Note that is not permitted pursuant to this Section 4.3. Failure to comply with Section 4.3(b) shall cause any purported transfer to be null and void.
- (g) If the transferee is a "Commercial Bank" as defined in the Responsible Banking Ordinance, the transferee shall deliver to the Governmental Lender an executed Responsible Banking Ordinance Certificate in the form of Exhibit D hereto.

### **ARTICLE V**

## REPAYMENT OF THE FUNDING LOAN

## 5.1. Funding Loan Repayment.

- (a) The Funding Loan shall be evidenced by the Funding Loan Note which shall be executed by the Governmental Lender in the form attached hereto as Exhibit A. The Governmental Lender agrees to pay to the Fiscal Agent, but only from amounts received by the Governmental Lender from the Borrower pursuant to the Borrower Loan Agreement, the Borrower Note and the other Borrower Loan Documents, principal of, premium on (if any) and interest on the Funding Loan at the times, in the manner, in the amounts and at the rates of interest provided in the Funding Loan Note and this Funding Loan Agreement.
- (b) The Governmental Lender further agrees to cause the Borrower to pay, solely by the execution of the Borrower Loan Agreement and the assignment thereof to the Fiscal Agent under this Funding Loan Agreement, all fees, all late charges and prepayment premiums, fees and penalties as set forth in the Funding Loan Note, all taxes and assessments, general or special, including, without limitation, all ad valorem taxes, concerning or in any way related to the Development, or any part thereof, and any other governmental charges and impositions whatsoever, foreseen or unforeseen, and all utility and other charges and assessments; provided, however, that the Governmental Lender reserves the right to contest in good faith the legality of any tax or governmental charge concerning or in any way related to the Development and the Governmental Lender's obligations hereunder will be limited as provided in Sections 4.1, 5.2 and 6.14 hereof.

- (c) The Governmental Lender further agrees, subject to Sections 4.2, 5.2 and 6.13 hereof, to request the Borrower to pay to the Bank, solely by the execution of the Borrower Loan Agreement, on the Closing Date a loan fee equal to \$[ ].
- 5.2. Nature of the Governmental Lender's Obligations. The Governmental Lender shall repay the Funding Loan Note, but only from amounts received by the Governmental Lender, the Bank or the Fiscal Agent (in its capacity as Fiscal Agent under this Funding Loan Agreement) from the Borrower pursuant to the Borrower Loan Agreement, the Borrower Note and the other Borrower Loan Documents, pursuant to the terms of the Funding Loan Note irrespective of any rights of set-off, recoupment or counterclaim the Governmental Lender might otherwise have against the Fiscal Agent, the Bank or any other person. The Governmental Lender will not suspend, discontinue or reduce any such payment or (except as expressly provided herein) terminate this Funding Loan Agreement for any cause, including, without limiting the generality of the foregoing, (i) any delay or interruption in the acquisition, rehabilitation or operation of the Development; (ii) the failure to obtain any permit, order or action of any kind from any governmental agency relating to the Funding Loan or the Development; (iii) any event constituting force majeure; (iv) any acts or circumstances that may constitute commercial frustration of purpose; (v) any change in the laws of the United States of America, the State or any political subdivision thereof; or (vi) any failure of the Governmental Lender or the Fiscal Agent to perform or observe any covenant whether expressed or implied, or to discharge any duty, liability or obligation arising out of or connected with the Funding Loan Note; it being the intention of the parties that, as long as the Funding Loan Note or any portion thereof remains outstanding and unpaid, the Governmental Lender shall be obliged to repay the Funding Loan, but only from amounts received by the Governmental Lender or the Bank from the Borrower pursuant to the Borrower Loan Agreement, the Borrower Note and the other Borrower Loan Documents. This Section 5.2 shall not be construed to release the Governmental Lender from any of its obligations hereunder, or, except as provided in this Section 5.2, to prevent or restrict the Governmental Lender from asserting any rights which it may have against the Bank under the Funding Loan Note or under any provision of law or to prevent or restrict the Governmental Lender from prosecuting or defending any action or proceeding by or against the Fiscal Agent, the Bank or the Borrower or taking any other action to protect or secure its rights, or to prevent or restrict the Bank from asserting any rights which it may have against the Borrower.

Notwithstanding the foregoing, neither any past, present or future member of its governing body, its officers, attorneys, accountants, financial advisors, agents or staff or the officers, attorneys, accountants, financial advisors, agents or staff of any successor public entity, as such, either directly or through the Governmental Lender or any successor public entity, under any rule of law or penalty of otherwise shall be personally liable for the amounts owing under this Funding Loan Agreement, the Funding Loan Note or any of the other Funding Loan Documents, and the Bank's and the Fiscal Agent's remedies in the event of a default under the Funding Loan shall be limited to those remedies set forth in the Borrower Loan Documents and, if a default also exists under the Borrower Loan Agreement or the Borrower Note, to commence foreclosure under the Deed of Trust and the other Borrower Loan Documents and the exercise of the power of sale or other rights granted thereunder. In the event of a default hereunder or under the Funding Loan Note, the Fiscal Agent shall not have the right to proceed directly against the Governmental Lender or the right to obtain a deficiency judgment from the Governmental Lender after foreclosure.

Nothing contained in the foregoing shall limit any rights or remedies the Governmental Lender or the Fiscal Agent may have against the Borrower.

#### ARTICLE VI

## **FURTHER AGREEMENTS**

- 6.1. <u>Successor to the Governmental Lender</u>. The Governmental Lender will at all times use its best efforts to maintain the powers, functions, duties and obligations now reposed in it pursuant to law or assure the assumptions of its obligations hereunder by any public trust or political subdivision succeeding to its powers.
- 6.2. <u>Additional Instruments</u>. The Governmental Lender hereby covenants to execute and deliver, or cause to be executed and delivered, at the expense of the Borrower, such additional instruments and to perform such additional acts, or cause the performance of such additional acts, as may be necessary, in the written opinion of the Fiscal Agent or the Bank, each acting in good faith, to carry out the intent of this Funding Loan Agreement and the Funding Loan Note or to perfect or give further assurances of any of the rights granted or provided for in this Funding Loan Agreement, the Assignment of Deed of Trust or the other Funding Loan Documents.
- 6.3. Books and Records. The Governmental Lender shall, solely by the execution of the Borrower Loan Agreement and the assignment thereof to the Fiscal Agent, and subject to the provisions of Sections 4.1, 5.2 and 6.13 hereof, cause the Borrower to permit the Fiscal Agent or the Funding Lender or its duly Authorized Representatives access during normal business hours to the books and records of the Borrower pertaining to the Borrower Loan and the Development, and to make such books and records available for audit and inspection, at reasonable times and under reasonable conditions to the Governmental Lender, the Fiscal Agent, the Funding Lender and their duly Authorized Representatives, and at the sole expense of the Borrower.
- 6.4. Notice of Certain Events. The Governmental Lender hereby covenants to advise the Fiscal Agent and the Funding Lender promptly in writing of the occurrence of any Event of Default under and as defined in the Borrower Loan Agreement, the Regulatory Agreement or the other Funding Loan Documents of which it has received written notice, or any event which, with the passage of time or service of notice, or both, would constitute an Event of Default thereunder of which it has received written notice, specifying the nature and period of existence of such event and the actions being taken or proposed to be taken with respect thereto. In [Section 7.23] of the Borrower Loan Agreement, the Borrower has agreed to advise the Governmental Lender and the Funding Lender promptly in writing of the occurrence of any Event of Default (as defined in the Borrower Loan Agreement).
- 6.5. <u>Compliance With Usury Laws</u>. Notwithstanding any other provision of this Funding Loan Agreement, it is agreed and understood that in no event shall this Funding Loan Agreement, with respect to the Funding Loan Note, be construed as requiring the Governmental Lender or any other person to pay interest and other costs or considerations that constitute interest under any applicable law which are contracted for, charged or received pursuant to this Funding Loan Agreement in an amount in excess of the maximum amount of interest allowed under any applicable law.

In the event of any acceleration of the payment of the principal amount of the Funding Loan Note, that portion of any interest payment in excess of the maximum legal rate of interest, if any, provided for in this Funding Loan Agreement or related documents shall be cancelled automatically as of the date of such acceleration, or if theretofore paid, credited to the principal amount.

The provisions of this Section prevail over any other provision of this Funding Loan Agreement.

6.6. No Untrue Statements. Neither this Funding Loan Agreement nor any other document, certificate or written statement furnished to the Fiscal Agent or the Funding Lender by the Governmental Lender contains to the best of its knowledge any untrue statement of a material fact or omits to state a material fact necessary in order to make the statements contained herein and therein not misleading or incomplete as of the date hereof. It is specifically understood by the Governmental Lender that all such statements, representations and warranties shall be deemed to have been relied upon by the Funding Lender as an inducement to make the Funding Loan and that if any such statements, representations and warranties were materially incorrect at the time they were made, the Funding Lender may consider any such misrepresentation or breach an Event of Default.

No document, certificate or written statement furnished to the Governmental Lender by the Funding Lender contains to the best of its knowledge any untrue statement of a material fact or omits to state a material fact necessary in order to make the statements contained therein not misleading or incomplete as of the date hereof.

- 6.7. No Arbitrage. Solely in reliance upon the covenants and representations of the Borrower in the Borrower Loan Agreement, in the Regulatory Agreement and in the Tax Certificate, the Governmental Lender shall not take, permit or suffer to be taken any action with respect to the proceeds of the Funding Loan Note which, if such action had been reasonably expected to have been taken, or had been deliberately and intentionally taken, on the Closing Date would have caused the Funding Loan Note to be an "arbitrage bond" within the meaning of Section 148 of the Code and the Regulations promulgated thereunder. The Governmental Lender covenants, solely in reliance upon the covenants and representations of the Borrower in the Borrower Loan Agreement, in the Regulatory Agreement and in the Tax Certificate, to cause the Borrower to rebate to the United States Treasury any amounts which are required to be rebated thereto pursuant to the Code and any regulations promulgated thereunder with respect to the Funding Loan and the Borrower shall cause payment of an amount equal to excess investment earnings with respect to the Funding Loan, to the United States in accordance with the Regulations, all at the sole expense of the Borrower.
- 6.8. <u>Limitation on Issuance Costs</u>. The Governmental Lender shall assure, solely in reliance upon the covenants and representations of the Borrower in the Borrower Loan Agreement, in the Regulatory Agreement and in the Tax Certificate, that, from the proceeds of the Funding Loan Note received from the Funding Lender and investment earnings thereon, an amount not in excess of 2% of the amount of the Funding Loan advanced by the Funding Lender shall be used to pay for or provide for the payment of costs associated with the issuance, execution and delivery of the Funding Loan Note. For this purpose, if the fees of the Funding Lender are retained as a

discount on the purchase of the Funding Loan Note, such retention shall be deemed to be an expenditure of proceeds of the Funding Loan for said fees.

- 6.9. <u>Federal Guarantee Prohibition</u>. The Governmental Lender shall take no action nor, solely in reliance upon the covenants and representations of the Borrower in the Borrower Loan Agreement, in the Regulatory Agreement and in the Tax Certificate, permit nor suffer any action to be taken if the result of the same would be to cause the Funding Loan Note to be "federally guaranteed" within the meaning of Section 149(b) of the Code.
- 6.10. Prohibited Facilities. The Governmental Lender, solely in reliance upon the covenants and representations of the Borrower in the Borrower Loan Agreement, in the Regulatory Agreement and in the Tax Certificate, shall assure that no portion of the proceeds of the Funding Loan shall be used to provide any airplane, skybox or other private luxury box, health club facility, facility primarily used for gambling, or store the principal business of which is the sale of alcoholic beverages for consumption off premises. The Governmental Lender, solely in reliance upon the covenants and representations of the Borrower in the Borrower Loan Agreement, in the Regulatory Agreement and in the Tax Certificate, shall assure that no portion of the proceeds of the Funding Loan shall be used for an office unless the office is located on the premises of the facilities constituting the Development and unless not more than a de minimis amount of the functions to be performed at such office is not related to the day-to-day operations of the Development.
- 6.11. <u>Use Covenant</u>. Solely in reliance upon the covenants and representations of the Borrower in the Borrower Loan Agreement, in the Regulatory Agreement and in the Tax Certificate, the Governmental Lender shall not use or knowingly permit the use of any proceeds of the Funding Loan or any other funds of the Governmental Lender, directly or indirectly, in any manner, and shall not take or permit to be taken any other action or actions, which would result in the Funding Loan Note not meeting the requirements of Section 142(d) of the Code as applicable to the Development.
- 6.12. <u>Limitation of Expenditure of Proceeds</u>. The Governmental Lender shall assure, solely in reliance upon the covenants and representations of the Borrower in the Borrower Loan Agreement, in the Regulatory Agreement and in the Tax Certificate, that not less than 97% of the amount advanced on the Funding Loan, plus premium (if any) paid on the purchase of the Funding Loan Note by the original purchaser thereof from the Governmental Lender, less any original discount, are used for Qualified Project Costs (as defined in the Regulatory Agreement) and that less than 25% of such amount is used for land or an interest in land.
- 6.13. <u>Tax-Exempt Status of Funding Loan</u>. The Governmental Lender covenants to and for the benefit of the Funding Lender that, notwithstanding any other provisions of this Funding Loan Agreement or any other instrument, it will:
  - (a) not knowingly take or cause to be taken any action or actions, or knowingly fail to take any action or actions, which would cause the interest payable on the Funding Loan Note to be includable in gross income for federal income tax purposes;
  - (b) whenever and so often as requested by the Funding Lender, at the sole cost and expense of the Borrower, the Governmental Lender shall do and perform all acts and

things permitted by law and necessarily desirable in order to assure the interest paid by the Governmental Lender on the Funding Loan Note will be excluded from the gross income of the owner of the Funding Loan Note for federal income tax purposes pursuant to Section 103 of the Code, except in the event where any owner of the Funding Loan Note is a "substantial user" of the facilities financed with the Funding Loan or a "related person" within the meaning of Section 147(a) of the Code; and

(c) not knowingly take any action nor, solely in reliance of the covenants and representations of the Borrower in the Borrower Loan Agreement, the Regulatory Agreement and the Tax Certificate, knowingly permit or suffer any action to be taken if the result of the same would be to cause the Funding Loan Note to be "federally guaranteed" within the meaning of Section 149(b) of the Code of the Regulations.

For purposes of this Section 6.13 the Governmental Lender's compliance shall be based solely on matters within the Governmental Lender's knowledge and control and no acts, omissions or directions of the Borrower, the Funding Lender or any other Persons shall be attributed to the Governmental Lender.

In complying with the foregoing covenants, the Governmental Lender may rely from time to time on an opinion of Tax Counsel.

# 6.14. <u>Immunities and Limitations of Responsibility of Governmental Lender</u>.

- The Governmental Lender shall be entitled to the advice of counsel, and the (a) Governmental Lender shall be wholly protected as to action taken or omitted in reliance on such advice. The Governmental Lender may rely conclusively on any written notice or other document furnished to it hereunder or under the Borrower Loan Agreement and reasonably believed by it to be genuine. The Governmental Lender shall in no event be liable for the application or misapplication of funds or for other acts or defaults by any person, except its own officers and employees. When any payment or consent or other action by it is called for hereby, it may defer such action pending receipt of such evidence (if any) as it may require in support thereof. The Governmental Lender shall not be required to take any remedial action (other than the giving of notice) hereunder or under any of the other Funding Loan Documents unless indemnity in a form acceptable to the Governmental Lender is furnished for any expense or liability to be incurred in connection with such remedial action. The Governmental Lender shall be entitled to reimbursement from the Borrower for its expenses reasonably incurred or advances reasonably made, with interest at the maximum rate of interest permitted under applicable law, in the exercise of its rights or the performance of its obligations hereunder, to the extent that it acts without previously obtaining indemnity. No permissive right or power to act which the Governmental Lender may have shall be construed as a requirement to act, and no delay in the exercise of a right or power shall affect its subsequent exercise of the right or power.
- (b) A default by the Borrower in any of its covenants, representations and agreements in the Borrower Loan Agreement, Regulatory Agreement or Tax Certificate on which the Governmental Lender is relying in the various sections of this Article VI shall not be considered a default hereunder by the Governmental Lender.

(c) The Borrower has indemnified the Governmental Lender against certain acts and events as set forth in [Section 7.24] of the Borrower Loan Agreement and Section 9 of the Regulatory Agreement. Such indemnities shall survive payment of the Funding Loan and discharge of this Funding Loan Agreement.

## **ARTICLE VII**

## **SECURITY**

- 7.1. Security for the Funding Loan. To secure the payment of the Funding Loan and the Funding Loan Note, the Governmental Lender hereby grants, bargains, sells, conveys, assigns, transfers, hypothecates, pledges and sets over to the Funding Lender (excepting only the Reserved Rights) a lien on and security interest in the following described property (collectively, the "Security"):
  - (a) All right, title and interest of the Governmental Lender in, to and under the Borrower Loan Agreement and the Borrower Note, including, without limitation, all rents, revenues and receipts derived by the Governmental Lender from the Borrower relating to the Development and, including, without limitation, all income, revenues, proceeds and other amounts which the Governmental Lender is entitled to derive from or in connection with the Development and the Borrower Loan Documents, including all amounts due under the Borrower Loan Agreement, the Borrower Note and the other Borrower Loan Documents and all amounts obtained after the exercise of the remedies provided in the Borrower Loan Documents and all receipts credited under the provisions of the Borrower Loan Agreement against said amounts payable;
  - (b) All right, title and interest of the Governmental Lender in, to and under the other Borrower Loan Documents, together with all rights, remedies, privileges and options pertaining to the Borrower Loan Documents, and all other payments, revenues and receipts derived by the Governmental Lender under and pursuant to, and subject to the provisions of, the Borrower Loan Documents;
  - (c) All right, title and interest of the Governmental Lender in and to: (i) the right to collect and receive net proceeds of any policy of insurance maintained pursuant to the Borrower Loan Documents; (ii) any award or payment becoming payable to the Governmental Lender under the Borrower Loan Documents by reason of any condemnation of the Development, any improvements located thereon or any conveyance in lieu of condemnation; and (iii) any bankruptcy, insolvency, reorganization or condemnation proceeding involving the Borrower or any Loan Party (as defined in the Borrower Loan Agreement) with respect to the Borrower Loan Documents;
  - (d) Any and all other real or personal property of every kind and nature or description which may from time to time hereafter, by delivery or by writing of any kind, be subject to the lien of this Funding Loan Agreement as additional security by the Governmental Lender or anyone on its part or with its consent or which pursuant to any of the provisions hereof or the Borrower Loan Documents may come into the possession or control of the Governmental Lender; and

(e) Any and all moneys and investments from time to time on deposit in, or forming a part of, all funds and accounts created and held under this Funding Loan Agreement, subject to the provisions of this Funding Loan Agreement permitting the application thereof for the purposes and on the terms and conditions set forth herein.

The pledge and assignment of and the security interest granted in the Security pursuant to this Section 7.1 for the payment of principal of, premium, if any, and interest on the Funding Loan Note, in accordance with its terms and provisions and for the payment of all other amounts due hereunder, shall attach and be valid and binding from and after the time of the delivery of the Funding Loan Note by the Governmental Lender. The Security so pledged and/or thereafter received by the Governmental Lender, the Bank or the Fiscal Agent shall immediately be subject to the lien of such pledge and security interest without any physical delivery or recording thereof or further act, and the lien of such pledge and security interest shall be valid and binding and prior to the claims of any and all parties having claims of any kind whether in tort, contract or otherwise against the Governmental Lender irrespective of whether such parties have notice thereof.

- 7.2. <u>Delivery of Security</u>. In connection with such pledge, assignment, transfer and conveyance of the Security pursuant to Section 7.1 herein, the Governmental Lender shall deliver to the Bank the following documents or instruments promptly following their execution and, to the extent applicable, their recordation or filing:
  - (a) The Borrower Note endorsed without recourse to the Funding Lender by the Governmental Lender;
  - (b) The Borrower Loan Agreement, Regulatory Agreement, Deed of Trust and other Borrower Loan Documents existing on the Closing Date and the Assignment of Deed of Trust assigning for security purposes and without recourse the Deed of Trust and Borrower Loan Documents from the Governmental Lender to the Funding Lender.
  - (c) Uniform Commercial Code financing statements or other chattel security documents giving notice of the Fiscal Agent's status as an assignee of the Governmental Lender's security interest in any personal property forming a part of the Development; and
  - (d) Uniform Commercial Code financing statements giving notice of the pledge by the Governmental Lender of the Security pledged under this Funding Loan Agreement, in forms provided by the Funding Lender.

The Governmental Lender shall deliver and deposit with the Funding Lender or the Fiscal Agent such additional documents, financing statements and instruments as the Funding Lender or the Fiscal Agent, at the direction of the Funding Lender, may reasonably request in writing from time to time for the purpose of better perfecting and assuring to the Fiscal Agent its lien and security interest in and to the Security in each case in forms provided by the Funding Lender and at the expense of the Borrower.

#### **ARTICLE VIII**

## **AGENCY**

- 8.1. <u>Assignment of Authority</u>. Solely for purposes of the administration of this Funding Loan Agreement, the Governmental Lender hereby vests the Fiscal Agent and the Funding Lender with full authority and power to act on its behalf for the purposes set forth herein as each is granted authority hereunder, and to do all other acts necessary or incidental to the performance and execution thereof, except for the Reserved Rights.
- 8.2. <u>Authority of the Fiscal Agent and Bank; Notices to Governmental Lender</u>. The Funding Lender is authorized and agrees to advance monies on behalf of the Governmental Lender to fund the Borrower Loan upon satisfaction of the conditions set forth in the Borrower Loan Agreement and otherwise to act on behalf of the Governmental Lender under the Borrower Loan Documents as expressly permitted hereunder and under the Borrower Loan Document, except for exercise of the Reserved Rights. Except for the Reserved Rights, the Funding Lender is hereby authorized, directed and empowered to exercise all the rights, powers or remedies of the Governmental Lender under the Borrower Loan Agreement and the other Borrower Loan Documents, and to make all determinations and exercise all options and elections thereunder, without the necessity of further advice or consultation with, or consent or authorization by, the Governmental Lender, and all actions taken by the Funding Lender under the Borrower Loan Agreement or any of the other Borrower Loan Documents shall be valid and shall have the same force and effect as if taken by the Governmental Lender. The Funding Lender shall have the right to exercise any rights and remedies conferred on the Governmental Lender pursuant to the Borrower Loan Documents (except for the Reserved Rights) as may be necessary or convenient to (i) enforce the payment of any amounts owing by the Borrower under the Borrower Loan Documents and prepayments thereof, or (ii) otherwise to protect the interest of the Governmental Lender, the Funding Lender or the Fiscal Agent upon a default by the Borrower under the Borrower The Fiscal Agent and the Funding Lender each agree to provide the Governmental Lender any notices given by it or delivered to it pursuant to the Borrower Loan Agreement regarding the occurrence of an Event of Default (as defined in the Borrower Loan Agreement), the acceleration of the Borrower Loan or the foreclosure of the Deed of Trust and shall provide written notice to the Governmental Lender of any amendment to the Borrower Note or the Borrower Loan Agreement. The Funding Lender shall have the right to collect all payments and other amounts received by the Governmental Lender from or on behalf of the Borrower pursuant to the Borrower Loan Agreement or the other Borrower Loan Documents, including prepayments thereof, except for payments of fees owing by the Borrower to the Governmental Lender in respect of the Reserved Rights.
- 8.3. <u>Successor Fiscal Agent</u>. Anything herein to the contrary notwithstanding, any corporation or association into which the Fiscal Agent may be converted or merged or with which it may be consolidated or to which it may sell or transfer its business and assets as a whole or substantially as a whole or any corporation or association resulting from any conversion, sale, merger, consolidation or transfer to which it is a party will, ipso facto, be and become the Fiscal Agent hereunder and vested with all of the title to the whole property and all the powers, discretion, immunities, privileges, obligations and all other matters as was its predecessor, without the

execution or filing of any instruments or any further act, deed or conveyance on the part of the parties hereto.

8.4. <u>Assignment of Rights</u>. The Governmental Lender agrees that the Funding Lender shall have the right to assign all of its rights under this Agreement, and under all instruments and documents executed by the Governmental Lender pursuant to this Agreement, to an Affiliate of the Funding Lender, or to a subsequent owner of all of the Funding Loan Note and the Funding Loan as permitted under Section 4.3 herein or an Affiliate thereof. The Funding Lender will advise the Governmental Lender in writing of any such proposed assignment and, following delivery of its consent, the Governmental Lender will execute and deliver to the Funding Lender any documents (at the expense of the Funding Lender) necessary to effectuate such assignment in forms provided by the Funding Lender.

# 8.5. [Reserved].

- 8.6. <u>Acceptance</u>. The Funding Lender hereby accepts the assignments and pledge made herein for the purpose of securing the payments due pursuant to the Funding Loan Agreement.
- 8.7. <u>Conditions</u>. This Article VIII shall confer no obligations or impose no duties upon the Funding Lender or the Fiscal Agent beyond those expressly provided in this Funding Loan Agreement and the Borrower Loan Agreement. This Article VIII shall confer no obligations or impose no duties upon the Governmental Lender beyond those expressly provided in this Funding Loan Agreement.

### ARTICLE IX

## **FUNDS AND ACCOUNTS**

- 9.1. <u>Authorization To Create Funds and Accounts</u>. Except as provided in Section 9.3 hereof, no funds or accounts shall be established in connection with the Funding Loan at the time of closing and origination of the Funding Loan. The Bank and the Fiscal Agent are authorized to establish and create from time to time such other funds and accounts or subaccounts as may be necessary for the deposit of moneys (including, without limitation, insurance proceeds and/or condemnation awards), if any, received by the Governmental Lender, the Bank or the Fiscal Agent, pursuant to the terms hereof or any of the other Funding Loan Documents and not immediately transferred or disbursed pursuant to the terms of the Funding Loan Documents and/or the Borrower Loan Documents.
- 9.2. <u>Investment of Funds</u>. Amounts held in any funds or accounts created under this Funding Loan Agreement shall be deposited in a deposit account, savings account or money market account with the Fiscal Agent at the direction of the Borrower, subject in all cases to the restrictions of Section 6.13 hereof and of the Tax Certificate.
- 9.3. <u>Establishment of Funds</u>. There are established with the Fiscal Agent the following funds and accounts:
  - (a) The Note Payment Fund;

- (b) The Project Fund;
- (c) The Expense Fund;
- (d) The Rebate Fund; and
- (e) The Closing Costs Fund.

All money required to be deposited with or paid to the Fiscal Agent for the account of any of the funds or accounts created by this Funding Loan Agreement shall be held by the Fiscal Agent in trust for the benefit of the owner of the Funding Loan Note, and, except for money held in the Expense Fund or the Rebate Fund, shall, while held by the Fiscal Agent, be subject to the lien hereof.

## 9.4. Note Payment Fund.

- (a) The Governmental Lender and the Borrower shall have no interest in the Note Payment Fund or the moneys therein, which shall always be maintained by the Fiscal Agent completely separate and segregated from all other moneys held hereunder and from any other moneys of the Governmental Lender and the Borrower.
- (b) The Fiscal Agent shall deposit into the Note Payment Fund any amounts received from the Borrower as payments of principal of or premium, if any, or interest on the Borrower Loan, any other amounts received by the Fiscal Agent that are subject to the lien and pledge of this Funding Loan Agreement and any funds pledged to the Bank hereunder not required to be deposited to the Expense Fund or not otherwise specifically directed in writing by the Funding Lender to be deposited into other funds created by this Funding Loan Agreement.
- (c) The Fiscal Agent shall apply all amounts on deposit in the Note Payment Fund in the following order of priority:

*First*, to pay or provide for the payment of the interest then due on the Funding Loan;

Second, to pay or provide for the payment or the prepayment of principal on the Funding Loan, and any premium thereon, provided moneys have been transferred or deposited into the Note Payment Fund for such purpose; and

*Third*, to pay or provide for the payment of the Funding Loan on its maturity date.

## 9.5. Expense Fund.

(a) The Fiscal Agent shall deposit in the Expense Fund the amounts required by the Regulatory Agreement or the Borrower Loan Agreement to be paid by the Borrower to the Governmental Lender or the Fiscal Agent. Amounts on deposit in the Expense Fund shall be used to pay the fees and expenses of the Governmental Lender and the Fiscal

Agent, as and when the same become due. In that regard, moneys in the Expense Fund shall be withdrawn or maintained, as appropriate, by the Fiscal Agent to pay (i) the Governmental Lender's ongoing fee pursuant to Section 7(n) of the Regulatory Agreement (the "Ongoing Governmental Lender Fee") to the Government Lender as and when due, (ii) the Fiscal Agent amounts due pursuant to the definition of the Fiscal Agent's Fees herein, (iii) upon receipt, to the Fiscal Agent, any amounts due to the Fiscal Agent which have not been paid, other than amounts paid in accordance with clause (ii) hereof, and (iv) upon receipt, to, or at the direction of, the Governmental Lender, any amounts owing the Governmental Lender by the Borrower and then due and unpaid, other than amounts paid in accordance with clause (i) hereof.

- (b) In the event that the amounts on deposit in the Expense Fund are not equal to the amounts payable from the Expense Fund as provided in Section 9.5(a) above on any date on which such amounts are due and payable, the Fiscal Agent shall give notice to the Borrower of such deficiency and of the amount of such deficiency and request payment within two Business Days to the Fiscal Agent of the amount of such deficiency.
- (c) Written notice of any insufficiency, which results in the Governmental Lender not receiving the Ongoing Governmental Lender Fee on the applicable due date, shall be provided by the Fiscal Agent to the Governmental Lender (with a copy to the Borrower and the Funding Lender) within 10 days of the respective due date. Upon payment by the Borrower of such deficiency, the amounts for which such deficiency was requested shall be paid by the Fiscal Agent to the Governmental Lender.
- (d) Notwithstanding anything herein to the contrary, the Fiscal Agent, on behalf of the Governmental Lender, shall prepare and submit a written invoice to the Borrower for payment of the Ongoing Governmental Lender Fee not later than 30 days prior to the due date for payment of such Ongoing Governmental Lender Fee, and shall remit moneys received from the Borrower to the Governmental Lender for payment of such fee.

## 9.6. Project Fund.

- (a) All proceeds of the Funding Loan provided by the Funding Lender shall be deposited to the Project Fund and disbursed as herein provided. The Fiscal Agent shall use moneys in the Project Fund for the acquisition, rehabilitation, construction and equipping of the Development, to pay other permitted development costs and to pay other costs related to the Development as provided herein.
- (b) Not less than 97% of the moneys deposited in and credited to the Project Fund, representing the proceeds of the Funding Loan, including any income earned thereon, will be expended for Qualified Project Costs (as defined in the Regulatory Agreement) (the "97% Requirement"). The amounts on deposit in the Project Fund shall not be applied to the payment of costs of issuance of the Funding Loan Note.
- (c) Before any payment representing Funding Loan Note proceeds shall be made from the Project Fund, the Regulatory Agreement and the Deed of Trust shall have been executed and recorded in the official records of the County of Los Angeles and there

shall be filed with the Fiscal Agent a written requisition of the Borrower substantially in the form attached hereto as Exhibit C ("Requisition") and approved by (i) the Funding Lender pursuant to the terms, conditions and provisions of the Borrower Loan Agreement and (ii) the Governmental Lender. The Governmental Lender agrees that if the Governmental Lender has not objected in writing to any disbursement from the Project Fund within five Business Days of receipt of a request for approval of such disbursement, the Governmental Lender shall be deemed to have approved such disbursement. Furthermore, if the Governmental Lender and the Bank disagree as to whether a particular disbursement from the Project Fund shall be approved or disapproved, they shall meet and confer in good faith, upon the request of either of them in an effort to resolve the matter, which meeting may be by telephonic or electronic means, or may be at a personal meeting. If they fail to agree upon the approval or disapproval of such a disbursement following such good faith efforts, the Funding Lender can approve the disbursement and the Fiscal Agent shall pay it from the Project Fund. The Fiscal Agent shall be entitled to conclusively rely upon any Requisition in determining whether to disburse amounts from the Project Fund.

- (d) In connection with a Requisition, except for a written request for amounts representing accrued interest due and payable on the Funding Loan Note:
  - (i) Only the signature of the Funding Lender and the Governmental Lender shall be required on a Requisition during any period in which an Event of Default by the Borrower has occurred and is then continuing under the Borrower Loan Agreement (notice of which default has been given in writing by the Funding Lender to the Fiscal Agent and the Governmental Lender, and the Fiscal Agent shall be entitled to conclusively rely on any such written notice as to the occurrence and continuation of such a default).
  - (ii) The Fiscal Agent shall disburse amounts in the Project Fund upon receipt of a Requisition signed only by the Funding Lender and the Governmental Lender, as applicable (and without any need for any signature by an Authorized Representative), so long as the amount to be disbursed is to be used solely to make payments of principal, interest and/or fees due under the Funding Loan Documents.
  - (iii) The Fiscal Agent may conclusively rely on all Requisitions, the execution of the Requisitions by the Borrower and the approval of all Requisitions by the Funding Lender and the Governmental Lender, as required by this Section, as conditions of payment from the Project Fund, which Requisitions constitute, as to the Fiscal Agent, irrevocable determinations that all conditions to payment of the specified amounts from the Project Fund have been satisfied. These documents shall be retained by the Fiscal Agent, subject at all reasonable times to examination by the Borrower, the Governmental Lender, the Funding Lender and the agents and representatives thereof upon reasonable notice to the Fiscal Agent. The Fiscal Agent is not required to inspect the Development or the rehabilitation or construction work or to make any independent investigation with respect to the matters set forth in any Requisition or other statements, orders, certifications and approvals received by the Fiscal Agent. The Fiscal Agent is not required to obtain

completion bonds, lien releases or otherwise supervise the acquisition, rehabilitation, construction, equipping, improvement and installation of the Development.

- (e) Upon receipt of each Requisition submitted by the Borrower and approved in writing by the Funding Lender and the Governmental Lender, as applicable, the Fiscal Agent shall promptly, but in any case within one Business Day, make payment from the Project Fund in accordance with such Requisition. The Fiscal Agent shall have no duty to determine whether any requested disbursement from the Project Fund complies with the terms, conditions and provisions of the Funding Loan Documents, constitutes payment of Qualified Project Costs or complies with the 97% Requirement. The approval in writing of a Requisition by the Funding Lender shall be deemed a certification and, insofar as the Fiscal Agent and the Governmental Lender are concerned, shall constitute conclusive evidence that all of the terms, conditions and requirements of the Funding Loan Documents applicable to such disbursement have been fully satisfied or waived and the Requisition from the Borrower shall, insofar as the Fiscal Agent and the Governmental Lender are concerned, constitute conclusive evidence that the costs described in the Requisition constitute Qualified Project Costs or other permitted development costs.
- The Fiscal Agent shall immediately provide written notice to the Borrower, the Funding Lender and the Governmental Lender if there are not sufficient funds available to or on deposit with the Fiscal Agent to make the transfers as and when required by Section 9.6(e) above. Except as provided in the next sentence, all such payments shall be made by check or draft payable, or by wire transfer, either (i) directly to the person, firm or corporation to be paid, (ii) to the Borrower and such person, firm or corporation, or (iii) upon receipt by the Funding Lender and the Governmental Lender of evidence that the Borrower has previously paid such amount and written direction to the Fiscal Agent as to such as evidenced by the Funding Lender's and the Governmental Lender's approval of the Requisition, to the Borrower. Upon the occurrence of an Event of Default of the Borrower of which the Fiscal Agent has knowledge as provided herein, which is continuing under the Funding Loan Documents, with the written consent of the Funding Lender, the Fiscal Agent may apply amounts on deposit in the Project Fund to the payment of principal of and interest on the Funding Loan. If a Requisition signed by the Borrower Representative and countersigned by the Funding Lender and the Governmental Lender is received by the Fiscal Agent, the requested disbursement shall be paid by the Fiscal Agent as soon as practicable, but in no event later than three Business Days following receipt thereof by the Fiscal Agent. Upon final disbursement of all amounts on deposit in the Project Fund, the Fiscal Agent shall close the Project Fund.
- (g) Immediately prior to any mandatory prepayment of the Funding Loan pursuant hereto, any amounts then remaining in the Project Fund shall, at the written direction of the Funding Lender, be applied to the prepayment of the Funding Loan pursuant hereto.
- (h) Investment income earned on amounts on deposit in the Project Fund shall be retained in and credited to and become a part of the amounts on deposit in the Project Fund.

- (i) Prior to the Conversion Date, the Funding Lender shall disburse the Funding Loan directly to the Funding Lender to pay accrued interest due and payable on the Funding Loan Note and will provide written notice of the amount of such disbursement to the Fiscal Agent within three Business Days of the disbursement.
- 9.7. <u>Rebate Fund</u>. The Fiscal Agent shall deposit or transfer to the credit of the Rebate Fund each amount delivered to the Fiscal Agent by the Borrower for deposit thereto and each amount directed by the Borrower to be transferred thereto.
  - (a) Within 15 days after each receipt or transfer of funds to the Rebate Fund, the Fiscal Agent shall withdraw from the Rebate Fund and pay to the United States of America the entire balance of the Rebate Fund.
  - (b) All payments to the United States of America pursuant to this Section shall be made by the Fiscal Agent for the account and in the name of the Governmental Lender and shall be paid through the United States mail (return receipt requested or overnight delivery), addressed to the appropriate Internal Revenue Service Center and accompanied by the appropriate Internal Revenue Service forms (such forms to be provided to the Fiscal Agent by the Borrower or the Rebate Analyst).
  - (c) The Fiscal Agent shall preserve all statements, forms and explanations received from the Borrower and delivered to the Fiscal Agent and all records of transactions in the Rebate Fund until six years after the retirement of the Funding Loan Note.
  - (d) The Fiscal Agent may conclusively rely on the instructions of the Borrower (based upon the report of the Rebate Analyst) with regard to any actions to be taken by it pursuant to this Section 9.7 and shall have no liability for any consequences of any failure of the Borrower or the Rebate Analyst to perform its duties or obligations or to supply accurate or sufficient instructions. Except as specifically provided in subsection (b) above, the Fiscal Agent shall have no duty or responsibility with respect to the Rebate Fund or the Borrower's duties and responsibilities with respect thereto except to follow the Borrower's specific written instruction related thereto.
  - (e) If at any time during the term of this Funding Loan Agreement the Governmental Lender, the Fiscal Agent or the Borrower desires to take any action which would otherwise be prohibited by the terms of this Section, such person shall be permitted to take such action if it shall first obtain and provide to the other persons named herein an opinion of Tax Counsel that such action shall be in compliance with the laws of the State and the terms of this Funding Loan Agreement and will not impair the exclusion of interest on the Funding Loan from gross income for purposes of federal income taxation.
  - (f) Moneys and securities held by the Fiscal Agent in the Rebate Fund shall not be deemed funds of the Governmental Lender and are not pledged or otherwise subject to any security interest in favor of the owners to secure the Funding Loan Note or any other obligations.

- (g) Moneys in the Rebate Fund may be separately invested and reinvested by the Fiscal Agent, at the request of and as directed in writing by the Borrower, subject to the Code. The Fiscal Agent shall sell and reduce to cash a sufficient amount of such investments whenever the cash balance in the Rebate Fund is insufficient for its purposes.
- (h) Notwithstanding anything to the contrary in this Funding Loan Agreement, no payment shall be made by the Fiscal Agent to the United States if the Borrower shall furnish to the Governmental Lender and the Fiscal Agent an opinion of Tax Counsel to the effect that such payment is not required under Section 148(d) and (f) of the Code in order to maintain the exclusion from gross income for federal income tax purposes of interest on the Funding Loan Note. In such event the Borrower shall be entitled to withdraw funds from the Rebate Fund to the extent the Borrower shall provide an opinion of Tax Counsel that such action will not impair the exclusion of interest on the Funding Loan from gross income for purposes of federal income taxation to the Governmental Lender and the Fiscal Agent with respect to such withdrawal.
- (i) The Fiscal Agent shall keep and make available to the Governmental Lender and the Borrower records concerning the investments of all funds held by the Fiscal Agent pursuant to the Funding Loan Agreement including date bought and sold, price and commission paid, and bids taken, if any, and shall keep all such records until six years after the date on which the Funding Loan Note is repaid in full in order to enable the Borrower to make the computations required under Section 148(f) of the Code.
- (j) Notwithstanding the foregoing, the computations and payments of rebate amounts referred to in this Section 9.7 need not be made to the extent that neither the Governmental Lender nor the Borrower will thereby fail to comply with any requirements of Section 148(f) of the Code based on an opinion of Tax Counsel that such action will not impair the excluding of interest on the Funding Loan from gross income for purposes of federal income taxation, a copy of which shall be provided to the Fiscal Agent.
- 9.8. Closing Costs Fund. On the Closing Date, the Borrower shall deposit or cause to be deposited with the Fiscal Agent, for deposit in the Closing Costs Fund, the amount of [\_\_\_\_\_]. Amounts in the Closing Costs Fund shall be disbursed by the Fiscal Agent to pay Closing Costs on the Closing Date or as soon as practicable thereafter as follows: (a) pay the Fiscal Agent its closing fee of \$[\_\_\_\_]; and (b) pay to the California Debt and Investment Advisory Commission ("CDIAC") the CDIAC fee of up to \$[\_\_\_\_] upon receipt of an invoice therefor. Any interest earnings on amounts on deposit in the Closing Costs Fund shall remain in such Fund. Any moneys remaining in the Closing Costs Fund (including investment proceeds) after the earlier of (i) the payment of all costs of issuance as certified in writing to the Fiscal Agent by the Borrower or (ii) a period of five months after the Closing Date shall be paid to the Borrower and the Closing Costs Fund shall be closed.

#### **ARTICLE X**

## THE FISCAL AGENT

10.1. <u>Appointment of Fiscal Agent; Acceptance</u>. The Governmental Lender hereby appoints [FISCAL AGENT] as Fiscal Agent hereunder. The Fiscal Agent shall signify its acceptance of the duties and obligations imposed upon it by this Funding Loan Agreement by executing this Funding Loan Agreement.

# 10.2. Certain Duties and Responsibilities of Fiscal Agent.

- (a) The Fiscal Agent undertakes to perform such duties and only such duties as are specifically set forth in this Funding Loan Agreement, and no implied covenants or obligations shall be read into this Funding Loan Agreement against the Fiscal Agent.
- (b) If an event of default exists hereunder or under any Borrower Loan Document, the Fiscal Agent shall exercise such of the rights and powers vested in it by this Funding Loan Agreement, and, subject to Section 10.2(c)(iv) hereof, use the same degree of care and skill in their exercise as a prudent corporate trust officer would exercise or use under the circumstances in the conduct of corporate trust business.
- (c) No provision of this Funding Loan Agreement shall be construed to relieve the Fiscal Agent from liability for its own negligent action, its own negligent failure to act, or its own willful misconduct, in each case, as finally adjudicated by a court of law, except that:
  - (i) This subsection shall not be construed to limit the effect of Section 10.2(a) hereof;
  - (ii) The Fiscal Agent shall not be liable for any error of judgment made in good faith, unless it shall be proved that the Fiscal Agent was negligent in ascertaining the pertinent facts;
  - (iii) The Fiscal Agent shall not be liable with respect to any action taken or omitted to be taken by it in accordance with the direction of the Funding Lender relating to the time, method and place of conducting any proceeding for any remedy available to the Fiscal Agent, or exercising any trust or power conferred upon the Fiscal Agent under this Funding Loan Agreement; and
  - (iv) No provision of this Funding Loan Agreement shall require the Fiscal Agent to expend or risk its own funds or otherwise incur any financial liability in the performance of any of its duties hereunder, or in the exercise of any of its rights or powers, if it shall have reasonable grounds for believing that repayment of such funds or adequate indemnity against such risk or liability is not assured to it in its sole discretion.
- (d) Whether or not therein expressly so provided, every provision of this Funding Loan Agreement and the other Funding Loan Documents relating to the conduct

or affecting the liability of or affording protection to the Fiscal Agent shall be subject to the provisions of this Section 10.2.

- (e) The Fiscal Agent may conclusively rely, as to the truth of the statements and the correctness of the opinions expressed therein, upon certificates or opinions furnished to the Fiscal Agent and conforming to the requirements of this Funding Loan Agreement; but in the case of any such certificates or opinions which by any provision hereof are specifically required to be furnished to the Fiscal Agent, the Fiscal Agent shall be under a duty to examine the same to determine whether or not they conform to the requirements of this Funding Loan Agreement.
- (f) The permissive rights of the Fiscal Agent to do things enumerated in this Funding Loan Agreement shall not be construed as a duty.
- 10.3. Notice of Defaults. Upon the occurrence of any default hereunder or under any Borrower Loan Document, and provided that the Fiscal Agent is aware of or has received written notice of the existence of such default, promptly, and in any event within 15 days, the Fiscal Agent shall transmit to the Governmental Lender, the Borrower, and the Funding Lender, in the manner and at the addresses for notices set forth in Section 12.2 hereof, notice of such default hereunder known to the Fiscal Agent pursuant to Section 10.4(g) hereof, unless such default shall have been cured or waived.

# 10.4. <u>Certain Rights of Fiscal Agent</u>. Except as otherwise provided in Section 10.1 hereof:

- (a) The Fiscal Agent may rely and shall be protected in acting or refraining from acting upon any resolution, certificate, statement, instrument, opinion, report, notice, request, direction, consent, order, bond, note, debenture, coupon or other paper or document believed by it to be genuine and to have been signed or presented by the proper party or parties;
- (b) Any request or direction of the Governmental Lender mentioned herein shall be sufficiently evidenced by a certificate or order executed by an Authorized Representative of the Governmental Lender;
- (c) Whenever in the administration of this Funding Loan Agreement the Fiscal Agent shall deem it desirable that a matter be proved or established prior to taking, suffering or omitting any action hereunder, the Fiscal Agent (unless other evidence be herein specifically prescribed) may, in the absence of bad faith on its part, rely upon a written certificate of the Funding Lender or the Borrower, as appropriate;
- (d) The Fiscal Agent shall be under no obligation to exercise any of the rights or powers vested in it by this Funding Loan Agreement at the request or direction of the Funding Lender, pursuant to this Funding Loan Agreement, unless the Funding Lender shall have offered to the Fiscal Agent in writing security or indemnity reasonably satisfactory to the Fiscal Agent against the costs, expenses and liabilities which might be incurred by it in compliance with such request or direction, except costs, expenses and liabilities which are adjudicated to have resulted from its own negligence or willful misconduct;

- (e) The Fiscal Agent shall not be bound to make any investigation into the facts or matters stated in any resolution, certificate, statement, instrument, opinion, report, notice, request, direction, consent, order, bond, note, debenture, coupon or other paper or document but the Fiscal Agent, in its discretion, may make such further inquiry or investigation into such facts or matters as it may see fit, and, if the Fiscal Agent shall determine to make such further inquiry or investigation, it shall be entitled to examine the books and records of the Governmental Lender, if any, and of the Borrower, in either case personally or by agent or attorney after reasonable notice and during normal business hours;
- (f) The Fiscal Agent may execute any of the trusts or powers hereunder or perform any duties hereunder either directly or by or through agents or attorneys and pay reasonable compensation thereto and the Fiscal Agent shall not be responsible for any misconduct or negligence on the part of any agent or attorney appointed with due care by it hereunder. The Fiscal Agent may act upon the advice of counsel of its choice concerning all matters hereof and the Fiscal Agent shall not be responsible for any loss or damage resulting from any action or inaction taken in good faith reliance upon said advice; and
- notice of any default hereunder or under any Borrower Loan Document except for failure by the Borrower to make payments of principal, interest, premium, if any, or fee owing to the Governmental Lender pursuant to the Regulatory Agreement when due, unless the Fiscal Agent shall be specifically notified by a written notice of such default by the Governmental Lender or the Funding Lender, and all notices or other instruments required by this Funding Loan Agreement or under any Borrower Loan Document to be delivered to the Fiscal Agent, must, in order to be effective, be delivered in writing to the Fiscal Agent, and in the absence of such written notice so delivered the Fiscal Agent may conclusively assume there is no default as aforesaid.

## 10.5. Not Responsible for Recitals.

- (a) The recitals contained herein and in the Funding Loan Note shall be taken as the statements of the Governmental Lender, and the Fiscal Agent assumes no responsibility for their correctness. The Fiscal Agent makes no representations as to the value or condition of the revenues pledged hereunder, the Security or any part thereof, or as to the title of the Governmental Lender thereto or as to the security afforded thereby or hereby, or as to the validity or sufficiency of this Funding Loan Agreement or of the Funding Loan.
- (b) The Fiscal Agent shall have no responsibility or liability with respect to any information, statement or recital in any offering memorandum or other disclosure material prepared or distributed with respect to the funding of the Funding Loan.
- (c) The Fiscal Agent shall not be required to monitor the financial condition of the Borrower or the physical condition of the Development. Unless otherwise expressly provided, the Fiscal Agent shall be under no obligation to analyze, review or make any credit decisions with respect to any financial statements, reports, notices, certificates or

documents received hereunder but shall hold such financial statements, reports, notices, certificates and documents solely for the benefit of, and review by, the Funding Lender and such other parties to whom the Fiscal Agent may provide such information pursuant to this Funding Loan Agreement.

- (d) The Fiscal Agent makes no representations as to and shall have no responsibility for the sufficiency of the insurance required under any of the Borrower Loan Documents.
- 10.6. <u>May Hold Funding Loan</u>. The Fiscal Agent in its individual or any other capacity may become the owner or pledgee of the Funding Loan and may otherwise deal with the Governmental Lender, the Funding Lender and the Borrower with the same rights it would have if it were not the Fiscal Agent.
- 10.7. Moneys Held in Trust. Moneys held by the Fiscal Agent in trust hereunder need not be segregated from other funds except to the extent required by law. The Fiscal Agent shall be under no liability for interest on any moneys received by it hereunder except as otherwise provided herein.

## 10.8. Compensation and Reimbursement.

- (a) Under the Borrower Loan Agreement, the Borrower has agreed, except as otherwise expressly provided herein, to reimburse the Fiscal Agent as provided in this Funding Loan Agreement or the Borrower Loan Agreement, upon its request for all reasonable expenses, disbursements and advances incurred or made by the Fiscal Agent in accordance with any provision of this Funding Loan Agreement (including the reasonable fees, expenses and disbursements of its agents and counsel), except any such expense, disbursement or advance as may be attributable to the Fiscal Agent's negligence or willful misconduct, both as finally adjudicated by a court of law.
- (b) When the Fiscal Agent incurs expenses or renders service in connection with any bankruptcy or insolvency proceeding, such expenses (including the fees and expenses of its counsel) and the compensation for such services are intended to constitute expenses of administration under any bankruptcy law or law relating to creditors rights generally.
- (c) The Governmental Lender has no obligation to pay the Fiscal Agent for services rendered.
- (d) As security for the performance of the obligations of the Borrower under this Section 10.8 and for the payment of such compensation, expenses, reimbursements and indemnity, the Fiscal Agent shall have the right to use and apply any moneys held by it as Security hereunder.
- (e) The Fiscal Agent's rights to compensation and reimbursement shall survive its resignation or removal, the payment of the Funding Loan or the Borrower Loan or the release of this Funding Loan Agreement.

10.9. Fiscal Agent Required; Eligibility. Any successor Fiscal Agent shall at all times be a trust company, a state banking corporation or a national banking association with the authority to accept trusts in the State of California approved in writing by the Governmental Lender and either (a) have a combined capital and surplus of at least \$50,000,000 as set forth in its most recent published annual report of condition, (b) be a wholly owned subsidiary of a bank holding company, or a wholly owned subsidiary of a company that is a wholly owned subsidiary of a bank holding company, having a combined capital surplus of at least \$50,000,000 as set forth in its most recent published annual report of condition, have at least \$500,000,000 of trust assets under management and have a combined capital surplus of at least \$2,000,000 as set forth in its most recent published annual report of condition, or (c) be otherwise acceptable to the Funding Lender and the Governmental Lender in their sole and absolute discretion.

# 10.10. Resignation and Removal; Appointment of Successor.

- (a) No resignation or removal of the Fiscal Agent hereunder and no appointment of a successor Fiscal Agent pursuant to this Article X shall become effective until the written acceptance by the successor Fiscal Agent of such appointment.
- (b) The Fiscal Agent may resign at any time by giving 60 days' written notice thereof to the Governmental Lender, the Borrower, and the Funding Lender. If an instrument of acceptance by a successor Fiscal Agent shall not have been delivered to the Fiscal Agent within 30 days after the giving of such notice of resignation, the resigning Fiscal Agent may petition any court of competent jurisdiction for the appointment of a successor Fiscal Agent.
- (c) The Fiscal Agent may be removed at any time with 30 days' notice by (i) the Governmental Lender, with the written consent of the Funding Lender in its sole and absolute discretion, (ii) the Borrower (unless the Borrower is in default under any of the Borrower Loan Documents), subject to applicable notice and cure periods, with the written consent of the Funding Lender and the Governmental Lender in their sole and absolute discretion, or (iii) the Funding Lender with the written consent of the Governmental Lender and written notice delivered to the Fiscal Agent and the Borrower.
- (d) If the Fiscal Agent shall resign, be removed or become incapable of acting, or if a vacancy shall occur in the office of the Fiscal Agent for any cause, the Governmental Lender shall promptly appoint a successor Fiscal Agent, with the consent of the Funding Lender. In case all or substantially all of the revenues pledged hereunder and Security shall be in the possession of a receiver or trustee lawfully appointed, such receiver or trustee may similarly appoint a successor to fill such vacancy until a new Fiscal Agent shall be so appointed by the Governmental Lender. If, within 60 days after such resignation, removal or incapability or the occurrence of such vacancy, the Governmental Lender has failed to so appoint a successor Fiscal Agent, then a successor Fiscal Agent shall be appointed by the Funding Lender (from any of the institutions approved by the Governmental Lender to serve as a fiscal agent or trustee) with written notice thereof delivered to the Governmental Lender, the Borrower, and the retiring Fiscal Agent, and the successor Fiscal Agent so appointed shall, forthwith upon its acceptance of such appointment, become the successor Fiscal Agent and supersede the successor Fiscal Agent appointed by such receiver or Fiscal

Agent. If no successor Fiscal Agent shall have been appointed by the Governmental Lender or the Funding Lender and accepted appointment in the manner hereinafter provided, the Fiscal Agent may petition any court of competent jurisdiction for the appointment of a successor Fiscal Agent.

(e) The retiring Fiscal Agent shall cause written notice of each resignation and each removal of the Fiscal Agent and each appointment of a successor Fiscal Agent to be provided to the Funding Lender. Each notice shall include the name of the successor Fiscal Agent and the address of the office of the successor Fiscal Agent.

## 10.11. Acceptance of Appointment by Successor.

- Every successor Fiscal Agent appointed hereunder shall execute, (a) acknowledge and deliver to the Governmental Lender and to the retiring Fiscal Agent an instrument accepting such appointment, and thereupon the resignation or removal of the retiring Fiscal Agent shall become effective and such successor Fiscal Agent, without any further act, deed or conveyance, shall become vested with all the estates, properties, rights, powers, trusts and duties of the retiring Fiscal Agent; notwithstanding the foregoing, on request of the Governmental Lender or the successor Fiscal Agent, such retiring Fiscal Agent shall, upon payment of its charges, execute and deliver an instrument conveying and transferring to such successor Fiscal Agent upon the trusts herein expressed all the estates, properties, rights, powers and trusts of the retiring Fiscal Agent, and shall duly assign, transfer and deliver to such successor Fiscal Agent all property and money held by such retiring Fiscal Agent hereunder. Upon request of any such successor Fiscal Agent, the Governmental Lender shall execute any and all instruments for more fully and certainly vesting in and confirming to such successor Fiscal Agent all such estates, properties, rights, powers and trusts.
- (b) No successor Fiscal Agent shall accept its appointment unless at the time of such acceptance such successor Fiscal Agent shall be qualified and eligible under this Article X, to the extent operative.
- 10.12. Merger, Conversion, Consolidation or Succession to Business. Any corporation into which the Fiscal Agent may be merged or with which it may be consolidated, or any corporation resulting from any merger, conversion or consolidation to which the Fiscal Agent shall be a party, or any corporation succeeding to all or substantially all of the corporate trust business of the Fiscal Agent, shall be the successor of the Fiscal Agent hereunder, provided such corporation shall be otherwise qualified and eligible under this Article X, to the extent operative, without the execution or filing of any paper or any further act on the part of any of the parties hereto. Notwithstanding the foregoing, any such successor Fiscal Agent shall cause written notice of such succession to be delivered to the Funding Lender and the Governmental Lender within 30 days of such succession.
- 10.13. Appointment of Co-Fiscal Agent. It is recognized that in case of litigation under this Funding Loan Agreement, the Borrower Loan Agreement, any other Borrower Loan Document or the Regulatory Agreement, and in particular in case of the enforcement of any of them on default, or in case the Fiscal Agent deems that by reason of any present or future law of

any jurisdiction it may not exercise any of the powers, rights or remedies herein granted to the Fiscal Agent or hold title to the properties, in trust, as herein provided, or take any other action which may be desirable or necessary in connection therewith, it may be necessary that the Fiscal Agent appoint an additional individual or institution as a separate or co-fiscal agent. The following provisions of this Section are adopted to these ends.

The Fiscal Agent is hereby authorized to appoint an additional individual or institution as a separate or co-fiscal agent hereunder, upon written notice to the Governmental Lender, the Funding Lender and the Borrower, and with the consent of the Governmental Lender and the Funding Lender, but without the necessity of further authorization or consent, in which event each and every remedy, power, right, claim, demand, cause of action, immunity, estate, title, interest and lien expressed or intended by this Funding Loan Agreement, any Borrower Loan Document, the Regulatory Agreement or the Borrower Loan Agreement to be exercised by or vested in or conveyed to the Fiscal Agent with respect thereto shall be exercisable by and vest in such separate or co-fiscal agent but only to the extent necessary to exercise such powers, rights and remedies, and every covenant and obligation necessary to the exercise thereof by such separate or co-fiscal agent shall run to and be enforceable by either of them.

Should any instrument in writing from the Governmental Lender be required by the separate fiscal agent or co-fiscal agent appointed by the Fiscal Agent for more fully and certainly vesting in and confirming to it such properties, rights, powers, trusts, duties and obligations, any and all such instruments in writing shall, on request of the Fiscal Agent, be executed, acknowledged and delivered by the Governmental Lender. In case any separate fiscal agent or co-fiscal agent, or a successor to either, shall die, become incapable of acting, resign or be removed, all the estates, properties, rights, powers, trusts, duties and obligations of such separate fiscal agent or co-fiscal agent, so far as permitted by law, shall vest in and be exercised by the Fiscal Agent until the appointment of a successor to such separate fiscal agent or co-fiscal agent.

10.14. No Recourse Against Officers or Employees of Fiscal Agent. No recourse with respect to any claim related to any obligation, duty or agreement contained in this Funding Loan Agreement or any other Funding Loan Document shall be had against any officer or employee, as such, of the Fiscal Agent, it being expressly understood that the obligations, duties and agreements of the Fiscal Agent contained in this Funding Loan Agreement and the other Funding Loan Documents are solely corporate in nature.

## **ARTICLE XI**

## **DEFAULTS UNDER BORROWER LOAN DOCUMENTS**

11.1. <u>Default Under Borrower Loan Agreement; Acceleration</u>. No default by the Borrower under the Borrower Loan Agreement shall constitute an event of default with respect to the Funding Loan Note. The Governmental Lender's, the Fiscal Agent's and the Funding Lender's remedies with respect to a default under the Borrower Loan Documents shall be as set forth under the Borrower Loan Documents. The Funding Lender may, upon the acceleration of the Borrower's obligations under the Borrower Loan Documents, accelerate, or direct the Fiscal Agent to accelerate, the maturity of the Funding Loan Note and apply any funds available hereunder for such purpose as provided herein (after paying the fees and expenses of the Fiscal Agent and the

Governmental Lender). The Funding Loan Note shall be deemed paid upon transfer, to or at the direction of the Funding Lender, of the Borrower Loan Documents and all Security therefor free and clear of the lien of this Funding Loan Agreement.

The Governmental Lender shall cooperate with the Funding Lender and the Fiscal Agent in exercising rights and remedies under the Funding Loan Documents and the Borrower Loan Documents, but only upon being satisfactorily indemnified by the Borrower for any fees or expenses relating thereto as provided in the Borrower Loan Agreement and Regulatory Agreement.

11.2. Actions Under Borrower Loan Documents. Whether or not an Event of Default has occurred, the Funding Lender, in its sole discretion, shall have the sole right to waive or forebear any term, condition, covenant or agreement in the Borrower Loan Documents applicable to the Borrower or any breach thereof, other than the covenant that would adversely impact the tax-exempt status of the interest on the Funding Loan Note and provided that the Funding Lender shall have no right to waive and the Governmental Lender may seek specific performance by the Borrower to enforce the Reserved Rights. With respect to any of its Reserved Rights, the Governmental Lender may exercise such rights as permitted by Section 18 of the Regulatory Agreement in connection with a default thereunder.

## **ARTICLE XII**

## **MISCELLANEOUS**

- 12.1. <u>Entire Agreement</u>. This Funding Loan Agreement, the Funding Loan Note and the other Funding Loan Documents constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, between the Governmental Lender and the Funding Lender with respect to the subject matter hereof.
- 12.2. <u>Notices</u>. All notices, certificates or other communications shall be in writing and shall be sufficiently given and shall be deemed given on the second day following the date on which the same have been personally delivered or mailed by first class mail postage prepaid, addressed as follows:

If to the Governmental Lender:	City of Los Angeles c/o Los Angeles Housing and Community Investment Department 8th Floor 1200 West 7th Street Los Angeles, CA 90017 Attention: Supervisor, Affordable Housing Bond Program HIMS# [] Facsimile: (213) 808-8918
with a copy to:	Los Angeles Housing and Community Investment

Department P.O. Box 532729

Los Angeles, CA 90053-2729

Attention: Supervisor, Affordable Housing Bond

Program

If to the Borrower: SP7 Apartments LP

c/o The Skid Row Housing Trust

1317 E. 7th St.

Los Angeles, CA 90021

with a copy to: Gubb & Barshay LLP

505 14<sup>th</sup> Street, Suite 450 Oakland, CA 94612 Attention: Scott Barshay Facsimile: (4150 781-6967

with a copy to: MUFG Union Bank, N.A.

200 Pringle Avenue, Suite 355 Walnut Creek, CA 94596 Attention: CDF Division Head Facsimile: (925) 947-2455

with a copy to: Rutan & Tucker LLP

611 Anton Boulevard, 14<sup>th</sup> Floor

Costa Mesa, CA 92626 Attention: Pat McCalla

If to the Funding Lender:

MUFG Union Bank, N.A.,

Loan Administration Department 3151 East Imperial Highway, 1st Floor

Brea, CA 92821 Attention: Manager

Facsimile: (323) 720-2433

with a copy to: MUFG Union Bank, N.A.,

Community Development Finance Department

1901 Avenue of the Stars, Suite 600

Los Angeles, CA 90067

Attention: Manager

Facsimile: (310) 551-8980

If to the Fiscal A	gent:	[FISCAL AGENT]

[\_\_\_\_\_]
Los Angeles, CA 90071

Attention: Global Corporate Trust Services

Ref: LA MF (SP7 Apartments 2018C)

Telephone: (213) [\_\_\_\_\_]
Facsimile: (213) [\_\_\_\_\_]

- 12.3. <u>Assignments</u>. Except as provided in Section 4.3 hereof, neither this Funding Loan Agreement nor any other Funding Loan Document may be assigned by any party hereto or thereto in whole or in part without the prior written consent of the other, which consent shall not be unreasonably withheld, and, in the case of the Governmental Lender, to the extent such assignment is not in contravention of its policies for tax-exempt debt.
- 12.4. <u>Severability</u>. If any provision of this Funding Loan Agreement shall be held or deemed to be or shall, in fact, be illegal, inoperative or unenforceable, the same shall not affect any other provision or provisions herein contained or render the same invalid, inoperative, or unenforceable to any extent whatever.
- 12.5. Execution of Counterparts. This Funding Loan Agreement may be executed in several counterparts, each of which shall be an original and all of which shall constitute but one and the same instrument.
- 12.6. <u>Amendments, Changes and Modifications</u>. Except as otherwise provided in this Funding Loan Agreement, this Funding Loan Agreement may not be effectively amended, changed, modified, altered or terminated without the written consent of the parties hereto. The Funding Lender may require, as a condition to any amendment, change or modification of this Funding Loan Agreement or the other Funding Loan Documents, that the Funding Lender shall have received, at the expense of the Borrower, an opinion of Tax Counsel that such amendment shall not adversely affect the exclusion of interest on the Funding Loan Note from gross income for purposes of federal income tax.
- 12.7. <u>Governing Law</u>. This Funding Loan Agreement shall be governed exclusively by and construed in accordance with the applicable laws of the State.
- 12.8. <u>Term of Agreement</u>. This Funding Loan Agreement shall be in full force and effect from the date hereof until such time as the Funding Loan shall have been fully paid or provision made for such payment. Time is of the essence in this Funding Loan Agreement.
- 12.9. <u>Survival of Agreement</u>. All agreements, representations and warranties made herein shall survive the making of the Funding Loan.
- 12.10. <u>Nondiscrimination and Affirmative Action</u>. The Fiscal Agent shall comply with the applicable nondiscrimination and affirmative action provisions of the laws of the United States of America, the State of California, and the City of Los Angeles, California (the "City"). The

Fiscal Agent shall not discriminate in its employment practices against any employee or applicant for employment regarding family and medical care leave, pregnancy disability leave, or reasonable accommodations because of such person's race, ancestry, color, citizenship, national origin, religion, sex, sexual orientation, gender identity/expression, transgender status, age, marital status, familial status, domestic partner status, physical handicap, mental disability, medical condition, political affiliation or belief. The Fiscal Agent shall comply with Executive Order 11246, entitled "Equal Employment Opportunity," as amended by Executive Order 11375, and as supplemented in Department of Labor regulations (41 C.F.R. Part 60). The Fiscal Agent shall comply with the provisions of the Los Angeles Administrative Code Sections 10.8 through 10.13, to the extent applicable hereto. The affirmative action program of the Fiscal Agent shall include the mandatory contract provisions set forth in the Los Angeles Administrative Code Section 10.8.4, and said provisions are incorporated herein by this reference. The Fiscal Agent shall also comply with all rules, regulations, and policies of the City's Board of Public Works, Office of Contract Compliance relating to nondiscrimination and affirmative action, including the filing of all forms required by the City. Any subcontract entered into by the Fiscal Agent relating to this Funding Loan Agreement, to the extent allowed hereunder, shall be subject to the provisions of this Section. No person shall on the grounds of race, ancestry, color, citizenship, national origin, religion, sex, sexual orientation, gender identity/expression, transgender status, age, marital status, familial status, domestic partner status, physical handicap, mental disability, medical condition, political affiliation or belief be excluded from participation in, be denied the benefit of, or be subjected to discrimination under this Funding Loan Agreement. For purposes of this Section, Title 24 Code of Federal Regulations Part 107 and Section 570.601(b) defines specific discriminatory actions that are prohibited and corrective action that shall be taken in a situation as defined therein.

- 12.11. <u>Business Tax Registration Certificate</u>. Subject to any exemption available to it, the Fiscal Agent represents that it has obtained or will obtain the Business Tax Registration Certificate(s) required by the City of Los Angeles' Business Tax Ordinance (Article 1, Chapter 2, Section 21.00 and following, of the Los Angeles Municipal Code). For the term covered by this Funding Loan Agreement, the Fiscal Agent shall maintain, or obtain as necessary, all such Business Tax Registration Certificates required of it under said Ordinance and shall not allow any such Business Tax Registration Certificate to be revoked or suspended.
- Section 10.10 of the Los Angeles Administrative Code, Child Support Assignment Orders Ordinance. Pursuant to this Ordinance, the Fiscal Agent certifies that: (a) it will fully comply with all State and federal employment reporting requirements applicable to Child Support Assignment Orders; (b) it will fully comply with all lawfully served Wage and Earnings Assignment Orders and Notices of Assignment in accordance with California Family Code Section 5230 et seq.; and (c) it will maintain such compliance throughout the term of this Funding Loan Agreement. Pursuant to Section 10.10 b of the Los Angeles Administrative Code, failure of the Fiscal Agent to comply with all applicable reporting requirements or to implement lawfully served Wage and Earnings Assignment Orders and Notices of Assignment or the failure of any principal owner(s) of the Fiscal Agent to comply with any Wage and Earnings Assignment Orders and Notices of Assignment applicable to them personally shall constitute a default by the Fiscal Agent under the terms of this Funding Loan Agreement subjecting the Fiscal Agent to the removal hereunder where such failure shall continue for more than 90 days after notice of such failure to the Fiscal Agent by the Governmental Lender. Any subcontract entered into by the Fiscal Agent relating to this

Funding Loan Agreement, to the extent allowed hereunder, shall be subject to the provisions of this paragraph and shall incorporate the provisions of the Child Support Assignment Orders Ordinance. Failure of the Fiscal Agent to obtain compliance of its subcontractors shall constitute a default by the Fiscal Agent under the terms of this Funding Loan Agreement subjecting the Fiscal Agent to the removal hereunder where such failure shall continue for more than 90 days after notice of such failure to the Fiscal Agent by the Governmental Lender.

The Fiscal Agent shall comply with the Child Support Compliance Act of 1998 of the State of California Employment Development Department. The Fiscal Agent hereby affirms that to the best of its knowledge it is fully complying with the earnings assignment orders of all employees, and is providing the names of all new employees to the New Hire Registry maintained by the Employment Development Department as set forth in subdivision (b) of Section 7110 of the Public Contract Code.

- 12.13. Americans with Disabilities Act. The Fiscal Agent shall be in full compliance with all federal and state laws, including those of the Americans with Disabilities Act, 42 U.S.C. 12101 et seq., and its implementing regulations and the American Disabilities Act Amendments Act (ADAAA) Pub. L.110 325 and all subsequent amendments (the "ADA"). Under the ADA, the Fiscal Agent shall provide for reasonable accommodations to allow qualified individuals access to and participation in their programs, services and activities in accordance with the ADA. In addition, the Fiscal Agent shall not discriminate against individuals with disabilities nor against persons due to their relationship or association with a person with a disability. Any subcontract entered into by the Fiscal Agent, relating to this Funding Loan Agreement, to the extent allowed hereunder, shall be subject to the provisions of this Section.
- 12.14. Responsible Banking Ordinance Filing. The Bank shall file with the City Treasurer of the Governmental Lender by July 1 of each year an annual statement of community reinvestment activities as required of a commercial bank under the Responsible Banking Ordinance. The Bank represents that it has, prior to the Closing Date, filed the report due by [July 1, 2019] under the Responsible Banking Ordinance for calendar year [2018].

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement, all as of the date first above written

date mist above written.	
	<b>CITY OF LOS ANGELES</b> , as Governmental Lender
	By Los Angeles Housing and Community Investment Department
	By:Name: Sean L. Spear Title: Assistant General Manager
Approved as to form:	
MICHAEL N. FEUER, City Attorney:	
By: Deputy/Assistant City Attorney	

[Issuer Signature Page to Funding Loan Agreement]

MUFG UNION BANK,	N.A.,	in its	capacity	as
Bank and Funding Lend	er			

By			
Name:			
Title:			

[Bank and Funding Lender Signature Page to Funding Loan Agreement]

By			
Name:			
Title:			

[Fiscal Agent Signature Page to Funding Loan Agreement]

#### **EXHIBIT A**

# CITY OF LOS ANGELES MULTIFAMILY MORTGAGE REVENUE NOTE (SP7 APARTMENTS) SERIES 2018C

FOR VALUE RECEIVED, the CITY OF LOS ANGELES, a charter city and municipal corporation of the State of California (the "Governmental Lender"), acknowledges itself indebted hereby promises to pay to the order of MUFG UNION BANK, N.A. (the "Bank"), or its successors and assigns, the sum of [THIRTY-TWO MILLION SEVEN THOUSAND EIGHT HUNDRED FOURTEEN] DOLLARS (\$[32,007,814]), together with interest on the advanced and unpaid principal amount of this City of Los Angeles Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C (this "Funding Loan Note") at the same interest rate as the applicable interest rate specified in the Borrower Loan Agreement (as hereinafter defined) with respect to the Borrower Loan, until the Governmental Lender's obligation to pay the Outstanding Balance (as hereinafter defined) shall be discharged. The "Outstanding Balance" as of any date of calculation shall mean the principal balance of the Funding Loan that has not been repaid by the Governmental Lender to the Bank as of the date of calculation of the Outstanding Balance. This Funding Loan Note shall be governed by and be payable in accordance with the terms and conditions of the Funding Loan Agreement dated as of [September] 1, 2018 (the "Funding Loan Agreement") among [FISCAL AGENT] (the "Fiscal Agent"), the Governmental Lender and the Bank, in its capacity as lender, pursuant to which the Bank has made the Funding Loan to the Governmental Lender.

This Funding Loan Note is issued to evidence the Funding Loan by the Bank to the Governmental Lender and the obligation of the Governmental Lender to repay the same, but only from amounts received by or on behalf of the Governmental Lender from SP7 Apartments LP, a California limited partnership (the "Borrower"), pursuant to a [Construction and Permanent Loan Agreement (Multifamily Housing Back to Back Loan Program)] dated as of [\_\_\_\_\_\_] 1, 2018, by and among the Governmental Lender, the Bank and the Borrower (the "Borrower Loan Agreement") and the other Borrower Loan Documents (as defined in the Borrower Loan Agreement).

Monthly payments of principal and interest shall be due and payable under this Funding Loan Note in the same amounts and on the same dates as payments of principal and interest are due and payable on the Borrower Note, as provided in the Borrower Loan Agreement. The Outstanding Balance of this Funding Loan Note shall be due and payable in its entirety on [ ].

The Funding Loan and this Funding Loan Note are pass-through obligations relating to the Borrower Loan made by the Governmental Lender from the proceeds of the Funding Loan to the Borrower under the Borrower Loan Agreement. Reference is hereby made to the Borrower Loan Agreement and to the Borrower Note for complete payment and prepayment terms of the Borrower Note.

In the event the Governmental Lender fails to make the timely payment of any monthly payment, and such payment remains unpaid for a period of 10 days subsequent to the established payment date, the Governmental Lender shall pay (solely from amounts received from the Borrower as late charges under the Borrower Loan Agreement) to the Bank a late charge in the amount specified in [Section 3.6] of the Borrower Loan Agreement. If the principal balance of this Funding Loan Note is accelerated following an Event of Default (as defined in the Funding Loan Agreement), the Bank may increase the interest rate on this Funding Loan to the Default Rate (as defined in the Borrower Loan Agreement).

The Governmental Lender may, at any time, prepay the principal amount of this Funding Loan Note to the same extent and subject to the same terms and conditions set forth in the Borrower Loan Agreement for the prepayment of the Borrower Loan (including any provisions for the payment of any prepayment premium, fee or penalty).

All sums due hereunder shall be paid in lawful money of the United States of America. Interest on this Funding Loan Note shall be computed as provided for the Borrower Loan in the Borrower Loan Agreement. All payments made hereunder shall be credited and applied as provided in the Funding Loan Agreement.

THIS FUNDING LOAN NOTE IS PAYABLE, AS TO PRINCIPAL, PREMIUM (IF ANY), AND INTEREST, SOLELY OUT OF THE PLEDGED REVENUES, FUNDS AND ASSETS OF THE GOVERNMENTAL LENDER AND THE OTHER SECURITY WHICH IS THE SOLE ASSET OF THE GOVERNMENTAL LENDER PLEDGED THEREFOR, AND THEN ONLY TO THE EXTENT PROVIDED IN THE FUNDING LOAN AGREEMENT AND NOT FROM ANY OTHER REVENUES, FUNDS OR ASSETS OF THE GOVERNMENTAL LENDER. EXCEPT TO THE LIMITED EXTENT PROVIDED ABOVE, THIS FUNDING LOAN NOTE SHALL NOT BE A DEBT OR AN OBLIGATION, EITHER GENERAL OR SPECIAL, OF THE GOVERNMENTAL LENDER, THE STATE POLITICAL SUBDIVISION THEREOF, AND NONE OF THE GOVERNMENTAL LENDER, THE STATE OR ANY POLITICAL SUBDIVISION THEREOF SHALL BE LIABLE THEREON. NEITHER THE FAITH, REVENUES, CREDIT NOR TAXING POWER OF THE GOVERNMENTAL LENDER, THE STATE OR ANY POLITICAL SUBDIVISION THEREOF SHALL BE PLEDGED TO THE PAYMENT OF THE PRINCIPAL OF, PREMIUM (IF ANY), OR INTEREST ON THIS FUNDING LOAN NOTE. NEITHER THE MEMBERS OF THE CITY COUNCIL OF THE GOVERNMENTAL LENDER NOR ANY PERSONS EXECUTING THIS FUNDING LOAN NOTE SHALL BE LIABLE PERSONALLY ON THIS FUNDING LOAN NOTE BY REASON OF THE ISSUANCE HEREOF.

THIS FUNDING LOAN NOTE HAS BEEN ISSUED PURSUANT TO THE LAW AND IN ACCORDANCE WITH THE ACT.

NO MEMBER, OFFICER, AGENT, EMPLOYEE OR ATTORNEY OF THE GOVERNMENTAL LENDER, INCLUDING ANY INDIVIDUAL EXECUTING THE FUNDING LOAN AGREEMENT OR THIS FUNDING LOAN NOTE, SHALL BE LIABLE PERSONALLY ON THIS FUNDING LOAN NOTE OR FOR ANY REASON RELATING TO THE ISSUANCE OF THIS FUNDING LOAN NOTE. NO RECOURSE

SHALL BE HAD FOR THE PAYMENT OF THE PRINCIPAL OF OR THE INTEREST ON THIS FUNDING LOAN NOTE, OR FOR ANY CLAIM BASED ON THIS FUNDING LOAN NOTE, OR OTHERWISE IN RESPECT OF THIS FUNDING LOAN NOTE, OR BASED ON OR IN RESPECT OF THE FUNDING LOAN AGREEMENT OR ANY SUPPLEMENT THERETO, AGAINST ANY MEMBER, OFFICER, EMPLOYEE, ATTORNEY OR AGENT, AS SUCH, OF THE OBLIGOR OR ANY SUCCESSOR, WHETHER BY VIRTUE OF ANY CONSTITUTION, STATUTE OR RULE OF LAW, OR BY THE ENFORCEMENT OF ANY ASSESSMENT OR PENALTY OR OTHERWISE, ALL SUCH LIABILITY BEING, BY THE ACCEPTANCE OF THIS FUNDING LOAN NOTE AND AS PART OF THE CONSIDERATION FOR THE ISSUE OF THIS FUNDING LOAN NOTE, EXPRESSLY WAIVED AND RELEASED.

THIS FUNDING LOAN NOTE AND THE REPAYMENT PROVISIONS CONTAINED HEREIN ARE SUBJECT TO THE PROVISIONS AND LIMITATIONS CONTAINED IN SECTIONS 4.1, 5.2 AND 6.14 OF THE FUNDING LOAN AGREEMENT.

No delay or omission on the part of the Bank in exercising any remedy, right or option under this Funding Loan Note or the Funding Loan Documents (as defined in the Funding Loan Agreement) shall operate as a waiver of such remedy, right or option. In any event a waiver on any one occasion shall not be construed as a waiver or bar to any such remedy, right or option on a future occasion. The rights, remedies and options of the Bank under this Funding Loan Note and the Funding Loan Documents are and shall be cumulative and are in addition to all the rights, remedies and options of the Bank at law or in equity or under any other agreement.

Presentment for payment, notice of dishonor, protest or notice of protest are hereby waived. The acceptance by the Bank of any amount after the same is due shall not constitute a waiver of the right to require prompt payment, when due, of all other amounts due hereunder. The acceptance by the owner hereof any sum and amount less than the amount then due shall be deemed an acceptance on account only and upon condition of the acceptance shall not constitute a waiver of the obligation of the Governmental Lender to pay the entire sum then due, and the Governmental Lender's failure to pay such amount then due shall be and continue to be at default notwithstanding such acceptance of such amount on account thereof. Consent by the Bank to any action of the Governmental Lender which is subject to approval of the Bank hereunder shall not be deemed a waiver of the right to require such consent or approval to future successive actions, waiver the right to assert the defense of any statute of limitations to any debt or obligation hereunder or consents to renewals and extensions of time for payment of any amounts due under this Funding Loan Note.

THIS FUNDING LOAN NOTE MAY BE TRANSFERRED ONLY IN ACCORDANCE WITH THE REQUIREMENTS OF SECTION 4.3 OF THE FUNDING LOAN AGREEMENT, AND ANY SUCH TRANSFER SHALL BE RECORDED IN THE NOTE REGISTER MAINTAINED BY THE FISCAL AGENT.

Capitalized terms used herein which are not defined herein shall have the meanings ascribed to them in the Funding Loan Agreement.

executed in its name and on its behalf all as	of the day of, 2018.
(SEAL)	CITY OF LOS ANGELES
City Treasurer	By Mayor
FORM OF CERTIFIC	CATE OF AUTHENTICATION
This is the Funding Loan Note Agreement and has been authenticated and	described in the within-mentioned Funding Loan registered on
	[FISCAL AGENT], as Fiscal Agent
	Bv

#### **EXHIBIT B**

## FORM OF REPRESENTATION LETTER

City of Los Angeles Los Angeles, California

Kutak Rock LLP Los Angeles, California

[FISCAL AGENT] Los Angeles, California

> \$[32,007,814] City of Los Angeles Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C

Ladies and Gentlemen:

The undersigned (the "Holder") hereby represents and warrants to you as follows:

- 1. The Holder [proposes to make] [has made] [has acquired] a loan in an aggregate principal amount not to exceed the amount set forth in the above-captioned note (the "Funding Loan Note") issued pursuant to that certain Funding Loan Agreement dated as of [September] 1, 2018 (the "Funding Loan Agreement") by and among the City of Los Angeles, California (the "Governmental Lender"), [FISCAL AGENT], as Fiscal Agent, and MUFG Union Bank, N.A., as lender (the "Bank"). The Holder understands that the Funding Loan Note is not rated by any rating agency and is secured only by the SP7 Apartments and the revenues therefrom, and will be sold to the Holder with the only above-addressed parties relying upon the representations and warranties of the Holder set forth herein. The Holder acknowledges that no offering document has been prepared in connection with the making of the Funding Loan. The Holder has requested and received all materials which the Holder has deemed relevant in connection with its making of the Funding Loan (the "Due Diligence Information"). The Holder has reviewed the documents executed in conjunction with the making of the Funding Loan Note, including, without limitation, the Funding Loan Agreement and the Borrower Loan Agreement.
- 2. The Holder hereby waives the requirement of any "due diligence investigation or inquiry" by the Governmental Lender, by each official of the Governmental Lender, by each employee of the Governmental Lender, by each member of the governing board of the Governmental Lender, and by counsel to the Governmental Lender, the Fiscal Agent, counsel to the Fiscal Agent, the Bank, counsel to the Bank and Tax Counsel in connection with the authorization, execution and delivery of the Funding Loan Note and Holder's making the Funding Loan. The Holder recognizes and agrees that the Governmental Lender, by each official of the Governmental Lender, each employee of the Governmental Lender, each member of the governing

board of the Governmental Lender, counsel to the Governmental Lender, the Fiscal Agent, counsel to the Fiscal Agent, the Bank, counsel to the Bank and Tax Counsel have made no representations or statements (expressed or implied) with respect to the accuracy or completeness of any of the materials reviewed by the Holder in connection with the Holder's making the Funding Loan. In making its decision to make the Funding Loan, the Holder is relying upon its own examination of the Governmental Lender, the Borrower, the Development and the terms of the Funding Loan.

- 3. The Holder has been provided an opportunity to ask questions of, and the Holder has received answers from, representatives of the Governmental Lender and the Borrower regarding the terms and conditions of the Funding Loan Note, and the Holder has obtained all additional information requested by it in connection with the Funding Loan Note.
- 4. The Holder has sufficient knowledge and experience in business and financial matters in general, and making loans such as the Funding Loan in particular, and is capable of evaluating the merits and risks involved in making the Funding Loan. The Holder is able to bear the economic risk of, and an entire loss of funds advanced under, the Funding Loan Note.
- 5. The Holder is [making] [purchasing] the Funding Loan solely for its own account and has no present intention to resell or distribute the Funding Loan Note, provided that the Holder reserves the right to transfer or dispose of the Funding Loan Note, at any time, and from time to time, in its complete and sole discretion, subject, however, to the restrictions described in paragraphs 6 through 8 of this Letter. The Holder hereby agrees that the Funding Loan Note may only be transferred in whole and in accordance with the Funding Loan Agreement, including Article IV thereof, to a single holder, which must execute and deliver to the parties addressed above a form of this Representation Letter.
- 6. The Holder agrees that it will offer, sell, pledge, transfer or exchange the Funding Loan Note (or any legal or beneficial interest therein) only (i) in accordance with an available exemption from the registration requirements of Section 5 of the Securities Act of 1933, as amended (the "Securities Act"), (ii) in accordance with any applicable state securities laws, and (iii) in accordance with the transfer restrictions set forth in the Funding Loan Note and the Funding Loan Agreement. The Holder acknowledges that written consent of the Governmental Lender is required in order to transfer the Funding Loan Note.
- 7. The Holder is: (a) a "qualified institutional buyer" as defined in Rule 144A ("Rule 144A") promulgated under the Securities Act; (b) an institutional "accredited investor" as defined in Sections 501(a)(1) through (3) of Regulation D promulgated under the Securities Act; (c) an entity that is directly or indirectly wholly owned or controlled by the Bank (being a financial institution described in (a) above); (d) an entity all of the investors in which are described in (a), (b) or (c) above; or (e) a custodian or trustee for a party described in (a), (b) or (c) above.
- 8. If the Holder transfers the Funding Loan (or any legal or beneficial interest therein), the Holder or its agent will obtain for the benefit of each of you from any subsequent holder an Investor Letter in the form of this Letter or such other materials as are required by the Funding Loan Note and the Funding Loan Agreement to effect such sale and purchase. The Holder understands and agrees that the Fiscal Agent is not authorized to register any transfer of the

Funding Loan Note prior to receipt of such letter and the written consent of the Governmental Lender.

- 9 Neither the Bank, the Fiscal Agent, Tax Counsel, counsel to the Governmental Lender, the Governmental Lender, its governing body, or any of its employees or agents will have any responsibility to the Holder for the accuracy or completeness of information obtained by the Holder from any source regarding the Development, the Governmental Lender or the Borrower or their financial conditions or regarding the Funding Loan Note, the provisions for payment thereof, or the sufficiency of any security therefor, including, without limitation, any information specifically provided by any of such parties contained in the Due Diligence Information. The Holder acknowledges that, as between Holder and all of such parties: (a) the Holder has assumed responsibility for obtaining such information and making such review as the Holder has deemed necessary or desirable in connection with its decision to make the Funding Loan; and (b) the Due Diligence Information and any additional information specifically requested from the Governmental Lender or the Borrower and provided to the Holder prior to closing constitute all the information and review, with the investigation made by Holder (including specifically the Holder's investigation of the Governmental Lender, the Development and the Borrower) prior to its making the Funding Loan, that Holder has deemed necessary or desirable in connection with its decision to make the Funding Loan.
- 10. The Holder understands that (a) the Funding Loan Note has not been registered with any federal or state securities agency or commission, and (b) no credit rating has been sought or obtained with respect to the Funding Loan Note, and the Holder acknowledges that the Funding Loan Note is speculative with a high degree of risk.
- The Holder acknowledges that the Funding Loan Note is a limited obligation of the Governmental Lender, payable solely from the revenues or other amounts provided by or at the direction of the Borrower, and is not an obligation payable from the general revenues or other funds of the Governmental Lender, the State of California or any political subdivision of the State of California. The Holder acknowledges that the Governmental Lender is issuing the Funding Loan Note on a conduit, nonrecourse basis and has no continuing obligations with respect thereto except as expressly set forth in the Funding Loan Agreement.
- 12. The Holder has the authority to make the Funding Loan and to execute this letter and other documents and instruments required to be executed by the Holder in connection with making the Funding Loan. The individual who is executing this letter on behalf of the undersigned is a duly appointed, qualified and acting officer of the Holder and authorized to cause the Holder to make the certifications, representations and warranties contained herein by the execution of this letter on behalf of the Holder.
- 13. The Holder acknowledges that no offering document has been produced in connection with the Funding Loan.
- 14. The Holder agrees to indemnify and hold harmless the Governmental Lender, the Governmental Lender's officials, officers, directors, employees, agents, attorneys, accountants, advisors, consultants, servants and the members of the governing board of the Governmental Lender past, present and future with respect to any claim asserted against any of them that is based

upon the Holder's sale, transfer or other disposition by it of the Funding Loan in violation of the provisions hereof or of the Funding Loan Agreement or any inaccuracy in any statement made by the Holder in this letter.

Capitalized terms used herein and not otherwise defined have the meanings given such terms in the Funding Loan Agreement.

Very truly yours,
MUFG UNION BANK, N.A., as Note Purchaser
By:
Name:
Title:

[Signature Page to Representation Letter]

#### **EXHIBIT C**

#### FORM OF PROJECT FUND REQUISITION

	Draw #
FISCAL AGENT]	
]	
Los Angeles, CA 90071	
Attention: Global Corporate Trust Services	

Re: City of Los Angeles Multifamily Mortgage Revenue Note (SP7 Apartments), Series 2018C

This requisition is being delivered to you in accordance with the Funding Loan Agreement dated as of [September] 1, 2018 (the "Funding Loan Agreement") among MUFG Union Bank, N.A. (the "Funding Lender"), the City of Los Angeles (the "Governmental Lender") and [FISCAL AGENT], as fiscal agent (the "Fiscal Agent") pursuant to which the above-referenced note (the "Funding Loan Note") was issued. Capitalized terms not defined herein shall have the meanings assigned thereto in the Funding Loan Agreement.

- 1. You are requested to disburse funds in the amount of \$\_\_\_\_\_ from the Project Fund as Draw #\_\_\_\_\_ pursuant to Section 9.6 of the Funding Loan Agreement in the amount(s), to the person(s) and for the purpose(s) set forth on Schedule I attached hereto and incorporated herein by reference.
  - 2. The undersigned certifies that:
  - (i) there has been received no notice (a) of any lien, right to lien or attachment upon, or claim affecting the right of the payee to receive payment of, any of the moneys payable under such requisition to any of the persons, firms or corporations named therein, and (b) that any materials, supplies or equipment covered by such requisition are subject to any lien or security interest, or if any notice of any such lien, attachment, claim or security interest has been received, such lien, attachment, claim or security interest has been released, discharged, insured or bonded over or will be released, discharged, insured or bonded over upon payment of the requisition;
  - (ii) this Requisition contains no items representing payment on account of any percentage entitled to be retained at the date of the certificate;
  - (iii) the obligation stated on this Requisition has been incurred in or about the acquisition, rehabilitation, construction or equipping of the Development, each item is a proper charge against the Project Fund, and the obligation has not been the basis for a prior requisition that has been paid;
  - (iv) this Requisition contains no items representing any costs of issuance of the Funding Loan Note or any other amount constituting an issuance cost under Section 147(g) of the Code and payment of the costs referenced herein will not violate any representation, warranty or covenant of the Borrower in the Borrower Loan Agreement, the Regulatory Agreement or the Tax Certificate;
  - (v) not less than 97% of the sum of (a) the amounts requisitioned by this Requisition to be funded from the Project Fund plus (b) all amounts previously disbursed

from the Project Fund have been or will be applied by the Borrower to pay Qualified Project Costs (as defined in the Regulatory Agreement);

- (vi) the Borrower acknowledges that fees, charges or profits (including, without limitation, developer fees) payable to the Borrower or a "related person" (within the meaning of Section 144(a)(3) of the Code) are not deemed to be Qualified Project Costs; and
- (vii) as of the date hereof, no event or condition has happened or is happening or exists that constitutes, or that with notice or lapse of time or both would constitute, an Event of Default under the Funding Loan Agreement or under the Borrower Loan Agreement.

of Defau	lt under the Funding L	oan Agreement or under the Borrower Loan Agreement.
Dated:	, 20	
		SP7 APARTMENTS LP, a California limited partnership
		By: SP7 Apartments GP LLC, a California limited liability company, its general partner
		By: The Skid Row Housing Trust, a California nonprofit public benefit corporation, its sole member/manager
		By: Ben Rosen, Real Estate Development Director
MUFG UNION	BANK, N.A.	
Title:		
	e Governmental Lender	
	al Lender consent requ of the Funding Loan Ag	
CITY OF LOS	ANGELES	
By:		
Name: Sean L. Title: Assistant		
Date:		

# SCHEDULE I TO PROJECT FUND REQUISITION

## **EXHIBIT D**

### FORM OF RESPONSIBLE BANKING ORDINANCE CERTIFICATE

# RESPONSIBLE BANKING ORDINANCE REPRESENTATIONS AND COVENANTS OF BANK

[
City of Los Angeles Los Angeles, California
\$[32,007,814] City of Los Angeles Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C
Ladies and Gentlemen:
The undersigned, on behalf of [] (the "Bank"), does hereby certify to the following as of the date of this certificate, relating to the Bank's purchase from the City of Los Angeles (the "City") of the City's \$[32,007,814] initial aggregate principal amount of Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C:
(a) The Bank shall file with the City Treasurer of the City by July 1 of each year ar annual statement of community reinvestment activities as required of a commercial bank under the City's Ordinance 182138 adopted May 25, 2012 (the "Responsible Banking Ordinance"); and
(b) The Bank represents that it has, prior to the date hereof, filed with the City Treasurer the report due by July 1, 20[] under the Responsible Banking Ordinance for calendar year 20[].
[Remainder of this page intentionally left blank.]

Very truly yours,	
[BANK]	
By:	
Name:	
Title:	

**HCIDLA** Request for Issuance of Bonds for the SP7 Project

# Attachment C

Borrower Loan Agreement for SP7 on next page.



# CONSTRUCTION AND PERMANENT LOAN AGREEMENT (Multifamily Housing Back to Back Loan Program) (SP7 Apartments)

**THIS AGREEMENT** is made as of the Contract Date by and among Borrower, Governmental Lender and Bank in connection with the following:

- **A.** Borrower has requested that Governmental Lender provide a construction and permanent loan to Borrower to finance the construction of the Improvements on the Real Property.
- **B.** Borrower owns or will own, concurrently with the first loan disbursement under this Agreement, the Real Property.
- **C.** Borrower intends to construct an affordable housing apartment project on the Real Property.
- D. Governmental Lender, Bank and Fiscal Agent have entered into the Funding Loan Agreement whereby Bank has agreed to make the Funding Loan to Governmental Lender for the sole purpose of making funds available to the Governmental Lender to make the Borrower Loan to Borrower pursuant to this Agreement in the manner and on the terms set forth in the Funding Loan Agreement, which terms include, without limitation, the obligation of the Governmental Lender to make loan payments to the Bank for amounts received by Governmental Lender from Borrower pursuant to this Agreement and the Borrower Note in repayment of the amounts loaned to Governmental Lender under the Funding Loan Agreement as evidenced by the Funding Loan Note. Governmental Lender has irrevocably pledged and assigned to Bank, as security for Governmental Lender's obligations to repay amounts due under the Funding Loan Note and its obligations under the Funding Loan Agreement, all right, title and interest to the Borrower Loan Documents (other than the Reserved Rights, as defined in the Funding Loan Agreement), including all rights to payments with respect to the Borrower Note. Upon the execution of the Funding Loan Note, all right, title and interest of Governmental Lender under and in the Borrower Loan (other than the Reserved Rights, as defined in the Funding Loan Agreement) will be assigned by Governmental Lender to Bank pursuant to the Funding Loan Agreement and the Assignment of Deed of Trust.
- E. All of the rights, powers, elections, determinations, remedies, duties and functions of Governmental Lender hereunder (other than the Reserved Rights, as defined in the Funding Loan Agreement) may be exercised and performed on behalf of Governmental Lender by Bank unless and until the assignment to Bank is terminated, modified, assigned, in whole or in part, or otherwise amended in accordance with the provisions of the Funding Loan Agreement.
- **F.** Subject to the execution of the Funding Loan Agreement and the terms and conditions of this Agreement, Governmental Lender is willing to make the Borrower Loan to Borrower.

THEREFORE, the parties hereto agree as follows:

#### 1. DEFINITIONS.

1.1 Acceptable Unit Lease. A lease agreement on a lease form approved by Bank which is entered into by and between Borrower and the lessee of a Unit and the terms (including the amount of rent payments) of which comply with the provisions of all Regulatory Agreements, the HAP Contract, the Subordinate Documents, the HCD Documents and the Density Bonus Restrictions.

1

**1.2** Act. As defined in the Funding Loan Agreement.

- **1.3** Advance. Each disbursement of proceeds of the Borrower Loan made pursuant to this Agreement.
  - **1.4** Aggregate Change Order Limit. \$150,000.
- **1.5** Agreement or Borrower Loan Agreement. This Construction and Permanent Loan Agreement (Multifamily Housing Back to Back Loan Program).
- **1.6 Agreement to Furnish Insurance**. The Agreement to Furnish Insurance dated as of the Contract Date executed by Borrower in favor of Governmental Lender and Bank, as the same may from time to time be amended, modified or supplemented.
  - **1.7 AHAP Contract.** Collectively, the AHAP Contract PBV and the AHAP Contract VASH.
- 1.8 AHAP Contract PBV. The Agreement to Enter Into Housing Assistance Payments Contract between Borrower and Contract Administrator effective as of [\_\_\_\_\_\_\_\_, 2018] [CHECK], with respect to fifty-nine (59) units at the Improvements, for a term of fifteen (15) years, and such other terms as are acceptable to the Bank.
- 1.9 AHAP Contract VASH. The Agreement to Enter Into Housing Assistance Payments Contract between Borrower and Contract Administrator effective as of [\_\_\_\_\_\_\_\_\_, 2018] [CHECK], with respect to twenty-two (22) units at the Improvements, for a term of fifteen (15) years, and such other terms as are acceptable to the Bank.
  - **1.10** AHP Lender. MUFG Union Bank, N.A.
- **1.11 AHP Loan.** That certain affordable housing program loan in an amount of not more than \$1,100,000 from AHP Lender.
- **1.12 AHP Documents.** The AHP Note, the AHP Deed of Trust, the AHP Subordination Agreement and all other documents and instruments evidencing, securing or pertaining to the AHP Loan.
- **1.13 AHP Note.** The promissory note, in form and content acceptable to Bank, made by Borrower to the order of the AHP Lender, evidencing all amounts to be disbursed under the AHP Loan.
- 1.14 AHP Subordination Agreement. A subordination agreement, in form and substance satisfactory to Bank, executed by AHP Lender and Borrower, pursuant to which the AHP Lender shall unconditionally subordinate the lien and effect of the AHP Deed of Trust to the lien and effect of the Deed of Trust.
- **1.15** Allocation Committee. The California Tax Credit Allocation Committee and any successor governmental agency appointed to carry out the obligations of the Allocation Committee.
  - **1.16** Amortization Date. The first day of the calendar month after the Outside Conversion Date.
- **1.17 Appraisal**. An appraisal or reappraisal of the Property (complying with Bank's appraisal policy) performed or to be performed by a certified real estate appraiser engaged by Bank.
- **1.18 Appraised Value**. The market value of the Property as determined by Bank in its business judgment, reasonably exercised, based upon an Appraisal.
- **1.19** Architect. [KFA Architecture] [CHECK], or such other architect as may be approved by Bank.

- **1.20** Architect's Agreement. The agreement between Borrower and Architect relating to the design and construction of the Improvements.
- **1.21** Assignment of AHAP Contract. Collectively, the Assignment of AHAP Contract PBV and the Assignment of AHAP Contract VASH.
- **1.22** Assignment of AHAP Contract PBV. The Assignment of Agreement to Enter Into Housing Assistance Payments Contract, on Bank's form, executed by Borrower in favor of Bank as additional collateral security for the performance of Borrower's obligations under the Loan Documents, assigning to Bank all of Borrower's rights under the AHAP Contract PBV together with that certain Consent to Assignment of AHAP Contract PBV attached thereto and executed by Contract Administrator.
- **1.23** Assignment of AHAP Contract VASH. The Assignment of Agreement to Enter Into Housing Assistance Payments Contract, on Bank's form, executed by Borrower in favor of Bank as additional collateral security for the performance of Borrower's obligations under the Loan Documents, assigning to Bank all of Borrower's rights under the AHAP Contract VASH together with that certain Consent to Assignment of AHAP Contract VASH attached thereto and executed by Contract Administrator.
- **1.24** Assignment of Construction Contract. The Assignment of Construction Contract dated as of the Contract Date executed by Borrower in favor of Governmental Lender and Bank, as the same may from time to time be amended, modified or supplemented.
- **1.25** Assignment of HAP Contract. Collectively, the Assignment of HAP Contract PBV and the Assignment of HAP Contract VASH.
- **1.26 Assignment of HAP Contract PBV**. The Assignment of Housing Assistance Payments Contract on Bank's form, to be executed by Borrower upon execution of the HAP Contract in favor of Bank as additional collateral security for the performance of Borrower's obligations under the Loan Documents, assigning to Bank all of Borrower's rights under the HAP Contract PBV together with that certain Consent to Assignment of HAP Contract attached thereto to and executed by Contract Administrator.
- **1.27** Assignment of HAP Contract VASH. The Assignment of Housing Assistance Payments Contract on Bank's form, to be executed by Borrower upon execution of the HAP Contract in favor of Bank as additional collateral security for the performance of Borrower's obligations under the Loan Documents, assigning to Bank all of Borrower's rights under the HAP Contract VASH together with that certain Consent to Assignment of HAP Contract attached thereto to and executed by Contract Administrator.
  - **1.28** Assignment of Hedge. As defined in Section 7.48.
- **1.29** Assignment of Partnership Interest (GP). An Assignment of Partnership Interest dated as of the Contract Date executed by General Partner in favor of Governmental Lender and Bank as additional collateral security for the performance of the Borrower's obligations under the Borrower Loan Documents, assigning to Governmental Lender and Bank all of such General Partner's rights as a general partner in Borrower.
- **1.30** Assignment of Plans and Specifications. The Assignment of Architect's Agreement, Plans and Specifications dated as of the Contract Date executed by Borrower, in favor of Governmental Lender and Bank, as the same may from time to time be amended, modified or supplemented.
- 1.31 Assignment of Tax Credits and Partnership Interests. An Assignment of Rights to Tax Credits and Partnership Interests dated as of the Contract Date executed by Borrower in favor of Governmental Lender and Bank as additional collateral security for the performance of Borrower's obligations under the Borrower Loan Documents, assigning to Governmental Lender and Bank all of

Borrower's rights under the Tax Credit Allocation Documents including, without limitation, the right to receive the Tax Credits set forth under the Tax Credit Allocation Documents and any interest Borrower may have in any partnership interest of Tax Credit Investor in the Borrower.

- **1.32** Bank. MUFG Union Bank, N.A. (i) acting in its capacity as owner of the Funding Loan Note and as assignee of and agent under this Agreement for the Governmental Lender pursuant to the Funding Loan Agreement, and (ii) its successors and assigns.
- **1.33 Bonded Work**. Offsite, common area, or other improvements required by a Governmental Authority or for which bonds may be required in connection with the development of the Real Property.
  - **1.34 Borrower**. SP7 Apartments LP, a California limited partnership.
- **1.35 Borrower's Equity**. As of any date of determination, Borrower's funds expended on Project costs in accordance with this Agreement as of such date, including Borrower's Funds and capital contributions made by the Tax Credit Investor, but excluding proceeds of the Borrower Loan, as determined by Bank in its sole discretion.
- **1.36 Borrower's Funds**. All funds of Borrower deposited into Borrower's Funds Account pursuant to the terms of this Agreement, to be disbursed in payment of Construction Costs as more particularly set forth in this Agreement.
- **1.37 Borrower's Funds Account**. An account with Bank into which Borrower's Funds shall be deposited as provided for in Section 7.2 or any other provision of this Agreement.
- **1.38 Borrower Loan**. The loan in the maximum principal amount of [\$32,007,000] **[CHECK]** made by the Governmental Lender to Borrower pursuant to this Agreement.
- **1.39 Borrower Loan Documents**. This Agreement, the Borrower Note, the Tax-Exempt Regulatory Agreement, the Deed of Trust, the Guaranty, the ECA, the Security Documents, the Financing Statements, the Agreement to Furnish Insurance, any Hedge Documents, the Indemnity Agreement and all other agreements, instruments and documents (together with amendments, supplements and replacements thereto) now or hereafter executed and delivered to Governmental Lender or Bank in connection with the Borrower Loan.
- **1.40 Borrower Note**. The Promissory Note (Multifamily Housing Back to Back Loan Program) dated as of the Contract Date from Borrower, as maker, in favor of Governmental Lender in the original principal amount of the Borrower Loan.
- **1.41 Business Day**. (i) Except as otherwise provided in clause (ii) below, a day which is not a Saturday or Sunday on which banks in the State of California are open for business for the funding of corporate loans, or (ii) for use only in connection with the definition of LIBOR Rate, a day which is both a New York Banking Day and a London Banking Day.
- **1.42 Capital Improvement Reserve Account**. A non-interest bearing account established with Bank by Borrower at the time of Conversion for the purpose of funding any capital improvements which are necessary for the continued operation of the Property.
- **1.43 CDC**. Community Development Commission of the CDC of Los Angeles, a public body corporate and politic.
- **1.44 CDC Deed of Trust**. The deed of trust executed by Borrower for the benefit of CDC, encumbering the Project and securing repayment of amounts owing under the CDC Note, the lien of which is to be subject and subordinate to the lien of the Deed of Trust.

- **1.45 CDC Documents**. The CDC Restrictions, the CDC Loan Agreement, the CDC Note, the CDC Deed of Trust, the CDC Subordination Agreement and all other documents and instruments evidencing, securing or pertaining to the CDC Loan.
- **1.46 CDC Loan**. The [\$1,500,000] **[CHECK]** loan to be made by CDC to Borrower pursuant to the terms of the CDC Loan Agreement.
- **1.47 CDC Loan Agreement**. The Loan Agreement entered into by and between Borrower and CDC pursuant to the terms of which CDC shall make to Borrower the CDC Loan.
- **1.48 CDC Note**. The promissory note made by Borrower to the order of the CDC, evidencing all amounts disbursed and to be disbursed under the CDC Loan.
- **1.49 CDC Restrictions**. The regulatory agreement executed by Borrower for the benefit of CDC in connection with the CDC's making the CDC Loan.
- **1.50 CDC Subordination Agreement**. A Subordination Agreement in form and substance satisfactory to Bank, executed by CDC and Borrower, pursuant to which CDC shall unconditionally subordinate the lien and effect of the CDC Restrictions and the CDC Deed of Trust to the lien and effect of the Deed of Trust.
- **1.51 Certification of Plans and Specifications**. The Certification of Plans and Specifications dated as of the Contract Date from Borrower, Contractor and Architect to Governmental Lender and Bank, as the same may from time to time be amended, modified or supplemented.
- **1.52 Change Order**. Any change or supplement to the Plans, Construction Contract or subcontract as permitted by this Agreement.
- **1.53** Closing Date. The date on which the Deed of Trust is recorded and the Initial Advance is made.
- **1.54 Code**. The Internal Revenue Code of 1986, as amended; including (a) any successor internal revenue law and (b) the applicable regulations promulgated thereunder whether final, temporary or proposed under the Code or such successor law.
- **1.55** Completion Date. The date of Project Completion, which date shall not be later than [November 1, 2020] [CHECK].
- **1.56 Conditions to Conversion**. The conditions precedent to Conversion as listed in Section 3.2.2 below.
- **1.57 Construction Contract**. The agreement between Borrower and Contractor relating to the construction of the Improvements.
- **1.58 Construction Costs**. All costs approved by Bank relating to the construction of the Improvements or otherwise pertaining to the Property, as set forth in the Detailed Cost Breakdown.
- **1.59 Construction Phase**. The period from the Closing Date through and including the date immediately preceding the Conversion Date.
  - **1.60 Contract Administrator**. Housing Authority of the City of Los Angeles.
  - 1.61 Contract Date. [September 1, 2018] [CHECK].

- **1.62 Contractor**. [\_\_\_\_\_\_] **[CHECK]**, or such other contractor as may be approved by Bank, or Borrower acting in the capacity of general contractor.
- **1.63 Conversion**. The conversion of the Borrower Loan from the Construction Phase to the Permanent Phase.
- **1.64 Conversion Date**. The date on which all Conditions to Conversion have been satisfied, as such date is established by Bank in the Conversion Notice. The Conversion Date shall be the first day of the calendar month following the month in which Bank issues the Conversion Notice, but in no event later than the Outside Conversion Date.
- **1.65 Conversion Election Notice**. Written notice delivered by Borrower to Bank that Borrower has elected to convert the Borrower Loan from the Construction Phase to the Permanent Phase.
- **1.66 Conversion Notice**. Written notice delivered by Bank to Borrower that the Conditions to Conversion have been fully satisfied.
- **1.67 Debt Coverage Ratio**. The ratio of (i) the annual stabilized Net Operating Income for the Property during a particular period of time, to (ii) the assumed combined interest and principal payment for the Permanent Phase that would be required based upon the projected outstanding principal balance of the Borrower Note, as of the Conversion Date, a fixed interest rate on the Borrower Note equal to the fixed rate of the Hedge (inclusive of the Margin) and monthly amortization payments on the Borrower Note based upon a one hundred eighty (180) month amortization period and any secondary financing permitted pursuant to the Borrower Loan Documents.
- **1.68 Deed of Trust**. The Deed of Trust, Assignment of Rents, Security Agreement and Fixture Filing (Construction Trust Deed) (Multifamily Housing Back to Back Loan Program) dated as of the Contract Date from Borrower, as trustor, for the benefit of Governmental Lender and Bank, as beneficiary, as the same may from time to time be amended, modified or supplemented.
- **1.69 Deed of Trust Assignment**. The Assignment of Deed of Trust and Related Documents dated as of the Contract Date by Governmental Lender in favor of Bank.
  - **1.70 Default Rate**. A rate equal to 5% more than the Variable Rate.
- **1.71 Density Bonus Restrictions.** That certain Density Bonus Housing Agreement between Borrower and HCID recorded against the Property concurrently with the Deed of Trust in a non-subordinated senior lien priority, in the form approved by Bank.
- **1.72 Detailed Cost Breakdown**. An itemized schedule on a component, unit and trade breakdown basis showing all costs and expenses required for construction of the Improvements in accordance with the Plans, which has been submitted to and approved by Bank.
  - **1.73 Developer**. The Skid Row Housing Trust, a California nonprofit public benefit corporation.
- **1.74 Disbursement Schedule**. The schedule or schedules for disbursement of the Advances and of Borrower's Funds, if any, set forth on <u>Exhibit B</u>, which may be amended from time to time by reallocations made in accordance with Section 5.5.
- 1.75 Draw Request. The certified invoice to be delivered by Borrower to Bank as a condition to Governmental Lender making an Advance, in such form and certified by such parties as required by Bank, together with such schedules, affidavits, releases, waivers, statements, invoices, bills, and other documents, certificates and information as may be required by Bank.

- **1.76 ECA**. The Environmental Compliance Agreement, dated as of the Contract Date by Borrower in favor of Governmental Lender and Bank, as the same may from time to time be amended, modified or supplemented.
  - 1.77 Event of Default. As defined in Section 8.
- **1.78** Extended Use Agreement. An "extended low-income housing commitment" as defined in Section 42(h)(6)(B) of the Code.
- **1.79 Financial Statements**. Balance sheets, income statements, statements of retained earnings with supporting schedules and such other financial reports as Bank may require, in form and content acceptable to Bank.
- **1.80 Financing Statements**. All UCC financing statements required in connection with the Borrower Loan.
  - 1.81 First Payment Date. [October 1, 2018] [CHECK].
- **1.82 Fiscal Agent**. The Fiscal Agent from time to time under the Funding Loan Agreement. Initially the Fiscal Agent shall be [\_\_\_\_\_\_] [CHECK].
  - **1.83 Funding Date**. The date on which the Initial Disbursement is made.
- **1.84 Funding Loan**. The loan in the maximum amount of [\$32,007,000] **[CHECK]** made by Bank to Governmental Lender pursuant to the Funding Loan Agreement.
- **1.85** Funding Loan Agreement. The Funding Loan Agreement dated as of the Contract Date among the Governmental Lender, the Bank and Fiscal Agent in connection with the issuance of the Funding Loan Note.
  - **1.86** Funding Loan Documents. As defined in the Funding Loan Agreement.
  - **1.87** Funding Loan Note. As defined in the Funding Loan Agreement.
  - 1.88 General Partner. SP7 Apartments GP LLC, a California limited liability company.
- **1.89 General Partner Equity Account.** A deposit account established with Bank in General Partner's name into which General Partner shall have deposited [\$4,182,935] **[CHECK]** pursuant to Section 5.7 below.
- **1.90 Governmental Authority**. Any federal, state or local governmental or quasi-governmental agency, authority, board, bureau, commission, department, instrumentality or public body, court, administrative tribunal or public or private utility having authority over the Property or its utilization.
- **1.91** Governmental Lender. City of Los Angeles, a charter city and a municipal corporation of the State of California.
- **1.92 Governmental Requirement**. Any law, statute, order, ordinance, rule, regulation, permit or act of a Governmental Authority.
- **1.93 Gross Operating Income**. The sum of any and all payments, fees, rentals, additional rentals (but specifically excluding any amounts received from tenant-based vouchers or other rent subsidies in excess of then maximum rents permitted under the Regulatory Agreements), expense reimbursements (including, without limitation, all reimbursements by tenants, subtenants, licensees and other users of the Property), income, interest, and other monies received directly or indirectly by or on behalf of Borrower from

any Person with respect to Borrower's ownership, use, development or operation of the Property, including, without limitation, any leasing or licensing of the Property. Gross Operating Income shall be computed on a cash basis and shall include for each monthly statement all amounts actually received in such month whether or not such amounts are attributable to a charge arising in such month.

- **1.94** Guarantor. Any Person who executes a Guaranty in connection with the Loan.
- **1.95** Guaranty. Bank's standard form Loan and Completion Guaranty, Loan Guaranty, Completion Guaranty or Interest and Maintenance Guaranty, as the case may be entered into in connection with the Borrower Loan.
  - **1.96 HAP Contract.** Collectively, the HAP Contract PBV and the HAP Contract VASH.
- **1.97 HAP Contract PBV.** The Housing Assistance Payments Contract to be entered into between Borrower and Contract Administrator with respect to fifty-nine (59) units at the Improvements, in the form attached to the AHAP Contract PBV and consistent with the terms of the final proposal attached to the AHAP Contract PBV.
- **1.98 HAP Contract VASH**. The Housing Assistance Payments Contract to be entered into between Borrower and Contract Administrator with respect to twenty-two (22) units at the Improvements, in the form attached to the AHAP Contract VASH and consistent with the terms of the final proposal attached to the AHAP Contract VASH.
- **1.99 HCID**. City of Los Angeles, a charter city and a municipal corporation of the State of California, acting by and through the Los Angeles Housing and Community Investment Department.
- **1.100 HCID Deed of Trust**. Collectively, the HCID HHH Deed of Trust, the HCID HOPWA Deed of Trust and the HCID Recast Deed of Trust.
- **1.101 HCID Documents.** Collectively, the HCID HHH Documents, the HCID HOPWA Documents and the HCID Recast Documents.
- **1.102 HCID HHH Deed of Trust**. The deed of trust to be executed by Borrower for the benefit of HCID, encumbering the Property and securing repayment of amounts owing under the HCID HHH Note, the lien of which shall be subject and subordinate to the lien of the Deed of Trust.
- **1.103 HCID HHH Documents**. The HCID HHH Loan Agreement, the HCID HHH Note, the HCID HHH Deed of Trust, the HCID HHH Restrictions, the HCID Subordination Agreement and all other documents and instruments evidencing, securing or pertaining to the HCID HHH Loan.
- **1.104 HCID HHH Loan**. The [\$12,000,000] **[CHECK]** loan to be made by HCID to Borrower pursuant to the terms of the HCID HHH Documents.
- **1.105 HCID HHH Loan Agreement**. The loan agreement to be entered into by and between Borrower and HCID pursuant to the terms of which HCID shall make to Borrower the HCID HHH Loan.
- **1.106 HCID HHH Note**. The promissory note to be made by Borrower to the order of HCID, evidencing all amounts disbursed and to be disbursed under the HCID HHH Loan.
- **1.107 HCID HHH Restrictions**. The regulatory agreement to be executed by Borrower for the benefit of HCID in connection with HCID's making the HCID HHH Loan.
- **1.108 HCID HOPWA Deed of Trust**. The deed of trust executed by Borrower for the benefit of HCID, encumbering the Property and securing repayment of amounts owing under the HCID HOPWA Note, the lien of which shall be subject and subordinate to the lien of the Deed of Trust.

- **1.109 HCID HOPWA Documents**. The HCID HOPWA Loan Agreement, the HCID HOPWA Note, the HCID HOPWA Deed of Trust, the HCID HOPWA Restrictions, the HCID Subordination Agreement and all other documents and instruments evidencing, securing or pertaining to the HCID HOPWA Loan.
- **1.110 HCID HOPWA Loan**. The [\$3,000,000] **[CHECK]** loan made by HCID to Borrower pursuant to the terms of the HCID HOPWA Documents.
- **1.111 HCID HOPWA Loan Agreement**. The loan agreement entered into by and between Borrower and HCID pursuant to the terms of which HCID shall make to Borrower the HCID HOPWA Loan.
- **1.112 HCID HOPWA Note**. The promissory note to be made by Borrower to the order of HCID, evidencing all amounts disbursed and to be disbursed under the HCID HOPWA Loan.
- **1.113 HCID HOPWA Restrictions.** The regulatory agreement to be executed by Borrower for the benefit of HCID in connection with HCID's making the HCID HOPWA Loan.
- **1.114 HCID Loan**. Collectively, the HCID HHH Loan, the HCID HOPWA Loan and the HCID Recast Loan.
- **1.115 HCID Recast Deed of Trust**. The deed of trust to be executed by Borrower for the benefit of HCID, encumbering the Property and securing repayment of amounts owing under the HCID Recast Note, the lien of which shall be subject and subordinate to the lien of the Deed of Trust.
- **1.116 HCID Recast Documents**. The HCID Recast Loan Agreement, the HCID Recast Note, the HCID Recast Deed of Trust, the HCID Recast Restrictions, the HCID Subordination Agreement and all other documents and instruments evidencing, securing or pertaining to the HCID Recast Loan.
- **1.117 HCID Recast Loan**. The [\$950,000] **[CHECK]** loan made by HCID to Borrower pursuant to the terms of the HCID Recast Documents.
- **1.118 HCID Recast Loan Agreement**. The loan agreement entered into by and between Borrower and HCID pursuant to the terms of which HCID shall make to Borrower the HCID Recast Loan.
- **1.119 HCID Recast Note**. The promissory note made by Borrower to the order of HCID, evidencing all amounts disbursed and to be disbursed under the HCID Recast Loan.
- **1.120 HCID Recast Restrictions.** The regulatory agreement executed by Borrower for the benefit of HCID in connection with HCID's making the HCID Recast Loan.
- **1.121 HCID Restrictions.** Collectively, the HCID HHH Restrictions, the HCID HOPWA Restrictions and the HCID Recast Restrictions.
- **1.122 HCID Subordination Agreement**. A subordination agreement in form and substance satisfactory to Bank, executed by HCID and Bank, and joined by Borrower, pursuant to which HCID shall unconditionally subordinate the lien and effect of the HCID Deed of Trust and HCID Restrictions to the lien and effect of the Deed of Trust.
- **1.123 HCD**. The Department of Housing and Community Development, a public agency of the State of California.
- **1.124 HCD Deed of Trust**. The deed of trust to be executed by Borrower for the benefit of HCD, encumbering the Property and securing repayment of amounts owing under the HCD Note, the lien of which shall be subject and subordinate to the lien of the Deed of Trust.

- **1.125 HCD Documents**. The Standard Agreement, the HCD Note, the HCD Deed of Trust, the HCD Restrictions, the HCD Subordination Agreement and all other documents and instruments evidencing, securing or pertaining to the HCD Loan.
- **1.126 HCD Loan**. The [\$4,657,670] **[CHECK]** loan to be made by HCD to Borrower pursuant to the terms of the HCD Documents.
- **1.127 HCD Loan Agreement**. The loan agreement to be entered into by and between Borrower and HCD pursuant to the terms of which HCD shall make to Borrower the HCD Loan.
- **1.128 HCD Note**. The promissory note to be made by Borrower to the order of HCD, evidencing all amounts disbursed and to be disbursed under the HCD Loan.
- **1.129 HCD Restrictions**. The regulatory agreement to be executed by Borrower for the benefit of HCD in connection with HCD's making the HCD Loan.
- 1.130 HCD Subordination Agreement. A subordination agreement in form and substance satisfactory to Bank, to be executed by HCD and Bank, and joined by Borrower, pursuant to which HCD shall unconditionally subordinate the lien and effect of the HCD Deed of Trust and HCD Restrictions to the lien and effect of the Deed of Trust.
  - **1.131 Hedge**. As defined in Section 7.48.
  - **1.132** Hedge Documents. As defined in Section 7.48.
- **1.133 Improvements**. A one hundred (100) unit multifamily special needs apartment project comprised of ninety-nine (99) low-income units and one (1) manager's unit and related appurtenances.
- **1.134 Indemnified Parties**. Collectively Governmental Lender, Fiscal Agent and Bank and each of their respective officers, members governing members or partners, directors, employees, attorneys and agents, past, present and future.
- 1.135 Indemnity Agreement. Any Indemnity Agreement entered into in connection with the Borrower Loan.
- **1.136 Initial Disbursement**. The initial Advance made by Governmental Lender to Borrower pursuant to this Agreement.
- **1.137** Interest Change Date. The First Payment Date and the first day of each calendar month thereafter.
- **1.138** Interest Period. The period of time from one Interest Change Date to (but excluding) the next Interest Change Date or the Maturity Date, as the case may be.
- **1.139 Interest Reserve**. The portion of the Project Budget allocated for the payment of interest due under this Agreement.
- **1.140** Late Charge. An amount equal to 6% of any delinquent payment of amounts due from Borrower under the Borrower Loan Documents.
- **1.141 Leases**. All leases of any portion of the Property and all amendments, guaranties and subleases relating thereto.
- **1.142 LIBOR Rate**. As of any given date, a per annum rate of interest equal to the greater of (a) 0.00% and (b) the rate for U.S. Dollar deposits for a period of one month or, for the Stub Period, for a period

equal to the number of days in the Stub Period which appears on the Reuters Screen LIBOR01 Page as of 11:00 a.m., London time, on the day that is two Business Days preceding such date. Should the LIBOR Rate cease to be available for any reason, then said rate shall be replaced by a rate which, in the sole discretion of Bank, most closely approximates the unavailable LIBOR Rate.

- **1.143 Liquid Assets**. Immediately available cash, bank deposits, accounts and mutual funds; obligations of or guaranteed by the U.S. government or an agency thereof; and stocks, bonds and other debt instruments regularly traded on the New York, American or NASDAQ stock exchange which can be readily converted into cash.
  - 1.144 Loan Fee. [\$224,049] [CHECK].
- **1.145** Loan Party. Any general partner, managing member, joint venturer, trustee or trustor of Borrower, as applicable and any Guarantor.
- **1.146** Loan-to-Value Ratio. The ratio of (i) then outstanding indebtedness in connection with the Borrower Loan to (ii) the Appraised Value of the Property.
- **1.147** London Banking Day. A day in which dealings in U.S. Dollar deposits in London, England may be carried on by Bank.
  - **1.148** Margin. 1.70% during the Construction Phase and 2.25% during the Permanent Phase.
  - 1.149 Maturity Date. [November 1, 2036] [CHECK].
  - **1.150 Maximum Lawful Rate**. As defined in the Funding Loan Agreement.
  - **1.151 Net Operating Income**. Gross Operating Income less Operating Expenses.
- **1.152 New York Banking Day**. A day which is not a Saturday or Sunday on which banks in New York City, New York are open for business for the funding of corporate loans.
- **1.153 Offsite Materials**. Materials to be incorporated into the Improvements or used in connection with the construction of the Improvements that are stored at a location other than the Real Property.
- **1.154 Onsite Materials**. Materials to be incorporated into the Improvements or used in connection with the construction of the Improvements that are stored on the Real Property.
- 1.155 Operating Deficits. The amount by which Gross Operating Income (including amounts recieved from tenant-based vouchers or other rent subsidies) for a particular period of time is exceeded by actual Operating Expenses plus required debt service payments on the Borrower Loan or any other approved secondary financing during the same period of time.
- 1.156 Operating Expenses. The following expenses to the extent that such expenses are reasonable in amount and customary for properties of a type similar to the Property, as determined by Bank in its sole discretion: (A) real property taxes and assessments imposed upon the Property, (B) premiums for insurance of the Property, including casualty and liability insurance, (C) reserves for capital expenditures, leasing commissions and tenant improvements, as determined by Bank in its business judgment, reasonably exercised, and (D) operating expenses actually incurred by Borrower in connection with the management, operation, cleaning, leasing, maintenance and repair of the Property or any part thereof. Operating Expenses shall be calculated on an accrual basis and shall not include any interest or principal payments due in respect of the Borrower Loan or any allowance for depreciation and similar noncash charges.

- **1.157 Operating Reserve Account.** An interest bearing account established with Bank by Borrower at the time of Conversion for the purpose of funding any Operating Deficits which are necessary for the continued operation of the Property.
- **1.158 Operating Statement**. A monthly, quarterly or annual statement that shows in detail the amounts and sources of Gross Operating Income, the amounts and nature of Operating Expenses, and Net Operating Income, in each case for the preceding calendar month, quarter or year. The Operating Statement shall be prepared in accordance with accounting practices and principles acceptable to Bank and consistently applied and in a form satisfactory to Bank.
- **1.159** Outside Conversion Date. [November 1, 2021] [CHECK], unless extended pursuant to Section 2.5 below.
- **1.160** Partnership Agreement. Borrower's agreement of limited partnership, as the same may be amended from time to time.
- 1.161 Paydown Amount. The amount by which (a) the current outstanding principal amount of the Note, plus all accrued but unpaid interest thereon, exceeds (b) the lesser of (i) the Projected Permanent Phase Loan Amount, (ii) the maximum outstanding principal balance of the Borrower Loan in order for the Property to satisfy the Debt Coverage Ratio pursuant to Section 3.2.2(o) as of the Conversion Date, which Paydown Amount shall be applied towards the payment of principal balance outstanding and accrued and unpaid interest under the Borrower Loan and all other amounts due and owing under the Borrower Loan Documents and the Funding Loan Documents, and (iii) the notional amount of the Hedge as of the Conversion Date.
  - 1.162 Permanent Phase. The period from the Conversion Date and ending on the Maturity Date.
- **1.163 Permitted Liens**. Any easements and restrictions listed in a schedule of exceptions to coverage in the Title Policy as required by the Borrower Loan Documents.
- **1.164 Person**. Any natural person or entity, including any corporation, partnership, joint venture, limited liability company, trust, trustee, unincorporated organization or Governmental Authority.
- **1.165 Personal Property**. Any tangible or intangible personal property described in the Deed of Trust or Security Documents that is security for the Borrower Loan.
- **1.166 Plans**. The final plans and specifications for construction of the Improvements (including any applicable general conditions), prepared by Architect and approved by Bank as required herein, and all amendments and modifications thereof made pursuant to Change Orders.
- **1.167 Preliminary Reservation**. That certain Tax-Exempt Reservation Letter dated March 21, 2018, issued by the Allocation Committee.
- **1.168 Project Budget**. The cost itemization (set forth in Exhibit B-1 hereto) of the total amount needed by Borrower to construct the Improvements and to perform Borrower's other obligations under the Borrower Loan Documents, which itemization may be amended from time to time in accordance with this Agreement.
- **1.169 Project Completion**. The date of completion of construction of the Project and issuance of all licenses and permits necessary for the occupancy and use of the Units such that the Project shall be considered "placed in service" for purposes of the provisions of Section 42 of the Code, which date of completion shall not be later than the Completion Date.

**1.170 Project Fund**. As defined in the Funding Loan Agreement.

- 1.171 Projected Permanent Phase Loan Amount. [\$3,216,500] [CHECK].
- **1.172** Property or Project. The Real Property, the Improvements and the Personal Property.
- **1.173 Qualified Allocation Plan.** The Qualified Allocation Plan adopted by the Allocation Committee from time to time in accordance with the provisions of Section 42(m) of the Code.
- **1.174 Qualified Occupancy**. The first date on which one hundred percent (100%) of the Units shall have been leased and physically occupied at least once by tenants qualifying the Units (except for the manager's unit) as low-income units within the meaning of Section 42(i)(3) of the Code.
- **1.175 R&T Code**. The California Revenue and Taxation Code, as amended from time to time thereto. Any reference to a particular provision of the R&T Code shall include any amendment of such provision.
  - **1.176** Real Property. That certain real property described in Exhibit A hereto.
- **1.177 Recorded Documents**. The Regulatory Agreements, the Deed of Trust, the Deed of Trust Assignment, the Subordination Agreements and the Subordinate Deeds of Trust.
- 1.178 Regulatory Agreements. All regulatory agreements and restrictions (including, without limitation, the Tax-Exempt Regulatory Agreement, the Extended Use Agreement, the Density Bonus Restrictions and the Subordinate Restrictions) now or hereafter encumbering the Property setting forth restrictions with respect to the leasing, maintenance and use of the Units.
- **1.179** Rent Restrictions. The occupancy and rent restrictions contained in the Regulatory Agreements.
- **1.180** Rental Assistance Operating Deficits. Any Operating Deficit which occurs as a result of any loss or decrease (whether by loss or reduction of appropriations, or otherwise) in the amount of the rental assistance provided by the Contract Administrator pursuant to the HAP Contract.
- 1.181 Security Documents. Any agreements granting a security interest in collateral securing the Borrower Loan and/or any Hedge provided by Bank other than the Deed of Trust, including without limitation, assignments and consents to assignments of the Architect's Agreement, Construction Contract, if any, Plans, any property management agreement or asset management agreement, the Assignment of Tax Credits and Partnership Interests, the Assignment of Partnership Interest (GP), the Assignment of Hedge (if any), the Assignment of AHAP Contract PBV, the Assignment of AHAP Contract VASH and the Assignment of HAP Contract VASH.
- **1.182 Set Aside Letter**. Any letter or letters to any Governmental Authority or Surety whereby Bank agrees to allocate proceeds of the Borrower Loan for construction of Bonded Work.
  - 1.183 Single Change Order Limit. \$50,000.
- **1.184 SNHP Deed of Trust.** A deed of trust to be executed by Borrower for the benefit of SNHP Lender encumbering the Project and securing repayment of amounts owing under the SNHP Note, the lien of which is to be subject and subordinate to the lien of the Deed of Trust as the same may from time to time be amended, modified or supplemented.
- **1.185 SNHP Documents.** The SNHP Note, the SNHP Deed of Trust, SNHP Restrictions, the SNHP Subordination Agreement and all other documents and instruments evidencing, securing or pertaining to the SNHP Loan.

- **1.186 SNHP Lender**. California Housing Finance Agency, a public instrumentality and political subdivision of the State of California.
- **1.187 SNHP Loan**. The [\$2,000,000] **[CHECK]** loan to be made by SNHP Lender to Borrower pursuant to the terms of the SNHP Note to cover, among other things, certain development costs for the Project.
- **1.188 SNHP Loan Agreement.** The loan agreement to be entered into by Borrower and SNHP Lender pursuant to the terms of which SNHP Lender shall make to Borrower the SNHP Loan.
- **1.189 SNHP Note**. The promissory note to be made by Borrower to the order of SNHP Lender, evidencing all amounts disbursed and to be disbursed under the SNHP Loan.
- **1.190 SNHP Restrictions.** The regulatory agreement executed by the Borrower for the benefit of SNHP Lender in connection with SNHP Lender making the SNHP Loan to Borrower.
- 1.191 SNHP Subordination Agreement. A subordination agreement in the form and substance satisfactory to Bank, executed by SNHP Lender and Bank and acknowledged by Borrower pursuant to which SNHP Lender shall unconditionally subordinate the lien in effect of the SNHP Deed of Trust and the SNHP Restrictions to the lien in effect of the Deed of Trust.
- **1.192 Standard Agreement.** That certain Standard Agreement entered into by HCD and Developer, pursuant to the terms of which HCD shall make available the HCD Loan to Borrower.
- **1.193 Stub Period**. The period from the Funding Date through (but excluding) the first day of the calendar month following such date.
- **1.194 Subordinate Documents**. Collectively, the CDC Documents, the HCID Documents, the SNHP Documents and the AHP Documents.
  - **1.195** Subordinate Lender. Collectively, the CDC, HCID, SNHP Lender and AHP Lender.
- **1.196 Subordinate Loans**. Collectively, the CDC Loan, the HCID Loan, the SNHP Loan and the AHP Loan.
- **1.197** Subordinate Deeds of Trust. Collectively, the CDC Deed of Trust, the HCID Deed of Trust, the SNHP Deed of Trust and the AHP Deed of Trust.
- **1.198 Subordination Agreements**. Collectively, the CDC Subordination Agreement, the HCID Subordination Agreement, the SNHP Subordination Agreement and the AHP Subordination Agreement.
  - **1.199** Surety. The bonding company that issues the bonds covering the Bonded Work.
  - **1.200** Tax Certificate. As defined in the Funding Loan Agreement.
  - **1.201** Tax Counsel. As defined in the Funding Loan Agreement.
- **1.202** Tax Credit Allocation Documents. The Tax Credit Application, the Preliminary Reservation, IRS Form 8609 to be hereafter executed by the Allocation Committee and all other documents heretofore and hereafter submitted to, and received by the Borrower from, the Allocation Committee, and all amendments, extensions and modifications thereto.
- **1.203 Tax Credit Application**. The 2018 Low-Income Housing Tax Credit Application submitted to the Allocation Committee to apply for Tax Credits with respect to the Project.

1.204	Tax Credit Investor.		] [CHECK].
-------	----------------------	--	------------

- **1.205** Tax Credits. Low income housing tax credits to be allocated under Section 42 of the Code pursuant to the terms of the Tax Credit Documents.
- **1.206 Tax-Exempt Regulatory Agreement**. The "Regulatory Agreement", as defined in the Funding Loan Agreement.
  - **1.207** Title Insurer. Commonwealth Land Title Company.
- **1.208 Title Policy**. An ALTA LP-10 Policy of Title Insurance or its equivalent acceptable to Bank, naming Governmental Lender and Bank as insured, with a liability limit of not less than the amount of the Borrower Loan, issued by Title Insurer, insuring that the Deed of Trust constitutes a valid first lien on the Real Property and Improvements, with only such exceptions from its coverage as shall have been approved in writing by Bank, with such reinsurance or coinsurance agreements or endorsements to such policy as Bank may require.
  - **1.209** Transfer. Any sale, lease or other transfer of any interest to any other Person.
- **1.210 Transition Reserve Account**. An interest bearing account established with Bank by Borrower at the time of Conversion for the purpose of funding any Rental Assistance Operating Deficits which are necessary for the continued operation of the Property.
- **1.211** Unit(s). The ninety-nine (99) low income apartment units and one (1) manager's unit constituting the Improvements.
- **1.212 Variable Rate**. A rate of Interest which bears interest with reference to a LIBOR Rate, pursuant to Section 3.1.2.
- **1.213** Variable Rate Principal. The outstanding principal balance of the Borrower Loan that is bearing interest at a Variable Rate.

#### 2. BORROWER LOAN.

- **2.1 Purpose**. The purpose of the Borrower Loan is to finance the acquisition of the Real Property and construction of the Improvements and other costs related thereto and to provide permanent financing for the Project.
- **2.2 Loan Terms and Conditions.** Subject to the terms and conditions contained in this Agreement, as may be modified by the provisions of <u>Exhibit C</u> and Section [3.4] **[CHECK]** of the Funding Loan Agreement, Governmental Lender agrees to make the Borrower Loan to Borrower. The repayment of all amounts due in connection with the Borrower Loan shall be secured by, among other things, the Deed of Trust, the Security Documents and such other collateral as may be required by Bank. Interest shall accrue and principal and interest shall be payable in accordance with the terms of this Agreement.
- **2.3 Loan Fee**. Borrower shall pay the Loan Fee to Bank in immediately available funds on or before the Closing Date. The Loan Fee shall be nonrefundable.
- **2.4 Full Payment and Reconveyance**. Upon Governmental Lender's and Bank's receipt, as applicable, of all sums owing and outstanding under the Borrower Loan Documents and under any other note or notes or any other obligation secured by the Deed of Trust, Bank shall issue a full reconveyance of the Property and Improvements from the lien of the Deed of Trust; <u>provided</u>, <u>however</u>, that all of the following conditions shall be satisfied at the time of, and with respect to, such reconveyance: (a) Bank shall have received all escrow, closing and recording costs, the costs of preparing and delivering such reconveyance and any sums then due and payable under the Borrower Loan Documents and the Funding

Loan Documents; and (b) Bank shall have received a written release satisfactory to Bank of any Set Aside Letter, letter of credit or other form of undertaking that Bank has issued to any Surety, Governmental Authority or any other party in connection with the Borrower Loan and/or the Property. As of the earlier of the last day of disbursement of the Funding Loan under Section 3.4(d) of the Funding Loan Agreement, or date of repayment in full of the Borrower Loan, Governmental Lender's obligation to make further disbursements under the Borrower Loan shall terminate as to any portion of the Borrower Loan undisbursed, and any commitment of Governmental Lender to lend any undisbursed portion of the Borrower Loan shall be cancelled.

- **2.5 Extension Term.** Borrower shall have the option to extend the Outside Conversion Date (for purposes of this Section, "Initial Outside Conversion Date") for an additional four (4) months ("Extension Term"), to and including [March 1, 2022] **[CHECK]** ("Extended Outside Conversion Date"), upon satisfaction of all of the following conditions, as determined by Bank:
- **2.5.1** Borrower shall provide Bank with Borrower's written request to extend the term of the Borrower Loan not less than ninety (90) days prior to the Initial Outside Conversion Date.
- **2.5.2** At the time of Bank's receipt of Borrower's written request to extend the term of the Borrower Loan, and as of the Initial Outside Conversion Date, no Event of Default shall have occurred and be continuing.
- **2.5.3** There shall have been no substantial deterioration in the financial condition of Borrower or any Loan Party, as determined by Bank in Bank's reasonable discretion.
- **2.5.4** Borrower and any Loan Party shall have executed such documents as Bank may require in connection with such extension, including any amendments to the Borrower Loan Documents.
- **2.5.5** Neither Borrower nor any Loan Party shall be in default under any promissory note, deed of trust, security agreement, guaranty or other agreement between Bank and any such party, and no event shall have occurred which would constitute a default or event of default thereunder.
- **2.5.6** Borrower shall have provided Bank with evidence that the construction of the Improvements shall be substantially completed in accordance with the Plans, as determined by Bank in its sole discretion.
- **2.5.7** Borrower shall have paid to Bank a loan extension fee equal to [\$112,024.50] **[CHECK]**.
- **2.5.8** Bank shall have the option, in its reasonable discretion, to re-balance the Interest Reserve to assure that there are sufficient funds in the Interest Reserve to pay the interest required under the terms of the Borrower Note during the Extension Term. In the event the Bank determines that the funds in the Interest Reserve are insufficient, Borrower shall pay into the Borrower's Funds Account such amount as is necessary, as determined by Bank in its sole discretion, to provide adequate funds to pay, at a minimum, the interest required under the terms of Section 3.1.2(a) during the Extension Term.
- **2.5.9** Borrower shall provide Bank with written evidence of HCD's consent to any such extension, including an amendment to the Standard Agreement evidencing the extension of the date by which funds are to be disbursed thereunder to not earlier than the Extended Outside Conversion Date.
- **2.5.10** Borrower shall pay all costs and expenses incurred by Bank in connection with extending the Initial Outside Conversion Date, including without limitation, documentation and/or recording fees, if any, and the cost of any title endorsements required by Bank.
- 2.6 Assignment of Borrower Loan Documents to Bank. Borrower acknowledges that the Governmental Lender has made an assignment to the Bank of all right, title and interest of the

Governmental Lender in this Borrower Loan Agreement (except for the Reserved Rights, as defined in the Funding Loan Agreement), the Borrower Note, the Deed of Trust and the other Borrower Loan Documents and has appointed the Bank as its agent to collect payments from the Borrower with respect to the Borrower Loan and to take all actions on behalf of Governmental Lender with respect to the Borrower Loan and the Borrower Loan Documents. Borrower hereby consents to all such assignments and the appointment of Bank as agent for the Governmental Lender.

#### 3. PAYMENTS; CONVERSION.

#### 3.1 Payments.

**3.1.1 General Obligation**. To induce Governmental Lender to issue the Funding Loan Note, Borrower shall pay to Bank all amounts, including principal, interest and premium (if any) that become due and payable on the Funding Loan Note, as and when such amounts become due and payable under the Funding Loan Note. Without limitation on the foregoing, Borrower shall also pay to Bank when due all other amounts described in this Agreement, as and when due and payable under this Agreement. Each such payment shall be made to the Fiscal Agent by deposit to such account as the Fiscal Agent shall designate by written notice to the Borrower.

#### 3.1.2 Interest.

- (a) At all times from and after the Funding Date to (but excluding) the Conversion Date, the outstanding principal balance of the Borrower Loan shall accrue interest at a rate which is 79% of the LIBOR Rate plus the Margin (applicable during the Construction Phase) for the then current Interest Period. The Variable Rate for the next Interest Period shall change on each Interest Change Date based on changes in the LIBOR Rate. There is no limit on the amount the Variable Rate may increase or decrease during the term of the Borrower Loan.
- (b) At all times from and after the Conversion Date, the outstanding principal balance of the Borrower Loan shall accrue interest at a rate which is 79% of the LIBOR Rate plus the Margin (applicable during the Permanent Phase) for the then current Interest Period. The Variable Rate for the next Interest Period shall change on each Interest Change Date based on changes in the LIBOR Rate. There is no limit on the amount the Variable Rate may increase or decrease during the term of the Borrower Loan.
- (c) At all times after the occurrence and during the continuance of an Event of Default, all principal outstanding under the Borrower Note shall accrue interest at the Default Rate.

#### 3.1.3 Monthly Payments.

- (a) Commencing on the First Payment Date and continuing on the 1st day of each calendar month thereafter through and including the Outside Conversion Date, payments in respect of the Borrower Loan shall be interest only, in arrears, on the outstanding principal of the Borrower Note at the Variable Rate. Interest shall be calculated on the basis of a year of 360 days, for actual days elapsed, prior to the Outside Conversion Date.
- (b) Commencing on the Amortization Date and on the 1st day of each calendar month thereafter through the Maturity Date, Borrower shall pay to Bank monthly installments of principal with respect to the Borrower Note as set forth on <u>Schedule 1</u> to be attached hereto and incorporated herein by this reference (the "Principal Payments"), plus interest accrued for the applicable Interest Period on the principal balance outstanding from time to time on the Borrower Note at the Variable Rate. Said principal and interest payments are hereinafter collectively referred to as the "Regular Payments" and are subject to change as and when the Variable Rate changes. The Regular Payments will be applied first to accrued but unpaid interest then due, and then to principal. A payment will be treated as made on the date it is received. At Conversion, <u>Schedule 1</u> shall be

prepared by Bank and attached to this Agreement and shall consist of a schedule of the monthly installments of principal required to fully amortize the outstanding principal balance of the Borrower Note owing on the Conversion Date, assuming equal monthly payments of principal and interest, an amortization period of one hundred eighty (180) months and a fixed rate of interest equal to the fixed rate or maximum interest rate of the Hedge in effect as of the Conversion Date. Bank shall provide Borrower with a copy of <a href="Schedule 1">Schedule 1</a> once it is prepared by Bank, but the effectiveness and date of such payment shall not be affected by such notice or lack thereof. Bank's determination of said Regular Payments shall be conclusive absent manifest error. All computations of interest shall be made on the basis of a year of 360 days, for actual days elapsed. On the Maturity Date, all principal and accrued interest then outstanding shall be immediately due and payable.

#### 3.2 Conversion; Termination.

- **3.2.1** Not later than 30 days prior to the earlier to occur of the proposed Conversion Date or the Outside Conversion Date, Borrower shall deliver the Conversion Election Notice to Bank. The Conversion Election Notice shall be accompanied by (a) a written certification by Borrower to Bank that all of the Conditions to Conversion have been fully satisfied; (b) a rent roll covering the Property for each of the three full calendar months immediately preceding the date of the Conversion Election Notice, certified by Borrower as true, correct and complete; (c) the Conversion Fee; and (d) operating statements for the Property for each of such three calendar months, in the form required by Bank, and certified by Borrower to be true, correct and complete.
- **3.2.2** The Conditions to Conversion specified in <u>Exhibit D</u> shall be applicable to the Conversion. Bank shall have the right to waive any Condition to Conversion set forth in <u>Exhibit D</u> in Bank's sole and absolute discretion.
- **3.2.3** If, based upon the information delivered pursuant to Section 3.2.1 and such other information as Bank may require as evidence of satisfaction of the Conditions to Conversion, Bank determines that the Conditions to Conversion have been fully satisfied, Bank shall deliver the Conversion Notice, which Conversion Notice shall state the Conversion Date and a copy of <u>Schedule 1</u> to be attached hereto setting forth the monthly installments of principal required to be paid by Borrower under the Borrower Note, as more particularly set forth in Sections 3.1.3(b) above.
- **3.2.4** Upon Conversion (and so long as all Conditions to Conversion are satisfied) the following documents shall be deemed automatically terminated and shall have no further force or effect without any further action by any Loan Party: (i) the Guaranty (except for the Indemnity Agreement); (ii) the Assignment of Tax Credits and Partnership Interests, and (iii) the Assignment of Partnership Interest (GP).
- **3.2.5** If the Conditions to Conversion have not been fully satisfied prior to the Outside Conversion Date, as such date may be extended in accordance with this Agreement, Borrower shall pay to Bank, on the Outside Conversion Date, the entire outstanding principal balance of the Borrower Loan, together with all accrued and unpaid interest thereon and other accrued and unpaid fees, costs and expenses owing under the Borrower Loan Documents and the Funding Loan Documents.
- 3.2.6 Non-Recourse After Conversion Date. From and after the Conversion Date, Governmental Lender and Bank agree that Governmental Lender's and Bank's recovery against Borrower in the event of a default under this Agreement, the Borrower Note or under any of the other Borrower Loan Documents shall be limited solely to, and Governmental Lender and Bank shall only proceed against, the Trust Estate (as defined in the Deed of Trust), together with the rents, issues, profits and income therefrom and proceeds and products thereof, and any other collateral given as security for Borrower's performance under the Borrower Loan Documents, and in no event shall (i) Borrower be personally liable for the payment of the Borrower Note or for the payment of any deficiency established upon foreclosure and the sale of the Trust Estate, or (ii) any other assets of Borrower (or any general partner of Borrower) be subject to levy, execution or other enforcement procedure in connection with any such default. Notwithstanding the

foregoing, Borrower (and General Partner) shall be fully and personally liable to Governmental Lender and Bank for the costs or damages arising from any of the following:

- (a) gross negligence, fraud, willful misrepresentation or waste by Borrower, to the full extent of Governmental Lender's and Bank's loss attributable thereto;
- (b) any inaccuracy in or breach of any representation` or warranty pertaining to any Hazardous Substances (as that term is defined in that certain Environmental Compliance Agreement (the "ECA") executed in favor of Governmental Lender and Bank by Borrower concurrently herewith), any failure in the due, prompt and complete observance and performance of any covenant or other obligation imposed under or pursuant to the ECA, or the presence of any Hazardous Substance on, under or about the Trust Estate, whenever arising:
- (c) failure to pay taxes, assessments or other charges which can create liens on any portion of the Trust Estate (to the full extent of any such taxes, assessments or other charges);
- (d) any loss which would have been covered by insurance required to be maintained under the terms of any of the Borrower Loan Documents, which Borrower failed to maintain;
- **(e)** failure to deliver to Bank any funds which should have been paid to Bank under the terms of the Borrower Loan Documents or the distribution of earnings or income from the Trust Estate in violation of the Borrower Loan Documents; or
- **(f)** any loss resulting from any claim or cause of action by a contractor, material supplier or other person or entity entitled to file a mechanic's lien against the Trust Estate.

In addition, Borrower and General Partner shall be fully and personally liable to Governmental Lender and Bank for the full amount of the Borrower Loan and all other obligations evidenced by the Borrower Loan Documents in the event (i) all or any part of the Trust Estate, other assets of Borrower or any ownership interest in Borrower is transferred in violation of the Borrower Loan Documents; (ii) any voluntary or involuntary proceeding under any laws relating to bankruptcy, insolvency, reorganization, arrangement, debt adjustment or debtor relief is commenced by or against Borrower or by or against any owner of the Property and, as to involuntary proceedings, is not dismissed within sixty (60) days; or (iii) Governmental Lender's or Bank's exercise of its rights and remedies under the Borrower Loan Documents is hindered, delayed, interfered with or prejudiced by or as a result of any act, omission, fraud or misrepresentation of Borrower or any other party now or hereafter liable for any part of the Borrower Loan.

The provisions hereof shall not be deemed to constitute a waiver of any obligation of Borrower or any other party or limitation of any kind of any right of Governmental Lender or Bank at law or equity or under any guaranty or other Borrower Loan Documents, provided that the assertion by Governmental Lender or Bank of any such right shall not result in a monetary claim upon the general unsecured assets of Borrower except as provided herein.

- **3.3 Maturity Date**. All unpaid principal and interest on the Borrower Loan and other amounts due under the Borrower Loan Documents and the Funding Loan Documents shall be due and payable in full on the Maturity Date, as such date may be extended or accelerated.
- **3.4 Application of Payments**. All payments and prepayments received by Governmental Lender or Bank pursuant to the terms hereof shall be applied in the following manner: first, to the payment of any Late Charge then due; second to the payment of all expenses, charges, costs and fees (including, but not limited to, the Prepayment Fee) incurred by or payable to Governmental Lender or Bank by Borrower pursuant to the terms of the Borrower Loan Documents (in such order and manner as Bank, in its sole discretion, may elect); third to the payment of all interest accrued to the date of such payment; and fourth, to the payment of principal. Notwithstanding anything to the contrary contained herein, after the occurrence

and during the continuation of an Event of Default, all amounts received by Governmental Lender and Bank from any party shall be applied in such order as Bank in its sole discretion, may elect.

- Acceleration. If any of the payments required by the terms hereof shall not be paid when 3.5 due and such failure shall continue beyond any applicable notice and cure periods, or if the payment due on the Maturity Date is not paid when due, whether by acceleration or otherwise, or if an Event of Default occurs, then, or at any time thereafter, the whole of the unpaid principal and interest owing on the Borrower Loan shall, at the option of Bank and without notice, become immediately due and payable. This acceleration option may be exercised at any time after any such event and the acceptance of one or more installments or other payments from any Person thereafter shall not constitute a waiver of Bank's acceleration option. Bank's failure to exercise such acceleration option in connection with any particular event or series of events shall not be construed as a waiver of the provisions hereof as regards such events or any subsequent events. The other Borrower Loan Documents may contain provisions that provide for the automatic acceleration of amounts owing in connection with the Borrower Loan upon the occurrence of certain specified events. Bank shall have, and be entitled to exercise, upon the occurrence of any Event of Default or other event described above, all rights and remedies available to Bank hereunder, under the other Borrower Loan Documents or Funding Loan Documents or at law or in equity. All such rights and remedies shall be cumulative.
- 3.6 Late Charge; Default Interest. Borrower recognizes that any default by Borrower in making the payments required under the Borrower Loan Documents when due will result in Governmental Lender and Bank incurring additional expense in servicing the Borrower Loan, in loss of the use of the money due and in frustration of meeting commitments under the Funding Loan Documents. Borrower agrees that, if for any reason Borrower fails to pay when due any payment due under this Agreement or under any of the other Borrower Loan Documents, any amount advanced under the Deed of Trust or the amount due on the Maturity Date, or the accelerated Maturity Date, whichever shall first occur, Bank shall be entitled to damages for the detriment caused thereby, but that it is extremely difficult and impractical to ascertain the extent of such damages. Borrower therefore agrees that a reasonable estimate of such damages to Bank is as follows:
- **3.6.1** In the event Borrower fails to pay any installment of principal and interest (other than payment on the Maturity Date) within ten days after the same is due, then Borrower shall pay to Bank a Late Charge.
- **3.6.2** In the event Borrower fails to reimburse Bank for any amount advanced under the Deed of Trust within ten days after written notice of such advance is made by Bank to Borrower, then such unreimbursed amount shall thereafter bear interest at the Default Rate until paid, such interest to be compounded annually.
- **3.6.3** In the event the payment of principal and accrued but unpaid interest due on the Maturity Date, or the accelerated Maturity Date, as applicable, is not made in full when due, then such amounts shall thereafter bear interest at the Default Rate, until paid, such interest to be compounded annually.
- **3.7 Prepayment**. Pursuant to the terms of this Section 3.7, the Borrower Loan may be prepaid by Borrower when and to the extent that the Funding Loan Note is susceptible to prepayment under the Funding Loan Documents, provided that Borrower shall in no event voluntarily or involuntarily prepay the Borrower Loan in whole or in part unless Borrower pays to Bank, concurrently with such prepayment, a prepayment fee as calculated below.
- **3.7.1 Variable Rate Principal**. Any Variable Rate Principal may be prepaid prior to the scheduled payment date, whether voluntary or involuntary, in whole or in part, provided Borrower has given Bank not less than five (5) business days prior written notice of Borrower's intention to make such prepayment and pays to Bank the prepayment fee due as a result. The prepayment fee shall be an amount equal to the present value of the product of: (i) the difference (but not less than zero) between (a) the

Variable Rate applicable to the principal amount which is being prepaid, and (b) the return which Bank could obtain if it used the amount of such prepayment of principal to purchase a bid price regularly quoted securities issued by the United States having a Maturity Date most closely coinciding with the relevant Interest Period and such securities were held by Bank until the relevant Interest Period ("Yield Rate"); (ii) a fraction, the numerator of which is the number of days in the period between the date of prepayment and the last day of the relevant Interest Period, and the denominator of which is 360, and (iii) the amount of principal so prepaid. The present value shall be determined by discounting the above product to present value using the Yield Rate as the annual discount factor. Bank shall provide Borrower a statement of the amount payable on account of prepayment. Borrower acknowledges that (i) Bank establishes a Variable Rate upon the understanding that it apply to the Variable Rate Principal for the entire Interest Period, and (ii) Governmental Lender would not lend to Borrower at a Variable Rate without Debtor's express agreement to pay the prepayment fee described above.

- **3.7.2** No Prepayment Fee Due. Notwithstanding Section 3.7.1 above, no prepayment fee shall be payable (i) in connection with the prepayment of Variable Rate Principal in connection with the prepayment of principal during the ninety (90) day period immediately preceding the Outside Conversion Date, as may be extended pursuant to Section 2.5 above, or the Maturity Date.
- **3.7.3** No Refund. In no event shall Bank be obligated to make any payment or refund to Borrower, nor shall Borrower be entitled to any setoff or other claim against Bank, should the return which Bank could obtain under the prepayment formula exceed the interest that Governmental Lender would have received if no prepayment had occurred.
- **3.7.4** Payment of Accrued Interest. All prepayments shall include payment of accrued interest on the principal amount so prepaid, shall be applied to payment of interest before application to principal, and shall be applied to the most remote principal installment or installments then unpaid (i.e., the principal balance due on the Maturity Date and then against installments due closest to the Maturity Date).
- 3.7.5 Involuntary Prepayment. Such prepayment fee shall also be payable if prepayment occurs as the result of any involuntary prepayment (e.g., proceeds of insurance or condemnation or any prepayment required in order to satisfy the Conditions to Conversion) or the acceleration of the principal hereof by Bank because of any default by Borrower (including any transfer or conveyance of any right, title or interest in the real property encumbered by the Deed of Trust) that gives Bank the right to accelerate the maturity of the Borrower Loan pursuant to the terms of the Deed of Trust. If, following any such acceleration, all or any portion of the unpaid principal is satisfied, whether through sale of the property encumbered by the Deed of Trust or other agreement securing the Borrower Loan at a foreclosure held thereunder or through the tender of payment at any time following such acceleration, but prior to such a foreclosure sale, then such satisfaction of principal shall be deemed an evasion of the prepayment provisions hereof, and Bank shall, automatically and without notice or demand, be entitled to receive, concurrently with such satisfaction of principal the prepayment fee set forth above, and the obligation to pay such prepayment fee shall be added to the principal hereof.

BORROWER HEREBY ACKNOWLEDGES AND AGREES THAT GOVERNMENTAL LENDER WOULD NOT LEND TO BORROWER THE BORROWER LOAN EVIDENCED BY THE BORROWER NOTE WITHOUT BORROWER'S AGREEMENT TO PAY BANK A PREPAYMENT FEE AS SET FORTH ABOVE. BORROWER EXPRESSLY WAIVES ANY RIGHT UNDER CALIFORNIA CIVIL CODE SECTION 2954.10 OR OTHERWISE TO PREPAY THE BORROWER LOAN WITHOUT A PREPAYMENT FEE AS HEREINABOVE SET FORTH. BORROWER ACKNOWLEDGES THAT PREPAYMENT OF THE BORROWER LOAN MAY RESULT IN GOVERNMENTAL LENDER AND BANK INCURRING ADDITIONAL COSTS, EXPENSES OR LIABILITIES. BORROWER THEREFORE AGREES THAT THE PREPAYMENT FEE HEREIN PROVIDED FOR REPRESENTS A REASONABLE ESTIMATE OF THE PREPAYMENT COSTS, EXPENSES OR LIABILITIES GOVERNMENTAL LENDER AND BANK MAY INCUR ON A PREPAYMENT. BORROWER AGREES THAT GOVERNMENTAL LENDER'S WILLINGNESS TO OFFER THE [FIXED] [VARIABLE] INTEREST RATE DESCRIBED ABOVE TO BORROWER IS SUFFICIENT AND INDEPENDENT CONSIDERATION, GIVEN INDIVIDUAL WEIGHT BY

GOVERNMENTAL LENDER AND BANK FOR THIS WAIVER. BORROWER UNDERSTANDS THAT GOVERNMENTAL LENDER WOULD NOT OFFER SUCH AN INTEREST RATE TO BORROWER ABSENT THIS WAIVER. BORROWER HAS CAUSED THOSE PERSONS SIGNING THIS AGREEMENT ON ITS BEHALF TO SEPARATELY INITIAL THIS PARAGRAPH BY PLACING THEIR INITIALS BELOW:

BORROWER INITIALS HERE:	<u></u>	 	 - <u>-                                    </u>

- **3.7.6 Certification.** A certificate as to the amount of any prepayment fee payable under this Section, setting forth the basis for such fee, prepared by Bank and submitted to Borrower shall be conclusive as to the matters set forth therein, and the Borrower Loan shall not be deemed to have been fully paid or satisfied until such fee shall have been paid.
- **3.7.7 Effect of Prepayment on Hedge.** Borrower and Bank hereby agree that, in accordance with and subject to the terms of any Hedge Documents, any Hedge entered into between Borrower and Bank in connection with the Borrower Loan shall, upon the making of any prepayment of amounts outstanding under the Borrower Loan, be subject to an Additional Termination Event (as defined in such Hedge Documents) and may be terminated as and to the extent more particularly provided in the documents and agreements evidencing such Hedge. Any amounts (which may be substantial) payable by Borrower to Bank, or by Bank to Borrower in respect of such full or partial termination of such Hedge shall be determined under the terms and conditions of the Hedge Documents relating to such Hedge.
- 3.8 Additional Fee Payment Obligations. All payments to fund taxes, insurance or any other escrow or reserve required to be established, funded or created pursuant to any Borrower Loan Document or Funding Loan Document, shall be due and payable by Borrower to Fiscal Agent the date monthly payments are due pursuant to Section 3.1.3 commencing in the month following the month in which the Conversion Date occurs in accordance with the applicable Borrower Loan Document or Funding Loan Document. Borrower shall pay to Fiscal Agent Fiscal Agent's Fees (as defined in the Funding Loan Agreement) in accordance with the terms of the Funding Loan Agreement.

#### 4. CONDITIONS PRECEDENT.

- **4.1** Conditions to Closing of the Borrower Loan. Prior to the Closing Date, Bank shall have received all of the following documents, instruments and other items (each of which, in the case of documents or instruments, shall be fully and properly executed and, where required by Bank, acknowledged by all parties thereto), each in form and content acceptable to Bank:
  - **4.1.1** The original Borrower Loan Documents.
- **4.1.2** Copies of organizational documents of Borrower and all Loan Parties, duly filed and/or recorded in the appropriate jurisdiction and certified as required by Bank, including without limitation, and as applicable, (a) articles of organization and operating agreements, (b) certificates of limited partnership, statements of partnership and partnership agreements, (c) statements of joint venture and joint venture agreements, (d) articles of incorporation, (e) trust agreements, and (f) any amendments to any of the foregoing.
- **4.1.3** Evidence that the insurance required by the Agreement to Furnish Insurance is in full force and effect.
  - **4.1.4** All Borrower's Funds required under this Agreement.
- **4.1.5** Copies of the Detailed Cost Breakdown, the Project Budget, the Plans, the Construction Contract (if any), the Architect's Agreement, and any other agreements that Bank determines are material to construction of the Improvements, all certified as required by Bank.

- **4.1.6** Copies of the building permits and any other authorizations required from any Governmental Authority in connection with construction of the Improvements.
- **4.1.7** If required by Bank, a current ALTA survey of the Real Property, including dimensions and delineation and location of all easements thereon, certified to and satisfactory to Bank and Title Insurer.
- **4.1.8** If required by Bank, letters from local utility companies and any Governmental Authority stating that electric, gas, sewer, water, cable and telephone facilities are or will be available to the Real Property upon completion of the Improvements.
- **4.1.9** Written results of such due diligence investigations with respect to Borrower, any Loan Party and the Property as Bank deems necessary, including without limitation, environmental reviews, engineering inspections, seismic studies and financial analysis.
- **4.1.10** An opinion of Borrower's counsel as to (a) the proper formation, valid existence and good standing of Borrower and all Loan Parties, (b) the due authorization and execution of all Borrower Loan Documents and any Hedge Documents with Bank by Borrower and all Loan Parties, (c) whether all necessary consents have been obtained with respect to the Borrower Loan and any Hedge Documents with Bank, (d) the absence of any threatened or pending actions, suits or proceedings against or affecting the Property, Borrower or any Loan Party, (e) the violation of any agreements to which Borrower or any Loan Party is bound, and (f) such other matters as Bank may determine to be necessary or appropriate.
- **4.1.11** A performance bond naming Governmental Lender and Bank as co-obligee and a labor and material payment bond, in an amount equal to the amount of the Construction Contract, or if there is no Construction Contract, then in such amounts as Bank may require, issued by a surety acceptable to Bank and otherwise in form and content acceptable to Bank. The performance and the labor and material bonds shall have been recorded in the official records of the county in which the Real Property is located prior to the commencement of work on the Improvements.
- **4.1.12** Such evidence (including, without limitation, the tax credit certifications attached to the Carryover Allocation as Exhibit A) as Bank may reasonably require to confirm the accuracy of the representations and warranties set forth in Section 6.28 of this Agreement.
- **4.1.13** Fully executed copies of the Subordinate Documents and the Density Bonus Restrictions, each in a form acceptable to Bank.
  - **4.1.14** Intentionally Omitted.
- **4.1.15** Borrower shall have entered into one or more Hedges, in form and content and from a counterparty complying with the provisions contained in Section 7.48, with respect to the Borrower Note in an amount not less than the Projected Permanent Phase Loan Amount, which provides for a fixed rate of interest on the Borrower Note not to exceed (or otherwise protects against the interest rate on the Borrower Note exceeding) [\_\_\_%] [CHECK] (including the Margin applicable during the Permanent Phase) for the period commencing on the Initial Outside Conversion Date (as defined in Section 2.5 above) through the Maturity Date.
- **4.1.16** Borrower shall have entered into the AHAP Contract on terms and conditions acceptable to Bank, and Contract Administrator shall have consented to the collateral assignment of the AHAP to Bank pursuant to the Assignment of AHAP Contract.
- **4.1.17** Borrower shall have delivered to Bank a fully executed copy of the Standard Agreement and an estoppel certificate executed by HCD with respect thereto, each in form and content acceptable to Bank.

- **4.1.18** Such other documentation, certifications, opinions and information as may be reasonably required by Governmental Lender or Bank.
- **4.1.19** Any special conditions set forth in the Special Conditions attached hereto as Exhibit C shall have been satisfied.
- **4.2 Conditions to Issuance of the Funding Loan Note**. Governmental Lender's obligation to execute the Funding Loan Note, and Governmental Lender's and Bank's obligation to enter into this Agreement, the other Borrower Loan Documents and the Funding Loan Documents, and to make the Initial Disbursement, are subject to the satisfaction, or waiver by Governmental Lender or Bank, as applicable, each of the conditions in Section 4.1 and of all of the following conditions precedent:
- **4.2.1** Governmental Lender and Bank shall have received fully executed originals of each of the Borrower Loan Documents and the Funding Loan Documents.
- **4.2.2** The Tax-Exempt Regulatory Agreement shall have been duly executed, acknowledged and delivered by Borrower to Governmental Lender and Bank.
- **4.2.3** Each of the Recorded Documents shall have been recorded in the Official Records of the county in which the Real Property is located.
- **4.2.4** The Financing Statements have been filed with the Secretary of State of California, and Bank shall have received a certificate of the Secretary of State showing such Financing Statements to be subject to no prior filings (other than filings perfecting Permitted Liens) except as otherwise agreed to by Bank.
  - **4.2.5** Title Insurer shall have committed to deliver to Bank the Title Policy.
- 4.2.6 Bank and Governmental Lender shall have received and approved an executed original of each of the following opinions, in each case addressed to each of Governmental Lender and Bank and in each case in form and substance approved by Governmental Lender and Bank: (a) the opinion of counsel to Borrower and the other Loan Parties, opining as to the due formation, qualification and good standing of Borrower and the other Loan Parties, the due authorization by Borrower and the Loan Parties of the execution, delivery and performance of the Borrower Loan Documents, and the enforceability of the Borrower Loan Documents, and covering such other matters as Bank may require; and (b) an opinion of Tax Counsel, opining as to the due organization and valid existence of the Governmental Lender, due execution and delivery by the Governmental Lender of the Funding Loan Agreement, and this Agreement, the enforceability of the Funding Loan Agreement and this Agreement, and the exclusion of interest on the Funding Loan Note from gross income for federal income tax purposes.
- **4.2.7** Bank shall have received and approved such Financial Statements and other financial information as it may require regarding the financial condition of Borrower, the Loan Parties and/or the Property.
- **4.2.8** Bank shall have received and approved a detailed sources and uses statement showing (i) all costs and expenses of issuance of the Funding Loan Note, and (ii) all sources for payment of such costs and expenses.
- **4.2.9** To the extent not funded from the Initial Disbursement, Borrower shall have paid to Governmental Lender and Bank, as applicable, in immediately available good funds (a) all costs and expenses incurred by Governmental Lender and Bank in connection with the Funding Loan, the making of the Borrower Loan and the negotiation, preparation and closing of the Borrower Loan Documents and Funding Loan Documents, (b) the Tax Counsel fees and expenses due and payable; (c) all of fees to Governmental Lender then due and payable; and (d) the initial Fiscal Agent's Fees (as defined in the Funding Loan Agreement).

- **4.2.10** Borrower shall have delivered to Bank, and Bank shall have approved such information, and/or documentation as Bank may require to evidence that paragraph (1) of Section 42(h) of the Code does not apply to the Tax Credits by virtue of the provisions set forth in subparagraph (4)(B) of Section 42(h) of the Code.
- **4.2.11** Any special conditions set forth in the Special Conditions attached hereto as Exhibit C shall have been satisfied.

#### 5. DISBURSEMENTS.

#### 5.1 Initial Disbursement.

- **5.1.1** Prior to the Initial Disbursement, the following conditions shall have been satisfied in addition to the conditions set forth in Sections 4.1 and 4.2, as determined by Bank:
  - (a) Borrower and all Loan Parties shall have performed to Bank's satisfaction all covenants required to be performed under this Agreement, the other Borrower Loan Documents and the Funding Loan Documents on or before the Funding Date.
  - **(b)** No change shall have occurred that could have a material adverse effect on Borrower, any Loan Party, the Property or Bank's right or ability to receive payment in full of the Borrower Loan, as determined by Bank in its sole discretion.
    - (c) No Event of Default shall exist.
  - (d) The representations and warranties of Borrower in this Agreement and the other Borrower Loan Documents shall be true and correct on and as of the date of the disbursement with the same effect as if made on such date.
  - **(e)** Bank shall have approved in its sole discretion, the Detailed Cost Breakdown, the Project Budget, the Plans, the Construction Contract (if any), the Architect's Agreement, and any other agreements that Bank determines are material to the construction of the Improvements.
  - (f) Bank shall have received satisfactory evidence that there are no liens on Personal Property, except as otherwise agreed to by Bank.
  - (g) If required by Bank, Bank shall have received a list of the names and addresses of all suppliers, laborers and subcontractors with whom agreements have been made with Contractor and/or Borrower to deliver materials and/or perform work on the Improvements.
  - **(h)** Such evidence as Bank may require evidencing expenditure of Borrower's Equity on Project costs in accordance with this Agreement is at least [\$1,742,975] **[CHECK]** in the aggregate.
  - (i) Any special conditions set forth in the Special Conditions attached hereto as Exhibit C shall have been satisfied.
- **5.1.2** Upon satisfaction of the conditions contained in Sections 4.1, 4.2 and 5.1.1, Bank, on behalf of Governmental Lender, shall make an Advance in accordance with the Project Budget and the Disbursement Schedule the amounts necessary to pay all costs, charges and expenses incurred or to be incurred (as estimated by Bank) in connection with the Borrower Loan or payable pursuant to this Agreement or the other Borrower Loan Documents, excluding direct costs of labor and materials related to the Improvements, but including without limitation, the Loan Fee, service charges, fees and expenses of the Fiscal Agent, title charges, tax and lien service charges, recording fees, escrow fees, appraisal fees, legal fees, real property taxes and assessments, insurance premiums any amounts required to pay existing

encumbrances affecting the Property, and any amounts required to complete purchase of the Real Property.

#### 5.2 Subsequent Disbursements.

- **5.2.1** Prior to making any Advances after the Initial Disbursement, except for the final Advance, the following additional conditions shall have been satisfied, as determined by Bank:
  - (a) All specific requirements for the disbursement set forth in the Disbursement Schedule shall have been satisfied.
    - (b) No Event of Default shall exist.
  - (c) The representations and warranties of Borrower in this Agreement and the other Borrower Loan Documents shall be true and correct on and as of the date of the disbursement with the same effect as if made on such date.
  - (d) The Improvements shall not have been damaged by fire or other casualty unless Bank has determined that Bank will receive proceeds sufficient in Bank's judgment to effect the satisfactory restoration of the Improvements and permit Project Completion prior to the Completion Date.
  - (e) If required by Bank, Bank shall have received confirmation to its satisfaction that (A) to date, the Improvements have been constructed in accordance with the Plans and the Construction Contract (if any), and (B) the present state of construction of the Improvements will, barring the unforeseen and unknown delays, permit Project Completion on or before the Completion Date.
  - (f) If Bank has determined that the undisbursed proceeds of the Borrower Loan, together with the undisbursed amount of the HCID Loan, the CDC Loan, the SNHP Loan and the AHP Loan designated for payment for construction of the Improvements and Borrower's Funds (if any), are insufficient to pay all costs to complete construction of the Improvements (and all other costs included within the Project Budget), Borrower shall have deposited into the Borrowers' Funds Account cash in the amount of such shortfall as provided in Section 7.2.
  - (g) If required by Bank, (A) Title Insurer shall have issued its continuation endorsement to the Title Policy indicating that since the last preceding disbursement, there: (1) has been no change in the condition of title to the Real Property; and (2) are no intervening liens that may now or hereafter take priority over the disbursement to be made, and (B) upon completion of the foundation, Title Insurer shall have issued its foundation endorsement to the Title Policy insuring Bank that the foundation is constructed wholly within the boundaries of the Real Property and does not encroach on any easements or violate any covenants, conditions or restrictions or any Governmental Requirement.
  - **(h)** Bank shall have received satisfactory evidence that there are no liens on Personal Property, except as otherwise agreed to by Bank.
  - (i) If required by Bank, Bank shall approve the subdivision map in final form as it was filed or recorded. If the Borrower Loan will close before the filing of a final subdivision map, Bank shall have received satisfactory evidence of Borrower's right to subdivide and develop the Real Property, including, but not limited to, copies of all documents submitted in connection with obtaining the final subdivision public report, including the declaration of covenants, conditions and restrictions.

- (j) If requested by Bank, (a) Tax Credit Investor shall have executed and delivered to Bank an estoppel certificate which shall contain such certifications as Bank shall reasonably require with respect to Tax Credit Investor's obligations under the Partnership Agreement, and (b) each Subordinate Lender shall have executed and delivered to Bank an estoppel certificate which shall contain such certifications as Bank shall reasonably require with respect to the applicable Subordinate Documents and/or the Density Bonus Restrictions.
- (k) All amounts deposited into the Borrower's Funds Account shall have been withdrawn by Borrower to cover Project costs in accordance with the terms and conditions of this Agreement, and [\$4,200,000] [CHECK] of the HCID HHH Loan (except for any holdback not exceeding [\$420,000] [CHECK]) and the entire amount of the HCID HOPWA Loan, the HCID Recast Loan, the CDC Loan and the SNHP Loan shall have been fully disbursed (except for any holdback not exceeding ten percent (10%) of the CDC Loan and the SNHP Loan) by the applicable Subordinate Lender to or for the account of Borrower and applied towards Project costs.
- (I) Any special conditions set forth in the Special Conditions attached hereto as Exhibit C shall have been satisfied.
- **5.2.2** Upon satisfaction of the conditions contained in Sections 5.2.1 and 5.4 (as applicable), on or about the first day of each calendar month following commencement of construction of the Improvements, Contractor shall submit to Borrower a Draw Request showing the estimated cost of labor performed on and materials incorporated into the Improvements, a pro-rata portion of Contractor's profit and that pro-rata portion of overhead of Contractor attributable to the construction of the Improvements. The original of such Draw Request, certified true and correct by Contractor and approved by Borrower, shall be submitted to Bank for payment. Upon verification of the accuracy of the Draw Request by inspection of the Real Property and Improvements (if required by Bank), Governmental Lender shall disburse the amount of the respective approved Draw Request in accordance with the Disbursement Schedule (i) directly to Borrower or, upon the occurrence and during the continuance of an Event of Default, directly to Contractor or to such Persons as have actually supplied labor, materials or services in connection with the construction of the Improvements (at Bank's option as to whom and in what amounts payments are to be made), or (ii) if specifically required by Bank, through a fund control service acceptable to Bank under a fund control agreement in form and content acceptable to Bank.

#### 5.3 Final Disbursement.

- **5.3.1** Prior to making the final Advance, the conditions set forth in Sections 5.1, 5.2 and 5.4 (as applicable) and the following conditions shall have been satisfied, as determined by Bank:
  - (a) Bank shall have received confirmation to its satisfaction that the Improvements have been completed in accordance with the Plans and the Construction Contract (if any).
  - **(b)** If required by Bank, Bank shall have received a copy of the final certificate of occupancy (or its equivalent as determined by Bank) issued by the appropriate Governmental Authority.
  - (c) Bank shall have received evidence that Borrower has recorded a notice of completion (or its equivalent as determined by Bank) with respect to the Improvements.
  - (d) Bank shall have received (A) such endorsements to the Title Policy as Bank may require which shall insure that the Improvements have been completed free of all mechanic's and materialmen's liens or claims thereof, or (B) such additional title policies with endorsements as Bank may require, with a liability limit of not less than the principal amount of the Loan, issued by Title Insurer, with coverage and in form satisfactory to Bank, insuring Governmental Lender's and Bank's interest under the Deed of Trust as a first lien on the Real Property, excepting only such items as shall have been approved in writing by Bank.

- **(e)** Such evidence as Bank may require evidencing expenditure of Borrower's Equity on Project costs in accordance with this Agreement is at least [\$2,162,975] **[CHECK]** in the aggregate.
- (f) The entire amount of the AHP Loan shall have been fully disbursed by AHP Lender to or for the account of Borrower and applied towards Project costs.
- **5.3.2** The final disbursement shall consist of the payment of any monies retained from progress payments or disbursements as set forth in this Agreement. Subject to the provisions of this Agreement, the final disbursement shall be made only after Borrower has satisfied the conditions of Sections 5.3.1 and 5.4 (as applicable).
- **5.4** Additional Conditions to Advances. Bank shall have the right to condition any Advance upon Bank's receipt and approval of the following, each in form and content acceptable to Bank:
  - **5.4.1** The Draw Request.
- **5.4.2** Bills, invoices, documents of title, vouchers, statements, receipts and any other documents evidencing the total amount expended, incurred or due for any requested line item shown in the Project Budget.
- **5.4.3** Evidence of Borrower's use of a lien release, joint check or voucher system acceptable to Bank for payments or disbursements to Contractor or to such Persons as have actually supplied labor, materials or services in connection with the construction of the Improvements.
- **5.4.4** Architect's, inspector's and/or engineer's periodic certifications of the percentage and/or stage of construction that has been completed and its conformance to the Plans and any Governmental Requirement based upon such architect's, inspector's and/or engineer's periodic physical inspections of the Real Property and Improvements.
- **5.4.5** Waivers and releases of any mechanic's lien, stop notice claim, equitable lien claim or other lien claim rights.
- **5.4.6** Any other documents, requirements, evidence or information that Bank may request under any provision of the Borrower Loan Documents.
- **5.4.7** Evidence that any goods, materials, supplies, fixtures or other work in progress for which disbursement is requested have been incorporated into the Improvements.
- **5.4.8** In the event any Draw Request includes the cost of Offsite Materials, such Draw Request shall include each of the following: (a) evidence that the Offsite Materials have been purchased by Borrower, have been segregated from other materials in the facility where they are stored and have been appropriately marked to indicate Borrower's ownership thereof and Bank's security interest therein; (b) evidence that the Offsite Materials are insured as required by this Agreement; and (c) at Bank's request, a security agreement, financing statement, acknowledgment, and/or subordination agreement in form and content satisfactory to Bank executed by the supplier of the Offsite Materials, and/or such other Persons as Bank determines may have an interest in or claim to the Offsite Materials, together with such other additional documentation and evidence as Bank may reasonably require to assure itself that it has a perfected first priority lien on the Offsite Materials.
- **5.4.9** In the event any Draw Request includes the cost of Onsite Materials, such Draw Request shall include each of the following: (a) evidence that the Onsite Materials have been purchased by Borrower; (b) evidence that the Onsite Materials are insured as required hereunder; and (c) evidence that the Onsite Materials are stored in an area on the Real Property for which adequate security is provided against theft and vandalism.

#### 5.5 Disbursement Limits.

- **5.5.1** Borrower hereby represents to Bank that, as of the date of this Agreement, the Project Budget represents the total amount needed by Borrower to construct the Improvements and to perform Borrower's obligations under the Borrower Loan Documents and Funding Loan Documents. Bank shall not be required to make any Advance for any Construction Costs or any other purpose that is not set forth in the Project Budget nor shall Bank be required to make any Advance for any line item in the Project Budget in an amount that when added to the sum of all prior Advances for that line item would exceed the sum allocated in the Project Budget for that line item.
- **5.5.2** Bank reserves and shall have the right to make Advances that are allocated to any line items in the Project Budget for such other purposes or in such different proportions as Bank may, in its sole discretion, deem necessary or advisable. Borrower shall have no right whatsoever to reallocate Advances from one line item in the Project Budget to another or otherwise amend the Project Budget without the prior consent of Bank.
- **5.5.3** All Advances shall be made in accordance with the applicable provisions of the Project Budget and the Disbursement Schedule. All funds disbursed to Borrower shall be received by Borrower in trust and Borrower agrees that such funds shall be used only for the payment of those items contemplated by the particular Advance.
- **5.5.4** Bank shall not be required to disburse an aggregate amount of the proceeds of the Borrower Loan for labor furnished to and materials incorporated into the Improvements during any stage of construction that exceeds the lesser of (a) the value of such labor and materials, and (b) the amount allocated to that stage of construction in the Project Budget. In any event, Bank shall not be required to disburse any amount that, in Bank's opinion, will reduce that portion of the undisbursed proceeds of the Borrower Loan designated for completion of the Improvements below the amount needed to pay for the labor and materials necessary to complete the Improvements.
- **5.5.5** Notwithstanding anything to the contrary contained herein, disbursements of the Borrower Loan shall be made from the Project Fund held by the Fiscal Agent pursuant to the Funding Loan Agreement.
- Disbursement into Project Fund. Notwithstanding anything to the contrary contained in 5.6 the Funding Loan Documents or the Borrower Loan Documents, if the Funding Loan has not been fully disbursed by December 1, 2021 and Conversion has not yet occurred, or in the event the Bank determines that legislative, judicial or other developments have occurred or other circumstances have emerged which could result in interest on any undisbursed portions of the Funding Loan (the "Remaining Undisbursed Funding Loan") not being excluded from gross income for federal income tax purposes, or otherwise determines that it is in the Bank's best interest to fully fund the Funding Loan in order to assure that interest on the Funding Loan will remain excluded from gross income for federal income tax purposes (each, a "Contingency Event"), then Bank may, in its discretion, upon five (5) days' written notice to Borrower, disburse all or any portion of the Remaining Undisbursed Funding Loan to the Fiscal Agent for deposit into the Project Fund established pursuant to the Funding Loan Agreement, at which time the proceeds so advanced shall constitute (i) an advance of the Funding Loan to the Governmental Lender, and (ii) an advance of the Borrower Loan by Governmental Lender to the Borrower, unless Bank receives an opinion of Tax Counsel to the effect that the draw of Funding Loan proceeds after the Contingency Event will not adversely affect the exclusion of interest on the Funding Loan Note from gross income for federal income tax purposes. The portion of the Borrower Loan disbursed into the Project Fund pursuant to this Section shall be deemed outstanding as of the date advanced into the Project Fund and will immediately commence to accrue interest as provided in Section 3.1.2. All funds disbursed into the Project Fund shall continue to be disbursed by Bank to pay Project costs pursuant to the provisions of the Funding Loan Agreement, this Section 5 and the Disbursement Schedule as if they were Advances of the Borrower Loan.

- **5.7 Disbursement of Funds from General Partner Equity Account.** Prior to Conversion, Borrower shall cause General Partner to (i) establish with Bank the General Partner Equity Account and (ii) deposit [\$4,182,935] **[CHECK]** into the General Partner Equity Account. On the Conversion Date, the entire amount of the proceeds in the General Partner Equity Account shall be applied to repay the Borrower Loan.
- 6. REPRESENTATIONS AND WARRANTIES OF BORROWER. Borrower makes the following representations and warranties for the benefit of Governmental Lender and Bank, each of which is material and is relied upon by Governmental Lender in making the Borrower Loan and Governmental Lender and Bank in executing this Agreement. Each of the following representations and warranties shall be true and accurate as of the Contract Date, the Closing Date and upon disbursement of the Initial Disbursement and each Advance. Borrower agrees that such representations and warranties shall survive and continue until full and final payment of all sums owed under the Borrower Loan Documents.
- **6.1 Formation/Authority**. Borrower has complied with all laws and regulations concerning Borrower's organization, existence and the transaction of Borrower's business, and is in good standing in each state in which Borrower conducts business. Borrower is authorized to execute, deliver and perform Borrower's obligations under each of the Borrower Loan Documents and the Funding Loan Documents, and Borrower is authorized to construct the Improvements and to own and operate the Property.
- **6.2 No Defaults Under Existing Agreements.** The transactions contemplated hereby and the performance by Borrower of Borrower's obligations under the Borrower Loan Documents and the Funding Loan Documents will not result in any breach of or default under any deed of trust, mortgage, lease, loan, security agreement or any other agreement to which Borrower is a party or may be bound or affected.
- 6.3 No Actions. There are no actions, suits or proceedings pending or, to the best knowledge of Borrower, threatened against or affecting Borrower or the Property or involving the validity, priority or enforceability of the Deed of Trust or any other Loan Document or Funding Loan Documents or affecting Bank's right to receive payment in full of all amounts outstanding under this Agreement, the other Borrower Loan Documents or the Funding Loan Documents. Borrower is not in default with respect to any order, writ, injunction, decree or demand of any court or any Governmental Authority. There (a) is no completed, pending or threatened bankruptcy, reorganization, receivership, insolvency or like proceeding, whether voluntary or involuntary, affecting the Property, the Borrower, or any Loan Party, and (b) has been no assertion or exercise of jurisdiction over the Property, the Borrower or any Loan Party by any court empowered to exercise bankruptcy powers. Borrower is not presently under any cease or desist order or other orders of a similar nature, temporary or permanent, of any Governmental Authority that would have the effect of preventing or hindering performance of its duties under this Agreement, any other Borrower Loan Documents or any Funding Loan Documents, nor are there any proceedings presently in progress or to its knowledge contemplated that would, if successful, lead to the issuance of any such order.
- **6.4 Other Liens**. Borrower has made no contract or arrangement of any kind, the performance of which by the other party thereto would give rise to a lien on the Property, except for its arrangements with the Architect, the Contractor or the subcontractors if there is no Contractor.
- **6.5 Leases**. All Leases are in full force and effect, there are no defaults under any of the provisions thereof by any party thereto, and all conditions to the effectiveness or continuing effectiveness of the Leases required to be satisfied as of the date hereof have been satisfied.
- 6.6 Financial Statements. The Financial Statements delivered to Bank by Borrower and any Loan Party are true and correct in all material respects, have been prepared in accordance with accounting practices and principles acceptable to Bank and consistently applied, and fairly present the financial condition(s) of the Person(s) referred to therein as of the respective dates; no materially adverse change has occurred in the financial condition reflected in any such financial statement since the date shown thereon, and no additional material liabilities have been incurred by any such Person since the date thereof

other than the borrowing contemplated hereby or other borrowing disclosed in writing to and approved by Bank.

- **6.7 Compliance With Laws**. The Property and the actual use thereof by Borrower complies in all material respects with all Governmental Requirements. Borrower has received no notices of violations of any Governmental Requirement.
- **6.8 Permits, Approvals, Licenses**. Borrower has obtained all licenses, permits and approvals necessary for the ownership, construction operation and management of the Property, including all approvals essential to the transactions contemplated by this Agreement, the Funding Loan Documents, the Borrower Loan Documents and any other documents contemplated hereby or thereby
- **6.9 Ownership of Real Property**. Borrower has, or as of the Closing Date will have, and will continue to have fee simple title to Real Property, subject only to the Permitted Liens. The Borrower is the sole borrower under the Borrower Loan. Borrower shall make no changes to the Property, when it is built, or to the operation thereof that would affect the qualification of the Property under the Act. The Borrower intends to utilize the Property as multifamily rental housing during the Qualified Project Period (as defined in the Tax-Exempt Regulatory Agreement).
- **6.10** Ownership of Personal Property. Borrower owns directly all of the Personal Property free and clear of all liens, encumbrances and adverse claims and the security interest of Bank in the Personal Property shall be a first lien thereon.
- **6.11 Other Financing**. Except for the Subordinate Loans and the HCD Loan or as otherwise disclosed in writing to Bank and approved by Bank in writing prior to the Closing Date, Borrower has not received other financing for either the acquisition of the Property or the construction and installation of the Improvements.
- **6.12 Plans, Defects**. The Plans are satisfactory to Borrower, and to the extent required by any Governmental Requirement or any effective restrictive covenant, have been approved by all applicable Governmental Authorities and the beneficiaries of any such covenant respectively; the Plans so approved have been approved by Borrower and Contractor as set forth in the Certification of Plans and Specifications delivered to Bank by Borrower.
- **6.13 Utilities**. All utility services necessary for the construction of the Improvements and the operation thereof for their intended purpose are either available at the boundaries of the Real Property or all necessary steps have been taken by Borrower and applicable Governmental Authorities to assure the complete construction and installation thereof, including water supply, storm drain and sanitary sewer facilities, and gas, electric, cable and telephone facilities.
- **6.14 Roads**. All roads necessary for the full use of the Improvements for their intended purposes have been completed or the necessary rights-of-way therefore have either been acquired by the applicable Governmental Authority or dedicated to public use and accepted by such Governmental Authority. All necessary steps have been taken by Borrower and such Governmental Authority to assure the complete construction thereof.
- **6.15 CC&Rs, Zoning**. Borrower has examined, is familiar with, and the Improvements will in all respects conform to and comply with, all covenants, conditions, restrictions, reservations and zoning ordinances affecting the Property.
- **6.16 Finder's Fees**. Borrower has not dealt with any Person who is or may be entitled to any finder's fee, brokerage commission, loan commission or other sum in connection with the execution of this Agreement, consummation of the transactions contemplated hereby, or the making of the Borrower Loan to Borrower.

- **6.17 Draw Request**. Each Draw Request shall be true, complete and accurate and the submission of same shall constitute a reaffirmation of the representations, warranties and covenants contained herein.
- 6.18 Other Information. No information, statement or report furnished in writing to Governmental Lender or Bank by Borrower, any Loan Party or any of their respective representatives in connection with this Agreement, the Funding Loan Documents or the other Borrower Loan Documents or the consummation of the transactions contemplated hereby and thereby (including, without limitation, any information furnished by Borrower in connection with the preparation of any materials related to the issuance, delivery or offering of the Funding Loan Note) contains any material misstatement of fact or omits to state a material fact necessary to make the statements contained therein, in the light of the circumstances under which they were made, not misleading; and the representations and warranties of Borrower and the statements, information and descriptions contained in Borrower's closing certificates, as of the Closing Date, are true, correct and complete, do not contain any untrue statement or misleading statement of a material fact, and do not omit to state a material fact required to be stated therein or necessary to make the certifications, representations, warranties, statements, information and descriptions contained therein, in the light of the circumstances under which they were made, not misleading; and the estimates and the assumptions contained herein and in any certificate of Borrower delivered as of the Closing Date are reasonable and based on the best information available to Borrower.
- **6.19 No Default**. No event has occurred and no condition exists with respect to Borrower, any Loan Party or the Property that would constitute an Event of Default or with the giving of notice or passage of time, or both, if not cured would become an Event of Default.
- **6.20** Tax Certificate. Borrower has complied with all terms and conditions of the Tax Certificate, including the terms and conditions of the exhibits thereto, and the representations set forth in the Tax Certificate pertaining to Borrower and the Property are true and accurate.
- **6.21** Regulatory Agreement. Borrower is not in default under the Regulatory Agreements. The Property is, as of the Closing Date, in compliance with all requirements of the Regulatory Agreements, including all applicable requirements of the Act and the Code. Borrower intends to cause the residential units at the Property to be rented or available for rental on a basis that satisfies the requirements of the Regulatory Agreements, including all applicable requirements of the Act and the Code. All Leases will comply with all Governmental Requirements and the Regulatory Agreements. The Property meets the requirements of this Agreement, the Regulatory Agreements, the Act and the Code with respect to multifamily rental housing.
- **6.22 No Governmental Lender Relationships**. To the best knowledge of Borrower, no member, officer, agent or employee of Governmental Lender has been or is in any manner interested, directly or indirectly, in that Person's own, name or in the name of any other Person, in the Funding Loan Note, the Funding Loan Documents, the Borrower Loan Documents, Borrower, any Loan Party or the Property, in any contract for property or materials to be furnished or used in connection with the Property, or in any aspect of the transactions contemplated by the Funding Loan Documents or the Borrower Loan Documents.
- **6.23** Authorizations and Consents. No authorization, consent, approval, order, registration declaration or withholding of objection on the part of or filing of or with any Governmental Authority not already obtained or made (or to the extent not yet obtained or made Borrower has no reason to believe that such authorizations, consents, approvals, orders, registrations or declarations will not be obtained or made in a timely fashion) is required for the execution and delivery or approval, as the case may be, of this Agreement, the Funding Loan Documents, the Borrower Loan Documents or any other documents contemplated by this Agreement, the Funding Loan Documents or the Borrower Loan Documents, or the performance of the terms and provisions hereof or thereof by the Borrower.

- **6.24 No Reliance**. Borrower acknowledges, represents and warrants that it understands the nature and structure of the transactions relating to the financing of the Property; that it is familiar with the provisions of all of the documents and instruments relating to such financing to which it or Governmental Lender is a party or of which it is a beneficiary including, without limitation, the Funding Loan Agreement; that it understands the risks inherent in such transactions, including, without limitation, the risk of loss of the Property; and that it has not relied on the Governmental Lender or Bank for any guidance or expertise in analyzing the financial or other consequences of the transactions contemplated by this Agreement, the Funding Loan Agreement or otherwise relied on Governmental Lender, Bank or Bank in any manner.
- 6.25 Environmental Matters. Borrower has not received any notice that it or the Property is not in compliance with all provisions of the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended ("CERCLA"); the Resource Conservation and Recovery Act; the Superfund Amendments and Reauthorization Act of 1986; the Toxic Substances Control Act and all environmental laws of the State (collectively "Environmental Laws"), or with any rules, regulations and administrative orders of any Governmental Authority, or with any judgments, decrees or orders of any court of competent jurisdiction with respect thereto; and Borrower has not received any assessment, notice (primary or secondary) of liability or financial responsibility, and no notice of any action, claim or proceeding to determine such liability or responsibility, or the amount thereof, or to impose civil penalties with respect to a site listed on any federal or state listing of sites containing or believed to contain "hazardous materials" (as defined in the Environmental Laws), nor has Borrower received notification that any hazardous substances (as defined under CERCLA) that it has disposed of have been found in any site at which any governmental agency is conducting an investigation or other proceeding under any Environmental Law.
- **6.26 ERISA**. Borrower has not received any notice that it is not in full compliance with the Employment Retirement Income Security Act of 1974, as amended, and the Department of Labor regulations thereunder, with the Code and with terms of such plan or plans with respect to each pension or welfare benefit plan to which Borrower is a party or makes any employer contributions with respect to its employees, for the current or prior plan years of such plans.
- **6.27 Funding Loan Note**. The weighted average maturity of the Funding Loan Note does not exceed 120% of the average reasonably expected economic life of the Property financed with the proceeds of the Borrower Loan. The Funding Loan Note is not and shall not be "federally guaranteed" as defined in Section 149(b) of the Code. Borrower intends to hold the Property for its own account and has no current plans to sell and has not entered into any agreement to sell all or any portion of the Property.
- **6.28** Tax Credit Allocation Documents Effective. The Tax Credit Allocation Documents are in full force and effect and have not been revoked, amended or modified in any way. Borrower knows of no reason why Project Completion could not occur on or before the Completion Date.
- 6.29 Satisfaction of Conditions under Tax Credit Allocation Documents, Subordinate Documents, Density Bonus Restrictions, AHAP Contract and HAP Contract. Each and every covenant, condition and obligation contained in the Tax Credit Allocation Documents, the Subordinate Documents, the Density Bonus Restrictions, the AHAP Contract and the HAP Contract required to be performed or satisfied as of the date hereof, and each and every matter required to be approved thereunder as of the date hereof, has been satisfied or approved, as applicable.
- **6.30** Tax Credits Not Subject to State Ceiling. Fifty Percent (50%) or more of the aggregate basis of the Improvements and Borrower's interest in the Property will be financed with proceeds from the Funding Loan Note and, therefore, paragraph (1) of Section 42(h) of the Code will not apply to the Tax Credits by virtue of the provisions set forth in subparagraph (4)(B) of Section 42(h) of the Code.
- **6.31 Standard Agreement.** The Standard Agreement is unmodified, in full force and effect, and all conditions to the effectiveness or continuing effectiveness of the Standard Agreement required to be satisfied by the date hereof have been satisfied.

- **6.32** Additional Representations and Warranties. Borrower also makes the representations and warranties set forth in Section 5.5.4 of the Special Conditions attached hereto as Exhibit C.
- **7. BORROWER'S COVENANTS**. Borrower covenants and agrees with Governmental Lender and Bank that until the full and final payment of all sums owed under the Borrower Loan Documents and the Funding Loan Documents, unless Bank waives compliance in writing:
- **7.1 Application of Advances**. Borrower shall receive the Advances made hereunder in trust, strictly for the purpose of paying the costs identified in the request for such Advance.
- **7.2 Borrower's Funds**. At the time and in amounts required by Bank, Borrower shall deposit Borrower's Funds into the Borrower's Funds Account. Should it appear at any time in Bank's judgment that the sum of undisbursed proceeds of the Borrower Loan and the then balance of the Borrower's Funds Account are insufficient to provide the financing for completion of the Improvements, Borrower shall pay to Bank, within ten days following receipt of written demand by Bank, an amount equal to such deficiency for deposit into the Borrower's Funds Account.
- **7.3** Lien Priority. At Borrower's sole cost and expense, Borrower shall maintain the Deed of Trust as a first lien on the Property.

#### 7.4 Construction Start and Completion.

- **7.4.1** Borrower shall not commence construction of the Improvements, including, but not limited to, grading and site clearance, and shall not undertake any other act on the Real Property prior to recordation of the Deed of Trust, the result of which would cause any mechanics' or materialmen's lien thereafter filed to take priority over the lien of the Deed of Trust, unless prior arrangements satisfactory to both Bank and Title Insurer have been made.
- **7.4.2** Borrower shall cause construction of the Improvements to be commenced not more than 30 days after the recordation of the Deed of Trust.
- **7.4.3** Borrower shall cause (a) the Improvements to be constructed in a good and workmanlike manner, with materials of high quality, and in accordance with the Plans, Governmental Requirements and sound building and engineering practices, (b) the construction of the Improvements to be prosecuted with diligence and continuity and completed in accordance with the Plans and to otherwise cause Project Completion to occur on or before the Completion Date, free and clear of liens or claims for liens, and (c) all licenses and permits necessary for the occupancy, use or sale of the Improvements to be issued. Borrower shall promptly commence and diligently proceed with the Project.
- **7.4.4** Borrower shall complete the construction of the Improvements on or before the Completion Date. The construction of the Improvements shall be considered complete for purposes of this Agreement only when (a) the construction of the Improvements has been completed substantially in accordance with the Plans and has been fully paid for subject to Borrower's obligations to pay and discharge or cause the release of any mechanics' lien, (b) all work requiring inspection or certification by any Governmental Authority has been completed and all requisite certificates, approvals and other necessary authorizations (including any required certificates of occupancy) have been obtained, and (c) streets and offsite utilities located within or pertaining to the Property have been completed to the satisfaction of all applicable authorities.

#### 7.5 Change Orders.

**7.5.1** Borrower shall not permit any change in the Plans without Bank's prior consent if any such change (a) constitutes a material change in material or equipment specifications, architectural or structural design, or the value or quality of the Improvements, or (b) would result in an increase or decrease

in the cost of construction of the Improvements in excess of the Single Change Order Limit for any single change, or in excess of the Aggregate Change Order Limit for all changes.

- **7.5.2** Borrower shall submit any proposed change in the Plans to Bank not later than ten Business Days prior to the commencement of construction relating to such change.
- **7.5.3** Borrower shall deliver to Bank in connection with any proposed change requiring Bank's prior written consent (a) a written request therefor, together with working drawings and a written description of the proposed change, submitted on a change order form acceptable to Bank and executed by Borrower, Architect and Contractor, and (b) evidence satisfactory to Bank as to the cost and time necessary to complete the proposed change.
- **7.5.4** Prior to permitting any change in the Plans requiring Bank's consent, Borrower shall satisfy any condition of Bank's consent, including, but not limited to, depositing funds to cover any increased Construction Costs into the Borrower's Funds Account as required by Bank, which Bank is authorized to disburse in accordance with the Project Budget and the Disbursement Schedule for payment of such Change Orders upon completion of such changes to Bank's satisfaction.
- **7.6 Detailed Cost Breakdown**. Borrower shall not modify the Project Budget or the Detailed Cost Breakdown without Bank's prior written consent, which consent may be conditioned upon, among other things, (a) Bank's receipt of evidence satisfactory to Bank that the change in the Project Budget or the Detailed Cost Breakdown is reasonably necessary, and (b) Bank's confirmation that, in the opinion of Bank, sufficient funds remain in the undisbursed proceeds of the Borrower Loan (and in the Borrower's Funds Account, if any) to pay for all remaining direct or indirect costs to complete construction of the Improvements.
- **7.7 Contractor Covenants**. Borrower shall (a) require from the Contractor (i) covenants similar to the covenants made by Borrower in Sections 7.3, 7.4 and 7.5, and (ii) a covenant that Contractor will, upon request, deliver to Bank the names of all Persons with whom Contractor has contracted or intends to contract for construction of the Improvements or for furnishing of labor or materials therefore; and (b) cause the Contractor (or if no Contractor, the subcontractors) to cooperate with Bank.
- **7.8** Construction Contract Only. Borrower shall not execute any contract or become party to any arrangement for the performance of work on the Real Property with any Person except Contractor, and if there is no Contractor, Borrower shall contract only with major subcontractors approved by Bank for the performance of work on the Real Property.
- **7.9 Paid Vouchers**. Borrower shall deliver to Bank, on demand, any contracts, bills of sale, statements, receipted vouchers or agreements under which Borrower claims title to any materials, fixtures or articles incorporated in the Improvements.
- **7.10 Application of Disbursements**. Borrower shall receive the disbursements to be made hereunder in trust, strictly for the purpose of paying the costs identified in the request for such disbursement.
- **7.11 Foundation Completion**. Borrower shall notify Bank immediately upon completion of the foundation of the Improvements and, if required by Bank, deliver to Bank, promptly after completion of the foundation, a foundation survey in form satisfactory to Bank and Title Insurer.
- **7.12 Personal Property Installation**. Without Bank's written consent, Borrower shall not install materials, personal property, equipment, or fixtures subject to any security agreement or other agreement or contract giving any Person other than Borrower any right or title to such property.
- **7.13 Defect Corrections**. Upon demand of Bank, Borrower shall correct any defect in the Improvements or any departure from the Plans not approved by Bank.

- **7.14 Stop Notices; Mechanic's Liens**. If (a) a bonded stop notice is received by Bank that Bank believes requires the withholding of funds from any Advance or from any disbursement of proceeds from the Borrower's Funds Account, or (b) a mechanics' lien, material supplier's lien or other construction lien is recorded against the Real Property, then Borrower shall within 20 days of such receipt or recordation or within five days of Bank's demand (whichever first occurs):
  - 7.14.1 pay and discharge same;
- **7.14.2** effect the release of same by recording a surety bond in sufficient form and amount issued by a surety acceptable to Bank; or
- **7.14.3** provide Bank with such other assurance as Bank, in its sole discretion, deems to be satisfactory for the payment of, and protection of Bank from, such lien or bonded stop notice.
- **7.15** Record Keeping, Financial and Other Information. Borrower shall keep and maintain full and complete books of account and other records reflecting the results of operations of the Property in accordance with accounting practices and principles acceptable to Bank and consistently applied, and shall furnish or cause to be furnished to Bank such financial information concerning Borrower, each Loan Party and the Property as Bank may require, including but not limited to:
- **7.15.1** within 45 days after the close of each quarter, except for the final quarter of each year, Borrower's and Guarantor's Financial Statement as of the close of such period,
- **7.15.2** within 180 days of the close of each fiscal year-end, the annual Financial Statements for Borrower and each Loan Party,
- **7.15.3** within 30 days after written request by Bank, a copy of the most recent filed Federal income tax returns for Borrower and each Loan Party, together with all supporting schedules,
- **7.15.4** within 30 days after written request by Bank, the Financial Statements of all affiliates and subsidiaries of Borrower and each Loan Party,
- **7.15.5** within 45 days after the close of each quarter, including the final quarter of each year, a certified statement of Liquid Assets for Guarantor,
- **7.15.6** within 30 days of the final quarter of each year, a projected cash flow statement for the next succeeding calendar year for Guarantor, and
- **7.15.7** Promptly, upon request, any other financial information requested by Governmental Lender and Bank.
- **7.16 Post-Construction Financial Reporting**. Upon completion of construction of the Improvements, Borrower shall furnish to Bank, without prior request or demand:
- **7.16.1** Within thirty (30) days after the close of each calendar month prior to Conversion and, thereafter within thirty (30) days of written request by Bank, a monthly or quarterly (as applicable) Operating Statement, a current rent roll and, if retail property, a schedule of gross sales; and
- **7.16.2** Within thirty (30) days of written request by Bank, an annual Operating Statement for the preceding operating year for the Property; provided that Borrower shall not be obligated to deliver such annual Operating Statement earlier than one hundred eighty (180) days after the close of the preceding operating year.
- **7.17** Audit and Inspection Rights. Borrower shall permit any representative of Governmental Lender or Bank, at any reasonable time, to inspect, audit and examine and copy the books and records of Borrower and each Loan Party.

- **7.18 Dividends, Distributions**. Following the occurrence and during the continuance of an Event of Default, Borrower shall not (a) make any distribution either in cash, stock or any other property, (b) redeem, retire, repurchase or otherwise acquire any shares or interest in Borrower, or (c) repay any outstanding indebtedness or other advance to any shareholder, partner, member or, if a trust, any trustor or beneficiary of Borrower.
- **7.19 Payment of Lawful Claims**. Borrower shall pay or discharge all lawful claims, including taxes, assessments and governmental charges or levies imposed upon Borrower or Borrower's income or profits or upon any property belonging to Borrower prior to the date upon which any penalties attach; provided that Borrower shall not be required to pay any such tax, assessment, charge or levy, the payment of which is being contested in good faith and by proper proceedings and for which Borrower is maintaining adequate reserves in accordance with generally accepted accounting principles.
- **7.20** Payment of Costs. Borrower shall pay all costs and expenses incurred by Bank in connection with the enforcement by Bank of any of Borrower's obligations under this Agreement or the other Borrower Loan Documents, and the preparation of this Agreement and the other Borrower Loan Documents, including but not limited to (a) all appraisal fees, cost engineering and inspection fees, legal fees and expenses (including the fees and costs of in-house counsel and legal staff), accounting fees, environmental consultant fees and costs of title insurance, survey, seismic, escrow and other fees and charges, and (b) all taxes and recording expenses, including stamp taxes, if any.
- **7.21** Approval of Easements and Other Documents. Borrower shall submit to Bank for Bank's approval all prospective easements, private or public dedications, and declarations of covenants, conditions and restrictions intended to affect the Real Property and Bank's approval shall be obtained in writing prior to the execution or granting thereof by Borrower. Borrower's request for approval of any prospective easement or private or public dedication shall be accompanied by a drawing or survey showing the precise location of such prospective easement or private or public dedication. Borrower's request for approval of any prospective declaration of covenants, conditions and restrictions shall be accompanied by a description of the property affected thereby.
- **7.22** Compliance with Laws; Preservation of Rights. Borrower shall comply promptly with all Governmental Requirements, and shall obtain, preserve and maintain in good standing, as applicable, all rights, privileges and franchises necessary or desirable for the operation of the Property and the conduct of Borrower's business thereon and therefrom. If payment of the indebtedness secured by the Deed of Trust or any of the other Security Documents is to be insured or guaranteed by any governmental agency, Borrower shall comply with all rules, regulations, requirements and statutes relating thereto or provided in any commitment issued by any such agency to insure or guarantee payment of such indebtedness.
  - 7.23 Notices. Borrower shall promptly notify Bank and the Governmental Lender in writing of:
    - **7.23.1** the occurrence of any Event of Default;
- **7.23.2** any litigation affecting Borrower, any Loan Party or the Property, or any other circumstance, event or occurrence that may reasonably be expected to result in a material adverse change in (a) the financial condition of Borrower or any Loan Party, (b) Borrower's ability to timely perform any of Borrower's obligations under any of the Borrower Loan Documents and the Funding Loan Documents, (c) the physical condition or operation of the Property; or (d) the tax exempt status of the interest payable on the Funding Loan Note. and
- **7.23.3** any notice that the Improvements or construction thereof, the Property or Borrower's business fails in any respect to comply with any applicable Governmental Requirement.
- **7.23.4** any notice of default under the Regulatory Agreements, the Subordinate Documents, the Density Bonus Restrictions, the HCD Documents, the AHAP Contract or the HAP Contract.

## 7.24 Indemnity.

- **7.24.1** Borrower shall indemnify, defend and hold each of the Indemnified Parties harmless from and against any and all liabilities, claims, actions, proceedings, damages, costs and expenses (including all attorney's fees, including, but not limited to, the fees and costs of any of such party's in-house counsel and legal staff) arising out of or resulting from:
  - (a) The Borrower Loan, the Borrower Loan Documents, the Funding Loan Documents, the Regulatory Agreements, the Subordinate Documents, the Density Bonus Restrictions, the HCD Documents, the AHAP Contract, the HAP Contract or the execution or amendment or performance thereof or in connection with the transactions contemplated therein, including the issuance, sale and/or resale of the Funding Loan Note.
  - **(b)** Any finder's fee, brokerage commission, loan commission or other sum in connection with the consummation of the transactions contemplated hereby.
  - **(c)** The development of the Property, construction of the Improvements or the ownership, operation or use of the Property
  - (d) Any declaration of taxability of interest on the Funding Loan Note, or allegations (or regulatory inquiry) that interest on the Funding Loan Note is taxable, for federal tax purposes.
  - **(e)** The issuance any Set Aside Letter, whether such matters are based on theories of derivative liability, comparative negligence or otherwise, at Borrower's own cost and with counsel approved by Bank, unless Bank elects to conduct its own defense at the expense of Borrower.
  - (f) Any act or omission of the Borrower or any of its agents, contractors, servants, employees or licensees in connection with the Borrower Loan or the Project, the operation of the Project, or the condition, environmental or otherwise, occupancy, use, possession, conduct or management of work done in or about, or from the planning, design, acquisition, installation or construction of, the Project or any part thereof.
  - (g) Any lien or charge upon payments by the Borrower to the Governmental Lender and/or the Bank hereunder, or any taxes (including, without limitation, all ad valorem taxes and sales taxes), assessments, impositions and other charges imposed on the Governmental Lender in respect of any portion of the Project.
  - **(h)** Any violation of any environmental law, rule or regulation with respect to, or the release of any toxic substance from, the Project or any part thereof.
    - (i) The defeasance and/or redemption, in whole or in part, of the Funding Loan Note.
  - (j) Any untrue statement or misleading statement or alleged untrue statement or alleged misleading statement of a material fact by the Borrower contained in any closing certificate, offering statement or disclosure or continuing disclosure document for the Funding Loan Note or any of the documents relating to the Funding Loan Note to which the Borrower is a party, or any omission or alleged omission from any offering statement or disclosure or continuing disclosure document for the Funding Loan Note of any material fact necessary to be stated therein in order to make the statements made therein by the Borrower, in the light of the circumstances under which they were made, not misleading.
  - **(k)** The Bank's acceptance or administration of the Funding Loan Agreement, or the exercise or performance of any of its powers or duties as Bank thereunder or under any of the documents relating to the Funding Loan Note to which it is a party.

- **7.24.2** The liability of Borrower under this indemnity shall not be limited or impaired in any way by (a) the release, reconveyance, foreclosure or other termination of the Deed of Trust, the payment in full of the Borrower Loan, any bankruptcy or other bankruptcy proceeding, or any other event whatsoever; (b) any provision in the Borrower Loan Documents or the Funding Loan Documents or applicable law limiting Borrower's liability or any Indemnified Party's recourse or rights to a deficiency judgment; or (c) any change, extension, release, inaccuracy, breach or failure to perform by any party under the Borrower Loan Documents or the Funding Loan Documents. Borrower's liability hereunder is direct and primary and not secondary as a guarantor or surety.
- **7.24.3** This indemnity is not intended to give rise to, and shall not give rise to, a right of Bank to claim payment of the principal and accrued interest with respect to the Borrower Loan as a result of an claim under this Section 7.24.
- **7.24.4** In the event that any action or proceeding is brought against any Indemnified Party with respect to which indemnity may be sought hereunder, Borrower, upon written notice from such Indemnified Party, shall assume the investigation and defense thereof, including the employment of counsel selected by such Indemnified Party, and shall assume the payment of all expenses related thereto, with full power to litigate, compromise or settle the same in its sole discretion; provided that such Indemnified Party shall have the right to review and approve or disapprove any such compromise or settlement. Each Indemnified Party shall have the right to employ separate counsel in any such action or proceeding and participate in the investigation and defense thereof, and Borrower shall pay the reasonable fees and expenses of such separate counsel; provided, however, that such Indemnified Party may only employ separate counsel at the expense of Borrower if, in the judgment of such Indemnified Party, a conflict of interest exists by reason of common representation or if all parties commonly represented do not agree as to the action (or inaction) of counsel.
- **7.24.5** Notwithstanding any transfer of the Property to another owner in accordance with the provisions of this Agreement, Borrower shall remain obligated to indemnify each Indemnified Party pursuant to this Section 7.24 if such subsequent owner fails to indemnify any party entitled to be indemnified hereunder, unless such Indemnified Party has consented to such transfer and to the assignment of the rights and obligations of Borrower hereunder.
- **7.24.6** The rights of any persons to indemnity hereunder and to payment of fees and reimbursement of expenses pursuant to this Agreement shall survive the final payment or defeasance of the Borrower Loan and the Funding Loan Note and, in the case of Bank, any resignation. The provisions of this Section 7.24 shall survive the termination of this Agreement.
- **7.25 Performance of Acts**. Upon request by Bank, Borrower shall perform all acts that may be necessary or advisable to perfect any lien or security interest provided for in the Borrower Loan Documents or the Funding Loan Documents to carry out the intent of the Borrower Loan Documents or the Funding Loan Documents.
- **7.26 Notice of Change**. Borrower shall give Bank prior written notice of any change in the location of Borrower's place of business (or Borrower's chief executive office if Borrower has more than one place of business) or Borrower's name, business structure or place of incorporation or other formation, and, unless otherwise approved by Bank in writing, Borrower shall maintain all tangible Personal Property (other than the books and records) at the Real Property and all books and records at Borrower's place of business (or chief executive office if Borrower has more than one place of business).
- **7.27 Tax Certificate**. Borrower shall timely comply with all of its obligations under the Tax Certificate (which Tax Certificate is hereby incorporated herein as fully as if set forth at length herein).
- **7.28 Funding Loan Documents**. Borrower shall timely perform its obligations under the Funding Loan Documents.

- **7.29 Regulatory Agreements**. Borrower hereby covenants and agrees (a) to comply with all provisions of the Regulatory Agreements; to advise Bank and Governmental Lender in writing promptly upon learning of any default with respect to the covenants, obligations and agreements of Borrower set forth in the Regulatory Agreements; (b) upon written direction by Governmental Lender, to cooperate fully and promptly with Governmental Lender in enforcing the terms and provisions of the Tax-Exempt Regulatory Agreement; and (c) to file in accordance with the time limits established by the Regulatory Agreements all reports and certificates required thereunder, and the Certification to the Secretary of the Treasury required by the Tax-Exempt Regulatory Agreement. Neither Governmental Lender nor Bank shall incur any liability in the event of any breach or violation of any of the Regulatory Agreements by Borrower, and Borrower agrees to indemnify the Indemnified Parties from any claim or liability for any such breach under the Regulatory Agreements.
  - 7.30 Prohibited Activities. Without Bank's prior written consent Borrower shall not:
- **7.30.1** Engage in any business activities substantially different from Borrower's present business or liquidate or dissolve Borrower's business;
- **7.30.2** Suffer or permit any liens or encumbrances to be placed on the Property other than the Permitted Liens.
- 7.30.3 Transfer any interest in the Property (other than the lease of residential units within the Property for a term of one-year or less and otherwise in compliance with the Regulatory Agreements and dispositions of Personal Property expressly permitted by the Borrower Loan Documents) without the prior written consent of Bank, which consent may be withheld in Bank's absolute discretion. In connection with the foregoing consent requirements, Borrower acknowledges that Bank relied upon Borrower's particular expertise in entering into this Agreement and continues to rely on such expertise to ensure the satisfactory completion and operation of the Property. Transfers requiring Bank's prior written consent shall include, without limitation, (a) involuntary transfers and transfers by operation of law; (b) liens and assignments as security for obligations, whether voluntary or involuntary; and (c) except as otherwise expressly permitted by the terms of the Deed of Trust, the issuance, sale, assignment, disposition, encumbering or other transfer of any direct or indirect ownership interest in Borrower, any Loan Party or any general partner, member or shareholder of any Loan Party, whether voluntary or involuntary, by operation of law or otherwise. No sale, lease or other transfer shall relieve Borrower from primary liability for its obligations under the Borrower Loan Documents or the Funding Loan Documents, and Borrower shall deliver to Bank all documents reasonably required by Governmental Lender to evidence its continuing liability. No consent by Bank in connection with any Transfer shall constitute (x) a consent by Governmental Lender under the Tax-Exempt Regulatory Agreement to any sale, assignment, encumbrance, transfer or other disposition of all or any part of the Property, or any direct or indirect interest therein, or (y) a waiver by Governmental Lender of any term or condition of the Tax-Exempt Regulatory Agreement. Notwithstanding the foregoing, (a) Tax Credit Investor may transfer its limited partnership interests in Borrower to any limited partnership or limited liability company in which Tax Credit Investor or an affiliate thereof is the general partner or managing member: provided that following such transfer. Tax Credit Investor shall remain jointly and severally liable for all contributions to be made by Tax Credit Investor under the Partnership Agreement, (b) the transfer of limited partnership interests or non-managing membership interest in Tax Credit Investor shall not constitute a "transfer" hereunder, and (c) subject to Bank's consent, which shall not be unreasonably withheld. Tax Credit Investor may remove and replace the General Partner in accordance with the Partnership Agreement following a default by the General Partner thereunder.
- **7.30.4** Amend or modify in any material respect any organizational documents pertaining to Borrower or any Loan Party.
- **7.30.5** Cause or otherwise consent to the formation of any community facilities district that includes the Property or any part of the Property pursuant to the Mello-Roos Community Facilities Act of 1982, any assessment district that includes the Property or any part of the Property pursuant to the Municipal Improvement Act of 1913, or any other comparable or similar district, area or territory that includes

the Property or any part of the Property pursuant to any Law, or cause or otherwise consent to the levying of special taxes by any community facilities district against the Property or any part thereof, the levying of assessments by any such assessment district against the Property or any part thereof, or the levying of assessments, taxes and/or other impositions by any such district, area or territory.

- **7.30.6** Enter into any new Funding Loan Documents, Subordinate Documents or HCD Documents, or amend, modify, supplement, cancel or terminate any Funding Loan Documents, Subordinate Documents, the Density Bonus Restrictions, HCD Documents, the AHAP Contract (including the form of HAP Contract attached as an exhibit to the AHAP Contract) or the HAP Contract.
- **7.30.7** Take, or omit to take, any action that, if taken or omitted, would jeopardize or adversely affect the tax-exempt status of the interest payable on the Funding Loan Note.
- **7.30.8** Accept any deed or other restriction or enter into any regulatory or other similar agreement regulating or restricting the use or operation of the Property or restricting the tenant income and/or rent levels for the Property in connection with the allocation to the Property of federal low-income housing tax credits or otherwise.
- **7.31 Set Aside Letters**. In the event Bank issues, at Borrower's request, any Set Aside Letter, Borrower represents, warrants and agrees as follows:
- **7.31.1** The sum that Borrower requests Bank to allocate for Bonded Work shall be sufficient to pay for the costs of construction and completion of the Bonded Work in accordance with any agreement between Borrower and the Governmental Authority and a copy of such agreement shall be furnished to Bank by Borrower as a condition precedent to the issuance by Bank of any Set Aside Letter;
- **7.31.2** Bank is irrevocably and unconditionally authorized to disburse to the Governmental Authority or Surety all or any portion of proceeds of the Borrower Loan upon a demand of the Governmental Authority or Surety made in accordance with the terms and conditions of the Set Aside Letter;
- **7.31.3** Any disbursement or payments that Bank makes or may be obligated to make under any Set Aside Letter, whether made directly to the Governmental Authority, Surety, or to others for completion of all or part of the Bonded Work, shall be deemed an Advance to or for the benefit of Borrower; and
- **7.31.4** Bank shall have no obligation to release any security under the Borrower Loan Documents unless and until Bank has received a full and final written release of its obligations under each Set Aside Letter.
- **7.31.5** The fee for issuing each Set Aside Letter hereunder shall be determined when each Set Aside Letter is issued by Bank.
- **7.32 Management of Property**. Borrower shall not enter into any agreement providing for the management or operation of the Real Property or the Improvements without the prior written consent of Bank.

### 7.33 Leases.

**7.33.1 Negative Covenants**. In addition to the provisions of the Deed of Trust, and regardless of whether or not Bank's prior written approval is required, Borrower shall not, without Bank's prior written consent: (a) grant to any tenant any right or option to purchase the Property or any portion thereof, or any other present or future interest in any portion of the Property other than the right to use and occupy the leased premises, (b) grant to any tenant the right to terminate its lease if the lease of one or more other tenant is terminated, or (c) accept payment of rent from any tenant in any form other than cash or cash equivalent.

- **7.33.2** Affirmative Covenants. In addition to the provisions of the Deed of Trust, Borrower shall (a) document all Leases covering any portion of the Property or the Improvements on a standard lease form approved by Bank (with no material change), (b) not enter into any lease for any Unit with a potential tenant unless such lease is an Acceptable Unit Lease and the rent charged thereunder complies with the Subordinate Documents, the Density Bonus Restrictions, the HCD Documents, the HAP Contract and all Regulatory Agreements and is consistent with the rent proforma submitted by Borrower and approved by Bank (c) enter into Leases only with bona fide third party tenants in an arm's length transaction at the then current rates for comparable space in accordance with the Rent Restrictions and on such other terms and conditions as are reasonably acceptable to Bank, (d) whether or not Bank's prior written approval is required, deliver to Bank, within ten days of written request, all Leases (together will all financial information obtained by Borrower regarding the tenant) and all modifications, amendments and consents to assignment or subletting of existing Leases, and (e) promptly notify Bank in writing of (i) the termination, abandonment, or surrender of any Lease, and (ii) claims of any breach of any of Borrower's obligations as landlord under any Lease.
- **7.34 Compliance**. Upon the request of Bank from time to time and at any time certification of the matters set forth below is provided to Governmental Lender or any Governmental Authority, Borrower shall promptly provide to Bank the following:
- **7.34.1** Borrower's certification of the Property's compliance with the rules qualifying the interest payable on the Funding Loan Note for federal tax exemption pursuant to Section 142(d) of the Code and the regulations issued under Section 142(d) and the requirements of the Regulatory Agreements;
- **7.34.2** Property has received or receives a tax credit allocation, Borrower's certification of the Property's compliance with the requirements of Section 42 of the Code and the regulations issued under Section 42 and if the tax credits have not yet been syndicated, Borrower's report regarding progress in syndicating the tax credit allocation until the syndication is completed; and
- **7.34.3** Such other documents, certificates and other information as may be deemed necessary or appropriate to enable Bank to perform the functions under this Agreement or the Funding Loan Agreement.
- **7.35 Property Reserves**. Borrower shall establish and maintain such operating, replacement and/or tenant improvement reserves for the Property as required by Bank, and Borrower hereby grants to Bank a security interest in all such reserves. Borrower agrees to execute such supplemental security documentation as Bank may request confirming such security interest.
- **7.36 Standard Agreement**. Prior to or concurrently with the Conversion Date, Borrower shall comply with all conditions of the Standard Agreement, and shall execute all documents necessary to close the HCD Loan.

#### 7.37 Establishment of Capital Improvement Reserve Account.

7.37.1 Concurrently with the Conversion and as a condition precedent thereto, Borrower shall: (i) establish with Bank the Capital Improvement Reserve Account and Borrower shall execute such documents as are necessary to evidence same and to create and perfect in favor of Bank a security interest therein for the purpose of paying for any capital improvements which are necessary for the continued operation of the Property and which capital improvements are approved by Bank, which approval will not be unreasonably withheld ("Capital Improvements"); and (ii) commencing after the Conversion Date, on the first day of the month in which Borrower is required to make its first principal and interest payment under the Borrower Note, and continuing on the first day of every month thereafter, deposit or cause to be deposited into the Capital Improvement Reserve Account an amount equal to no less than [\$4,166.67] [CHECK] each month.

- 7.37.2 Borrower shall be entitled to withdraw funds from the Capital Improvement Reserve Account from time to time (but no more often than once every thirty (30) days and in an amount of no less than \$1.000 for each such withdrawal) to cover Capital Improvements, but only upon ten (10) days prior written notice from Borrower to Bank requesting to withdraw such funds and only so long as no Event of Default exists and no event has occurred that, with the giving of notice or the passage of time, or both, would constitute an Event of Default. Said written request shall set forth the amount of funds Borrower wishes to withdraw from the Capital Improvement Reserve Account, shall set forth with specificity those Capital Improvements for which the funds are to be used and shall be accompanied by copies of invoices or other evidence satisfactory to Bank confirming the cost of such Capital Improvements. Bank may also condition the withdrawal of funds from the Capital Improvement Reserve Account upon delivery by Borrower of such contractor's affidavits, owner's sworn statements, partial and final waivers of lien and other additional documentation Bank may require to insure that the Capital Improvements have been completed free and clear of any claims of lien, and in a good and workmanlike manner and otherwise in accordance with all applicable legal requirements. The disbursement of funds withdrawn from the Capital Improvements Reserve Account may be made, in Bank's discretion, either directly to the parties entitled thereto or to Borrower to pay the same. If such funds are disbursed directly to Borrower, Borrower shall provide Bank with evidence of the payment of the cost of the Capital Improvements within ten (10) days after the date such funds are withdrawn from the Capital Improvement Reserve Account.
- 7.37.3 Borrower shall diligently pursue completion of all Capital Improvements upon the commencement of the same. All Capital Improvements shall be made in a good and workmanlike manner and shall be completed free and clear of any mechanic's or materialman's liens and encumbrances. Borrower shall pay all costs necessary for completion of all Capital Improvements without regard to the sufficiency of the funds in the Capital Improvement Reserve Account. Borrower shall not commence construction of any Capital Improvement or other work prior to obtaining a building permit and all other governmental authorizations required with respect thereto, which Borrower shall provide to Bank upon request. Once any construction work has commenced, Borrower shall cause same to be completed in accordance with the plans and specifications therefor and in compliance with all restrictive covenants applicable thereto, free and clear of liens or claims for liens, and shall correct all defects therein. No disbursement of funds from the Capital Improvement Reserve Account shall constitute a waiver of Bank's right to require compliance with the foregoing covenants.
- **7.38 Rent Restrictions**. Borrower shall comply, and cause the tenants occupying the Units to comply, with the Rent Restrictions, including, without limitation, maintaining all appropriate records.
- **7.39** Preservation of Tax Credits. Borrower shall observe and perform all obligations imposed on Borrower for the purpose of obtaining, maintaining and utilizing the maximum amount of Tax Credits allocated pursuant to the Tax Credit Allocation Documents and to operate the Project, or to cause the appropriate parties to operate the Project, in accordance with all applicable provisions of the Code and the R&T Code, if applicable, and all other statutes and regulations governing the Tax Credits including, without limitation, the monitoring and reporting requirements set forth in the Qualified Allocation Plan.
- **7.40** Election of Credit Period. Borrower shall not make its election (electing whether the first year of the ten (10) year "credit period" shall commence in the year the Project is placed in service or the following year) under Section 42(f) of the Code without first obtaining Bank's prior written consent to Borrower's election, which consent shall not be unreasonably withheld or delayed.
- 7.41 Compliance with Subordinate Documents, Density Bonus Restrictions, HCD Documents, Regulatory Agreements, AHAP Contract and HAP Contract. Borrower shall observe and comply with all of the terms and conditions set forth in the Subordinate Documents, the Density Bonus Restrictions, the HCD Documents, all Regulatory Agreements, the AHAP Contract and the HAP Contract.
- **7.42** Payment of Development Fee. Borrower shall not pay Developer more than (i) [\$420,000] [CHECK] of its development fee in the aggregate on or prior to the Closing Date, (ii) [\$700,000] [CHECK] of its development fee in the aggregate on or prior to the date Bank determines fifty percent (50%)

of the Project has been completed, (iii) [\$840,000] **[CHECK]** of the development fee in the aggregate on or prior to the date Bank determines Project Completion has occurred, (iv) [\$5,442,935] **[CHECK]** of the development fee in the aggregate on or prior to the date the Project has achieved Qualified Occupancy, (v) [\$5,542,935] **[CHECK]** of its development fee in the aggregate prior to the Conversion Date, and (iii) [\$5,592,935] **[CHECK]** of its development fee in the aggregate prior to receipt of IRS Form 8609 from the Allocation Committee.

- **7.43 IRS Form 8609**. Borrower shall deliver to Bank the IRS Form 8609 within five (5) business days following Borrower's receipt of the same from the Allocation Committee.
- **7.44** Obtaining and Maintaining Real Property Tax Exemption. Borrower shall cause Developer to maintain its status as an "eligible non-profit corporation" (as such term is used in Section 214(g) of the R&T Code) and take all actions and provide such certifications as may be necessary from time to time so that the Project shall be exempt from the payment of real property taxes in accordance with the provisions of Section 214(g) of the R&T Code.
- 7.45 Draws under Subordinate Loan and Disbursement of Borrower's Funds. Borrower shall request and receive disbursements of [\$4,200,000] [CHECK] of the HCID HHH Loan (except for any holdback not exceeding [\$420,000] [CHECK] of the HCID HHH Loan) and the entire amount of the HCID HOPWA Loan, the HCID Recast Loan, the CDC Loan and the SNHP Loan (except for any holdback not exceeding ten percent (10%) of the CDC Loan and the SNHP Loan) prior to requesting disbursements of Borrower's Funds.
- **7.46 Draw Requests**. Borrower shall furnish to Bank such statements and other financial data as Bank shall from time to time reasonably request in writing with respect to disbursements made under the Subordinate Lender Loan, if any. Borrower shall deliver, or cause to be delivered, to Bank (concurrently with the delivery of the same to Subordinate Lender) copies of all draw requests (and accompanying back-up documentation), if any, submitted to the Subordinate Lender with respect to disbursements made under the Subordinate Lender Loan from time to time.
- 7.47 Progress Reports and Annual Project Status Reports; Allocation Committee Notices. Borrower shall promptly deliver to Bank copies of all "Progress Reports" all "Annual Project Status Reports" and all other reports delivered by Borrower to the Allocation Committee or Subordinate Lender from time to time including, without limitation, those reports required by the terms and conditions of the Qualified Allocation Plan or as otherwise required under the terms of the Tax Credit Allocation Documents; such reports shall be delivered to Bank concurrently with the delivery of the same to the Allocation Committee. Borrower shall promptly deliver to Bank copies of all notices and\or correspondence it receives from time to time from the Allocation Committee to the extent the same relate to the allocation of Tax Credits as evidenced by the Carryover Allocation.

### 7.48 Hedge.

7.48.1 As a condition precedent to making the Borrower Loan, the Borrower shall enter into one or more interest rate caps, collars, swaps, swaptions, forward swaps or similar transactions designed to protect against fluctuations in the interest rate of each Borrower Note commencing no later than the Outside Conversion Date and expiring no earlier than the Maturity Date with a counterparty acceptable to Bank (which counterparty may, but is not required to be, Bank) (together, as modified from time to time, the "Hedge"). The notional amount of the Hedge must be the outstanding principal amount of the Borrower Note as of the Outside Conversion Date or, if later, the effective date of the Hedge. The Hedge shall provide for a fixed rate of interest not to exceed (or otherwise protect against the interest rate on the Borrower Note exceeding) [\_\_\_%] [CHECK] (inclusive of the Margin applicable during the Permanent Phase). The cost of the Hedge must be paid in full on its effective date. The identity of the counterparty and the form and substance of the documents and agreements evidencing, securing, guarantying or otherwise governing the Hedge, including, without limitation, any ISDA Master Agreement and Schedule thereto, and any confirmations evidencing the Hedge (together, the "Hedge Documents"),

shall be acceptable to Bank in the Bank's sole discretion. In no event shall the counterparty have a rating by a national rating agency which is less than the rating assigned by such rating agency to Bank. No Hedge Document shall be secured by the Project unless expressly consented to in writing by Bank, which consent may be withheld in Bank's sole discretion.

- **7.48.2** On the Closing Date, the Borrower shall acquire Hedges complying with the requirements of this Section 7.48 and Sections 4.1.14 and 4.1.15. As a condition to the Conversion, the Hedge shall comply with the requirements of this Section 7.48 and subsection (t) of Exhibit D.
- **7.48.3** The Borrower shall timely perform all of its obligations under the Hedge Document in accordance with its terms, including payment of all breakage and termination fees due under the applicable Hedge Documents. Unless Bank is the counterparty, the Borrower may not exercise any right or remedy under any Hedge Document without the Bank's prior written consent and shall exercise its rights and remedies under the Hedge Documents as directed by the Bank in writing.
- 7.48.4 So long as the Borrower is required to maintain a Hedge, the Borrower shall not terminate, transfer or consent to any termination or transfer of the Hedge without the Bank's prior written consent, which consent may be withheld in Bank's sole discretion. No Hedge shall be terminated for any reason unless Borrower enters into a new Hedge complying with the requirements of this Section 7.48; provided, that no Hedge undertaken with Bank may be terminated, terminated and replaced or transferred by the Borrower without the consent of Bank, which consent may be withheld by Bank in its sole discretion Each replacement Hedge must have a term which commences no later than the later of the Outside Conversion Date or the termination date of the preceding Hedge. If Borrower desires to transfer or terminate a Hedge, Borrower shall provide Bank for Bank's approval written notice thereof at least sixty (60) days prior to termination of the existing Hedge, together with a description of the terms proposed for the replacement Hedge and the identity of the financial institutions who will bid to be the counterparty on the replacement Hedge. In addition, the Borrower shall provide the Bank for Bank's approval the identity of the counterparty and copies of the proposed replacement Hedge Documents at least fourteen (14) business days prior to the termination of the existing Hedge; provided, however, that if a Hedge unexpectedly and unavoidably terminates on a date other than its scheduled expiration date, the Borrower shall, within fourteen (14) business days of such termination, obtain a new Hedge satisfying the requirements of this Section 7.48; provided that if such terminated Hedge is one provided by Bank, Bank shall be under no obligation to permit such replacement Hedge to be entered into or to forbear from exercising its creditor remedies during such time.
- **7.48.5** If Bank is not (or is no longer) the counterparty to the Hedge, the Borrower shall assign each Hedge in effect from time to time to Governmental Lender and Bank pursuant to an assignment of hedge ("Assignment of Hedge") in a form and content acceptable to Bank in its sole discretion. The Assignment of Hedge must be entered into on or before the effective date of the Hedge. The Hedge Documents and the Assignment of Hedge shall direct the counterparty to make any payments on the Hedge directly to Bank to be applied by Bank to payments due under the Borrower Loan, provided that after the occurrence of an Event of Default, Bank may apply such payments as may determine in its discretion.
- **7.49** Funding of HCD Loan and HCID HHH Loan. Upon funding (partial or full) of the HCD Loan and the HCID HHH Loan (other than the initial [\$4,200,000] [CHECK] of the HCID HHH Loan), Borrower shall promptly deliver to Bank the net proceeds of such loan funding to paydown the outstanding balance of the Borrower Loan.
- **7.50 Establishment of Transition Reserve Account.** Concurrently with the Conversion and as a condition precedent thereto, Borrower shall: (i) establish with Bank the Transition Reserve Account and Borrower shall execute such documents as are necessary to evidence same and to create and perfect in favor of Bank a security interest therein for the purpose of paying for any Rental Assistance Operating Deficits as approved by Bank, which approval will not be unreasonably withheld; and (ii) deposit into the Transition Reserve Account the sum of [\$495,056] **[CHECK]** on or before the Conversion Date. Borrower shall be entitled to withdraw funds from the Transition Reserve Account from time to time (but no more

often than once every thirty (30) days and in an amount of no less than \$1,000 for each such withdrawal) to cover Rental Assistance Operating Deficits, but only upon ten (10) days prior written notice from Borrower to Bank requesting to withdraw such funds and only so long as no Event of Default exists and no event has occurred that, with the giving of notice or the passage of time, or both, would constitute an Event of Default. Said written request shall set forth the amount of funds Borrower wishes to withdraw from the Transition Reserve Account and shall be accompanied by an Operating Statement for the applicable period of time and such other evidence satisfactory to Bank confirming the amount of such Rental Assistance Operating Deficits.

- as a condition precedent thereto, Borrower shall: (i) establish with Bank the Operating Reserve Account and Borrower shall execute such documents as are necessary to evidence same and to create and perfect in favor of Bank a security interest therein for the purpose of paying for any Operating Deficits as approved by Bank, which approval will not be unreasonably withheld; and (ii) deposit into the Operating Reserve Account the sum of [\$286,617] [CHECK] on or before the Conversion Date. Borrower shall be entitled to withdraw funds from the Operating Reserve Account from time to time (but no more often than once every thirty (30) days and in an amount of no less than \$1,000 for each such withdrawal) to cover Operating Deficits, but only upon ten (10) days prior written notice from Borrower to Bank requesting to withdraw such funds and only so long as no Event of Default exists and no event has occurred that, with the giving of notice or the passage of time, or both, would constitute an Event of Default. Said written request shall set forth the amount of funds Borrower wishes to withdraw from the Operating Reserve Account and shall be accompanied by an Operating Statement for the applicable period of time and other evidence satisfactory to Bank confirming the amount of such Operating Deficits.
- Americans with Disabilities Act. Borrower hereby certifies that it and any contractor and 7.52 subcontractor will comply with the Accessibility Requirements (as defined in Exhibit I of the Tax-Exempt Regulatory Agreement). Borrower and any contractor and subcontractor will provide reasonable accommodations to allow qualified individuals with disabilities to have access to and to participate in its programs, services and activities in accordance with the applicable provisions of: (i) the Americans with Disabilities Act, 42 U.S.C. 12101 et seq., and its implementing regulations and the American Disabilities Act Amendments Act (ADAAA) Pub. L.110 325 and all subsequent amendments; (ii) Section 504 of the Rehabilitation Act of 1973, 29 U.S.C. §794, the implementing regulations at 24 C.F.R. Part 8, as well as the requirements of Uniform Federal Accessibility Standards 24 C.F.R §40, Appendix A; (iii) the Fair Housing Act, 42 U.S.C. §§3601-3620; 24 C.F.R. Parts 100, 103, and 104, and its implementing regulations; and (iv) applicable California building codes. Borrower and any contractor and subcontractor will not discriminate against persons with disabilities or against persons due to their relationship to or association with a person with a disability. Any contract and subcontract entered into by Borrower, relating to this Agreement, to the extent allowed hereunder, shall be subject to the provisions of this Section.
- **7.53 Business Tax Registration Certificate.** Subject to any exemption available to it, Borrower and Bank each represent that it has obtained or will obtain the Business Tax Registration Certificate(s) required by the Governmental Lender's Business Tax Ordinance (Article 1, Chapter 2, Section 21.00 and following, of the Los Angeles Municipal Code). For the term covered by this Agreement, Borrower and Bank each shall maintain, or obtain as necessary, any such Business Tax Registration Certificate(s) required of it under said Ordinance and shall not allow any such Business Tax Registration Certificate(s) to be revoked or suspended.
- 7.54 Child Support Assignment Orders. This Agreement is subject to Section 10.10 of the Los Angeles Administrative Code, Child Support Assignment Orders Ordinance. Pursuant to this Ordinance, Borrower certifies that it will (a) fully comply with all State and Federal employment reporting requirements applicable to Child Support Assignment Orders; (b) that the principal owner(s) of Borrower are in compliance with any Wage and Earnings Assignment Orders and Notices of Assignment applicable to them personally; (c) fully comply with all lawfully served Wage and Earnings Assignment Orders and Notices of Assignment in accordance with California Family Code Section 5230 et seq.; and (d) maintain such compliance throughout the term of the Tax-Exempt Regulatory Agreement. Pursuant to Section

10.10.b of the Los Angeles Administrative Code, failure of Borrower to comply with all applicable reporting requirements or to implement lawfully served Wage and Earnings Assignment Orders and Notices of Assignment or the failure of any principal owner(s) of Borrower to comply with any Wage and Earnings Assignment Orders and Notices of Assignment applicable to them personally shall constitute a default by Borrower as appropriate under the terms of this Agreement, subjecting Borrower to the remedies provided herein where, in either case, such failure shall continue for more than ninety (90) days after notice of such failure to Borrower by the Governmental Lender. Any subcontract entered into by Borrower relating to this Agreement, to the extent allowed hereunder, shall be subject to the provisions of this paragraph and shall incorporate the provisions of the Child Support Assignment Orders Ordinance. Failure of Borrower to obtain compliance of its subcontractors shall constitute a default by Borrower under the terms of this Agreement, subjecting Borrower to the remedies provided herein where such failure shall continue for more than 90 days after notice of such failure to Borrower by the Governmental Lender. Borrower shall comply with the Child Support Compliance Act of 1998 of the State of California Employment Development Department. Borrower hereby affirms that to the best of its knowledge it is fully complying with the earnings assignment orders of all employees, and is providing the names of all new employees to the New Hire Registry maintained by the Employment Development Department as set forth in subdivision (b) of Section 7110 of the Public Contract Code.

- 7.55 Nondiscrimination and Affirmative Action. Borrower shall comply with the applicable nondiscrimination and affirmative action provisions of the laws of the United States of America, the State of California, and the Governmental Lender. Borrower shall not discriminate in its employment practices against any employee or applicant for employment denial of family and medical care leave; denial of pregnancy disability leave or reasonable accommodations against any employee or applicant for employment because of such person's race, ancestry, color, citizenship, national origin, religion, sex, sexual orientation, gender identity/expression, transgender status, age, marital status, familial status, domestic partner status, physical handicap, mental disability, medical condition, political affiliation or belief. Borrower shall comply with Executive Order 11246, entitled "Equal Employment Opportunity", as amended by Executive Order 11375, and as supplemented in Department of Labor regulations (41 C.F.R. Part 60). Borrower shall comply with the provisions of the Los Angeles Administrative Code Sections 10.8 through 10.13, to the extent applicable hereto. The affirmative action program of Borrower shall include the mandatory contract provisions set forth in the Los Angeles Administrative Code Section 10.8.4, and said provisions are incorporated herein by this reference. Borrower shall also comply with all rules, regulations, and policies of the Governmental Lender's Board of Public Works, Office of Contract Compliance relating to nondiscrimination and affirmative action, including the filing of all forms required by the Governmental Lender. Any subcontract entered into by Borrower relating to this Agreement, to the extent allowed hereunder, shall be subject to the provisions of this section. No person shall on the grounds of race, ancestry, color, citizenship, national origin, religion, sex, sexual orientation, gender identity/expression, transgender status, age, marital status, familial status, domestic partner status, physical handicap, mental disability, medical condition, political affiliation or belief be excluded from participation in, be denied the benefit of, or be subjected to discrimination under this Agreement. For purposes of this Section, Title 24 Code of Federal Regulations Part 107 and Section 570.601(b) defines specific discriminatory actions that are prohibited and corrective action that shall be taken in a situation as defined therein.
- 8. EVENTS OF DEFAULT. The occurrence of any one or more of the following events shall constitute an "Event of Default" hereunder and at Bank's option, exercisable in its sole discretion, shall terminate any obligation of Bank to make any Advance or disbursement of Borrower's Funds. Upon the occurrence of an Event of Default, Bank shall also have the option, exercisable in its sole discretion, to declare the Borrower Loan immediately due and payable, without notice of default, presentment or demand for payment, protest or notice of nonpayment or dishonor, or other notices or demand of any kind or character; provided, however, upon the occurrence of any Event of Default that, under the terms of any Borrower Loan Document or Funding Loan Document results in the Borrower Loan becoming automatically due and payable, such occurrence shall result in automatic acceleration of payments of all principal and interest due under the Borrower Loan:

- **8.1** Borrower fails to (a) pay when due any sums payable under any Borrower Loan Document or Funding Loan Document after giving effect to any express curative provisions (if any) provided herein or therein, or (b) deposit with Bank any of Borrower's Funds as and when required under this Agreement.
- **8.2** Borrower has breached, or defaulted under, any term, condition or provision contained in (a) any Borrower Loan Document that is not specifically referred to in this Section 8, (b) the Tax-Exempt Regulatory Agreement or other Funding Loan Document, or (c) any ground lease, if the Property is a leasehold estate.
- **8.3** Borrower or Contractor does not (a) commence construction of the Improvements within the time period required in this Agreement, (b) proceed diligently and continuously with the construction of the Improvements, or the construction of the Improvements is otherwise discontinued for a period of five consecutive Business Days or more, for any reason, or (c) complete the construction of the Improvements and cause the issuance of all licenses and permits necessary for the occupancy and use of the Improvements, on or before the Completion Date.
- **8.4** Any representation or warranty by Borrower or any Loan Party made hereunder or under any other Borrower Loan Document proves to be materially false or misleading.
- **8.5** Any person obtains an order or decree in any court of competent jurisdiction prohibiting the construction of the Improvements or Borrower or Governmental Lender and Bank from performing this Agreement, and such order or decree is not vacated within ten days after the granting thereof.
- **8.6** Borrower neglects, fails or refuses to keep in full force and effect any permit or approval with respect to the construction of the Improvements or the use and occupancy thereof.
- **8.7** Any bonded notice to withhold in connection with the Borrower Loan is validly served on Governmental Lender or Bank and within five days of the receipt of such service (a) is not discharged, or (b) if the amount claimed is disputed in good faith by Borrower or Contractor, an appropriate counter bond or equivalent acceptable to Bank is not provided to Bank.
- **8.8** The imposition, voluntary or involuntary, of any lien or encumbrance upon the Property without Bank's written consent, unless an adequate counter bond is provided and such lien is accordingly released within ten days of the imposition of such lien.
- **8.9** Bank fails to have an enforceable first lien on or security interest in any property given as security for the Borrower Loan, except as permitted by Bank in writing.
- **8.10** An event or condition occurs or arises that materially impairs Borrower's intended use of the Property.
- **8.11** Borrower neglects, fails or refuses to keep in force and effect any insurance coverage required by Bank.
- **8.12** Any of the Funding Loan Documents, the Subordinate Documents, the Density Bonus Restrictions, the HCD Documents, the AHAP Contract (including the form of HAP Contract attached as an exhibit to the AHAP Contract), the HAP Contract, or the Regulatory Agreements are modified, amended or terminated without Bank's prior written consent.
- **8.13** Interest on the Funding Loan Note is no longer excludable from the gross income of the holder thereof for federal income tax purposes.
- **8.14** The occurrence of an event of default by Borrower under the Subordinate Documents, the Density Bonus Restrictions, the HCD Documents, the AHAP Contract, the HAP Contract, or the Regulatory Agreements (following the expiration of any curative periods set forth therein).

- **8.15** The failure of Borrower to comply with any of the terms and conditions of the Tax Credit Allocation Documents, the failure of Borrower to cause Project Completion to occur on or before the Completion Date, or the failure of Borrower to comply with any of the monitoring or reporting requirements set forth in the Qualified Allocation Plan.
- **8.16** The determination by Bank (in Bank's reasonable opinion) at any time that (i) paragraph (1) of Section 42(h) of the Code will apply to the allocation of the Tax Credits or (ii) Project Completion will not occur on or before the Completion Date.
- **8.17** The maximum amount of Tax Credits reserved by the Allocation Committee under the Preliminary Reservation is reduced by the Allocation Committee which results in a reduction of the Tax Credit Investor's capital contributions to Borrower which, together with other financing or equity investment permitted under the Borrower Loan Documents, would prevent Borrower from making the full Paydown Amount on or before the Outside Conversion Date, as determined by Bank in its sole discretion.
- **8.18** Borrower shall fail to obtain the Hedge in accordance with the terms and provisions of Section 7.48. Borrower shall fail to perform any of its obligations under any agreement relating to any Hedge or Hedge Documents following the expiration of any applicable curative provision.
- **8.19** Borrower modifies, amends or terminates the Standard Agreement or otherwise fails to consummate the closing of the HCD Loan in accordance with the terms of the Standard Agreement or takes any action that might or does result in modification, amendment, termination or expiration of the Standard Agreement, in each case without Bank's written consent.

Tax Credit Investor shall have the right to cure any default by Borrower hereunder within the time periods (if any) set forth herein for such cure and Bank agrees to accept such cure as if cured by Borrower.

- **9. REMEDIES**. If an Event of Default occurs under this Agreement:
- **9.1** Governmental Lender and Bank may exercise any right or remedy that it has under any of the Borrower Loan Documents, or that is otherwise available at law or in equity or by statute (which may be exercised directly or by directing the actions of the Fiscal Agent), and all of Governmental Lender's and Bank's rights and remedies shall be cumulative.
- 9.2 Bank shall have the right, in its sole discretion, to enter the Property and take possession of it, whether in person, by agent or by court-appointed receiver, to perform any and all work and labor necessary to complete the Improvements substantially in accordance with the Plans, and to collect rents and otherwise protect its collateral and exercise its rights and remedies under the Borrower Loan Documents. If Bank exercises any of the rights or remedies provided in this Section, that exercise shall not make Bank a partner or joint venturer of Borrower. All sums that are expended by Bank in completing the Improvements or in preserving Bank's collateral for the Borrower Loan shall be considered an additional loan to Borrower secured by the Deed of Trust and Security Documents and shall bear interest at the Default Rate.
- **9.3** Notwithstanding the exercise of any remedy described above or the existence of any Event of Default, Bank, at its option, may make any Advance or disburse any or all of Borrower's Funds without (a) waiving Bank's right to demand payment of the Borrower Loan, (b) incurring liability to make any other or further Advances, and (c) waiving Bank's right to require compliance with Borrower's covenant to correct any defect in the Improvements or departure from the Plans not approved by Bank.
- **10. POWER OF ATTORNEY**. Borrower hereby constitutes and appoints Bank as Borrower's true and lawful attorney in fact with the power and authority, including full power of substitution upon the occurrence and during the continuance of an Event of Default, as follows:
  - **10.1** To take possession of the Property and complete the Improvements.

- **10.2** To use any of Borrower's Funds and any undisbursed proceeds of the Borrower Loan for the purpose of completing the Improvements and for other costs related thereto.
- **10.3** To make such additions and changes and corrections in the Plans as may be necessary or desirable, as Bank, in Bank's sole discretion, deems proper to complete the Improvements.
- **10.4** To employ such contractors, subcontractors, agents, architects, engineers and inspectors as are required to complete the Improvements.
  - **10.5** To employ security personnel to protect the Property from damage.
- **10.6** To pay, settle or compromise all existing bills and claims against Borrower's Funds or any undisbursed proceeds of the Borrower Loan as may be necessary or desirable or as Bank deems proper, in Bank's sole discretion, for the completion of the Improvements, or for the protection or clearance of title to the Property, or for the protection of Bank's interest with respect thereto.
- **10.7** To prosecute and defend all actions and proceedings in connection with the construction of the Improvements.
- **10.8** To record any notices of completion, cessation of labor and other notices that Bank deems necessary to protect any interest of Bank under the provisions of this Agreement, the Deed of Trust, any of the Security Documents, or any other Borrower Loan Document.
- **10.9** To execute, acknowledge, and deliver all instruments and documents in the name of Borrower that may be necessary or desirable or as Bank deems proper, in Bank's sole discretion, and to perform any and every act with respect to the construction of the Improvements that Borrower might perform on Borrower's own behalf.

This Power of Attorney is a power coupled with an interest and cannot be revoked. Any costs or expenses incurred by Bank in connection with any acts performed by Bank under or pursuant to this Section shall be paid by Borrower. If such costs are not paid by Borrower upon demand of Bank, interest shall accrue thereon at the Default Rate. Any such advances made or costs or expenses incurred by Bank shall be secured by the Deed of Trust and Security Documents.

### 11. MISCELLANEOUS.

- 11.1 Disclaimer. WHETHER OR NOT GOVERNMENTAL LENDER OR BANK ELECT TO EMPLOY ANY OR ALL OF THE REMEDIES AVAILABLE TO GOVERNMENTAL LENDER OR BANK UPON THE OCCURRENCE OF AN EVENT OF DEFAULT, NEITHER GOVERNMENTAL LENDER NOR BANK SHALL BE LIABLE FOR THE CONSTRUCTION OF, OR FAILURE TO CONSTRUCT, COMPLETE OR PROTECT, THE IMPROVEMENTS.
- 11.2 Notices. All notices, demands, requests or other communications (including communications by facsimile transmission or e-mail) provided for or allowed hereunder shall be in writing and shall be effective only if the same is delivered by personal service, mailed (postage prepaid, return receipt requested), faxed, or e-mailed to the address given with the signatures at the end of this Agreement. Any such notice shall be deemed to have been received by the addressee, (a) if mailed, on the third day following the date of such mailing, or (b) if faxed or e-mailed, upon telephone confirmation of receipt. Any party may at any time change its address for such notices by delivery or mailing the other parties to this Agreement a notice of such change.
- 11.3 Waivers. Any forbearance, failure or delay by Bank in exercising any right, power or remedy shall not be deemed a waiver thereof and any single or partial exercise of any power, right or remedy shall not preclude any further exercise thereof. No waiver of or consent to any breach of any of the covenants or conditions of this Agreement or any other Borrower Loan Document shall be construed to be

a waiver of or a consent to any previous or subsequent breach of the same or any other condition or covenant. No waiver or consent shall be effective under any Borrower Loan Document unless it is in writing and signed by an officer of Bank.

## 11.4 Governmental Lender's and Bank's Expenses; Rights of Governmental Lender and Bank.

- 11.4.1 Borrower shall promptly pay to Governmental Lender and Bank, upon demand, with interest thereon from the date of demand at the Default Rate, reasonable attorneys' fees (including the fees and costs of Governmental Lender's, Fiscal Agent's and Bank's in-house counsel and legal staff) and all costs and other expenses paid or incurred by Governmental Lender, Fiscal Agent and Bank in exercising its rights or remedies provided for in this Agreement or any other Loan Document. If at any time Borrower fails to perform any of its obligations hereunder, Bank shall have the right, but not the obligation, to perform such obligations at the expense of Borrower. The amount of any monies so expended or obligations so incurred by Governmental Lender, Fiscal Agent and Bank, together with interest thereon at the Default Rate, shall be repaid to Governmental Lender, Fiscal Agent and Bank promptly upon demand and payment thereof shall be secured by the Deed of Trust and Security Documents. The obligations of this Section 11.4.1 and those in Section 7.24 (Indemnity) shall remain valid and in effect notwithstanding repayment of the Borrower Loan or the Funding Loan Note or termination of this Borrower Loan Agreement or the Funding Loan Agreement.
- 11.4.2 Governmental Lender and Bank, and any of Governmental Lender's and Bank's representatives, shall have the right, at any time and from time to time, and without notice, to enter upon the Property, to inspect the Improvements and all materials to be used in the construction thereof and to examine the Plans and all detailed plans and shop drawings that are or may be kept at the construction site.
- 11.5 No Third Party. This Agreement is made for the sole benefit of Borrower, Governmental Lender, Bank and Governmental Lender's and Bank's successors and assigns, and no other Person shall have any rights or remedies under or by reason of this Agreement or any right to exercise any right or power of Governmental Lender and Bank hereunder or arising from any default by Borrower. Governmental Lender and Bank shall owe no duty whatsoever to any claimant for labor performed or material furnished in connection with the construction of the Improvements nor any duty whatsoever to apply any undisbursed proceeds of the Borrower Loan to the payment of any such claim or to exercise any right or power of Bank hereunder or arising from any default by Borrower.
  - **11.6 Time of Essence**. Time is of the essence of this Agreement and every part hereof.
- 11.7 Successors and Assigns. Neither this Agreement nor any right of Borrower to receive any sums, proceeds or disbursements hereunder, may be assigned, pledged, hypothecated, anticipated or otherwise encumbered by Borrower without the prior written consent of Bank. Subject to the foregoing restriction and the restrictions contained in the Deed of Trust, this Agreement shall inure to the benefit of Governmental Lender and Bank and Governmental Lender's and Bank's successors and assigns and shall bind Borrower and Borrower's successors and assigns.
- 11.8 Participation or Syndication. Bank shall have the right, in its sole discretion, to assign all or any part of Bank's rights in the Borrower Loan and under the Borrower Loan Documents or the Funding Loan Documents, either through direct assignment or through participating interests, subject to the provisions of Section [4.3] [CHECK] of the Funding Loan Agreement. Bank is hereby authorized to disclose to any prospective assignee or participant in the Borrower Loan any and all information regarding Borrower, any Loan Party, the Property or the Borrower Loan.
- **11.9 Governing Jurisdiction**. This Agreement shall be governed by and construed in accordance with the laws of the state in which the Property is located.

- **11.10 Entire Agreement**. This Agreement and all other Borrower Loan Documents and the Funding Loan Documents constitute the entire understanding between the parties hereto with respect to the subject matter hereof, superseding all prior written or oral understandings. This Agreement and the other Borrower Loan Documents may be modified, amended or terminated only in writing signed by all parties hereto or thereto.
- 11.11 Joint and Several Liability. If Borrower consists of more than one Person, each shall be jointly and severally liable to Bank for the performance of this Agreement and the other Borrower Loan Documents.
- 11.12 Publicity, Signs. Borrower hereby agrees that Bank, at Bank's expense, may publicize the financing of the Property (including the name of Borrower) and, in connection therewith, may use the project name and address, and a description, photograph or other illustrative drawing of the Property. Borrower hereby grants Bank the right to erect or cause to be erected Bank's sign or signs in size and location desired by Bank on the Property so long as such sign or signs do not interfere with the construction of the Improvements. Borrower will exercise, and will cause Contractor and subcontractors to exercise, due care to protect said sign or signs from damage.
- 11.13 Credit Information and Reports. Borrower authorizes Bank to release information concerning Borrower's financial condition to suppliers, other creditors, credit bureaus and other credit reporters, and to obtain such information from any third party at any time.
- **11.14 Headings**. The various headings of this Agreement are included for convenience only and shall not affect the meaning or interpretation of this Agreement or any provision hereof.
- **11.15 Severability**. Should any one or more provisions of this Agreement be determined to be illegal or unenforceable, all other provisions nevertheless shall be effective.
- 11.16 Counterparts. This Agreement and each other Borrower Loan Document may be executed in two or more counterparts, each of which shall be deemed an original but taken together shall be one and the same document.
- 11.17 USA Patriot Act. Bank is subject to the USA Patriot Act and hereby notifies Borrower that pursuant to the requirements of that Act, Bank is required to obtain, verify and record information that identifies Borrower, which information includes the name and address of Borrower and other information that will allow Bank to identify Borrower in accordance with the Act.
- 11.18 Waiver of Jury Trial. To the extent permitted by law, in connection with any action or proceeding, whether brought in state or federal court, the Borrower and Bank hereby expressly, intentionally and deliberately waive any right they may otherwise have to trial by jury of any Claim (as defined below). For purposes of clarity, this provision shall not constitute a waiver of any right to trial by jury held by Governmental Lender.
- 11.19 Judicial Reference. If the waiver of jury trial set forth hereinabove is not enforceable under the laws of the state in which the Property is located, then Borrower and Bank hereby agree that all Claims, including any and all questions of law or fact relating thereto, shall, at the written request of any party, be determined by Reference (as hereinafter defined) as set forth hereinbelow:
- **11.19.1 Selection Or Appointment Of Referee**. Bank and Borrower shall select a single neutral referee, who shall be a retired state or federal judge. In the event that Bank and Borrower cannot agree upon a referee, the referee shall be appointed by the court.
- 11.19.2 Conduct Of Reference. Except as otherwise provided in this Agreement, the Reference shall be conducted pursuant to the laws of the state in which the Property is located. The referee shall determine all issues relating to the applicability, interpretation, legality and enforceability of the

Borrower Loan Documents or Funding Loan Documents. The referee shall report a statement of decision to the court. The Bank and Borrower shall equally bear the fees and expenses of the referee, unless the referee otherwise provides in the statement of decision.

- 11.19.3 Provisional Remedies, Self-Help And Foreclosure. No provision of this Agreement shall limit the right of any party to (i) exercise self-help remedies including, without limitation, set-off, (ii) foreclose against or sell any collateral, by power of sale or otherwise or (iii) obtain or oppose provisional or ancillary remedies from a court of competent jurisdiction before, after or during the pendency of the Reference. The exercise of, or opposition to, any such remedy does not waive the right of any party to a Reference pursuant to this Agreement.
- 11.19.4 No Decision By Jury. Borrower and Bank hereby acknowledge that if a referee is selected or appointed to determine the Claims, then the Claims in accordance with this Section 11.19 will not be decided by a jury.
- 11.19.5 Miscellaneous. In the event that multiple Claims are asserted, some of which are not subject to this Section, Borrower and Bank agree to stay the proceedings of the Claims not subject to this Section until all other Claims are resolved in accordance with this Section. In the event that Claims are asserted against multiple parties, some of whom are not subject to this Section, the Parties agree to sever the Claims subject to this Section and resolve them in accordance with this Section.
- 11.19.6 Claim. "Claim" shall mean any claim, cause of action, action, dispute or controversy between or among the parties, whether sounding in contract, tort or otherwise, which arises out of or relates to: (i) any of the Borrower Loan Documents or the Funding Loan Documents; (ii) and negotiations or communications relating to any of the Borrower Loan Documents or the Funding Loan Documents, whether or not incorporated into the Borrower Loan Documents or the Funding Loan Documents or any indebtedness evidenced thereby; or (iii) any alleged agreements, promises, representations or transactions in connection therewith.
- **11.19.7 Reference**. "Reference" shall mean a judicial reference conducted pursuant to this Agreement and in accordance with the laws of the state in which the Property is located, as in effect at the time the referee is selected or appointed.
- **11.20 Limitation on Damages**. In the event that punitive damages are permitted under the laws of the state in which the Property is located, the amount thereof shall not exceed a sum equal to three times the amount of actual damages.
- **11.21 Exhibits**. All exhibits attached hereto are incorporated herein as if fully set forth within this Agreement.

[Signatures on following page]

IN WITNESS WHEREOF, the parties have executed this Construction and Permanent Loan Agreement as of the date and year first above written.

#### **BORROWER:**

SP7 APARTMENTS LP, a California limited partnership

By: SP7 Apartments GP LLC,

a California limited liability company,

its general partner

By: The Skid Row Housing Trust,

a California nonprofit public benefit corporation,

its sole member/manager

By:\_\_\_\_\_

Ben Rosen, Real Estate Development Director

54

### **Address for Notice to Borrower:**

SP7 Apartments LP c/o The Skid Row Housing Trust
Attn: Ben Rosen, Real Estate Development Director
1317 E. 7th Street
Los Angeles, CA 90021
Phone No. (213) 683-0522
Fax No. (213) 683-0781

E-mail address: ben.rosen@skidrow.org

## With copies to:

Gubb & Barshay Attn: Scott Barshay, Esq. 505 14<sup>th</sup> Street, Suite 1050 Oakland, CA 94612 Phone No. (415) 781-6600 Fax No. (415) 781-6967

National Equity Fund, Inc. 500 S. Grand Avenue, Suite 2300 Los Angeles, CA 90071 Attn: Todd Fabian, Regional Manager Phone No. (213) 240-3144 Fax No. (213) 250-0014

E-mail address: tfabian@nefinc.org

#### and

Barnes & Thornburg LLP 41 S. High Street, Suite 3300 Columbus, OH 43215-6104 Attn: Holly Heer, Esq. Phone No. (614) 628-1458 Fax No. (614) 628-1433

E-mail address: holly.heer@btlaw.com

cla.doc

## **BANK:**

## MUFG UNION BANK, N.A.

By:	
Name:	
Title:	

#### **Addresses for Notice to Bank:**

MUFG Union Bank, N.A.
Attn: Manager
Commercial Credit Loan Administration
3151 E. Imperial Highway, 1<sup>st</sup> Floor
Brea, CA 92821
Phone No. (714) 985-2469
Fax No. (949) 752-836
E-mail address: maria.trevias@unionbank.com

MUFG Union Bank, N.A.
Attn: Zennia Harris, Vice President
Community Development Finance
1901 Avenue of the Stars, Suite 600
Los Angeles, CA 90067
Phone No. (310) 551-8951
Fax No. (310) 284-5791
E-mail address: zennia.harris@unionbank.com

## **GOVERNMENTAL LENDER:**

#### **CITY OF LOS ANGELES**

By: \_\_\_\_\_\_
Name: Helmi A. Hisserich
Title: Assistant General Manager
Approved as to form:
MICHAEL N. FEUER, City Attorney:

By: Los Angeles Housing and Community Investment Department

#### Addresses for Notice to Governmental Lender:

Deputy/Assistant City Attorney

Los Angeles Housing and Community Investment Department 1200 West 7<sup>th</sup> Street, 8th Floor Los Angeles, CA 90017 Attention: Supervisor, Affordable Housing Bond Program

Facsimile: (213) 808-8918

with a copy to:

By:\_

Los Angeles Housing and Community Investment Department P.O. Box 532729 Los Angeles, CA 90053-2729

Attn: Supervisor, Affordable Housing Bond Program

## JOINDER REGARDING DEVELOPMENT FEE

The undersigned hereby acknowledges and agrees that the undersigned shall not be entitled to receive more than (i) [\$420,000] [CHECK] of the development fee in the aggregate on or prior to the Closing Date, (ii) [\$700,000] [CHECK] of the development fee in the aggregate on or prior to the date Bank determines fifty percent (50%) of the Project has been completed, (iii) [\$840,000] [CHECK] of the development fee in the aggregate on or prior to the date Bank determines Project Completion has occurred, (iv) [\$5,442,935] [CHECK] of the development fee in the aggregate on or prior to the date the Project has achieved Qualified Occupancy, (v) [\$5,542,935] [CHECK] of the development fee in the aggregate prior to the Conversion Date, and (vi) [\$5,592,935] [CHECK] of the development fee in the aggregate prior to receipt of IRS Form 8609 from the Allocation Committee; any portion of such development fee received by the undersigned while any amounts are outstanding or remain undisbursed under the Borrower Loan, shall be remitted to MUFG Union Bank, N.A. to be held as additional collateral for the Borrower Loan and, upon an event of default with respect thereto, applied in reduction of amounts outstanding under the Borrower Loan in such amounts and in such order as MUFG Union Bank, N.A. shall elect in its sole and absolute discretion.

[SIGNATURE PAGE FOLLOWS]

THE SKID ROW HOUSING TRUST, a California nonprofit public benefit corporation
By:Ben Rosen, Real Estate Development Director

## Schedule 1

## Post-Conversion Date Installments of Principal and Interest

[To be attached on the Conversion Date]

# EXHIBIT A LEGAL DESCRIPTION

This **Exhibit A** is attached to and a part of that certain Construction and Permanent Loan Agreement dated September 1, 2018 by and among SP7 Apartments LP, a California limited partnership, MUFG Union Bank, N.A. and the City of Los Angeles.

[TO BE INSERTED]

# EXHIBIT B DISBURSEMENT SCHEDULE

This **Exhibit B** is attached to and a part of that certain Construction and Permanent Loan Agreement dated September 1, 2018 by and among SP7 Apartments LP, a California limited partnership, MUFG Union Bank, N.A. and the City of Los Angeles.

[TO BE INSERTED]

B-1

## EXHIBIT B-1 PROJECT BUDGET

This <u>Exhibit B-1</u> is attached to and a part of that certain Construction and Permanent Loan Agreement dated September 1, 2018 by and among SP7 Apartments LP, a California limited partnership, MUFG Union Bank, N.A. and the City of Los Angeles.

[TO BE INSERTED]

B-1-1

## EXHIBIT C SPECIAL CONDITIONS

- 1. The following representations and warranties are incorporated by reference in Section 6 of this Agreement:
  - (a) The Project is located wholly within the jurisdiction of the City of Los Angeles.
- (b) The Borrower shall make no changes to the Project or to the operation thereof which would affect the qualification of the Project under the Act or impair the exclusion from gross income for federal income tax purposes of the interest on the Funding Loan Note. The Borrower intends to utilize the Project as multifamily rental housing during the Qualified Project Period (as defined in the Tax-Exempt Regulatory Agreement).
- (c) Not in excess of two percent (2%) of the proceeds of the Borrower Loan will be used to pay costs of issuance of the Funding Loan Note.
- (d) The acquisition, construction and operation of the Project in the manner presently contemplated and as described herein and in the Tax-Exempt Regulatory Agreement will not conflict with any zoning, water or air pollution or other ordinance, order, law or regulation applicable thereto. The Borrower will cause the Project to be constructed and operated in all material respects in accordance with all applicable federal, state and local laws or ordinances (including rules and regulations) relating to zoning, building, safety and environmental quality.
- (e) The Borrower acknowledges, represents and warrants that it understands the nature and structure of the Project; that it is familiar with the provisions of all of the documents and instruments relating to the financing of the Project to which it is a party; that it understands the risks inherent in such transactions, including without limitation the risk of loss of the Project; and that it has not relied on the Governmental Lender for any guidance or expertise in analyzing the financial or other consequences of such financing transactions or otherwise relied on the Governmental Lender in any manner except to issue the Funding Loan Note in order to provide funds for the Borrower Loan.
- (f) The Borrower intends to hold the Project for its own account, has no current plans to sell and has not entered into any agreement to sell the Project.
- (g) The Borrower has contacted all "related persons" thereof (within the meaning of Section 147(a) of the Code) of which it is aware; and none of them shall, at any time, pursuant to any arrangement, formal or informal, acquire any interest in the Funding Loan Note in an amount related to the amount of the Borrower Loan.
- (h) All of the proceeds from the Borrower Loan plus any income from the investment of the proceeds of the Borrower Loan will be used to pay or reimburse the Borrower for Project Costs (as defined in the Tax-Exempt Regulatory Agreement), and at least 97% of the proceeds of the Borrower Loan will be used to pay or reimburse the Borrower for Qualified Project Costs (as defined in the Tax-Exempt Regulatory Agreement) and less than 25% of such amount will be used to pay or reimburse the Borrower for the cost of land or any interest therein. The Borrower shall assure that the proceeds of the Borrower Loan are expended so as to cause the Funding Loan Note to constitute a "qualified residential rental bond" within the meaning of Section 142(d) of the Code.
- (i) The Borrower has not knowingly taken or permitted to be taken and will not knowingly take or permit to be taken any action which would have the effect, directly or indirectly, of causing interest on the Funding Loan Note to be included in the gross income of the owners thereof for purposes of federal income taxation.

C-1

- (j) The Borrower hereby represents and warrants that, within the meaning of Section 147(a)(14) of the Code, the weighted average maturity of the Funding Loan Note does not exceed 120 percent of the average reasonably expected economic life of the facilities being financed with the proceeds of the Funding Loan Note.
- (k) The Borrower represents and warrants that no portion of the proceeds of the Borrower Loan will be used to provide any airplane, skybox or other private luxury box, health club facility, facility primarily used for gambling, or store the principal business of which is the sale of alcoholic beverages for consumption off premises, and no portion of the proceeds of the Borrower Loan will be used for an office unless (i) the office is located on the premises of the facilities constituting the Project and (ii) not more than a de minimus amount of the functions to be performed at such office is not related to the day-to-day operations of the Project.
- 2. The following covenants of Borrower are incorporated by reference in Section 7 of this Agreement:
- (a) The Borrower covenants that it shall not take, or permit or suffer to be taken by the Governmental Lender or otherwise, any action with respect to the proceeds of the Funding Loan Note which if such action had been reasonably expected to have been taken, or had been deliberately and intentionally taken, on the date of issuance of the Funding Loan Note would have caused the Funding Loan Note to be an "arbitrage bond" within the meaning of Section 148(a) of the Code.
  - (b) Payment of Governmental Lender Fees and Expenses.
- (i) Borrower hereby agrees to pay to the Governmental Lender the amounts described in [Sections 7(n) and (o)] **[CHECK]** of the Tax-Exempt Regulatory Agreement.
- (ii) The Borrower agrees to pay to the Governmental Lender, within fifteen (15) days after receipt of request for payment thereof, all reasonable out-of-pocket expenses of the Governmental Lender related to the Project and the financing thereof that are not otherwise required to be paid by the Borrower under the terms of this Agreement and are not paid from disbursements of the Borrower Loan, including, without limitation, legal fees and expenses incurred in connection with the interpretation, performance, enforcement or amendment of any documents relating to the Project or the Funding Loan.
  - (c) <u>Tax Exempt Status of the Funding Loan Note</u>.
- (i) It is the intention of the Governmental Lender, Bank and the Borrower that interest on the Funding Loan Note shall be and remain excludable from gross income for federal income taxation purposes, and to that end the covenants and agreements of the Borrower in this Section are for the benefit of the Bank and the Governmental Lender.
- (ii) The Borrower covenants and agrees that it will not knowingly and willingly use or permit the use of any of the funds provided by the Governmental Lender or the Bank hereunder or any other funds of the Borrower, directly or indirectly, in such manner as would, or enter into, or allow any "related person" (as defined in Section 147(a)(2) of the Code) to enter into, any arrangement, formal or informal, for the purchase of the Funding Loan Note that would, or take or omit to take any other action that would cause the Funding Loan Note to be an "arbitrage bond" within the meaning of Section 148 of the Code or "federally guaranteed" within the meaning of Section 156(b) of the Code and applicable regulations promulgated from time to time thereunder.
- (iii) In the event that at any time the Borrower is of the opinion or becomes otherwise aware that for purposes of this Section it is necessary to restrict or to limit the yield on the investment of any moneys held under the Funding Loan Agreement, the Borrower Loan Documents or otherwise by the Bank, the Borrower shall determine the limitations and so instruct the Bank in writing and cause the Bank to comply with those limitations.

C-2

- (iv) The Borrower will take such action or actions as may be reasonably necessary in the opinion of Tax Counsel or of counsel to the Governmental Lender, or of which it otherwise becomes aware, to fully comply with Section 148 of the Code.
- (v) The Borrower further agrees that it shall not discriminate on the basis of race, creed, color, sex, sexual preference, source of income (e.g. AFDC, SSI), physical disability, national origin or marital status in the lease, use or occupancy of the Project or in connection with the employment or application for employment of persons for the operation and management of the Project, to the extent required by applicable State or federal law.
- (vi) The Borrower further warrants and covenants that it has not executed and will not execute any other agreement, or any amendment or supplement to any other agreement, with provisions contradictory to, or in opposition to, the provisions of this Borrower Loan Agreement and of the Tax-Exempt Regulatory Agreement, and that in any event, the requirements of this Borrower Loan Agreement and the Tax-Exempt Regulatory Agreement are paramount and controlling as to the rights and obligations herein set forth and supersede any other requirements in conflict herewith and therewith.
- (vii) The Borrower shall not purchase, and shall use its best efforts to prevent any guarantor of the Borrower from purchasing, pursuant to an arrangement, formal or informal, any interest in the Funding Loan Note in an amount related to the amount of the Borrower Loan.
- (viii) The Borrower will use due diligence to complete the acquisition and construction of the Project and reasonably expects to fully expend the portion of the Borrower Loan by [the Outside Conversion Date] [CHECK].
- (ix) The Borrower will calculate or cause to be calculated, at the times required by the Code, any rebate due to the federal government in respect of the Funding Loan Note, and will make timely payment of any rebate amount due to the federal government.
- (d) **Federal Guarantee Prohibition**. The Borrower shall take no action, nor permit nor suffer any action to be taken if the result of the same would be to cause the Funding Loan Note to be "federally guaranteed" within the meaning of Section 149(b) of the Code.
- (e) **Limited Liability**. The Governmental Lender shall not be liable for any costs, expenses, losses, damages, claims or actions, of any conceivable kind on any conceivable theory, under or by reason of or in connection with this Borrower Loan Agreement, the Funding Loan Note or any of the other Borrower Loan Documents, except only to the extent amounts are received for the payment thereof from the Borrower under this Borrower Loan Agreement. All obligations and any liability of the Governmental Lender shall be further limited as provided in Sections [4.1, 5.2 and 6.14] **[CHECK]** of the Funding Loan Agreement.

C-3

## EXHIBIT D CONDITIONS TO CONVERSION

The following shall be the conditions precedent to conversion:

	Conditions to Conversion	Check When Satisfied
(a)	The final disbursement shall have occurred.	
(b)	All indebtedness incurred by the Borrower in connection with the Project, including, but not limited to, the Borrower Loan and any subordinate financing, shall be completely funded and, if applicable, converted to permanent financing.	
(c)	No material adverse change has occurred in the financial condition of Borrower or any other Loan Party, as evidenced by current Financial Statements provided by Borrower to Bank.	
(d)	All representations and warranties made by Borrower in the Borrower Loan Documents and the Funding Loan Documents shall be true and correct on and as of the Conversion Date as if made on and as of the Conversion Date (and, if required by Bank, Bank shall have received a certificate of Borrower to that effect).	
(e)	The Improvements shall not have been materially injured or damaged by fire or other casualty.	
<b>(f)</b>	Bank shall have received (A) such endorsements to the Title Policy as Bank may require which shall insure that the Improvements have been completed free of all mechanic's and materialmen's liens or claims thereof, or (B) such additional title policies with endorsements as Bank may require, with a liability limit of not less than the principal amount of the Borrower Loan, issued by Title Insurer, with coverage and in form satisfactory to Bank, insuring Governmental Lender's and Bank's interest under the Deed of Trust as a first lien on the Real Property, excepting only such items as shall have been approved in writing by Bank.	
(g)	Borrower delivers to Bank fully executed copies of any amendments or assignments affecting the formation documents of Borrower and, if applicable, its constituent general partners or members, to the extent not previously provided to and approved by Bank.	
(h)	Borrower provides Bank with current evidence of the insurance coverage required pursuant to this Agreement, provided that Borrower need not provide evidence of course of construction insurance and Borrower shall in addition provide evidence of business interruption and/or rental interruption insurance, as applicable.	
(i)	Bank shall have received the Paydown Amount in cash or current funds.	

D-1

	Conditions to Conversion	Check When Satisfied
(j)	During each month of the three-month period immediately preceding the Conversion Date; at least ninety percent (90%) of the Units within the Property shall have been leased to, and occupied by, third-party residential tenants under Acceptable Unit Leases executed by Borrower in strict compliance with the terms and conditions of this Agreement and the Regulatory Agreements.	
(k)	The Improvements shall have been completed in substantial accordance with the Plans free and clear of all liens other than Permitted Liens and Bank shall have received copies of the final certificates of occupancy for each Unit within the Property.	
(1)	As of the Conversion Date, no Event of Default and no other event or condition that, with the giving of notice or the passage of time, or both, would become an Event of Default, shall have occurred and be continuing.	
(m)	If required by Bank, a current survey of the Real Property, including dimensions and delineation of all the Improvements and location of all easements thereon, certified to and satisfactory to Bank and Title Insurer.	
(n)	Guarantor has executed and delivered to Governmental Lender and Bank the Indemnity Agreement.	
(0)	During each month of the three-month period immediately preceding the Conversion Date, the Debt Coverage Ratio for the Property shall have been at least 1.15 to 1.00.	
(p)	Borrower shall have established with Bank the Capital Improvement Reserve Account and collaterally assigned such account to Bank.	
(q)	Borrower shall have paid to Bank all reasonable costs and expenses incurred by Bank and Fiscal Agent in connection with the Conversion.	
(r)	Borrower delivers to Bank such other documentation, certifications, opinions and information as may be required by Bank.	
(s)	If requested by Bank, (a) each Subordinate Lender shall have executed and delivered to Bank estoppel certificates which shall contain such certifications as Bank shall reasonably require with respect to the applicable Subordinate Documents and/or the Density Bonus Restrictions, and (b) Tax Credit Investor shall have executed and delivered to Bank an estoppel certificate in form and substance reasonably acceptable to the Bank, which shall contain such certifications as Bank shall reasonably require with respect to Tax Credit Investor's obligations under the Partnership Agreement.	
(t)	Borrower shall have entered into one or more Hedges, in form and content and from a counterparty complying with the provisions contained in Section 7.48, which shall provide for the Borrower to pay a fixed rate of interest no greater than [%] [CHECK] (including the Margin applicable during the Permanent Phase), on an amount not more than the entire outstanding principal balance of the Borrower Note as of the Conversion Date for the period commencing no later than the Initial Outside Conversion Date through the Maturity Date.	

D-2

	Conditions to Conversion	Check When Satisfied
(u)	Such evidence as Bank may require evidencing expenditure of Borrower's Equity on Project costs in accordance with this Agreement is at least [\$17,389,745] [CHECK] in the aggregate.	
(v)	Borrower shall have entered into the HAP Contract on terms and conditions acceptable to Bank, and the HAP Contract shall have been collaterally assigned by the Borrower to the Bank pursuant to the Assignment of HAP Contract and Contract Administrator shall have consented to such assignment of the HAP Contract pursuant to the Assignment of HAP Contract.	
(w)	Borrower shall have delivered to Bank fully executed copies of the HCD Documents, each in form and content acceptable to Bank, and the HCD Loan shall have been closed and fully funded, or shall close and fully fund concurrently with Conversion.	
(x)	The Loan-to-Value Ratio shall not exceed eighty percent (80%). Bank shall have the right to require an Appraisal of the Property. Any and all costs, fees and expenses incurred in connection with such Appraisal shall be paid by Borrower.	
(y)	Borrower shall have established the Transition Reserve Account, deposited not less than [\$495,056] <b>[CHECK]</b> in the Transition Reserve Account and collaterally assigned such account to Bank.	
(z)	Borrower shall have established the Operating Reserve Account, deposited not less than [\$286,617] <b>[CHECK]</b> in the Operating Reserve Account and collaterally assigned such account to Bank.	
(aa)	The undisbursed portion of the HCID HHH Loan shall have been fully funded, or shall fully fund concurrently with Conversion.	
(bb)	The amount of [\$4,182,935] <b>[CHECK]</b> held in the General Partner Equity Account shall have been applied to repay the Borrower Loan in accordance with Section 5.7 of this Agreement.	
(cc)	Borrower shall have delivered to Bank updated economic projections for the Project in a form reasonably acceptable to Bank.	

D-3



## CONSTRUCTION AND PERMANENT LOAN AGREEMENT (MULTIFAMILY HOUSING BACK TO BACK LOAN PROGRAM)

by and among

CITY OF LOS ANGELES,

as Governmental Lender,

MUFG UNION BANK, N.A.,

as Bank

and

SP7 APARTMENTS LP, a California limited partnership,

as Borrower

Dated: [September 1, 2018] [CHECK]

Relating to

[\$32,007,000] [CHECK]
City of Los Angeles
Multifamily Mortgage Revenue Note
(SP7 Apartments)
Series 2018C

### **TABLE OF CONTENTS**

ı	P	а	a	e

1.	DEFIN	IITIONS	1
	1.1	Acceptable Unit Lease	
	1.2	Act	
	1.3	Advance	
	1.4	Aggregate Change Order Limit	
	1.5	Agreement or Borrower Loan Agreement	
	1.6	Agreement to Furnish Insurance	
	1.7	Allocation Committee	
	1.8	Amortization Date	
	1.9	Appraisal	
	1.10	Appraised Value	
	1.11	Architect	
	1.12	Architect's Agreement	
	1.13	Assignment of Construction Contract	
	1.14	Assignment of Hedge	
	1.15	Assignment of Partnership Interest (GP	
	1.16	Assignment of Plans and Specifications	
	1.17	Assignment of Tax Credits and Partnership Interests	
	1.18	Bank	3
	1.19	Bonded Work	3
	1.20	Borrower	3
	1.21	Borrower's Equity	3
	1.22	Borrower's Funds	3
	1.23	Borrower's Funds Account	3
	1.24	Borrower Loan	3
	1.25	Borrower Loan Documents	3
	1.26	Borrower Note	3
	1.27	Business Day	3
	1.28	Capital Improvement Reserve Account	
	1.29	Certification of Plans and Specifications	
	1.30	Change Order	
	1.31	Closing Date	
	1.32	Code	
	1.33	Completion Date	
	1.34	Conditions to Conversion	
	1.35	Construction Contract	
	1.36	Construction Costs	
	1.37	Construction Phase	
	1.38	Contract Date	
	1.39	Contractor	
	1.40	Conversion	
	1.41	Conversion Date	
	1.42	Conversion Election Notice	
	1.42	Conversion Notice	
	1.44 1.45	Debt Coverage Ratio	
		Deed of Trust Assignment	
	1.46	Deed of Trust Assignment	
	1.47	Default Rate	
	1.48	Detailed Cost Breakdown	
	1.49	Disbursement Schedule	
	1.50	Draw Request	
	1.51	ECA	5

		<b>Page</b>
1.52	Event of Default	5
1.53	Extended Use Agreement	5
1.54	Financial Statements	5
1.55	Financing Statements	5
1.56	First Payment Date	5
1.57	Fiscal Agent	5
1.58	Funding Date	5
1.59	Funding Loan	5
1.60	Funding Loan Agreement	5
1.61	Funding Loan Documents	
1.62	Funding Loan Note	
1.63	General Partner	
1.64	Governmental Authority	
1.65	Governmental Lender	
1.66	Governmental Requirement	
1.67	Gross Operating Income	
1.68	Guarantor	
1.69	Guaranty	
1.70	Hedge	
1.71	Hedge Documents	
1.72	Improvements	
1.73	Indemnified Parties	
1.74	Indemnity Agreement	
1.75	Initial Disbursement	
1.76	Interest Change Date	
1.77	Interest Period	
1.78	Interest Reserve	
1.79 1.80	Late Charge Leases	
1.80 1.81	LIBOR Rate	
1.82	Liquid Assets	
1.83	Loan Fee	
1.83 1.84	Loan Party	
1.85	Loan-to-Value Ratio	
1.86	London Banking Day	
1.87	Margin	
1.88	Maturity Date	
1.89	Maximum Lawful Rate	
1.90	Net Operating Income	
1.91	New York Banking Day	
1.92	Offsite Materials	
1.93	Onsite Materials	
1.94	Operating Deficits	
1.95	Operating Expenses	
1.96	Operating Statement	
1.100	Permanent Phase	
1.101	Permitted Liens	
1.102	Person	
1.103	Personal Property	
1.104	Plans	8
1.105	Preliminary Reservation	
1.108	Project Fund	8
1.109	Projected Permanent Phase Loan Amount	8
1.110	Property or Project	8
1.111	Qualified Allocation Plan	9

		<u>Page</u>
1.112	R&T Code	9
1.113	Real Property	9
1.114	Recorded Documents	9
1.115	Regulatory Agreements	
1.117	Rental Assistance Operating Deficits	
1.118	Security Documents Security Documents	
1.119	Set Aside Letter	
1.120	Single Change Order Limit	
1.121	Stub Period	
1.122	Subordinate Lender	
.123	[Subordinate Lender Deed of Trust	
.123	[Subordinate Lender Deed of Trust	
1.125	[Subordinate Lender Loan[Subordinate Lender Loan Agreement	10
1.126		
1.127	[Subordinate Lender Note	
1.128	[Subordinate Lender Restrictions	
1.129	[Subordinate Lender Subordination Agreement	
1.130	Subordination Agreement(s	
1.131	Surety	
1.132	Tax Certificate	
1.133	Tax Counsel	
1.134	Tax Credit Allocation Documents	10
.135	Tax Credit Application	10
1.136	Tax Credit Investor	
1.137	Tax Credits	
.138	Tax-Exempt Regulatory Agreement	
.139	Title Insurer	
1.140	Title Policy	
1.141	Transfer	
.142	Transition Reserve Account	
1.143	Unit(s	
1.144	Variable Rate	
1.145	Variable Rate Principal	
1.143	Valiable Rate Fillicipal	11
BORR	OWER LOAN	11
2.1	Purpose	11
2.2	Loan Terms and Conditions	11
2.3	Loan Fee	11
2.4	Full Payment and Reconveyance	
	ENTS; CONVERSION	
3.1	Payments	12
3.2	Conversion; Termination	13
3.3	Maturity Date	14
3.4	Application of Payments	14
3.5	Acceleration	
3.6	Late Charge; Default Interest	
3.7	Prepayment	
3.8	Additional Fee Payment Obligations	
COND	ITIONS PRECEDENT	
4.1	Conditions to Closing of the Borrower Loan	
4.2	Conditions to Issuance of the Funding Loan Note	18

2.

3.

4.

			Page
5.	DISBL	JRSEMENTS	20
•	5.1	Initial Disbursement	
	5.2	Subsequent Disbursements	
	5.3	Final Disbursement	
	5.4	Additional Conditions to Advances	
	5.5	Disbursement Limits.	
	5.6	Disbursement into Project Fund	
6.		RESENTATIONS AND WARRANTIES OF BORROWER	
	6.1	Formation/Authority	
	6.2	No Defaults Under Existing Agreements	
	6.3	No Actions	
	6.4	Other Liens	
	6.5	Leases	
	6.6	Financial Statements	
	6.7	Compliance With Laws	
	6.8	Permits, Approvals, Licenses	
	6.9	Ownership of Real Property	
	6.10	Ownership of Personal Property	26
	6.11	Other Financing	
	6.12	Plans, Defects	26
	6.13	Utilities	26
	6.14	Roads	26
	6.15	CC&Rs, Zoning	26
	6.16	Finder's Fees	26
	6.17	Draw Request	26
	6.18	Other Information	26
	6.19	No Default	27
	6.20	Tax Certificate	27
	6.21	Regulatory Agreement	27
	6.22	No Governmental Lender Relationships	
	6.23	Authorizations and Consents	
	6.24	No Reliance	
	6.25	Environmental Matters	
	6.26	ERISA	
	6.27	Funding Loan Note	28
	6.28	Tax Credit Allocation Documents Effective	
	6.29	Satisfaction of Conditions under Tax Credit Allocation Documents and	
		Subordinate Documents	28
	6.30	Tax Credits Not Subject to State Ceiling	
	6.31	Additional Representations and Warranties	
7	DODE	20MEDIO COMENANTO	00
7.		ROWER'S COVENANTS	
	7.1 7.2	Application of Advances	
		Borrower's Funds	
	7.3	Lien Priority	
	7.4	Construction Start and Completion	
	7.5	Change Orders	
	7.6	Detailed Cost Breakdown	
	7.7	Contractor Covenants	
	7.8	Construction Contract Only	
	7.9	Paid Vouchers	
	7.10	Application of Disbursements	
	7.11	Foundation Completion	
	7.12	Personal Property Installation	30

			<b>Page</b>
	7.13	Defect Corrections	30
	7.14	Stop Notices; Mechanic's Liens	30
	7.15	Record Keeping, Financial and Other Information	
	7.16	Post-Construction Financial Reporting	
	7.17	Audit and Inspection Rights	
	7.18	Dividends, Distributions	
	7.19	Payment of Lawful Claims	
	7.20	Payment of Costs	
	7.21	Approval of Easements and Other Documents	
	7.22	Compliance with Laws; Preservation of Rights	
	7.23	Notices	
	7.24	Indemnity	
	7.25	Performance of Acts	
	7.26	Notice of Change	
	7.20 7.27	Tax Certificate	
	7.27 7.28		
		Funding Loan Documents	
	7.29	Regulatory Agreements	
	7.30	Prohibited Activities	
	7.31	Set Aside Letters	
	7.32	Management of Property	
	7.33	Leases	
	7.34	Compliance	
	7.35	Property Reserves	
	7.36	Establishment of Capital Improvement Reserve Account	
	7.37	Rent Restrictions	
	7.38	Preservation of Tax Credits	38
	7.39	Election of Credit Period	38
	7.40	Compliance with Subordinate Documents and Regulatory Agreements	38
	7.41	Payment of Development Fee	38
	7.42	IRS Form 8609	38
	7.43	Obtaining and Maintaining Real Property Tax Exemption	
	7.44	Draws under Subordinate Loan and Disbursement of Borrower's Funds	
	7.45	Draw Requests	
	7.46	Progress Reports and Annual Project Status Reports; Allocation Committee	
		Notices	38
	7.48	Hedge	38
	7.48	Establishment of Transition Reserve Account	40
8.	EVEN <sup>-</sup>	TS OF DEFAULT	40
9.	REME	DIES	42
		R OF ATTORNEY	
10.			
11.		ELLANEOUS	
	11.1	Disclaimer	43
	11.2	Notices	43
	11.3	Waivers	
	11.4	Bank's Expenses; Rights of Bank	
	11.5	No Third Party	44
	11.6	Time of Essence	44
	11.7	Successors and Assigns	44
	11.8	Participation or Syndication	
	11.9	Governing Jurisdiction	
	11.10	Entire Agreement	

		<u>Pag</u> e
11.11	Joint and Several Liability	44
	Publicity, Signs	
	Credit Information and Reports	
	Headings	
11.15	Severability	44
11.16	Counterparts	45
	USA Patriot Act	
11.18	Waiver of Jury Trial	45
11.19	Judicial Reference	45
11.20	Limitation on Damages	46
	Exhibits	

## Attachment D

Bond Regulatory Agreement for SP7 (519 E. 7th Street) on next page.

RECORDING REQUESTED BY AND WHEN RECORDED RETURN TO:

KUTAK ROCK LLP 1650 FARNAM STREET OMAHA, NE 68102 ATTENTION: J. TOGER SWANSON, ESQ.

# REGULATORY AGREEMENT AND DECLARATION OF RESTRICTIVE COVENANTS

by and among

CITY OF LOS ANGELES,

as City

and

[FISCAL AGENT],

as Fiscal Agent

and

SP7 APARTMENTS LP,

as Borrower

relating to

\$[32,007,814] City of Los Angeles Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C

Dated as of [September] 1, 2018

519 East 7th Street

### TABLE OF CONTENTS

		Page
Section 1.	Definitions and Interpretation	2
Section 2.	Acquisition, Construction and Equipping of the Project	8
Section 3.	Residential Rental Property	10
Section 4.	Low Income Tenants; Records and Reports	12
Section 5.	Tax-exempt Status of the Note	15
Section 6.	Additional Requirements of the Act	16
Section 7.	Additional Requirements of the City	18
Section 8.	Modification of Covenants	24
Section 9.	Indemnification	24
Section 10.	Consideration	26
Section 11.	Reliance	26
Section 12.	Project in the City	27
Section 13.	Sale or Transfer of the Project; Equity Interests	27
Section 14.	Term	28
Section 15.	Covenants To Run With the Land	29
Section 16.	Burden and Benefit	30
Section 17.	Uniformity; Common Plan	30
Section 18.	Default; Enforcement	30
Section 19.	The Fiscal Agent	32
Section 20.	Recording and Filing.	32
Section 21.	Governing Law	32
Section 22.	Amendments	32
Section 23.	Notices	33
Section 24.	Severability	35
Section 25.	Multiple Counterparts	35
Section 26.	Nondiscrimination and Affirmative Action	35
Section 27.	Business Tax Registration Certificate	35
Section 28.	Financial Obligations Personal to Borrower	36
Section 29.	[Reserved]	36
Section 30.	Child Support Assignment Orders	36

Section 31.	Americans with Disabilities Act	37
Section 32.	Slavery Disclosure Ordinance	37
Section 33.	Requirements of CDLAC	37
EXHIBIT A	DESCRIPTION OF PROJECT SITE	
EXHIBIT B	FORM OF CERTIFICATE OF CONTINUING PROGRAM COMPLIA	NCE
EXHIBIT C	FORM OF INCOME CERTIFICATION	
EXHIBIT D	FORM OF ANNUAL TENANT INCOME RECERTIFICATION	
EXHIBIT E	FORM OF CERTIFICATE OF CDLAC PROGRAM COMPLIANCE	
EXHIBIT F	FORM OF CONSTRUCTION COMPLETION CERTIFICATE	
EXHIBIT G	CDLAC RESOLUTION	
EXHIBIT H	FORM OF SLAVERY DISCLOSURE ORDINANCE CERTIFICATE	
EXHIBIT I	ACCESSIBILITY COVENANTS	
EXHIBIT J	FORM OF CERTIFICATE OF QUALIFIED PROJECT PERIOD	

# REGULATORY AGREEMENT AND DECLARATION OF RESTRICTIVE COVENANTS

THIS REGULATORY AGREEMENT AND DECLARATION OF RESTRICTIVE

COVENANTS (this "Agreement" or this "Regulatory Agreement") is made and entered into and dated as of [September] 1, 2018 by and among the CITY OF LOS ANGELES, a charter city and municipal corporation in the State of California (together with any successor to its rights, duties and obligations, the "City"), [FISCAL AGENT], a national banking association in its capacity as Fiscal Agent (the "Fiscal Agent") under the Funding Loan Agreement dated as of [September] 1, 2018 (the "Funding Loan Agreement") by and among the City, MUFG Union Bank, N.A., as Funding Lender and the Fiscal Agent, with an office in Los Angeles, California, and SP7 APARTMENTS LP, a California limited partnership (the "Borrower").

#### WITNESSETH:

WHEREAS, pursuant to Section 248 of the City Charter of the City and Article 6.3 of Chapter 1 of Division 11 of the Los Angeles Administrative Code, as amended (collectively, the "Law"), and in accordance with Chapter 7 of Part 5 of Division 31 (commencing with Section 52075) of the Health and Safety Code of the State of California, as amended (the "Act"), the City is empowered to issue notes and other evidence of indebtedness to finance the acquisition, construction and equipping of multifamily rental housing; and

WHEREAS, on August 25, 2017, the City indicated its intent to provide for the issuance of a revenue bond or note to finance a portion of the acquisition, rehabilitation, construction and equipping of SP7 Apartments, a multifamily residential rental housing project a portion of which will be located in the City of Los Angeles at 519 East 7th Street on the site more particularly described in Exhibit A hereto (the "Project") and the City Council of the City subsequently adopted a resolution (the "Resolution") authorizing the issuance of a note for such purpose; and

WHEREAS, in furtherance of the purposes of the Law, the Act and the Resolution, and as a part of the City's program of financing housing, the City has issued \$[32,007,814] maximum aggregate principal amount of its Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C (the "Note") the proceeds of which will be used to fund a loan (the "Loan") to the Borrower to finance a portion of the acquisition, construction and equipping of the Project and the acquisition, rehabilitation and equipping of that companion multifamily housing project located at 647 South San Pedro (the "Companion Project"); and

WHEREAS, in order for interest on the Note to be excluded from gross income for federal income tax purposes under the Internal Revenue Code of 1986 (the "Code"), and the below-defined Regulations and rulings with respect to the Code, and in order to comply with the Law, the Act and the policies with respect to the City's housing program, the use and operation of the Project must be restricted in certain respects; and

WHEREAS, the City, the Fiscal Agent and the Borrower have determined to enter into this Regulatory Agreement in order to set forth certain terms and conditions relating to the acquisition, construction and equipping of the Project and in order to ensure that the Project will be used and

operated in accordance with the Code, the Law, the Act and the additional requirements of the City and the California Debt Limit Allocation Committee ("CDLAC");

NOW, THEREFORE, in consideration of the mutual covenants and undertakings set forth herein, and other good and valuable consideration, the receipt and sufficiency of which hereby are acknowledged, the City, the Fiscal Agent and the Borrower hereby agree as follows:

**Section 1. Definitions and Interpretation**. Terms not otherwise defined herein shall have the meanings assigned thereto in the Funding Loan Agreement or Loan Agreement, as applicable. The following terms shall have the respective meanings assigned to them in this Section 1 unless the context in which they are used clearly requires otherwise:

"Act" means Chapter 7 of Part 5 of Division 31 of the Health and Safety Code of the State of California, as the same may be amended from time to time (but only to the extent any such amendments, by their terms or by appropriate election of the City, apply to the Note outstanding as of the effective date of such amendments).

"Adjusted Income" means the adjusted income of a person (together with the adjusted income of all persons who intend to reside with such person in one residential unit) calculated pursuant to Section 142(d)(2)(B) of the Code.

"Affiliated Party" means a limited or general partner or member of the Borrower, a person whose relationship with the Borrower would result in a disallowance of losses under Section 267 or 707(b) of the Code or a person who, together with the Borrower, is a member of the same controlled group of corporations (as defined in Section 1563(a) of the Code, except that "more than 50 percent" shall be substituted for "at least 80 percent" each place it appears therein).

"Agreement" or "Regulatory Agreement" means this Regulatory Agreement and Declaration of Restrictive Covenants, as it may be amended from time to time.

"Area" means the Los Angeles Primary Metropolitan Statistical Area.

"Authorized Borrower Representative" means any person who, at any time and from time to time, may be designated as the Borrower's authorized representative by written certificate furnished to the City and the Fiscal Agent containing the specimen signature of such person and signed on behalf of the Borrower by or on behalf of any authorized general partner of the Borrower if the Borrower is a general or limited partnership, by any authorized managing member of the Borrower if the Borrower is a limited liability company, or by any authorized officer of the Borrower if the Borrower is a corporation, which certificate may designate an alternate or alternates, or in the event that such term shall refer to successors or assigns of the Borrower, any authorized general partner if the successor or the assignee is a general or limited partnership, any authorized managing member if the successor or assignee is a limited liability company or any authorized officer if the successor or the assignee is a corporation. The Fiscal Agent may conclusively presume that a person designated in a written certificate filed with it as an Authorized Borrower Representative is an Authorized Borrower Representative until such time as the Borrower files with it (with a copy to the City) a written certificate identifying a different person or persons to act in such capacity.

"Bond Counsel" means an attorney at law or firm of attorneys of nationally recognized standing in matters pertaining to the validity of, and the Tax-exempt nature of interest on, obligations issued by states and their political subdivisions, selected by the City and duly admitted to the practice of law before the highest court of any state of the United States of America or the District of Columbia but shall not include counsel for the Borrower or the Fiscal Agent.

"Borrower" means SP7 Apartments LP, a California limited partnership, and its successors and assigns.

"CDLAC" means the California Debt Limit Allocation Committee or its successors.

"CDLAC Conditions" has the meaning given such term in Section 33 hereof.

"CDLAC Resolution" means CDLAC Resolution No. 18-011, adopted on March 21, 2018, attached to this Regulatory Agreement as Exhibit G and related to the Project as such resolution may be modified or amended from time to time.

"Certificate of CDLAC Program Compliance" means the Certification of Compliance II for Qualified Residential Rental Projects to be filed with the City at the times specified in Section 33(a) of this Regulatory Agreement, such report to contain the information set forth in and to be in substantially the form attached hereto as Exhibit E or such other form required or otherwise provided by CDLAC from time to time.

"Certificate of Continuing Program Compliance" means the Certificate of Continuing Program Compliance and Statistical Report to be filed by the Borrower with the City and the Fiscal Agent at the times specified in Sections 4(d) and (f) of this Regulatory Agreement, such report to contain the information set forth in and to be in substantially the form attached hereto as Exhibit B or such other form as may from time to time be prescribed by the City.

"Certificate of Qualified Project Period" means the certificate to be filed by the Borrower upon commencement of the Qualified Project Period in substantially the form attached hereto as Exhibit J.

"City" means the City of Los Angeles, a charter city and municipal corporation of the State of California.

"Closing Date" or "Note Closing Date" means the date upon which the Note is initially funded in an amount equal to at least \$50,001.

"Code" means the Internal Revenue Code of 1986; each reference to the Code shall be deemed to include (a) any successor internal revenue law and (b) the applicable regulations whether final, temporary or proposed under the Code or such successor law.

"Companion Project" means the parcel or parcels of real property located at 647 South San Pedro in the City of Los Angeles, California, and all rights and appurtenances thereunto appertaining, as more particularly described in Exhibit A to the Companion Project Regulatory Agreement.

"Companion Project Regulatory Agreement" means that certain Regulatory Agreement and Declaration of Restrictive Covenants of even date herewith, by and among the City, the Fiscal Agent and the Borrower executed in connection with the issuance of the Note and restricting the real property comprising the Companion Project.

"Completion Date" means the date of the completion of the acquisition, construction and equipping of the Project, as that date shall be certified as provided in Section 2(i) hereof as specified in the Construction Completion Certificate.

"Compliance Period" means the period beginning on the first day of the Qualified Project Period and ending on the later of the end of the Qualified Project Period or such later date as set forth in Section 33 of this Regulatory Agreement.

"Construction Completion Certificate" means a written certification signed by an Authorized Borrower Representative certifying among other things to the substantial completion of the Project and delivered to the City, the Fiscal Agent and CDLAC not more than 30 months after the Closing Date, in substantially the form of Exhibit F hereto or such other form required or otherwise provided by CDLAC from time to time.

"Costs of Issuance" means costs of issuing the Note as set forth in the Funding Loan Agreement.

"Determination of Taxability" means either (a) refusal by the Borrower to consent to any amendment or supplement hereto or to the Funding Loan Agreement which, in the written opinion of Bond Counsel delivered to the City, the Fiscal Agent, the Funding Lender and the Borrower, is necessary or advisable to maintain the exclusion of interest on the Note from gross income for federal income tax purposes; or (b) any of (i) the enactment of applicable legislation of which the Fiscal Agent has actual knowledge, (ii) a final judgment or order of a court of original or appellate jurisdiction of which the Fiscal Agent has actual knowledge, (iii) a final ruling or decision of the Internal Revenue Service of which the Fiscal Agent has actual knowledge or (iv) the filing with the Fiscal Agent of an opinion of Bond Counsel, in each case to the effect that the interest on the Note (other than interest on the Note for any period during which such Note is held by a "substantial user" of any facility financed with the proceeds of the Note or a "related person," as such terms are used in Section 147(a) of the Code) is includable in the gross incomes of all recipients thereof for federal income tax purposes. With respect to the foregoing, a judgment or order of a court or a ruling or decision of the Internal Revenue Service shall be considered final only if no appeal or action for judicial review has been filed and the time for filing such appeal has expired.

"Fiscal Agent" means [FISCAL AGENT] in its capacity as fiscal agent under the Funding Loan Agreement, together with its successors and assigns.

"Funding Loan Agreement" means the Funding Loan Agreement dated as of [September] 1, 2018 by and among the City, the Funding Lender and the Fiscal Agent relating to the issuance of the Note, as amended, modified, supplemented or restated from time to time.

"Hazardous Materials" means petroleum and petroleum products and compounds containing them, including gasoline, diesel fuel and oil; explosives; flammable materials;

radioactive materials; polychlorinated biphenyls ("PCBs") and compounds containing them; lead and lead-based paint; asbestos or asbestos-containing materials in any form that is or could become friable; underground or above-ground storage tanks, whether empty or containing any substance; any substance the presence of which on the Project is prohibited by any federal, state or local authority; any substance that requires special handling and any other material or substance now or in the future that (i) is defined as a "hazardous substance," "hazardous material," "hazardous waste," "toxic substance," "toxic pollutant," "contaminant," or "pollutant" by or within the meaning of any Hazardous Materials Law, or (ii) is regulated in any way by or within the meaning of any Hazardous Materials Law.

"Hazardous Materials Laws" means all federal, state, and local laws, ordinances and regulations and standards, rules, policies and other governmental requirements, administrative rulings and court judgments and decrees in effect now or in the future and including all amendments, that relate to Hazardous Materials or the protection of human health or the environment and apply to the Borrower or to the Project. Hazardous Materials Laws include, but are not limited to, the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. Section 9601, et seq., the Resource Conservation and Recovery Act of 1976, 42 U.S.C. Section 6901, et seq., the Toxic Substance Control Act, 15 U.S.C. Section 2601, et seq., the Clean Water Act, 33 U.S.C. Section 1251, et seq., and the Hazardous Materials Transportation Act, 49 U.S.C. Section 5101 et seq., and their state analogs.

"Housing Act" means the United States Housing Act of 1937, as amended, or its successor.

"Income Certification" means, initially, an Income Certification in the form attached hereto as Exhibit C or in such other form as may from time to time be provided by the City to the Borrower and, with respect to recertifications, the Annual Tenant Income Recertification attached hereto as Exhibit D or such other form as may, from time to time, be provided by the City to the Borrower.

"Inducement Date" means August 25, 2017.

"Loan" means the loan of the sale proceeds of the Note by the City to the Borrower pursuant to the Loan Agreement for the purpose of providing funds for the acquisition, construction or rehabilitation and equipping of the Project and the Companion Project.

"Loan Agreement" means the [Construction and Permanent Loan Agreement (Multifamily Housing Back to Back Loan Program)] dated as of [\_\_\_\_\_] 1, 2018 by and among the City, MUFG Union Bank, N.A. and the Borrower, as amended or supplemented from time to time.

"Low Income Tenant" means a tenant whose Adjusted Income does not exceed limits determined in a manner consistent with determinations of lower-income families under Section 8 of the Housing Act, except that the percentage of median gross income that qualifies as lower income shall be [60]% of median gross income for the Area with adjustments for family size. Except as otherwise provided herein, the occupants of a unit in the Project shall not be considered to be Low Income Tenants if all the occupants of a unit are students (as defined in Section 152(f)(2) of the Code) and any one of those students is not (1) a single parent living with his/her children; (2) a student receiving assistance under Title IV of the Social Security Act (Temporary Assistance for Needy Families); (3) a student enrolled in a job training program receiving assistance under the Job Training Partnership Act or under other similar federal, state, or local laws; (4) a student

who was previously under the care and placement responsibility of a foster care program (under part B or E of Title IV of the Social Security Act); or (5) a student who is married and files a joint return. Single parents described in (1) above may not be dependents of another individual and their children may not be dependents of another individual other than their parents. The determination of a tenant's status as a Low Income Tenant shall be made by the Borrower upon initial occupancy of a unit in the Project by such Tenant and annually thereafter and at any time the Borrower has knowledge that the number of occupants in that unit has increased, on the basis of an Income Certification executed by the tenant.

"Low Income Units" means the units in the Project required to be rented to, or held available for occupancy by, Low Income Tenants pursuant to Sections 4(a), 4(b) and 6(a) hereof.

"Net Proceeds" means the total proceeds derived from the issuance, sale and delivery of the Note, representing the total purchase price of the Note, including any premium paid as part of the purchase price of the Note, but excluding the accrued interest, if any, on the Note paid by the initial purchaser of the Note.

"Note" means the note authorized, authenticated and delivered under the Funding Loan Agreement.

"Note Documents" means the Funding Loan Agreement, the Loan Agreement, this Regulatory Agreement, the Companion Project Regulatory Agreement, the Tax Certificate and any other document now or hereafter executed by the Borrower, City, Fiscal Agent or Noteholder in connection with the Note.

"Noteholder" or "Owner" or "Holder" means the party identified as the owner of the Note on the registration books maintained by the Fiscal Agent on behalf of the Governmental Lender.

"Project" means the Project Facilities and the Project Site.

"Project Costs" means, to the extent authorized by the Code, the Regulations, the Law and the Act, any and all costs incurred by the Borrower with respect to the acquisition and construction and equipping, the credit enhancement fees, if any, attributable to the period of the construction of the Project, whether paid or incurred prior to or after the Inducement Date, including, without limitation, costs for site preparation, the planning of housing, related facilities and improvements, the acquisition of property, the removal or demolition of existing structures, the construction or rehabilitation of housing and related facilities and improvements, and all other work in connection therewith, including Qualified Project Costs, and all costs of financing, including, without limitation, the cost of consultant, accounting and legal services, other expenses necessary or incident to determining the feasibility of the Project, contractors' and developer's overhead and supervisors' fees and costs directly allocable to the Project, administrative and other expenses necessary or incident to the Project and the financing thereof (including reimbursement to any municipality, county or other entity or person for expenditures made, for the Project).

"Project Facilities" means the buildings, structures and other improvements on the Project Site to be acquired, constructed, equipped or improved by the Borrower, and all fixtures and other property owned by the Borrower and located on, or used in connection with, such buildings, structures and other improvements constituting the Project. Project Facilities do not include retail

sales facilities, leased office space, commercial facilities or recreational, fitness, parking or business facilities available to members of the general public.

"Project Site" means the parcel or parcels of real property having the street address of 519 East 7th Street in the City of Los Angeles, California and all rights and appurtenances thereunto appertaining, as more particularly described in Exhibit A hereto.

"Qualified Project Costs" means the Project Costs (excluding issuance costs) incurred not earlier than the date 60 days prior to the Inducement Date which either constitute land or property of a character subject to the allowance for depreciation under Section 167 of the Code, or are chargeable to a capital account with respect to the Project for federal income tax and financial accounting purposes, or would be so chargeable either with a proper election by the Borrower or but for the proper election by the Borrower to deduct those amounts; provided, however, that only such portion of the interest accrued on the Note during the construction of the Project shall constitute Qualified Project Costs as bear the same ratio to all such interest or fees, as applicable, as the Qualified Project Costs bear to all Project Costs; and provided further that interest accruing on or after the Completion Date shall not be Qualified Project Costs; and provided finally that if any portion of the Project is being constructed by the Borrower or an Affiliated Party (whether as a general contractor or a subcontractor), "Qualified Project Costs" shall include only (a) the actual out-of-pocket costs incurred by the Borrower or such Affiliated Party in constructing the Project (or any portion thereof), (b) any reasonable fees for supervisory services actually rendered by the Borrower or such Affiliated Party (but excluding any profit component) and (c) any overhead expenses incurred by the Borrower or such Affiliated Party which are directly attributable to the work performed on the Project, and shall not include, for example, intercompany profits resulting from members of an affiliated group (within the meaning of Section 1504 of the Code) participating in the construction of the Project or payments received by such Affiliated Party due to early completion of the Project (or any portion thereof). Qualified Project Costs do not include Costs of Issuance. Notwithstanding anything herein to the contrary, no Project Costs relating to the acquisition of the Project or any assets relating thereto (including, without limitation, rights and interests with respect to development of the Project) shall constitute "Qualified Project Costs" unless, at the time Note proceeds are expended to pay such costs, the Borrower and the seller of such assets are not "related parties" as such term is defined in Section 1.150-1(b) of the Regulations.

"Qualified Project Period" means the period beginning on the first day on which 10% of the dwelling units in the Project are first occupied and ending on the latest of (a) the date which is 15 years after the date on which 50% of the dwelling units in the Project are first occupied, (b) the first date on which no Tax-exempt private activity bond (as that phrase is used in Section 142(d)(2) of the Code) issued with respect to the Project is outstanding or (c) the date on which any assistance provided with respect to the Project under Section 8 of the Housing Act terminates. The CDLAC Conditions apply for a period which, in some cases, exceeds the Qualified Project Period.

"Regulations" means the Income Tax Regulations promulgated or proposed (if deemed appropriate in the opinion of Bond Counsel) by the Department of the Treasury pursuant to the Code from time to time.

"Tax Certificate" means the Tax Certificate as to Arbitrage and the Provisions of Sections 103 and 141-150 of the Internal Revenue Code of 1986 dated the Closing Date, executed

and delivered by the City and the Borrower, as amended, modified, supplemented or restated from time to time.

"Tax-exempt" means, with respect to interest on any obligations of a state or local government, including the Note, that such interest is excluded from gross income for federal income tax purposes (other than interest on the Note for any period during which the Note is held by a "substantial user" of any facility financed with the proceeds of the Note or a "related person," as such terms are used in Section 147(a) of the Code); provided, however, that such interest may be includable as an item of tax preference or otherwise includable directly or indirectly for purposes of calculating other tax liabilities, including any alternative minimum tax or environmental tax, under the Code.

Unless the context clearly requires otherwise, as used in this Regulatory Agreement, words of the masculine, feminine or neuter gender shall be construed to include each other gender and words of the singular number shall be construed to include the plural number, and vice versa. This Regulatory Agreement and all the terms and provisions hereof shall be construed to effectuate the purposes set forth herein and to sustain the validity hereof.

The defined terms used in the preamble and recitals of this Regulatory Agreement have been included for convenience of reference only, and the meaning, construction and interpretation of all defined terms shall be determined by reference to this Section I, notwithstanding any contrary definition in the preamble or recitals hereof. The titles and headings of the sections of this Regulatory Agreement have been inserted for convenience of reference only, and are not to be considered a part hereof and shall not in any way modify or restrict any of the terms or provisions hereof or be considered or given any effect in construing this Regulatory Agreement or any provisions hereof or in ascertaining intent, if any question of intent shall arise.

In the event of any conflict between this Regulatory Agreement and the CDLAC Conditions, the most restrictive requirement shall govern.

**Section 2. Acquisition, Construction and Equipping of the Project**. The Borrower hereby represents as of the date hereof, covenants and agrees with the City and the Fiscal Agent as follows:

- (a) The Borrower has incurred, or will incur within six months after the Closing Date, a substantial binding obligation to commence the acquisition, construction and equipping of the Project, pursuant to which the Borrower is or will be obligated to expend at least 5% of the proceeds of the Loan financed from proceeds of the Note.
- (b) The Borrower's reasonable expectations respecting the total cost of the acquisition, construction and equipping of the Project are accurately set forth in the Borrower Cost Certificate (the "Borrower Cost Certificate") submitted to the City on the Closing Date.
- (c) The Borrower has acquired the Project Site and will, within six months following the Note Closing Date, commence the construction of the Project and will proceed with due diligence to complete the same. Notwithstanding anything herein to the contrary, no Project Costs relating to the acquisition of the Project or any assets relating

thereto (including, without limitation, rights and interests with respect to development of the Project) shall constitute "Qualified Project Costs" unless, at the time Note proceeds are expended to pay such costs, the Borrower and the seller of such assets are not "related parties" as such term is defined in Section 1.150-1(b) of the Regulations. The Borrower reasonably expects to complete the acquisition, construction and equipping of the Project and to expend the full amount of the proceeds of the Loan for Project Costs prior to the date which is 30 months after the Closing Date.

(d) The Borrower agrees that the full amount of each disbursement of Note proceeds pursuant to the Funding Loan Agreement and the Loan Agreement will be applied to pay or to reimburse the Borrower for the payment of Project Costs as set forth in the Borrower Cost Certificate and that, after taking into account each such disbursement, (i) the aggregate disbursements of Note proceeds will have been applied to pay or to reimburse the Borrower for the payment of Qualified Project Costs in an aggregate amount equal to 97% or more of the aggregate disbursements of the Loan; provided, however, that if the Borrower provides the Fiscal Agent with an opinion of Bond Counsel to the effect that the Tax-exempt status of interest on the Note will not be adversely affected if less than the aforesaid percentage, but not less than 95%, is disbursed for such purpose, then the certificate may refer to such lesser percentage as may be specified by Bond Counsel; and (ii) less than 25% of the proceeds of the Note expended relative to the Project Site will be disbursed to pay or to reimburse the Borrower for the cost of acquiring land or rights with respect to land relative to the Project Site (exclusive of the cost of acquiring improvements on such land).

### (e) [Reserved].

(f) No proceeds of the Note will be used to pay or reimburse any cost (i) incurred more than 60 days prior to the Inducement Date, or (ii) incurred more than three years prior to such payment or reimbursement. Any allocation of Note proceeds to the reimbursement of previously incurred costs shall be made not later than 18 months after the later of (i) the date the original expenditure was paid or (ii) the date the Project is placed in service or abandoned. The acquisition, construction and equipping of the Project by the Borrower commenced less than 60 days prior to the Inducement Date, and as of 60 days prior to the Inducement Date (A) neither the Borrower nor any "related person" (as such phrase is used in Section 147(a)(2) of the Code) has made any expenditure in connection with the acquisition, construction or equipping of the Project, (B) no on-site work has been commenced by the Borrower or any related person in connection with the construction of the Project, and (C) no off-site fabrication of any portion of the Project has been commenced by the Borrower or any related person. The Project consists, and shall at all times consist, of property which is land or is subject to the allowance for depreciation provided in Section 167 of the Code.

### (g) [Reserved].

(h) The Borrower (and any Affiliated Party) will not take or omit to take, as is applicable, any action if such action or omission would in any way cause the proceeds from the Loan to be applied in a manner contrary to the requirements of this Regulatory

Agreement, nor will it take or omit to take any such action if the Borrower (or any Affiliated Party) knows that such action or omission may cause the proceeds from the sale of the Note to be applied in a manner contrary to the Funding Loan Agreement, the Loan Agreement, the Law, the Act or the Code.

- (i) The Borrower shall, on the Completion Date, evidence the Completion Date by providing a Construction Completion Certificate to CDLAC, the Fiscal Agent and the City, signed by the Authorized Borrower Representative. Notwithstanding the foregoing, such certificate may state that it is given without prejudice to any rights of the Borrower against third parties for the payment of any amount not then due and payable which exist at the date of such certificate or which may subsequently exist. The Construction Completion Certificate shall be delivered to the Fiscal Agent no later than the date 36 months from the Closing Date unless the Borrower delivers to the Fiscal Agent a certificate of the City consenting to an extension of such date, accompanied by an opinion of Bond Counsel to the effect that such extension will not result in interest on the Note being included in gross income for federal income tax purposes.
- (j) The Borrower agrees to spend additional moneys for payment of any costs of the Project sufficient to reduce the portion of Note proceeds (A) spent on land by the Borrower relative to the Project Site to an amount that is less than 25% of the amount of Note proceeds spent by the Borrower relative to the Project Site for all purposes and (B) spent on costs of the Project paid or incurred by or on account of the Borrower or any "related person" (as such term is used in Section 147(a)(2) of the Code) on or after the date 60 days prior to the Inducement Date and chargeable to the capital account of the Project (or so chargeable either with a proper election by the Borrower to deduct such amounts, within the meaning of Treasury Regulation Section 1.103-8(a)(1)) so that the amount of Note proceeds expended on such Qualified Project Costs are at least 97% of the amount of Note proceeds spent for all purposes related to the Project, except that, upon receipt by the Borrower, the Fiscal Agent and the City of an approving opinion of Bond Counsel, the percentage of such amounts so used may be 95%.
- (k) No Note proceeds shall be expended to acquire any structures other than buildings within the meaning of Section 147(d) of the Code.
- **Section 3. Residential Rental Property**. The Borrower hereby acknowledges and agrees that the Project is to be owned, managed and operated as a "qualified residential rental project" (within the meaning of Section 142(d) of the Code) for a term equal to the Qualified Project Period. To that end, and for the Qualified Project Period, the Borrower hereby represents, covenants, warrants and agrees as follows:
  - (a) The Project Facilities will be developed for the purpose of providing multifamily residential rental property, and the Borrower will own, manage and operate the Project Facilities as a project to provide multifamily residential rental property comprising a building or structure or several interrelated buildings or structures, together with any functionally related and subordinate facilities, and no other facilities in accordance with Section 142(d) of the Code and Section 1.103-8(b) of the Regulations, the Law and the Act, and in accordance with such requirements as may be imposed thereby on

the Project from time to time. For purposes of this Section 3(a), the term "functionally related and subordinate facilities" includes facilities for use by the tenants (for example, swimming pools, other recreational facilities and parking areas) and other facilities which are reasonably required for the Project, for example, heating and cooling equipment, trash disposal equipment and units for resident managers and maintenance personnel. Substantially all of the Project will contain such units and functionally related and subordinate facilities.

- (b) All of the dwelling units in the Project will be similarly constructed units, and each Low Income Unit in the Project will contain complete separate and distinct facilities for living, sleeping, eating, cooking and sanitation for a single person or a family, including a sleeping area, bathing and sanitation facilities and cooking facilities equipped with a cooking range and oven, a sink and a refrigerator. Each of the Accessible Housing Units (as defined in Exhibit I hereto) shall also comply with the requirements of Exhibit I. Notwithstanding the foregoing, a unit shall not fail to be treated as a residential unit merely because such unit is a single-room occupancy unit within the meaning of Section 42(i)(3)(B)(iv) of the Code even though such housing may provide eating, cooking and sanitation facilities on a shared basis
- (c) None of the dwelling units in the Project will at any time be utilized on a transient basis or will ever be used as a hotel, motel, dormitory, fraternity house, sorority house, rooming house, nursing home, hospital, sanitarium, rest home or trailer court or park. Notwithstanding the foregoing, single-room occupancy units provided under Section 42(i)(3)(B)(iv) of the Code shall not be considered to be utilized on a transient basis.
- (d) No part of the Project will at any time be owned by a cooperative housing corporation, nor shall the Borrower take any steps in connection with a conversion to such ownership or uses. Other than filing a condominium map and a final subdivision map on the Project and obtaining a Final Subdivision Public Report from the California Department of Real Estate, the Borrower shall not take any steps in connection with a conversion of the Project to condominium ownership during the Qualified Project Period.
- (e) All of the dwelling units (which shall not include any manager units) in the Project will be available for rental on a continuous basis to members of the general public, and the Borrower will not give preference to any particular class or group in renting the dwelling units in the Project, except to the extent of the following: (1) any dwelling units that are required to be leased or rented to Low Income Tenants, (2) the requirements of any regulatory agreement executed between the Borrower and HUD or between the Borrower and a subordinate lender (including the City), (3) the requirements of any Section 8 Housing Assistance Payments Contract with respect to the Project, (4) any preference Borrower gives to a class of persons permitted to be given preference pursuant to the Code, State law and other applicable federal law and (5) Accessible Housing Units shall be made available to persons with disabilities as provided in Exhibit I.
- (f) The Project Site consists of a parcel or parcels that are contiguous except for the interposition of a road, street or stream, and all of the Project Facilities comprise a

single geographically and functionally integrated project for residential rental property, as evidenced by the ownership, management, accounting and operation of the Project.

- (g) No dwelling unit in the Project shall be occupied by the Borrower; provided, however, that if the Project contains five or more dwelling units, this subsection shall not be construed to prohibit occupancy of not more than one dwelling unit by one or more resident managers or maintenance personnel any of whom may be the Borrower.
- (h) The Project shall be maintained in conformity with the habitability and fire codes of the City of Los Angeles.
- (i) The Project shall be managed in a manner consistent with prudent property management standards and in compliance with all state and local laws, ordinances and regulations relating thereto.
- (j) Should involuntary noncompliance with the provisions of Regulations Section 1.103-8(b) be caused by fire, seizure, requisition, foreclosure, transfer of title by deed in lieu of foreclosure, change in a federal law or an action of a federal agency after the Closing Date which prevents the City from enforcing the requirements of the Regulations, or condemnation or similar event, the Borrower covenants that, within a "reasonable period" determined in accordance with the Regulations, and subject to the provisions of the Funding Loan Agreement and the Loan Agreement, it will either prepay the Loan or apply any proceeds received as a result of any of the preceding events to reconstruct the Project to meet the requirements of Section 142(d) of the Code and the Regulations.

**Section 4. Low Income Tenants; Records and Reports**. Pursuant to the requirements of the Code and the City, the Borrower hereby represents, warrants and covenants as follows:

- (a) The Project will be developed for the purpose of providing multifamily residential rental property, and the Borrower will own, manage and operate the Project as a project to provide multifamily residential rental property comprising a building or structure or several interrelated buildings or structures, together with any functionally related and subordinate facilities, and no other facilities in accordance with the Law and the Act, and in accordance with such requirements as may be imposed thereby on the Project from time to time. Within 30 days after the date on which 50% of the dwelling units in the Project are occupied by tenants providing an Income Certification, the Borrower shall execute and deliver to the City (with a copy to the Los Angeles Housing and Community Investment Department, Occupancy Monitoring Section, 1200 West 7th Street, 8th Floor, Los Angeles, CA 90017), and a copy to CDLAC and the Fiscal Agent a Certificate of Qualified Project Period.
- (b) Commencing on the first day of the Qualified Project Period, Low Income Tenants shall occupy at least [40]% of all completed and occupied units in the Project (excluding units occupied by property managers) before any additional units are occupied by persons who are not Low Income Tenants; and for the Qualified Project Period no less than [40]% of the total number of completed units of the Project (excluding units occupied

by property managers) shall at all times be rented to and occupied by Low Income Tenants. For the purposes of this subsection (b), a vacant unit which was most recently occupied by a Low Income Tenant is treated as rented and occupied by a Low Income Tenant until reoccupied, other than for a temporary period of not more than 31 days, at which time the character of such unit shall be redetermined. In determining whether the requirements of this subsection (b) have been met, fractions of units shall be treated as entire units.

- (c) No tenant qualifying as a Low Income Tenant shall be denied continued occupancy of a unit in the Project because, after admission, such tenant's Adjusted Income increases to exceed the qualifying limit for Low Income Tenants; provided, however, that should a Low Income Tenant's Adjusted Income, as of the most recent determination thereof, exceed 140% of the then applicable income limit for a Low Income Tenant of the same family size, the next available unit of comparable or smaller size must be rented to (or held vacant and available for immediate occupancy by) a Low Income Tenant; and provided further that, until such next available unit is rented to a tenant who is not a Low Income Tenant, the former Low Income Tenant who has ceased to qualify as such shall be deemed to continue to be a Low Income Tenant for purposes of the [40]% requirement of paragraph (b) of this Section 4 (if applicable). If the Project consists of more than one building, this requirement shall apply on a building-by-building basis.
- The Borrower will obtain, complete and maintain on file Income Certifications from each Low Income Tenant, including (i) an Income Certification dated no later than the day prior to the initial occupancy of such Low Income Tenant in the Project and (ii) thereafter, annual Income Certifications dated as of the anniversary date of each initial Income Certification. The Borrower will obtain such additional information as may be required in the future by the State of California, by the City and by Section 142(d) of the Code, as the same may be amended from time to time, or in such other form and manner as may be required by applicable rules, rulings, policies, procedures, Regulations or other official statements now or hereafter promulgated, proposed or made by the Department of the Treasury or the Internal Revenue Service with respect to obligations which are Tax-exempt under Section 142(d) of the Code. A copy of the most recent Income Certification for Low Income Tenants commencing or continuing occupation of a Low Income Unit (and not previously filed with the City) shall be attached to the Certificate of Continuing Program Compliance which is to be filed with the City no later than the fifteenth day of each month until such report indicates compliance with Section 4(b) and thereafter on the fifteenth day of each [ ] and [ 1 until the end of the Qualified Project Period. The Borrower shall make a good-faith effort to verify that the income information provided by an applicant in an Income Certification is accurate by obtaining the acceptable forms of verification enumerated in Chapter 3 of the most current, amended edition of HUD Handbook 4350.3, or such instruction by HUD that may supersede this handbook, and any additional documentation that the City shall deem relevant, such as the two most recent years' tax returns or other forms of independent verification satisfactory to the City.
- (e) The Borrower will use its best efforts to maintain complete and accurate records pertaining to the Low Income Units, and will with reasonable notice permit any duly authorized representative of the City, the Fiscal Agent, the Department of the Treasury

or the Internal Revenue Service to inspect the books and records of the Borrower pertaining to the Project during regular business hours, including those records pertaining to the occupancy of the Low Income Units.

- The Borrower will prepare and submit to the City and the Fiscal Agent, no (f) later than the fifteenth day of each month following the receipt by the Fiscal Agent of the Construction Completion Certificate to and including the month in which such report indicates that [40]% of the occupied units (excluding units occupied by property managers) are occupied by Low Income Tenants, and thereafter no later than the fifteenth day of each until the end of the Qualified Project Period, a Certificate of Continuing Program Compliance executed by the Borrower stating (i) the percentage of the dwelling units of the Project which were occupied or deemed occupied, pursuant to subsection (b) of this Section 4, by Low Income Tenants during such period; (ii) that either (A) no unremedied default has occurred under this Regulatory Agreement, or (B) a default has occurred, in which event the certificate shall describe the nature of the default in detail and set forth the measures being taken by the Borrower to remedy such default; and (iii) that, to the knowledge of the Borrower, no Determination of Taxability has occurred, or if a Determination of Taxability has occurred, setting forth all material facts relating thereto.
- (g) On or before each February 15 during the Qualified Project Period, the Borrower will submit to the City a draft of the completed Internal Revenue Service Form 8703 or such other annual certification required by the Code to be submitted to the Secretary of the Treasury as to whether the Project continues to meet the requirements of Section 142(d) of the Code. On or before each March 31 during the Qualified Project Period, the Borrower will submit such completed form to the Secretary of the Treasury, regardless of whether or not the City has responded to such draft.
- (h) Subject to the requirements of any Section 8 Housing Assistance Payments Contract with respect to the Project, each lease or rental agreement pertaining to a Low Income Unit shall contain a provision to the effect that the Borrower has relied on the Income Certification and supporting information supplied by the Low Income Tenant in determining qualification for occupancy of the Low Income Unit and that any material misstatement in such certification (whether or not intentional) will be cause for immediate termination of such lease or rental agreement. Each such lease or rental agreement shall also provide that the tenant's income is subject to annual certification in accordance with Section 4(c) hereof and to recertification if the number of occupants in the units changes for any reason (other than the birth of a child to an occupant of such unit) and that if upon any such certification such tenant's Adjusted Income exceeds 140% of the then applicable income limit for a Low Income Tenant of the same family size, such tenant may cease to qualify as a Low Income Tenant, and such tenant's rent is subject to increase. Notwithstanding anything in this Section 4(h) to the contrary, such tenant's rent may be increased only pursuant to Section 7(l) hereof.
- (i) Pursuant to the CDLAC Conditions attached hereto and for the entire term of the Regulatory Agreement, the Project, together with the Companion Project, shall consist of 81 units, including one manager unit, of which at least 9 qualified residential

units shall be rented or held vacant for rental for persons or families whose income is at or below 50% of the area median income and 71 qualified residential units shall be rented or held vacant for rental for persons or families whose income is at or below 60% of the area median income as shown in the chart below:

Unit Type	Units at 30% AMI	Units at 40% AMI	Units at 50% AMI	Market Rate (Manager's Unit)	Total Number of Units
Studio	25	16	39		80
Two Bedroom				1	1
Total	25	16	39	1	81

**Section 5. Tax-exempt Status of the Note**. The Borrower and the City make the following representations, warranties and agreements for the benefit of the holder of the Note from time to time:

- (a) The Borrower and the City will not knowingly take or permit actions within their control, or omit to take or cause to be taken, as is appropriate, any action that would adversely affect the Tax-exempt nature of the interest on the Note and, if either should take or permit, or omit to take or cause to be taken, any such action, it will take all lawful actions necessary to rescind or correct such actions or omissions promptly upon obtaining knowledge thereof, provided that the Borrower shall not have violated these covenants if the interest on the Note becomes taxable to a person solely because such person is a "substantial user" of the Project or a "related person" within the meaning of Section 147(a) of the Code.
- (b) The Borrower and the City will take such action or actions as may be necessary, in the written opinion of Bond Counsel filed with the City and the Fiscal Agent, with a copy to the Borrower and the Funding Lender, to comply fully with all applicable rules, rulings, policies, procedures, Regulations or other official statements promulgated, proposed or made by the Department of the Treasury or the Internal Revenue Service pertaining to obligations the interest on which is Tax-exempt under Section 142(d) of the Code.
- (c) The Borrower and the City will file or record such documents and take such other steps as are necessary, in the written opinion of Bond Counsel filed with the City and the Fiscal Agent, with a copy to the Borrower and the Funding Lender, in order to insure that the requirements and restrictions of this Regulatory Agreement will be binding upon all owners of the Project, including, but not limited to, the execution and recordation of this Regulatory Agreement in the real property records of the County of Los Angeles.
- (d) The Borrower will not knowingly enter into any agreements which would result in the payment of principal or interest on the Note being "federally guaranteed" within the meaning of Section 149(b) of the Code.
- (e) Subject to Section 14 hereof, the Borrower hereby covenants to include the requirements and restrictions contained in this Regulatory Agreement in any documents

transferring any interest in the Project prior to the expiration of the Qualified Project Period to another person to the end that such transferee has notice of, and is bound by, such restrictions, and to obtain the agreement from any transferee to abide by all requirements and restrictions of this Regulatory Agreement; provided, however, that so long as any former Borrower has no remaining interest in the Project, such former Borrower shall have no obligation to monitor such transferee's compliance with such restrictions, and such former Borrower shall incur liability if such transferee fails to comply with such restrictions only in proportion to its then remaining interest.

- (f) The Borrower and any related party (as defined in Section 1.150-1(b) of the Regulations) thereto shall not acquire the Note in an amount related to the amount of the Loan.
- **Section 6.** Additional Requirements of the Act. In addition to the requirements set forth in Sections 2 through 5, and without limiting any additional requirements in Section 7, during the Qualified Project Period, the Borrower and the City hereby agree to comply with each of the requirements of the Act, and, without limiting the foregoing, the Borrower hereby specifically agrees to comply with each of the requirements set forth in this Section 6, as follows:
  - (a) As provided in Section 52097.5 of the Act, not less than [40]% of the total number of units in the Project (excluding units occupied by property managers) shall be reserved for occupancy by tenants whose adjusted gross income does not exceed [60]% of the median adjusted gross income for the Area, adjusted for family size, as determined pursuant to Section 8 of the Housing Act.
  - (b) The rents paid by the tenant for the units reserved pursuant to paragraph (a) of this Section (excluding any supplemental rental assistance from the State, the federal government, or any other public agency to those occupants or on behalf of those units) shall not exceed the amount derived by multiplying 30% times [60]% of the median gross income for the Area, adjusted for family size, as determined pursuant to Section 8 of the Housing Act.
  - (c) During the Qualified Project Period the Borrower shall file Certificates of Continuing Program Compliance in the form and at the time required by Sections 4(d) and (f) hereof that shall contain sufficient information to allow the City to file any annual report required by the Act or pursuant to California Government Code Section 8855.5, and the Borrower shall provide to the California Debt and Investment Advisory Commission any annual report information required by California Government Code Section 8855(k)(1) until the later of the date the Note is no longer outstanding or the proceeds of the Note have been fully spent.
  - (d) No portion of the Note shall be used to finance the acquisition, construction, rehabilitation, equipping, refinancing or development of commercial property for lease.
  - (e) The Borrower shall not apply selection criteria to certificate holders under Section 8 of the United States Housing Act of 1937, as amended, that are more burdensome than the criteria applied to all other prospective tenants.

- (f) Following the expiration or termination of the Qualified Project Period with respect to the Project, except in the event of foreclosure and redemption of the Note, deed in lieu of foreclosure, eminent domain or action of a federal agency preventing enforcement, units required to be reserved for occupancy pursuant hereto shall remain available to any eligible household occupying a reserved unit at the date of expiration or termination, at a rent not greater than the amount set forth in (b) above, until the earliest of any of the following occur:
  - (i) The household's income exceeds 140% of the maximum eligible income specified herein;
  - (ii) The household voluntarily moves or is evicted for "good cause." "Good cause," for the purposes of this Section, means the nonpayment of rent or allegation of facts necessary to prove major, or repeated minor, violations of material provisions of the occupancy agreement which detrimentally affect the health and safety of other persons or the structure, the fiscal integrity of the Project, or the purposes or special programs of the Project;
  - (iii) Thirty (30) years after the date of the commencement of the Qualified Project Period relative to the Project; and
  - (iv) The Borrower pays the relocation assistance and benefits to tenants as provided in subdivision (b) of Section 7264 of the California Government Code.
- (g) During the three years prior to expiration of the Qualified Project Period, the Borrower shall continue to make available to Low Income Tenants reserved units that have been vacated to the same extent that nonreserved units are made available to tenants other than Low Income Tenants.

Notwithstanding Section 1461 of the California Civil Code, the provisions hereof shall run with the land and may be enforced either in law or in equity by any resident, local agency, entity or any other person adversely affected by the Borrower's failure to comply with this Regulatory Agreement.

- **Section 7. Additional Requirements of the City**. In addition to, and not in derogation of, the requirements set forth in the preceding and following sections of this Regulatory Agreement, each of which is hereby incorporated in this Section as a specific requirement of the City, whether or not required by California or federal law, the Borrower represents, warrants, covenants and agrees as follows:
  - (a) The Borrower shall promptly provide to the City such information with respect to the Project or the Note as the City shall from time to time request. The Borrower shall provide written notice to the City of receipt of a certificate of occupancy or other official authorization to occupy the Project immediately upon receipt.
  - (b) The Low Income Units shall be of comparable quality to all other units in the Project, shall be dispersed throughout the Project, and shall offer a range of size and number of bedrooms comparable to those units which are available to other tenants; and

Low Income Tenants shall have access to and enjoyment of all common areas and facilities of the Project on the same basis as tenants of other units.

(c) The Borrower agrees that it will not discriminate in the rental of units or in its employment practices against any employee or applicant for employment because of the applicant's race, religion, national origin, ancestry, sex, age, sexual orientation, gender identity/expression, transgender status, disability (except to give priority to persons with disabilities for the occupancy of Accessible Housing Units), marital status, domestic partner status or medical condition. All contracts entered into by the Borrower which relate to the Project shall contain a like provision. The Borrower shall comply with the provisions of Sections 10.8.2 and 10.8.4 of the Administrative Code of the City, the provisions of which are hereby incorporated by reference.

#### (d) [Reserved].

- (e) For the Qualified Project Period, the Borrower will comply with the provisions of the Unruh Civil Rights Act, including, without limitation, Sections 51.2 and as applicable, 51.3 of the California Civil Code, as amended, and Sections 45.50 et seq. of the Los Angeles Municipal Code, as amended.
- The lease to be utilized by the Borrower in renting any residential units in (f) the Project to Low Income Tenants shall provide for termination of the lease and consent by such person to immediate eviction, subject to applicable provisions of California law, for any tenant who fails to qualify as a Low Income Tenant and who has made a material misrepresentation on the Income Certification as to such tenant's qualification as a Low Income Tenant. All such leases shall contain clauses, among others, wherein each individual lessee (i) certifies the accuracy of the statements made in the Income Certification and (ii) agrees that the family income, family composition and other eligibility requirements shall be deemed substantial and material obligations of the lessee's tenancy; that the lessee will comply promptly with all requests for information with respect thereto from the Borrower or the City; and that the lessee's failure to provide accurate information in the Income Certification or refusal to comply with a request for information with respect thereto shall be deemed a violation of a substantial obligation of the lessee's tenancy and shall be a default thereunder. Additionally, such lease shall contain provisions informing any tenant of the possibility of rental payment increases in accordance with the terms of this Regulatory Agreement.
- (g) All Income Certifications will be maintained on file at the Project or, with the prior written consent of the City, at the principal place of business of the Borrower or the property manager of the Project, so long as this Regulatory Agreement is in effect and for five years thereafter with respect to each Low Income Tenant who occupied a residential unit in the Project during the Qualified Project Period.
- (h) The Borrower will accept as tenants, on the same basis as all other prospective tenants, persons who are recipients of federal certificates for rent subsidies pursuant to the existing program under Section 8 of the Housing Act, or its successor. The

Borrower shall not apply selection criteria to Section 8 certificate or voucher holders that are more burdensome than criteria applied to all other prospective tenants.

- (i) The Borrower shall submit to the City (i) at the times specified in Sections 4(d) and (f) herein, a Certificate of Continuing Program Compliance, which shall include the information called for therein, including occupancy records for all units in the Project, and (ii) within 15 days after receipt of a written request, any other information or completed forms requested by the City, in each case in order to comply with reporting requirements of the Internal Revenue Service or the State of California, including, without limitation, information necessary for the City to file any periodic report, or any other information concerning the Project as the City may reasonably request.
- (i) All workers performing construction or rehabilitation work for the Project employed by the Borrower or by any contractor or subcontractor shall be compensated in an amount no less than the greatest of (i) the general prevailing rate of per diem wages ("Prevailing Wages") as determined pursuant to Labor Code Sections 1770-1781 and implementing regulations of the Department of Industrial Relations, (ii) the general prevailing rate of per diem wages as determined by the U.S. Labor Department pursuant to the Davis-Bacon Act under 40 U.S.C.S. 3141-3148 and implementing regulations ("Davis-Bacon Wages"), if applicable; and (iii) the "Living Wage" as determined by the policies and procedures of the City of Los Angeles. The Borrower shall comply with all reporting and recordkeeping requirements of the City's prevailing wage policy. The Borrower shall, and shall cause the contractors and subcontractors to, submit data and documents related to Prevailing Wages or Davis-Bacon Wages, if applicable, using the LCP Tracker or comparable HCIDLA-approved program. The fee for the LCP Tracker, or comparable HCIDLA-approved program, will be in the amount equal to 0.03% of the total construction cost, which fee shall be paid in full to the City within 30 days of execution of this Agreement.
- (k) The City may, at its option and at its expense, at any time appoint an administrator to administer this Regulatory Agreement and to monitor performance by the Borrower of the terms, provisions and requirements hereof. Following any such appointment, the Borrower shall comply with any request by the City to deliver to such administrator, in addition to or instead of the City, any reports, notices or other documents required to be delivered pursuant hereto, and upon reasonable notice to the Borrower to make the Project and the books and records with respect thereto available for inspection during regular business hours by such administrator as an agent of the City.
- (l) If upon the annual certification or recertification required in Section 4(d) a tenant's Adjusted Income exceeds 140% of the then applicable income limit for a Low Income Tenant of the same family size, all rental limits herein previously applicable to the unit occupied for such tenant shall continue to apply until the next available unit is rented to a tenant who is a Low Income Tenant.

- (m) The Borrower shall give written notice to Low Income Tenants at the following four points in time:
  - (i) Upon initial move-in/lease execution, the Borrower shall give written notice to all tenants of Low Income Units, of the duration of the rent restrictions under this Regulatory Agreement. The Borrower must maintain, in its files, a copy of each notice containing each tenant's signed acknowledgment of the notice required hereunder. The notice shall, at the least, contain language that the rent restrictions under this Regulatory Agreement shall be for a term equal to the later of the expiration of: (A) the Qualified Project Period; or (B) the CDLAC Conditions. Upon termination of the rent restriction period under this Regulatory Agreement, rents may be set at market rates unless otherwise restricted by some other legal, regulatory, or contractual requirement.
  - (ii) Twelve months prior to the termination of the rent restriction period under this Regulatory Agreement, the Borrower must give written notice to its tenants of the termination of the restrictions on the Low Income Units before their rents may be raised to market rent levels. The Borrower must also give written notice, pursuant to California Government Code Section 65863.10, to the Mayor of the City of Los Angeles, the Housing Authority of the City of Los Angeles, the California Department of Housing and Community Development and the Los Angeles Housing and Community Investment Department.
  - (iii) Six months prior to the termination of the rent restriction period under this Regulatory Agreement, the Borrower must give written notice to its tenants of the termination of the restrictions on the Low Income Units before their rents may be raised to market rent levels. The Borrower must also give written notice, pursuant to California Government Code Section 65863.10, to the Mayor of the City of Los Angeles, the Housing Authority of the City of Los Angeles, the California Department of Housing and Community Development and the Los Angeles Housing and Community Investment Department.
  - (iv) Ninety days prior to the termination of the rent restriction period under this Regulatory Agreement, the Borrower must again give written notice to its tenants of the termination of the restrictions on the Low Income Units before their rents may be raised to market rent levels.

Unless the Borrower meets the requirements of California Government Code 65863.13, pursuant to California Government Code 65863.11, prior or concurrent with the 12-month notice referenced above in (ii), the Borrower must provide notice of the opportunity to offer to purchase the assisted housing development to all qualified entities on the list maintained by the California Department of Housing and Community Development as well as to those qualified entities that contact the Borrower directly. The notice shall conform to the requirements of California Government Code 65863.11(h) and shall be sent to the entities by registered or certified mail, return receipt requested. The Borrower shall also post a copy of the notice in a conspicuous place in the common area of the Project.

- The Borrower shall, on the Note Closing Date, pay to the City its initial fee and thereafter pay to the City its ongoing fees with respect to the issuance of the Note as follows. The Borrower shall pay the City an initial fee immediately upon issuance of the Note equal to \$[80,019.54] (.25% of the aggregate maximum principal amount of the Note issuable under the Funding Loan Agreement (\$[32,007,814])). In addition, the Borrower shall, as compensation for the City's monitoring of the provisions of this Regulatory Agreement, pay to the City, semiannually in arrears, prorated for the initial payment, on the first day of each [ ] and [ ] commencing 1, 201[9] for the period from the date of issuance of the Note through the later of: (i) the end of the Qualified Project Period; or (ii) the termination of the CDLAC Conditions, prorated for the initial and any subsequent partial period, a semiannual amount equal to the greater of: (A) during the period from the Closing Date to the Conversion Date, \$1,250 or one-half of 0.125% of the maximum principal amount of Note issuable under the Funding Loan Agreement (\$[32,007,814]); and (B) from and after the Conversion Date \$1,250 or one half of 0.125% of the outstanding principal amount of the Note Outstanding under the Funding Loan Agreement immediately after the Conversion Date; or in either case, such lesser amount as shall be necessary in the opinion of Bond Counsel to preserve the exemption of interest on the Note from gross income for federal income tax purposes. Throughout the term of this Agreement, the Fiscal Agent, or the City, as applicable, shall provide an invoice to the Borrower at least 30 days prior to the due date of each such payment (and, if applicable, a copy of which shall be provided to the City) and shall collect such payments from the Borrower and immediately remit such funds to the City. In the event of any prepayment of the Note in whole, prior to the later of: (i) the end of the Qualified Project Period; or (ii) the termination date of the CDLAC Conditions, the Borrower, at its election, shall either: (A) pay to the City, on or before such payment, an amount equal to the present value of the remaining City fees payable hereunder, as calculated by the City, using a discount rate equal to the yield on the date of prepayment on the United States treasury security maturing on the date nearest the later of: (1) the end of the Qualified Project Period or (2) the termination date of the CDLAC Conditions, or such lesser amount as shall be necessary in the opinion of Bond Counsel to preserve the exemption of interest on the Note from gross income for federal income tax purposes; or (B) enter into a fiscal agent agreement with a corporate fiscal agent acceptable to the City requiring the fiscal agent appointed thereunder to bill and collect from the Borrower and to pay to the City on an annual basis, in arrears on or before each [ 11, the annual fee described above. The Borrower shall bear the cost of such fiscal agent through the term of this Regulatory Agreement. The Borrower shall not be required to pay the fee described in the preceding sentences if the Note is prepaid in whole under circumstances which permit termination of this Regulatory Agreement pursuant to Section 14 hereof. Provisions identical to the foregoing appear in Section 7(n) of the Companion Project Regulatory Agreement. Such fees are calculated and due in connection with the Project and Companion Projects as a whole. To the extent that the foregoing fees are paid in full pursuant to this Regulatory Agreement, no additional fees shall be due under the Companion Project Regulatory Agreement.
- (o) The Borrower shall pay to the City a processing fee equal to: (i) prior to the Conversion Date the greater of \$5,000 or 0.125% of the maximum principal amount

of the Note issuable under the Funding Loan Agreement; and (ii) following the Conversion Date the greater of \$5,000 or 0.125% of the principal amount of the Note outstanding under the Funding Loan Agreement immediately after the Conversion Date, plus any expenses incurred by the City, including, without limitation, Bond Counsel, City attorney and financial advisor fees, as a condition to the consideration and receipt of any consent, approval, amendment, transfer or waiver requested of the City with respect to the Project, the Project Site or the Note. The City shall provide an invoice directly to the Borrower for such amounts. Provisions identical to the foregoing appear in Section 7(o) of the Companion Project Regulatory Agreement. Such fees are calculated and due in connection with the Project and Companion Projects as a whole. To the extent that the foregoing fees are paid in full pursuant to this Regulatory Agreement, no additional fees shall be due under the Companion Project Regulatory Agreement.

- (p) The Borrower shall pay the City its then-current fees in connection with any consent, approval, transfer, amendment or waiver requested of the City, together with any expenses incurred by the City in connection therewith.
- (q) The Fiscal Agent shall report to the City in writing semiannually, within 10 days of each June 30 and December 31, the principal amount of the Note outstanding as of such June 30 or December 31, as appropriate.
- (r) The Borrower shall promptly provide the City such information with respect to the Project or the Note as the City shall from time to time request.
- (s) The Borrower shall include the City as an additional insured on all liability insurance policies relating to the Borrower or the Project.
- (t) The Borrower shall not rent any Low Income Unit to: (i) any individual who (A) holds an ownership interest in the Borrower or any general partner or member (or owner of such general partner or member) of the Borrower, (B) is an officer, board member, employee or agent of, or consultant to, the Borrower or any general partner or member thereof or owner of such general partner or member or (C) is a developer of the Project (collectively, an "Owner/Developer"); (ii) any Immediate Family Member of an Owner/Developer ("Immediate Family Members" consists of: (A) spouses, (B) children, (C) parents and grandparents, (D) siblings, (E) in-laws, including brothers/sisters-in-law and mothers/fathers-in-law and sons/daughters-in-law or (F) significant others or domestic partners); or (iii) any elected official or his or her spouse/partner who participated in the deliberative process, vote or consideration of legislative action regarding the issuance of the Note or other loan in support of the Project, unless such person otherwise qualifies for tenancy under this Agreement and such tenancy is approved in writing by the City.
- (u) The Borrower shall include a certification in each tenant application that the applicant is not an Owner/Developer, an elected official who participated in the issuance of the Note or an Immediate Family Member thereof. The Borrower recognizes

and agrees that the penalty for violation of the above covenant shall be a fine of \$5,000 per violation/ per unit.

(v) Neither the Borrower nor any general partner thereof shall issue any publicity release or other communication to any print, broadcast or online media, post any sign or in any other way identify the City as the source of the financing provided for the Project, without the prior written approval of the City (provided that nothing herein shall prevent the Borrower or any general partner thereof from identifying the City as the source of such financing to the extent that the Borrower or any general partner thereof is required to do so by disclosure requirements applicable to publicly held companies).

Any of the foregoing requirements of the City may be expressly waived by the City in writing in the City's sole discretion, but (i) no waiver by the City of any requirement of this Section 7 shall, or shall be deemed to, extend to or affect any other provision of this Regulatory Agreement, including particularly but without limitation the provisions of Sections 2 through 6 hereof, except to the extent the City has received an opinion of Bond Counsel that any such provision is not required by the Act or the Law and may be waived without adversely affecting the exclusion from gross income of interest on the Note for federal income tax purposes; and (ii) any requirement of this Section 7 shall be void and of no force and effect if the City and the Borrower receive a written opinion of Bond Counsel to the effect that compliance with any such requirement would cause interest on the Note to become includable in gross income for federal income tax purposes, if such opinion is accompanied by a copy of a ruling from the Internal Revenue Service to the same effect or to the effect that compliance with such requirement would be in conflict with the Act or the Law.

**Section 8. Modification of Covenants**. The Borrower, the Fiscal Agent and the City hereby agree as follows:

- (a) To the extent any amendments to the Law, the Act, the Regulations or the Code shall, in the written opinion of Bond Counsel filed with the City, the Fiscal Agent and the Borrower (with a copy to the Funding Lender), impose requirements upon the ownership or operation of the Project more restrictive than those imposed by this Regulatory Agreement in order to maintain the Tax-exempt status of interest on the Note, this Regulatory Agreement shall be deemed to be automatically amended, without the consent or approval of any other person, to impose such additional or more restrictive requirements. The parties hereto hereby agree to execute such amendment hereto as shall be necessary to document such automatic amendment hereof.
- (b) To the extent that the Law, the Act, the Regulations or the Code, or any amendments thereto, shall, in the written opinion of Bond Counsel filed with the City, the Fiscal Agent and the Borrower (with a copy to the Funding Lender), impose requirements upon the ownership or operation of the Project less restrictive than imposed by this Regulatory Agreement, this Regulatory Agreement may be amended or modified to provide such less restrictive requirements but only by written amendment signed by the City, the Fiscal Agent and the Borrower and approved by the written opinion of Bond Counsel to the effect that such amendment is permitted by the Law and the Act and will

not affect the Tax-exempt status of interest on the Note. The City shall be under no obligation to agree to any such amendment, it being understood that each of the requirements of this Regulatory Agreement is a specific requirement of the City, whether or not required by California or federal law.

(c) The Borrower, the City and, if applicable, the Fiscal Agent shall execute, deliver and, if applicable, file or record any and all documents and instruments necessary to effectuate the intent of this Section 8, and the City hereby appoints the Fiscal Agent as its true and lawful attorney-in-fact to execute, deliver and, if applicable, file or record on behalf of the City, as is applicable, any such document or instrument (in such form as may be approved in writing by Bond Counsel) if the City defaults in the performance of its obligations under this subsection (c); provided, however, that unless directed in writing by the City, the Fiscal Agent shall take no action under this subsection (c) without first notifying the City and without first providing the City an opportunity to comply with the requirements of this Section 8. Nothing in this Section 8(c) shall be construed to allow the Fiscal Agent to execute an amendment to this Regulatory Agreement on behalf of the City.

Section 9. Indemnification. The Borrower shall defend, indemnify and hold harmless the City and the Fiscal Agent and the respective officers, members, supervisors, directors, officials and employees, counsel, attorneys and agents, past, present and future of each of them (collectively, the "Indemnified Parties"), against all loss, costs, damages, expenses, suits, judgments, actions and liabilities of whatever nature (including, without limitation, reasonable attorneys' fees, litigation and court costs, amounts paid in settlement, and amounts paid to discharge judgments) directly or indirectly resulting from or arising out of or related to (a) the design, construction, installation, operation, use, occupancy, maintenance, financing or ownership of the Project (including compliance with laws, ordinances and rules and regulations of public authorities relating thereto), (b) any written statements or representations with respect to the Borrower, the Project or the Note made or given to the City or the Fiscal Agent, or any underwriters or purchaser of the Note, or any tenants or applicants for tenancy in the Project or any other person, by the Borrower, or any Authorized Borrower Representative, including, but not limited to, statements or representations of facts, financial information or limited partnership affairs, (c) the Note or the Tax-exempt status of interest on the Note, (d) the failure or alleged failure of any person or entity (including the Borrower or its contractor or subcontractors) to pay the general prevailing rate of per diem wages as determined pursuant to Labor Code Sections 1770-1781 and implementing regulations of the Department of Industrial Relations in connection with the construction of the improvements or any other work undertaken or in connection with the Project, or (e) any actual or alleged violation of any Hazardous Materials Law or with respect to the presence of Hazardous Materials on or under the Project or in any of improvements or on or under any property of the Borrower that is adjacent to the Project (whether before or after the date of this Agreement and whether or not the Borrower knew of the same); provided, however, that this provision shall not require the Borrower to indemnify the Indemnified Parties from any claims, costs, fees, expenses or liabilities arising from its active negligence or willful misconduct or, in the case of the Fiscal Agent, its negligence, fraud or willful misconduct. The Borrower also shall pay and discharge and shall indemnify and hold harmless the City and the Fiscal Agent from (i) any lien or charge upon payments by the Borrower to the City and the Fiscal Agent hereunder or under the Note Documents and (ii) any taxes (including, without limitation, all ad valorem taxes and sales taxes), assessments, impositions and other charges in respect of any portion of the Project.

If any such claim is asserted, or any such lien or charge upon payments, or any such taxes, assessments, impositions or other charges are sought to be imposed, the City or the Fiscal Agent shall give prompt notice to the Borrower and the Borrower shall, as provided in the following paragraph, have the right to assume the defense thereof, with full power to litigate, compromise or settle the same in its sole discretion, provided that the City and the Fiscal Agent shall have the right to review and approve or disapprove any such compromise or settlement. In addition thereto, the Borrower will pay upon demand all of the reasonable fees and expenses paid or incurred by the Fiscal Agent and/or the City in enforcing the provisions hereof against the Borrower. The Borrower shall also pay the City its standard fees and reimburse the City for its expenses in connection with any consent, approval, amendment, waiver or other action taken at the request or for the benefit of the Borrower in connection with this Regulatory Agreement, the Note or any other document or agreement relating thereto. In the event of any audit or inquiry regarding the Note or the Project from any governmental entity, the Borrower shall, at the election of the City, be responsible for responding to and resolving such audit or inquiry at the expense of the Borrower.

Promptly after receipt by any party entitled to indemnification under this Section 9 of notice of the commencement of any suit, action or proceeding, such Indemnified Party shall, if a claim in respect thereof is to be made against the indemnifying party under this Section 9, notify the indemnifying party in writing of the commencement thereof; but the omission so to notify the indemnifying party shall not relieve the indemnifying party from any liability which it may have to any Indemnified Party otherwise than under this Section 9 or from any liability under this Section 9 unless the failure to provide notice prejudices the defense of such suit, action or proceeding. In case any such action is brought against any Indemnified Party, and it notifies the indemnifying party, the indemnifying party shall be entitled to participate in, and to the extent that it may elect by written notice delivered to the Indemnified Party within five Business Days after receiving the aforesaid notice from such Indemnified Party (but shall not be required) to assume, the defense thereof, with counsel reasonably satisfactory to such Indemnified Party, provided, however, if the defendants in any such action include both the Indemnified Party and the indemnifying party and the Indemnified Party shall have reasonably concluded that there are legal defenses available to it and/or other Indemnified Parties which are different from or additional to those available to the indemnifying party, the Indemnified Party or parties shall have the right to select separate counsel to assert such legal defenses and otherwise to participate in the defense of such action on behalf of such Indemnified Party or parties. Upon the indemnifying party's receipt of notice from the Indemnified Party of such Indemnified Party's election so to assume the defense of such action and selection by the Indemnified Party of counsel, the indemnifying party shall not be liable to such Indemnified Party under this Section 9 for any attorneys' fees or expenses subsequently incurred by such Indemnified Party for the engagement of separate counsel in connection with defense thereof unless (i) the Indemnified Party shall have employed separate counsel in connection with the assertion of legal defenses in accordance with the proviso to the next preceding sentence, (ii) the indemnifying party shall not have employed counsel reasonably satisfactory to the Indemnified Party to represent the Indemnified Party or shall not have employed such counsel within a reasonable time after notice of commencement of the action or (iii) the indemnifying party has authorized the employment of separate counsel to represent the Indemnified Party at the expense of the indemnifying party. Notwithstanding the foregoing, the Fiscal Agent shall not be indemnified for income tax, franchise tax or similar tax liability relating to the Fiscal Agent's own income and operations.

**Section 10. Consideration**. The City has issued the Note to provide funds to finance the acquisition, construction and equipping of the Project, all for the purpose, among others, of inducing the Borrower to acquire, construct and equip the Project. In consideration of the issuance of the Note by the City, the Borrower has entered into this Regulatory Agreement and has agreed to restrict the uses to which this Project can be put on the terms and conditions set forth herein.

Section 11. Reliance. The City and the Borrower hereby recognize and agree that the representations and covenants set forth herein may be relied upon by all persons interested in the legality and validity of the Note and in the exemption from federal income taxation and California personal income taxation of the interest on the Note. In performing their duties and obligations hereunder, the City and the Fiscal Agent may rely upon statements and certificates of the Low Income Tenants and upon audits of the books and records of the Borrower pertaining to the Project. In addition, the City and the Fiscal Agent may consult with counsel, and the written opinion of such counsel shall be full and complete authorization and protection in respect of any action taken or suffered by the City or the Fiscal Agent hereunder in good faith and in conformity with such opinion. In determining whether any default or lack of compliance by the Borrower exists under this Regulatory Agreement, the Fiscal Agent may, but shall not be required to, conduct any investigation into or review of the operations or records of the Borrower and may rely solely on any written notice or certificate delivered to the Fiscal Agent by the Borrower or the City with respect to the occurrence or absence of a default unless it knows that the notice or certificate is erroneous or misleading.

**Section 12. Project in the City**. The Borrower hereby represents and warrants that the Project will be located entirely within the City.

Section 13. Sale or Transfer of the Project; Equity Interests. The Borrower hereby covenants and agrees not to voluntarily (which term shall not be interpreted to include a foreclosure of any security for the Loan, the granting by the Borrower of a deed in lieu of foreclosure, or any other comparable conversion of the Loan) sell, transfer or otherwise dispose of the Project, or any portion thereof (other than for individual tenant use as contemplated hereunder), equity interests in the Borrower aggregating more than 50% of the equity interest in the Borrower, or any general partner interests in the Borrower, without obtaining the prior written consent of the City, which consent shall not be unreasonably withheld by the City and shall be given by the City if (a) the Borrower is not in default hereunder or under the Loan Agreement, (b) the purchaser or assignee is not in default under any obligations it may have to the City and is not the subject of any legal or enforcement actions by the City, and the purchaser or assignee certifies that the continued operation of the Project will comply with the provisions of this Regulatory Agreement; (c) evidence reasonably satisfactory to the City is presented to establish that the purchaser or assignee is willing to comply and capable of complying with the terms and conditions of this Regulatory Agreement; (d) either (i) evidence satisfactory to the City is presented to establish that the purchaser or assignee has at least three years' experience in the ownership, operation and management of rental housing projects, without any record of material violations of discrimination restrictions or other state or federal laws or regulations applicable to such projects, (ii) the purchaser or assignee agrees to retain a property management firm which the City determines has the experience and record described in subclause (i) above or (iii) the City determines that it has no reason to believe that the purchaser or assignee is incapable, financially or otherwise, of complying with, or may be unwilling to comply with, the terms of all agreements binding on such

purchaser or assignee and relating to the Project; (e) the City and the Fiscal Agent shall have received (i) with respect to any transfer of the Project, reasonable evidence satisfactory to the City that the Borrower's purchaser or transferee has assumed in writing and in full the Borrower's duties and obligations under this Regulatory Agreement and the Loan Agreement, (ii) with respect to any transfer of the Project to a new Borrower, an opinion of counsel to the transferee that the transferee has duly assumed the obligations of the Borrower under this Regulatory Agreement and that such obligations and this Regulatory Agreement are binding on the transferee, (iii) unless waived by the City, an opinion of Bond Counsel that such transfer will not adversely affect the Tax-exempt nature of the interest on the Note, (iv) from the Borrower, a Certificate of Continuing Program Compliance (and a "bring-down" certificate, if necessary) current as of the date of transfer and (v) evidence satisfactory to the City that the purchaser or assignee does not have pending against it, nor does it have a history of, building or fire code violations as identified by the City, the State of California or federal regulatory agencies; (f) the purchaser or assignee complies with the provisions of the Los Angeles Administrative Code Section 10.8.4 Affirmative Action Program Provisions; (g) the Borrower or transferee pays all costs of the transfer of title, including, but not limited to, the cost of meeting the conditions specified in this Section 13; and (h) such other conditions are met as the City may reasonably impose to assure compliance by the Project with the requirements of this Regulatory Agreement. It is hereby expressly stipulated and agreed that, except for any such sale, transfer or disposition agreed to by the City in a separate writing, any sale, transfer or other disposition of the Project in violation of this Section 13 shall be null, void and without effect, shall cause a reversion of title to the Borrower, and shall be ineffective to relieve the Borrower of its obligations under this Regulatory Agreement. Upon any sale or other transfer which complies with this Regulatory Agreement, the Borrower shall be fully released from its obligations hereunder, but only to the extent such obligations have been assumed by the transferee of the Project, without the necessity of further documentation. Any transfer of the Project to any entity, whether or not affiliated with the Borrower, shall be subject to the provisions of this Section 13.

Nothing contained in this Section 13 shall affect any provision of the Deed of Trust, or any of the other Loan Documents to which the Borrower is a party, which requires the Borrower to obtain the consent of the Funding Lender as a precondition to sale, transfer or other disposition of, or any direct or indirect interest in, the Project or of any direct or indirect interest in the Borrower or which otherwise gives the Funding Lender the right to accelerate the maturity of the Loan or any obligations of the Borrower under the Loan Documents, or to take some other similar action with respect to the Loan or any obligations of the Borrower under the Loan Documents, upon the sale, transfer or other disposition of the Project or any such other interest. Notwithstanding anything contained in this Section 13 to the contrary, neither the consent of the City nor the delivery of items (a) through (h) of the preceding paragraph shall be required in the case of, and nothing in this Section 13 shall otherwise affect the right of the Funding Lender, the Fiscal Agent or a designee or third-party purchaser to effect, a foreclosure or deed in lieu of foreclosure, whereby the Funding Lender, the Fiscal Agent or a designee or third-party purchaser forecloses on the Project or accepts a deed in lieu of a foreclosure or effects a comparable conversion of the Loan or the Loan Documents. Consent of the City and delivery of items (a) through (h) of the preceding paragraph shall be required for any transfer of the Project subsequent to the purchase at a foreclosure or transfer pursuant to deed in lieu of foreclosure as described in the preceding sentence.

Notwithstanding the foregoing, if the Fiscal Agent acquires title to the Project by foreclosure or deed in lieu of foreclosure, no consent of the City shall be required to such transfer under this Regulatory Agreement and no other conditions shall be required to be satisfied. However, if the Fiscal Agent acquires title to the Project by foreclosure or deed in lieu of foreclosure and this Regulatory Agreement has not been terminated pursuant to Section 14 below, consent of the City and delivery of items (a) through (h) above shall be required for any transfer of the Project subsequent to the Fiscal Agent's acquisition of the Project by foreclosure or deed in lieu of foreclosure.

Notwithstanding anything to the contrary contained herein, the interest of the Borrower's limited partners shall, with prior written notice to the City, be transferable under this Regulatory Agreement to any affiliate of the limited partners of the Borrower, without the consent of the City and/or Fiscal Agent but with prior written notice thereto.

The Borrower acknowledges and recognizes that in addition to the above requirements, the consent of CDLAC in the manner and to the extent as may at the time be required by CDLAC, among other parties, may be required in connection with any transfer of the Project.

**Section 14. Term**. This Regulatory Agreement and all and each of the provisions hereof shall become effective upon its execution and delivery and shall remain in full force and effect for the periods provided herein and, except as otherwise provided in this Section 14, shall terminate in its entirety at the end of the Qualified Project Period (or such later date provided in Section 33 hereof pursuant to the CDLAC Resolution, which imposes restrictions for a term of at least 55 years), it being expressly agreed and understood that the provisions hereof are intended to survive the retirement of the Note, discharge of the Loan and termination of the Funding Loan Agreement and the Loan Agreement.

Notwithstanding the foregoing, the provisions of Section 9 hereof shall, in the case of the Fiscal Agent, survive the term of this Regulatory Agreement or the replacement of the Fiscal Agent, but only as to claims arising from events occurring during the term of this Regulatory Agreement or the Fiscal Agent's tenure as Fiscal Agent under the Funding Loan Agreement, and shall, in the case of the City, survive the term of this Regulatory Agreement, but only as to claims arising from events occurring during the term of this Regulatory Agreement.

The terms of this Regulatory Agreement to the contrary notwithstanding, this Regulatory Agreement and all the requirements set forth herein (except Section 9 as aforesaid) shall terminate and be of no further force and effect in the event of (a) involuntary noncompliance with the provisions of this Regulatory Agreement caused by fire, seizure, requisition, change in a federal law or an action of a federal agency after the Closing Date which prevents the City or the Fiscal Agent from enforcing the provisions hereof, or (b) condemnation, foreclosure, delivery of a deed in lieu of foreclosure or a similar event, but only if, within a reasonable period thereafter, either the portion of the Note attributable to the affected portion of the Project is retired or amounts received as a consequence of such event are used to provide a project which meets the requirements of the Code set forth in Sections 2 through 6 of this Regulatory Agreement and provided that, in either case, an opinion of Bond Counsel (unless waived by the City) is delivered to the Fiscal Agent to the effect that the exclusion from gross income for federal income tax purposes of interest on the Note will not be adversely affected thereby. The provisions of the preceding sentence shall

cease to apply and the requirements referred to therein shall be reinstated if, at any time during the Qualified Project Period after the termination of such requirements as a result of involuntary noncompliance due to foreclosure, transfer of title by deed in lieu of foreclosure or similar event, the Borrower or any "related party" (within the meaning of Section 1.150-1(b) of the Regulations) or "related person" (defined in Section 1.103-10(e) of the Regulations) obtains an ownership interest in the Project for tax purposes. The Borrower hereby agrees that, following any foreclosure, transfer of title by deed in lieu of foreclosure or similar event, neither the Borrower nor any related party as described above will obtain an ownership interest in the Project for tax purposes.

Upon the termination of this Regulatory Agreement, the parties hereto agree to execute, deliver and record appropriate instruments of release and discharge of the terms hereof; provided, however, that the execution and delivery of such instruments shall not be necessary or a prerequisite to the termination of this Regulatory Agreement in accordance with its terms.

Section 15. Covenants To Run With the Land. The Borrower hereby subjects the Project (including the Project Site) to the covenants, reservations and restrictions set forth in this Regulatory Agreement. The City and the Borrower hereby declare their express intent that the covenants, reservations and restrictions set forth herein shall be deemed covenants running with the land and shall pass to and be binding upon the Borrower's successors in title to the Project; provided, however, that on the termination of this Regulatory Agreement said covenants, reservations and restrictions shall expire. The City and, if necessary, the Fiscal Agent agree to execute a quitclaim deed or other documents required to remove this Regulatory Agreement from title after the covenants, agreements and restrictions herein have expired. Each and every contract, deed or other instrument hereafter executed covering or conveying the Project or any portion thereof shall conclusively be held to have been executed, delivered and accepted subject to such covenants, reservations and restrictions, regardless of whether such covenants, reservations and restrictions are set forth in such contract, deed or other instruments.

No breach of any of the provisions of this Regulatory Agreement shall impair, defeat or render invalid the lien of any security instrument, deed of trust or like encumbrance made in good faith and for value encumbering the Project or any portion thereof.

**Section 16. Burden and Benefit**. The City and the Borrower hereby declare their understanding and intent that the burden of the covenants set forth herein touch and concern the land in that the Borrower's legal interest in the Project is rendered less valuable thereby. The City and the Borrower hereby further declare their understanding and intent that the benefit of such covenants touch and concern the land by enhancing and increasing the enjoyment and use of the Project by Low Income Tenants, the intended beneficiaries of such covenants, reservations and restrictions, and by furthering the public purposes for which the Note was issued. Notwithstanding the foregoing or any other provision of this Regulatory Agreement, no person, other than the parties hereto, shall have any rights of enforcement of this Regulatory Agreement.

**Section 17. Uniformity; Common Plan**. The covenants, reservations and restrictions hereof shall apply uniformly to the entire Project in order to establish and carry out a common plan for the use, development and improvement of the Project Site.

Section 18. Default; Enforcement. If the Borrower defaults in the performance or observance of any covenant, agreement or obligation of the Borrower set forth in this Regulatory Agreement, and if such default remains uncured for a period of 60 days after notice thereof shall have been given by the City to the Borrower, then the City shall declare an "Event of Default" to have occurred hereunder; provided, however, that if the default stated in the notice is of such a nature that it cannot be corrected within 60 days, such default shall not constitute an Event of Default hereunder so long as (i) the Borrower institutes corrective action within said 60 days and diligently pursues such action until the default is corrected and (ii) in the opinion of Bond Counsel, the failure to cure said default within 60 days will not adversely affect the Tax-exempt status of interest on the Note. The Fiscal Agent hereby consents to any correction of the default by the City on behalf of the Borrower. The City hereby consents to any correction of a default on the part of the Borrower hereunder made by the Borrower's limited partners on behalf of the Borrower within the time periods provided in this Section. Copies of any notices sent to the Borrower hereunder shall simultaneously be sent to the Borrower's limited partners at the addresses set forth in Section 23

Following the declaration of an Event of Default hereunder, the Fiscal Agent, as directed by the City and subject to the provisions of the Funding Loan Agreement relative to the Fiscal Agent's duty to exercise remedies generally, or the City, may, at its option, take any one or more of the following steps:

- (a) by mandamus or other suit, action or proceeding at law or in equity, including injunctive relief, require the Borrower to perform its obligations and covenants hereunder or enjoin any acts or things which may be unlawful or in violation of the rights of the City or the Fiscal Agent hereunder;
- (b) have access to and inspect, examine and make copies of all or any portion of the books and records of the Borrower pertaining to the Project; and
- (c) take such other action at law or in equity as may appear necessary or desirable to enforce the obligations, covenants and agreements of the Borrower hereunder.

During the Qualified Project Period, the Borrower hereby grants to the City the option, upon either (a) the expiration of 60 days after the giving of the notice to the Borrower referred to in the first paragraph of this Section 18 of the Borrower's default under this Regulatory Agreement or (b) the vacancy of a Low Income Unit for more than six months and the submission by the City to the Borrower during such six-month or longer period of at least five proposed tenants which meet the qualifications of Low Income Tenants and the qualifications of a reasonable landlord, to lease up to [40]% of the units in the Project for a rental of \$1.00 per unit per year for the sole purpose of subleasing such units to Low Income Tenants for a period of not less than six months, but only to the extent necessary to comply with the provisions of Sections 2 through 7 of this Regulatory Agreement and to insure full occupancy of the Low Income Units. The option granted in the preceding sentence shall be effective only if the Borrower or the Fiscal Agent has not instituted corrective action before the end of such 60-day period referenced in (a) above, or the Borrower has not rented the unit during the six-month or longer period referenced in (b) above, to a qualified Low Income Tenant. The option and any leases to the City under this provision shall terminate with respect to each default upon the achievement, by the Borrower, the Fiscal Agent or

the City, of compliance with the requirements of Sections 2 through 7 hereof, and any subleases entered into pursuant to the City's option shall be deemed to be leases from the Borrower. The City shall make diligent effort, but shall not be required, to rent Low Income Units to Low Income Tenants at the highest rents practicable, subject to the limits of Sections 5, 6 and 7 hereof. Any rental paid under any such sublease shall be paid to the Borrower after the City has been reimbursed for any reasonable expenses incurred in connection with such sublease, provided that if the Borrower is in default under the Loan Agreement, such rental shall be paid to the Fiscal Agent for credit against payments due under the Loan Agreement. The Fiscal Agent shall have the right, as directed by the City, in accordance with this Section 18 and the provisions of the Funding Loan Agreement, to exercise any or all of the rights or remedies of the City hereunder, provided that prior to taking any such action the Fiscal Agent shall give the City written notice of its intended action. All reasonable fees, costs and expenses of the City and the Fiscal Agent incurred in taking any action pursuant to this Section 18 shall be the sole responsibility of the Borrower.

After the Funding Loan Agreement has been discharged, the City may act on its own behalf to declare an "Event of Default" to have occurred and to take any one or more of the steps specified hereinabove to the same extent and with the same effect as if taken by the Fiscal Agent.

The obligations of the Borrower hereunder are not secured by a lien on the Project and the Loan shall not be accelerated as a result of any default hereunder. The Borrower hereby agrees that specific enforcement of the Borrower's agreements contained herein is the only means by which the City may obtain the benefits of such agreements made by the Borrower herein and the Borrower therefore agrees to the imposition of the remedy of specific performance against it in the case of any default by the Borrower hereunder.

The occurrence of a Determination of Taxability shall not, in and of itself, constitute a default hereunder.

**Section 19. The Fiscal Agent**. The Fiscal Agent shall act as specifically provided herein and in the Funding Loan Agreement. The Fiscal Agent is entering into this Regulatory Agreement solely in its capacity as fiscal agent under the Funding Loan Agreement, and the duties, powers, rights and liabilities of the Fiscal Agent in acting hereunder shall be subject to the provisions of the Funding Loan Agreement.

The City shall be responsible for monitoring and verifying compliance by the Borrower with the terms of this Regulatory Agreement. The Fiscal Agent may at all times assume compliance with this Regulatory Agreement unless otherwise notified in writing by the City or unless it has actual knowledge of noncompliance.

After the date on which no principal of the Note remains outstanding as provided in the Funding Loan Agreement, the Fiscal Agent shall no longer have any duties or responsibilities under this Regulatory Agreement and all references to the Fiscal Agent in this Regulatory Agreement shall be deemed references to the City.

**Section 20. Recording and Filing**. The Borrower shall cause this Regulatory Agreement, and all amendments and supplements hereto and thereto, to be recorded and filed in the real

property records of the County of Los Angeles and in such other places as the City or the Fiscal Agent may reasonably request. The Borrower shall pay all fees and charges incurred in connection with any such recording. This Regulatory Agreement shall be recorded in the grantor-grantee index to the name of the Borrower as grantor and the City as grantee.

**Section 21. Governing Law**. This Regulatory Agreement shall be governed by the laws of the State of California. The Fiscal Agent's rights, duties and obligations hereunder are governed in their entirety by the terms and provisions of the Funding Loan Agreement.

Section 22. Amendments. Except as provided in Section 33(e), this Regulatory Agreement shall be amended only by a written instrument executed by the parties hereto or their successors in title, and duly recorded in the real property records of the County of Los Angeles, California, and only upon receipt by the City of an opinion from Bond Counsel that such amendment will not adversely affect the Tax-exempt status of interest on the Note and is not contrary to the provisions of the Law or the Act and with the written consent of the Fiscal Agent.

The City, the Fiscal Agent and the Borrower hereby agree to amend this Regulatory Agreement to the extent required, in the opinion of Bond Counsel (subject to the approval of the City Attorney of the City), in order that interest on the Note remains Tax-exempt. The party or parties requesting such amendment shall notify the other parties to this Regulatory Agreement of the proposed amendment, with a copy of such requested amendment to Bond Counsel and the City Attorney of the City and a request that such Bond Counsel render to the City an opinion as to the effect of such proposed amendment upon the Tax-exempt status of interest on the Note.

**Section 23.** Notices. Any notice required to be given hereunder shall be made in writing and shall be given by personal delivery, certified or registered mail, postage prepaid, return receipt requested, at the addresses specified below, or at such other addresses as may be specified in writing by the parties hereto:

If to the City:	City of Los Angeles
•	c/o Housing and Community Investment Department
	8th Floor
	1200 West 7th Street
	Los Angeles, CA 90017
	Attention: Supervisor, Affordable Housing Bond Program
	HIMS# [
	Facsimile: (213) 808-8918
with a copy to:	Los Angeles Housing and Community Investment Department
	P.O. Box 532729
	Los Angeles, CA 90053-2729
	Attention: Supervisor, Affordable Housing Bond Program
	HIMS# []

If to the Borrower:	SP7 Apartments LP c/o The Skid Row Housing Trust 1317 E. 7th St. Los Angeles, CA 90021 Attention: [] Facsimile: []
with a copy to:	[BORROWER'S COUNSEL]  []  []  Attention: []  Facsimile: []
with a copy to:	[INVESTOR LIMITED PARTNER] c/o [] [] [] Attention: [] Facsimile: []
with a copy to:	[INVESTOR LIMITED PARTNER'S COUNSEL]  []  []  Attention: []  Facsimile: []
with copy to:	MUFG Union Bank, N.A. 200 Pringle Avenue, Suite 355 Walnut Creek, CA 94596 Attention: CDF Division Head Facsimile: (925) 947-2455
with a copy to:	[LENDER'S COUNSEL] [] [] Attention: []
If to the Funding Lender:	
	MUFG Union Bank, N.A. Loan Administration Department 3151 East Imperial Highway, 1st Floor Brea, CA 92821 Attention: Manager Telephone (714) 990-7300 Facsimile: (323) 720-2433

[FISCAL AGENT]
[]
Attention: Global Corporate Trust Services
Ref: LA MF (SP7 Apartments 2018C)
Telephone: (213) [ ]
Facsimile: (213) []
California Debt Limit Allocation Committee
Room 311
915 Capitol Mall
Sacramento, CA 95814
Attention: Executive Director

Notice shall be deemed given three Business Days after the date of mailing.

A duplicate copy of each notice, certificate or other communication given hereunder by any party hereto to another party hereto shall also be given to all of the parties specified above. Failure to provide any such duplicate notice pursuant to the foregoing sentence, or any defect in any such duplicate notice so provided, shall not constitute a default hereunder. All other documents required to be submitted to any of the foregoing parties shall also be submitted to such party at its address set forth above. Any of the foregoing parties may, by notice given hereunder, designate any further or different addresses to which subsequent notices, certificates, documents or other communications shall be sent.

**Section 24. Severability**. If any provision of this Regulatory Agreement or if the applicability of any such provision shall be invalid, illegal or unenforceable, the validity, legality, enforceability or applicability with respect to the validity, legality and enforceability of the remaining portions hereof shall not in any way be affected or impaired thereby.

**Section 25. Multiple Counterparts**. This Regulatory Agreement may be simultaneously executed in multiple counterparts, all of which shall constitute one and the same instrument, and each of which shall be deemed to be an original.

Section 26. Nondiscrimination and Affirmative Action. The Fiscal Agent and the Borrower shall comply with the applicable nondiscrimination and affirmative action provisions of the laws of the United States of America, the State of California, and the City of Los Angeles. The Fiscal Agent and the Borrower shall not discriminate in their employment practices against any employee or applicant for employment; denial of family and medical care leave; denial of pregnancy disability leave or reasonable accommodations against any employee or applicant for employment because of such person's race, ancestry, color, citizenship, national origin, religion, sex, sexual orientation, gender identity/expression, transgender status, age, marital status, familial status, domestic partner status, physical handicap, mental disability, medical condition, political affiliation or belief. The Fiscal Agent and the Borrower shall comply with Executive Order 11246, entitled "Equal Employment Opportunity," as amended by Executive Order 11375, and as supplemented in Department of Labor regulations (41 C.F.R. Part 60). The Fiscal Agent and the Borrower shall comply with the provisions of the Los Angeles Administrative Code Sections 10.8

through 10.13, to the extent applicable hereto. The affirmative action program of the Fiscal Agent and the Borrower shall include the mandatory contract provisions set forth in Los Angeles Administrative Code Section 10.8.4, and said provisions are incorporated herein by this reference. The Borrower and the Fiscal Agent shall also comply with all rules, regulations, and policies of the City of Los Angeles' Board of Public Works, Office of Contract Compliance relating to nondiscrimination and affirmative action, including the filing of all forms required by the City. Any subcontract entered into by the Borrower or Fiscal Agent relating to this Regulatory Agreement, to the extent allowed hereunder, shall be subject to the provisions of this Section. No person shall on the grounds of race, ancestry, color, citizenship, national origin, religion, sex, sexual orientation, gender identity/expression, transgender status, age, marital status, familial status, domestic partner status, physical handicap, mental disability, medical condition, political affiliation or belief be excluded from participation in, be denied the benefit of, or be subjected to discrimination under, this Regulatory Agreement. For purposes of this Section, Title 24 Code of Federal Regulations Part 107 and Section 570.601(b) define specific discriminatory actions that are prohibited and corrective action that shall be taken in a situation as defined therein.

Section 27. Business Tax Registration Certificate. Subject to any exemption available to it, the Fiscal Agent and the Borrower each represent that they will obtain and hold the Business Tax Registration Certificate(s) required by the City's Business Tax Ordinance (Article 1, Chapter 2, Section 21.00 and following, of the Los Angeles Municipal Code). For the term covered by this Regulatory Agreement, the Fiscal Agent and the Borrower shall maintain, or obtain as necessary, all such Business Tax Registration Certificates required of them under said Ordinance and shall not allow any such Business Tax Registration Certificate to be revoked or suspended.

Section 28. Financial Obligations Personal to Borrower. The City acknowledges that the Project shall be encumbered by the Borrower Loan Documents. Notwithstanding any provisions of this Regulatory Agreement to the contrary, all obligations of the Borrower under this Regulatory Agreement for the payment of money and all claims for damages against the Borrower occasioned by breach or alleged breach by the Borrower of its obligations under this Regulatory Agreement, including indemnification obligations, shall not be a lien on the Project and no Person shall have the right to enforce such obligations other than directly against the Borrower as provided in Section 18 of this Regulatory Agreement, except that the City shall have the right at all times to enforce the rights contained in the third paragraph of Section 18 hereof. No subsequent owner of the Project shall be liable or obligated for the breach or default of any obligations of the Borrower under this Regulatory Agreement on the part of any prior Borrower, including, but not limited to, any payment or indemnification obligation. Such obligations are personal to the Person who was the Borrower at the time the default or breach was alleged to have occurred and such Person shall remain liable for any and all damages occasioned thereby even after such Person ceases to be the Borrower. Each Borrower shall comply with and be fully liable for all obligations of an "owner" hereunder during its period of ownership. Notwithstanding the foregoing, neither the Borrower nor its partners shall be personally liable for any indemnification obligation under the Borrower Loan Documents which would result in payment of principal of or interest on the Note.

### Section 29. [Reserved].

Section 30. Child Support Assignment Orders. This Regulatory Agreement is subject to Section 10.10 of the Los Angeles Administrative Code, Child Support Assignment Orders

Ordinance. Pursuant to this Ordinance, each of the Borrower and the Fiscal Agent certifies that (a) it will fully comply with all State and federal employment reporting requirements applicable to Child Support Assignment Orders; (b) the principal owner(s) of the Borrower are in compliance with any Wage and Earnings Assignment Orders and Notices of Assignment applicable to them personally; (c) it will fully comply with all lawfully served Wage and Earnings Assignment Orders and Notices of Assignment in accordance with California Family Code Section 5230 et seq.; and (d) it will maintain such compliance throughout the term of this Regulatory Agreement. Pursuant to Section 10.10.b of the Los Angeles Administrative Code, failure of the Borrower or the Fiscal Agent to comply with all applicable reporting requirements or to implement lawfully served Wage and Earnings Assignment Orders and Notices of Assignment or the failure of any principal owner(s) of the Borrower or the Fiscal Agent to comply with any Wage and Earnings Assignment Orders and Notices of Assignment applicable to them personally shall constitute a default by the Borrower or the Fiscal Agent, as appropriate, under the terms of this Regulatory Agreement, subjecting (i) the Borrower to the remedies provided herein and (ii) the Fiscal Agent to termination under the Funding Loan Agreement where, in either case, such failure shall continue for more than 90 days after notice of such failure to the Borrower or the Fiscal Agent by the City. Any subcontract entered into by the Borrower or the Fiscal Agent relating to this Regulatory Agreement, to the extent allowed hereunder, shall be subject to the provisions of this paragraph and shall incorporate the provisions of the Child Support Assignment Orders Ordinance. Failure of the Borrower or the Fiscal Agent to obtain compliance of its subcontractors shall constitute a default by the Borrower or the Fiscal Agent, as appropriate, under the terms of this Regulatory Agreement, subjecting (A) the Borrower to the remedies provided herein and (B) the Fiscal Agent to termination under the Funding Loan Agreement where such failure shall continue for more than 90 days after notice of such failure to the Borrower or the Fiscal Agent by the City.

The Borrower and the Fiscal Agent shall comply with the Child Support Compliance Act of 1998 of the State of California Employment Development Department. Each of the Borrower and the Fiscal Agent assures that to the best of its knowledge it is fully complying with the earnings assignment orders of all employees and is providing the names of all new employees to the New Hire Registry maintained by the Employment Development Department as set forth in subdivision (b) of the Public Contract Code Section 7110.

Section 31. Americans with Disabilities Act. Each of the Borrower and the Fiscal Agent hereby certifies that it and any contractor and subcontractor will comply with the Accessibility Requirements (as defined in Exhibit I). The Borrower and any contractor and subcontractor will provide reasonable accommodations to allow qualified individuals with disabilities to have access to and to participate in its programs, services, and activities in accordance with the applicable provisions of the ADA, the ADAAG, Section 504, the UFAS, the FHA (each as defined in Exhibit I) and all subsequent amendments. The Borrower, the Fiscal Agent and each and any contractor and subcontractor will not discriminate against persons with disabilities or against persons due to their relationship to or association with a person with a disability. Any contract and subcontract entered into by the Borrower or the Fiscal Agent, relating to this Regulatory Agreement and the Project, to the extent allowed hereunder, shall be subject to the provisions of this paragraph. The Borrower hereby agrees to observe all of the covenants contained in Exhibit I to this Regulatory Agreement as if contained herein.

- Section 32. Slavery Disclosure Ordinance. This Regulatory Agreement is subject to the Slavery Disclosure Ordinance, Section 10.41 of the Los Angeles Administrative Code, as it may be amended from time to time. The Borrower hereby agrees to execute and deliver a certificate in the form attached hereto as Exhibit H (or such other form as is required by the City) certifying that it has complied with the applicable provisions of this Ordinance. The Borrower acknowledges that failure to fully and accurately complete the affidavit may result in a default under this Regulatory Agreement.
- **Section 33. Requirements of CDLAC**. In addition to other requirements set forth herein and to the extent not prohibited by the requirements set forth in Sections 2 through 6 hereof, the Borrower hereby agrees to comply with each of the requirements of CDLAC set forth in this Section 33, as follows:
  - (a) The Borrower shall comply with the conditions set forth in Exhibit A to the CDLAC Resolution (the "CDLAC Conditions"), as they may be modified or amended from time to time, which conditions are incorporated herein by reference and made a part hereof and are attached hereto as Exhibit G. Notwithstanding anything to the contrary herein, the provisions of this Section 33 shall remain effective for the period specified in the CDLAC Conditions, unless this Regulatory Agreement shall terminate as otherwise provided in Section 14 hereof.
  - (b) The Borrower acknowledges that the City shall monitor the Borrower's compliance with the terms of the CDLAC Conditions. The Borrower will cooperate fully with the City in connection with such monitoring and reporting requirements as provided herein. Compliance with the terms of the CDLAC Conditions not contained within this Regulatory Agreement, but referred to in the CDLAC Conditions, is the responsibility of the Borrower to report to the City.
    - (i) The Borrower shall prepare and deliver a Certificate of CDLAC Program Compliance pursuant to the terms of the CDLAC Conditions. The Borrower acknowledges that the Borrower will prepare and submit to the City, not later than January 15 of each year, and the City will submit to CDLAC, not later than March 1 of each year, until the Borrower has submitted to the City and CDLAC a Construction Completion Certificate, and on March 1 every three years thereafter (such that the next succeeding year shall be the beginning of each such three-year period) until the end of the term of the CDLAC Conditions, a Certificate of CDLAC Program Compliance, executed by an Authorized Borrower Representative.
    - (ii) The Borrower shall prepare and deliver a Self-Certification Certificate pursuant to the terms of the CDLAC Conditions. The Borrower acknowledges that the Borrower will prepare and submit to the City, not later than January 15 of each year, and the City will submit to CDLAC, not later than March 1 of each year, until the Borrower has submitted to the City and CDLAC a Construction Completion Certificate, and on March 1 every three years thereafter (such that the next succeeding year shall be the beginning of each such three-year

period) until the end of the term of the CDLAC Conditions, a Self-Certification Certificate in the form provided by CDLAC.

- (iii) Within 30 days following the completion of the Project, the Borrower will prepare and submit to the City, the Fiscal Agent and CDLAC a Construction Completion Certificate. Following the submission of the Construction Completion Certificate, the Borrower will prepare and submit to the City, not later than January 15 every three years thereafter until the end of the Compliance Period, a California Tax Credit Allocation Committee Project Status Report or equivalent documentation in substantially the form required or otherwise provided by CDLAC from time to time.
- (c) Except as otherwise provided in Section 14 of this Regulatory Agreement, this Regulatory Agreement shall terminate on the date 55 years after the date on which at least 50% of the units in the Project are first occupied or such later date as the Qualified Project Period shall begin, as required by the CDLAC Conditions.
- (d) The Borrower shall notify CDLAC in writing of: (i) any change in ownership of the Project, (ii) any change in the issuer of the Note, (iii) any change in the name of the Project or the Project manager; (iv) any default under the Indenture, the Loan Agreement or this Regulatory Agreement; or (v) termination of this Regulatory Agreement.
- (e) Any of the foregoing requirements of CDLAC contained in this Section 33 may be expressly waived by CDLAC, in its sole discretion, in writing, but (i) no waiver by CDLAC of any requirement of this Section 33 shall, or shall be deemed to, extend to or affect any other provision of this Regulatory Agreement except to the extent the City has received an opinion of Bond Counsel that any such provision is not required by the Code, the Act and the Law and may be waived without adversely affecting the exclusion from gross income of interest on the Note for federal income tax purposes; and (ii) any requirement of this Section 33 shall be void and of no force and effect if the City and the Borrower receive a written opinion of Bond Counsel to the effect that compliance with any such requirement would cause interest on the Note to cease to be Tax-exempt or to the effect that compliance with such requirement would be in conflict with the Code, the Act, the Law or any other state or federal law.
- Agreement, and CDLAC shall have the right (but not the obligation) to enforce, separately or jointly with the City and/or the Fiscal Agent or to cause the City or the Fiscal Agent to enforce, the provisions of Section 33 of this Regulatory Agreement and to pursue an action for specific performance of such provisions or other available remedy at law or in equity in accordance with Section 18 hereof, provided that any such action or remedy shall not materially adversely affect the interests and rights of the Noteholders and shall otherwise be subject to the terms, conditions and limitations applicable to the enforcement of remedies under this Regulatory Agreement.
- (g) CDLAC shall have the right, but not the obligation, to deliver revised CDLAC Conditions to the Borrower after the Closing Date at any time, that are not more

restrictive than the original CDLAC Conditions; provided, however, that, with the prior written consent of the Noteholders, which will not be unreasonably withheld: (i) any changes in the terms and conditions of the CDLAC Conditions prior to the recordation against the Project in the real property records of Los Angeles County, California, of a regulatory agreement between the Borrower and the California Tax Credit Allocation Committee ("TCAC Regulatory Agreement") shall be limited to such changes as are necessary to correct any factual errors or to otherwise conform the CDLAC Conditions to any change in facts or circumstances applicable to the Borrower or the Project; and (ii) after recordation of the TCAC Regulatory Agreement, any changes in the terms and conditions of the CDLAC Conditions shall be limited to such changes as are necessary to conform Items 1, 6, 7, 10, 11, 12, 14, 15, 16, 18, 19, 20, 21, 22, 23, 24, 25, 26 or 37 of Exhibit A to the CDLAC Conditions to any change in terms and conditions requested by the Borrower and approved by CDLAC. The City may, in its sole and absolute discretion, require the Borrower to enter into an amendment to this Regulatory Agreement reflecting the revised CDLAC Conditions, which amendment shall be executed by the parties hereto or their successor in title and duly recorded in the real property records of the County of Los Angeles. The Borrower shall pay any costs and expenses in connection therewith and provide CDLAC with a copy of that recorded amendment reflecting the revised CDLAC Conditions.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the City, the Fiscal Agent and the Borrower have executed this Regulatory Agreement by duly authorized representatives, all as of the date first above written.

CITY OF LOS ANGELES, as City

By Los Angeles Housing and Community Investment Department

	By
	Sean L. Spear Assistant General Manager
	Assistant General Manager
Approved as to form:	
CITY OF LOS ANGELES MICHAEL N. FEUER, City Attorney	
Deputy/Assistant City Attorney	

[Signature Page to *SP7 Apartments* Regulatory Agreement]

By		
Name:		
Title:		

[FISCAL AGENT], as Fiscal Agent

[Signature Page to SP7 Apartments Regulatory Agreement]

SP7 APARTMENTS LP, a California limited partnership

By: SP7 Apartments GP LLC, a California limited liability company, its general partner

> By: The Skid Row Housing Trust, a California nonprofit public benefit corporation, its sole member/manager

By:	
	Ben Rosen,
	Real Estate Development Director

[Signature Page to SP7 Apartments Regulatory Agreement

State of California )
County of)
On, before me,, a Notary Public in and for
said State, personally appeared
(here insert name(s) of signers), who proved to me on the basis of satisfactory evidence to be the person(s) whose name(s)
is/are subscribed to the within instrument and acknowledged to me that he/she/they executed the same in his/her/their
authorized capacity(ies), and that by his/her/their signature(s) on the instrument the person(s), or the entity upon behalf
of which the person(s) acted, executed the instrument.
I certify under PENALTY OF PERJURY under the laws of the State of California that the foregoing paragraph is true and correct.
and correct.
WITNESS my hand and official seal.
Signature [SEAL]

State of California	)	
County of	)	
On	, before me,	, a Notary Public in and for
said State, personal	ly appeared	
(here insert name(s) of	signers), who proved to me on the basis of satisfactors	sfactory evidence to be the person(s) whose name(s)
is/are subscribed to the	e within instrument and acknowledged to me	that he/she/they executed the same in his/her/their
authorized capacity(ie	s), and that by his/her/their signature(s) on the	e instrument the person(s), or the entity upon behalf
of which the person(s)	acted, executed the instrument.	
I certify under PENAl and correct.	LTY OF PERJURY under the laws of the Stat	te of California that the foregoing paragraph is true
WITNESS my hand a	nd official seal.	
Signature	[SEA	L]

State of California	)		
County of)			
On	_, before me,	, a Notary Public in an	ıd for
said State, personally appear	ed		
(here insert name(s) of signers),	who proved to me on th	ne basis of satisfactory evidence to be the person(s) whose nar	ne(s)
is/are subscribed to the within is	nstrument and acknow	ledged to me that he/she/they executed the same in his/her/	/their
authorized capacity(ies), and that	nt by his/her/their signa	nture(s) on the instrument the person(s), or the entity upon be	ehalf
of which the person(s) acted, exe	ecuted the instrument.		
I certify under PENALTY OF Pand correct.	PERJURY under the la	ws of the State of California that the foregoing paragraph is	s true
WITNESS my hand and official	seal.		
Signature		[SEAL]	

State of California	)		
County of)			
On	_, before me,		, a Notary Public in and for
said State, personally appear	ed		
(here insert name(s) of signers),	who proved to me on th	e basis of satisfactory evid	dence to be the person(s) whose name(s)
is/are subscribed to the within in	nstrument and acknow	ledged to me that he/she/	they executed the same in his/her/their
authorized capacity(ies), and that	t by his/her/their signa	nture(s) on the instrument	the person(s), or the entity upon behalf
of which the person(s) acted, exe	ecuted the instrument.		
I certify under PENALTY OF Pand correct.	ERJURY under the la	ws of the State of Califor	nia that the foregoing paragraph is true
WITNESS my hand and official	seal.		
Signature		[SEAL]	

# **EXHIBIT A**

## **DESCRIPTION OF PROJECT SITE**

## 519 East 7th Street

Real property in the City of Los Angeles, County of Los Angeles, State of California, described as follows:

as follows.		
	[TO BE PROVIDED]	
Assessor's Parcel Number: [	]	

#### **EXHIBIT B**

### FORM OF CERTIFICATE OF CONTINUING PROGRAM COMPLIANCE

FOR THE [MONTH/QUARTER] ENDING	
\$[32,007,814]	
City of Los Angeles	
Multifamily Mortgage Revenue Note	
(SP7 Apartments)	
Series 2018C	

The undersigned, being the Authorized Borrower Representative of SP7 Apartments LP, a California limited partnership (the "Borrower"), has read and is thoroughly familiar with the provisions of the various loan documents associated with the Borrower's participation in the multifamily housing program of the City of Los Angeles (the "City"), including, without limitation, the Regulatory Agreement and Declaration of Restrictive Covenants, dated as of [September] 1, 2018 (the "Regulatory Agreement"), among the Borrower, the City and [FISCAL AGENT], as Fiscal Agent relative to the property located at 519 East 7th Street, Los Angeles, California.

As of the date of this Certificate, the following percentages of completed residential units in the Project (as such term is defined in the Regulatory Agreement) (i) are occupied by Low Income Tenants (as such term is defined in the Regulatory Agreement) or (ii) are currently vacant and being held available for such occupancy and have been so held continuously since the date a Low Income Tenant vacated such unit, as indicated:

Occupied by Low Income Tenants:	Unit Nos and size
Held vacant for occupancy continuously since last occupied by Low Income Tenant:	Unit Nos and size
Vacant Units:	%
Low Income Tenants who commenced Occupancy of units during the Preceding [month/quarter]:	Unit Nos

Attached is a separate sheet (the "Occupancy Summary") listing, among other items, the following information for each unit in the Project: the number of each unit, the occupants of each unit and the size, in square feet, of each unit. It also indicates which units are occupied by Low Income Tenants and which units became Low Income Units during the preceding [month/quarter]. The information contained thereon is true and accurate.

The undersigned hereby certifies that (1) a review of the activities of the Borrower during such [month/quarter] and of the Borrower's performance under the Loan Agreement has been made under the supervision of the undersigned; (2) to the best of the knowledge of the undersigned, based on the review described in clause (1) hereof, the Borrower is not in default under any of the terms and provisions of the above documents [OR DESCRIBE THE NATURE OF ANY DEFAULT IN DETAIL AND SET FORTH THE MEASURES BEING TAKEN TO REMEDY SUCH DEFAULT]; and (3) to the knowledge of the Borrower, no Determination of Taxability (as such term is defined in the Regulatory Agreement) has occurred [OR, IF A DETERMINATION OF TAXABILITY HAS OCCURRED, SET FORTH ALL MATERIAL FACTS RELATING THERETO].

SP7 APARTMENTS LP, a California limited partnership

By: SP7 Apartments GP LLC, a California limited liability company, its general partner

> By: The Skid Row Housing Trust, a California nonprofit public benefit corporation, its sole member/manager

By:	
	Ben Rosen,
	Real Estate Development Director

[Signature Page to SP7 Apartments Certificate of Program Compliance]

#### EXHIBIT C

### FORM OF INCOME CERTIFICATION

NOTE TO APARTMENT OWNER: This form is designed to assist you in computing Annual Income in accordance with the method set forth in the Department of Housing and Urban this form is at all times up to date with the HUD Regulations.

Development ("HUD") Regulations (24 C.F.R. Part 5 Subpart F). You should make certain that SP7 Apartments, 519 East 7th Street, Los Angeles, California Re: The undersigned hereby (certify) (certifies) that: 1. This Income Certification is being delivered in connection with the undersigned's application for occupancy of Apartment #\_\_\_\_\_ in the SP7 Apartments located at 519 East 7th Street, in Los Angeles, California. List all the occupants of the apartment, the relationship (if any) of the various occupants, their ages, and indicate whether they are students (for this purpose, a student is any individual who has been, or will be, a full-time student at an educational institution during five months (whether consecutive or not) of the year in which this application is submitted, other than a correspondence school, with regular facilities and students). **Social Security** Student Occupant Relationship (Yes or No) Number Age (e) \_\_\_\_\_ (f) 3. If all of the occupants are students, answer the following questions for each occupant Is any student listed in paragraph 2 above married and files a joint return for federal income tax purposes? List any such students.

No

Not Applicable

Yes

Name(s)

(b) Is any student listed in paragraph 2 above (i) a single parent living with his/her children, (ii) not a dependent of another individual and (iii) whose children are not dependents of an individual other than their parents? List any such students.

Yes Name(s) No Not Applicable

(c) Is any student listed in paragraph 2 above a student receiving assistance under Title IV of the Social Security Act (Temporary Assistance for Needy Families)? List any such students.

Yes Name(s) No Not Applicable

(d) Is any student listed in paragraph 2 above a student enrolled in a job-training program receiving assistance under the Job Training Partnership Act or under other similar federal, State or local laws? List any such students.

Yes Name(s) No Not Applicable

(e) Is any student listed in paragraph 2 a student who was previously under the care and placement responsibility of a foster care program (under Part B or E of Title IV of the Social Security Act)? List any such students.

Yes Name(s) No Not Applicable

4. The total anticipated income for each person listed in paragraph 2 above during the 12-month period commencing with the date occupancy will begin, including:

full amount, before any payroll deductions, of wages, salaries, overtime, commissions, fees, tips, and bonuses; net income from the operation of a business or profession or from the rental of real or personal property (without deducting expenditures for business expansion or amortization of capital indebtedness or any allowance for depreciation of capital assets); interest and dividends (including income from assets excluded below); the full amount of periodic payments from social security, annuities, insurance policies, retirement funds, pensions, disability or death benefits and other similar types of periodic payments, including any lump-sum payment for the delayed start of a periodic payment; payments in lieu of earnings, such as unemployment and disability compensation, workers' compensation and severance pay; all public assistance income; periodic and determinable allowances such as alimony and child support payments, and regular contributions or gifts received from persons not residing in the dwelling; all regular and special pay and allowances of members of the Armed Forces (whether or not living in the dwelling) who are the head of the family or spouse; and any earned income tax credit to the extent that it exceeds income tax liability;

### but excluding:

income from employment of children (including foster children) under the age of 18 years; payments received for the care of foster children or foster adults (usually individuals with disabilities, unrelated to the tenant family, who are unable to live alone); lump-sum additions to family assets, such as inheritances, insurance payments (including payments

under health and accident insurance and workers' compensation), capital gains and settlement for personal or property losses; amounts which are specifically for reimbursement of medical expenses; amounts of educational scholarships paid directly to the student or the educational institution, and amounts paid to a veteran for use in meeting the costs of tuition, fees, books and equipment, but in either case only to the extent used for such purposes; special pay to a serviceman head of a family who is away from home and exposed to hostile fire; amounts received under training programs funded by HUD; amounts received under Plan to Attain Self-Sufficiency; amounts for out-of-pocket expenditures incurred in connection with other public assistance programs; resident service stipend (not in excess of \$200 per month); amounts from state or local employment training programs; temporary, nonrecurring or sporadic income (including gifts); reparation payments paid by a foreign government to persons who were persecuted during the Nazi era; earnings in excess of \$480 for each full-time student 18 years old or older (excluding head of family and spouse); adoption assistance payments in excess of \$480 per adopted child; deferred periodic payments of supplemental social security income and benefits received in a lump sum; refunds or rebates of property taxes paid on the unit; payments from state agency to allow developmentally disabled family member to stay home; relocation payments under Title II of the Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970; foster child care payments; the value of coupon allotments for the purchase of food pursuant to the Food Stamp Act of 1964 which is in excess of the amount actually charged for the allotments; and payments to volunteers under the Domestic Volunteer Service Act of 1973; is as follows:

(	Occupant	Anticipated Annual Income	Source of Income or Employer
(a)			
(b)			
		<b>C</b>	
		Φ	
		_	
		\$	
TOTAL		\$	

5.(a) Do the persons whose income or contributions are included in paragraph 4 above have savings, stocks, bonds, equity in real property or other form of capital investment (excluding the values of necessary items of personal property such as furniture and automobiles and interest in Indian trust land)?

				_	Yes		No	
(b) disposed of a less than fair	ny asset:	s (other						ragraph 4 above ast two years at
				_	Yes		No	
(c) assets owned			to (a) or (b) by all such pe					alue of all such
				_	Yes		No	
(d)	If the a	answer	to (c) above i	s yes,				
	\$		insert the t	otal value	of all suc	h asset	s owned	or disposed of
		(ii)	state:					
			riod beginning					uch assets in the ne unit that you
	\$							
	above:		the amount	of such inco	me, if any	, that wa	as included	d in paragraph 4
	\$							
	NT - 141		16		- C 41:4	. I /		4 !41

- 6. Neither myself nor any other occupant of the unit I/we propose to rent is the owner of the rental housing project in which the unit is located (hereinafter, the "Borrower"), has any family relationship to the Borrower or owns directly or indirectly any interest in the Borrower. For purposes of this paragraph, indirect ownership by an individual shall mean ownership by a family member, ownership by a corporation, partnership, estate or trust in proportion to the ownership or beneficial interest in such corporation, partnership, estate or trust held by the individual or a family member and ownership, direct or indirect, by a partner of the individual.
- 7. This Income Certification is made with the knowledge that it will be relied upon by the Borrower to determine maximum income for eligibility to occupy the unit, and I/we declare that all information set forth herein is true, correct and complete and based upon information I/we deem reliable and that the statement of total anticipated income contained in paragraph 4 is reasonable and based upon such investigation as the undersigned deemed necessary.
- 8. I/we will assist the Borrower in obtaining any information or documents required to verify the statements made therein, including either an income verification from my/our present employer(s) or copies of federal tax returns for the immediately preceding calendar year.

- 9. I/we acknowledge that I/we have been advised that the making of any misrepresentation or misstatement (whether or not intentional) in this Income Certification will constitute a material breach of my/our agreement with the Borrower to lease the unit and will entitle the Borrower to prevent my/our occupancy of the unit and will be cause for immediate termination of such lease.
- 10. The undersigned hereby acknowledge and agree that on or before January 1 (or upon Lease renewal) of each year the undersigned and any other current residents of such apartment will complete and deliver a new Income Certification, in the form then in use, to the Borrower and that the undersigned's rent is subject to increase 30 days after written notice is given to the undersigned stating that the undersigned no longer qualifies as a Lower Income Tenant under the Tax Regulatory Agreement.
- 11. RESIDENT(S) STATEMENT: I/We certify that the statements are true and complete to the best of my/our knowledge and belief and are given under penalty of perjury. In the event this Income Certification is executed more than five days prior to the date I/we intend to occupy the unit, I/we hereby agree to update and recertify the accuracy of the information herein provided as of the date I/we first occupy the unit:

(a)	Date:	
(b)	Date:	
(c)	Date:	
(d)	Date:	
(e)	Date:	
(f)	Date:	
[The required]	signatures of all persons over the age of 18 years listed in po	aragraph 2 above
12.	Calculation of Eligible Income:	
(a)	Enter the amount entered for entire household in 4 above:	\$
(b)	Enter income derived from assets (line 5(d)(2)(A)):	\$
(c)	Subtract (b) from (a)	\$
(d)	Multiply the amount entered in 5(d)(1) by the current passbook savings rate to determine the total annual earnings on assets [5(d)(1)] if invested in passbook savings.  Passbook rate % X = \$	

are

(e)	Enter the greater of (b) or (d)	\$				
(f)	TOTAL ELIGIBLE INCOME (Line (e) + (c))	\$				
13.	The amount entered in 12(f):					
(a)	Qualifies the applicant(s) as a Lower Income T	enant(s).				
(b)	Does not qualify the applicant(s) as Lower Inco	ome Tenant(s).				
14.	Number of apartment unit assigned:					
	Bedroom size: Rent: \$					

	Tenan	t-paid I	Utilities:		
	Water		Gas	Electric _	
Trash		Other (list t	ype)		
		e aggre	gate anticipa	ted annual in	ccupied for a period of 31 consecutive days by come as certified in the above manner upon their d them as Lower Income Tenants?
				_	Yes No
	16.	Metho	od used to ver	rify applicant(	(s) income:
			Employer in	ncome verific	ation
			Social Secu	rity Administ	ration verification
			Department	t of Social Ser	vices verification
			Copies of ta	ax returns	
			Other (		
	17.	Metho	od used to ver	ify responses,	if any, in paragraph 3 of this Income Certification:
			Copies of ta	ax returns	
			Evidence of	f participation	in an enumerated program
named	in par	cumen agraph	tation submit 2 of this In	ted pursuant come Certific	Based on the representations herein and upon the to paragraph 8 hereof, the family or individual(s) cation is/are eligible under the provisions of the rictive Covenants to live in a unit in the Project.
Date _					Signature of Authorized Borrower Representative:
					ByName
					Title
	EXEC	UTION	OF PARAC	GRAPHS 19 A	AND 20
		IS	IS NOT N	ECESSARY	
	Initials	z.			

	me Certification was executed by me/us more than five days prior to unit, I/we hereby update and recertify the accuracy of the information
1 2	, 20 and state:
	No additional information is required to be provided to make this n true and correct on the date of this certification.
	The following information is provided to update the information in the Income Certification:
I	[Remainder of page intentionally left blank]

	(a)		Date:	
	(b)		Date:	
	(c)		Date:	
	(d)		Date:	
	(e)		Date:	
	(f)		Date:	
		ne Certificati	on have, pursuan	ENT: The family or individual(s) named in paragraph 2 at to paragraph 19 hereof, updated and recertified the eally set forth in paragraph 19 hereof.
Date _				Signature of Authorized Borrower Representative
				Ву
				Name
				Title

[Remainder of page intentionally left blank]

### **INCOME VERIFICATION**

(for employed persons)

The undersigned employee has applied for a rental unit located in a project financed by the issuance of a note by the City of Los Angeles for persons of low or moderate income. Every income statement of a prospective tenant must be stringently verified. Please indicate below the employee's current annual income from wages, overtime, bonuses, commissions or any other form of compensation received on a regular basis.

	Annual Wag	ges							
	Overtime								
	Bonuses								
	Commission	ns							
	Total Curren	nt Income							
knowl	-	ertify that the	statements	above are	e true and	l complete	to the	best	of my
Date									
				Ву					
				Name _					
				Title					

I hereby grant you permission to disclose my income to					
in order that they may determine my income eligibility for rental of an apartment located in their project which has been financed by an issuance of a note by the City of Los Angeles.					
Date Signature					
Please send form to:					

[Income verification signature page]

## **INCOME VERIFICATION**

(for self-employed persons)

I hereby attach copies of my individual federal and state (if applicable) income tax returns
for the immediately preceding calendar year and certify that the information shown in such income
tax returns is true and complete to the best of my knowledge.

Date		
	Signature	

#### **EXHIBIT D**

#### FORM OF ANNUAL TENANT INCOME RECERTIFICATION

## CITY OF LOS ANGELES ANNUAL TENANT INCOME RECERTIFICATION

Projec	ct name				
Apart	ment # Date of C	Original Certification_			
Resid	ent name				
This f which follow	HE RESIDENT: form is a continuation of the C was previously discussed with wing information each year whe so and the City to maintain this	h you. In order to ke en you renew your leas	ep you on the. The Borr	he qualifying list, you ower is required by the	will need to update the
1) 2)	chold Composition:  Please list all of those indiv Please list the anticipated and usemaker, or unemployed, etc.  If college or technical school	nnual income of all occ—please list as such).	cupants of yo		18 years of age or older
	NAME	SS#	AGE	ANTICIPATED ANNUAL INCOME*	OCCUPATION/STUDEN
1) 2)					
3)					
4)					
5) 6)					
7)					
DO Y	INCOME DEFINITION ATT OU OWN OR HAVE YOU A HE PAST YEAR?			POSED OF ANY AS	SETS OVER \$5,000.00

If all persons residing in your apartment are full-time students, please indicate for each such person whether they are: (1) a single parent living with his/her children; (2) a student receiving assistance under Title IV of the Social Security Act (Temporary Assistance for Needy Families); (3) a student enrolled in a job-training program receiving assistance under the Job Training Partnership Act or under other similar federal, state or local laws; (4) a student who was previously under the care and placement responsibility of a foster care program (under Part B or E of Title IV of the Social Security Act); or (5) a student who is married and files a joint return. Single parents described in (1) above may not be dependents of another individual and their children may not be dependents of another individual other than their parents.

#### Please have all occupants over the age of 18 sign this certification.

I/we acknowledge that I/we have been advised that the making of any misrepresentation or misstatement in this declaration will constitute a material breach of my/our agreement with the Borrower to lease the unit and will entitle the Borrower to prevent or terminate my/our occupancy of the unit by institution of an action for ejection or other appropriate proceedings.

	I/we declare under penalty of perjury that the foregoing is true and correct.					
	SIGNATURES:	DATE:				
1)						
2)						
3)						
4)						
MA	MANAGER'S SIGNATURE:					

#### **DEFINITION OF INCOME**

The full amount, before any payroll deductions, of wages, salaries, overtime, commissions, fees, tips, and bonuses; net income from the operation of a business or profession or from the rental of real or personal property (without deducting expenditures for business expansion or amortization of capital indebtedness or any allowance for depreciation of capital assets); interest and dividends (including income from assets excluded below); the full amount of periodic payments from social security, annuities, insurance policies, retirement funds, pensions, disability or death benefits and other similar types of periodic payments, including any lump-sum payment for the delayed start of a periodic payment; payments in lieu of earnings, such as unemployment and disability compensation, workers' compensation and severance pay; all public assistance income; periodic and determinable allowances such as alimony and child support payments, and regular contributions or gifts received from persons not residing in the dwelling; all regular and special pay and allowances of members of the Armed Forces (whether or not living in the dwelling) who are the head of the family or spouse; and any earned income tax credit to the extent that it exceeds income tax liability;

#### but excluding:

income from employment of children (including foster children) under the age of 18 years; payments received for the care of foster children or foster adults (usually individuals with disabilities, unrelated to the tenant family, who are unable to live alone); lump-sum additions to family assets, such as inheritances, insurance payments (including payments under health and accident insurance and workers' compensation), capital gains and settlement for personal or property losses; amounts which are specifically for reimbursement of medical expenses; amounts

of educational scholarships paid directly to the student or the educational institution, and amounts paid to a veteran for use in meeting the costs of tuition, fees, books and equipment, but in either case only to the extent used for such purposes; special pay to a serviceman head of a family who is away from home and exposed to hostile fire; amounts received under training programs funded by HUD; amounts received under Plan to Attain Self-Sufficiency; amounts for out-of-pocket expenditures incurred in connection with other public assistance programs; resident service stipend (not in excess of \$200 per month); amounts from state or local employment training programs; temporary, nonrecurring or sporadic income (including gifts); reparation payments paid by a foreign government to persons who were persecuted during the Nazi era; earnings in excess of \$480 for each full-time student 18 years old or older (excluding head of family and spouse); adoption assistance payments in excess of \$480 per adopted child; deferred periodic payments of supplemental social security income and benefits received in a lump sum; refunds or rebates of property taxes paid on the unit; payments from state agency to allow developmentally disabled family member to stay home; relocation payments under Title II of the Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970; foster child care payments; the value of coupon allotments for the purchase of food pursuant to the Food Stamp Act of 1964 which is in excess of the amount actually charged for the allotments; and payments to volunteers under the Domestic Volunteer Service Act of 1973.

## **EXHIBIT E**

### FORM OF CERTIFICATE OF CDLAC PROGRAM COMPLIANCE

Project Name Change:	No	Yes
(If project name has change as well as the new project na		allocation please note the original project name
If yes provide old and new I	Project Name:	
CDLAC Application No.:	18-316	
Bond Issuer Change:	No	Yes
(If Bond Issuer has changed please note the original Issu		result of refinance or refunding of an allocation ssuer.)
If yes provide the Name of	existing and New Issue	er
Contact Information		
Change in Borrower		
(If Borrower has changed sin Borrower as well as the new	_	g the CDLAC resolution please note the original
If yes provide the Name of t	he existing and New F	Borrower
Contact Information		
Change in Management Cor	npany No	Yes
If yes provide the Name of t	he New Management	Company
Has the Qualified Project Pe	eriod commenced?	No Yes
No Yes_	Already Sub:	mitted Certification
If yes please submit the Cor	struction Completion	Certificate (one time only)
Has the project been comple	eted and placed in serv	ice?
No Yes_	Already Sub	mitted Certification
If yes please submit Comple	etion Certification (one	e time only)

Have any of the following event limited to: defaults associated w Bond Default?		
No Yes	_	
If so, please describe and explain	1?	
Has a termination of the Regula year? Has proper noticing occur		termination planned in the next
No Yes	_	
If so, please describe and explain	1?	
Federally Bond Restricted Units (Reflected in in PSR)	Other Restrictions (reflected in PSR)	Total (Reported in CDLAC Resolution)
at 50% AMI	at 50% AMI	at 50% AMI
at 60% AMI	at 60% AMI	at 60% AMI
Total	Total	Total
Please attach a copy of the documentation.	project's TCAC Project Status	s Report (PSR) or equivalent
Please indicate the distribution o	of the CDLAC-restricted 10% of t	the 50% AMI units.
Bedroom Type	# of Units in PSR	# of Units in CDLAC Resolution
1 bedroom		
2 bedroom		
3 bedroom		

If the Project has committed to and is currently providing the service amenities for a term as specified in the CDLAC resolution, please verify the services are being provided: on a regular and ongoing basis, which are provided free of charge, and all hour requirements are being met:
After-school Programs
Educational, health and wellness, or skill building classes
Health and Wellness services and programs (not group classes)
Licensed Childcare provided for a minimum of 20 hours per week (Monday-Friday)
Bona Fide Service Coordinator/ Social Worker
Is the service being offered on an ongoing basis and provided free of charge (childcare excepted)?
No Yes If no please explain.
Are all hour requirements being met?  No Yes If no please explain.
Attach evidence demonstrating that the above listed services are being provided and have met the requirements in the CDLAC Resolution. Including but not limited to MOUs and or contracts associated with the services rendered, a 12-month schedule (current reporting year) of the services offered, flyers, sign-up sheets, etc.
Pursuant to Section 13 of Resolution No. 18-011 (the "Resolution"), adopted by the California Debt Limit Allocation Committee (the "Committee") on March 21, 2018, I, [], an Officer of the Borrower, hereby certify under penalty of perjury that, as of the date of this Certification, the above-mentioned Project is in compliance with the terms and conditions set forth in the Resolution. I further certify that I have read and understand the Resolution, which specifies that once the Note is issued, the terms and conditions set forth in the Resolution Exhibit A shall be enforceable by the Committee through an action for specific performance, negative points, withholding future allocations or any other available remedy.
Signature of Officer Date

Printed Name of Officer	
Title of Officer	

## **EXHIBIT F**

## FORM OF CONSTRUCTION COMPLETION CERTIFICATE

1) Project Name:	SP7 Apartments
as well as the new project name	ince the award of allocation please note the original project name e and request a change in the CDLAC resolution.)
2) CDLAC Application No.:	18-316
3) Name of Bond Issuer:	City of Los Angeles
new Borrower and request a ch	SP7 Apartments LP e since the award please note the original Borrower as well as the nange in the CDLAC resolution.)
5) The undersigned hereby cert, 20	ifies that all work on the Project was substantially completed as of
The undersigned hereby further	certifies that:
(a) the aggregate an	nount disbursed on the Loan to date is \$;
reimburse the undersign Regulatory Agreement)	bursed from proceeds of the Note have been applied to pay or ned for the payment of Project Costs (as that term is used in the and none of the amounts disbursed from the proceeds of the Note y or reimburse any party for the payment of costs or expenses other
applied to pay or reimb that term is used in the F from the proceeds of the	the amounts disbursed from the proceeds of the Note have been urse the Borrower for the payment of Qualified Project Costs (as Regulatory Agreement) and less than 25% of the amounts disbursed Note, exclusive of amounts applied to pay the costs of issuing the I to pay or reimburse the Borrower for the cost of acquiring land.
6) The undersigned hereby ce Project Period.	rtifies the Project meets the general federal rule for a Qualified
No Yes	<u> </u>
	lling units in the Project financed in part from the proceeds of the d on, 20 and
	lling units in the Project financed in part from the proceeds of the

7) If no to ( Qualified Pr	• • • •	ereby certifies the Project meets the special federal rule for a
No_	Yes	
	e for occupancy within	ition/rehabilitation where no more than 90% of the units were n 60 days of the earlier of the Project acquisition or the Note
(a)	Note was issued on _	
(b)	Property was acquire	ed on, 20
(c) of the		e units were available to occupy (within 60 days of the earlier ssuance) is
Signature of	Officer	Date
Printed Nam	e of Officer	_
Title of Office	cer	_
Phone Numb	per	_

## **EXHIBIT G**

## **CDLAC RESOLUTION**

## **EXHIBIT H**

### FORM OF SLAVERY DISCLOSURE ORDINANCE CERTIFICATE

#### **EXHIBIT I**

#### **ACCESSIBILITY COVENANTS**

The Accessibility Covenants (the "Covenants") herein are attached to the Regulatory Agreement as an exhibit as an exhibit and the Borrower hereby agrees to comply with each of the requirements of the City set forth as follows:

**Section 1. Definitions.** Terms not otherwise defined herein shall have the meanings assigned thereto in the Regulatory Agreement as applicable. The following terms shall have the respective meanings assigned to them in this Section 1 unless the context in which they are used clearly requires otherwise:

"Accessible," when used with respect to a Housing Unit or a Housing Development, means and refers to full compliance with the requirements of the Accessibility Standards.

"Accessible Housing Development" means a Housing Development that is Accessible, including Accessible public and common use areas, as well as the number and type of Accessible Housing Units that are required to be Accessible by this Agreement.

"Accessible Housing Units" or "Accessible Unit" refers collectively to Housing Units with Mobility Features and Housing Units with Hearing/Vision Features that are Accessible, on an accessible route, and in an Accessible Housing Development.

"Accessibility Laws" means Section 504 of the Rehabilitation Act of 1973, 29 U.S.C. §794 et seq, and the implementing regulations at 24 C.F.R. Part 8, as well as the requirements of the Uniform Federal Accessibility Standards ("UFAS"); the Americans with Disabilities Act, ("ADA"), 42 U.S.C. §12101, et seq. including (1) Title II of the ADA and the implementing standards ("2010 ADA Standards") at 28 C.F.R. Part 35 and the 2004 ADA Accessibility Guidelines ("ADAAG") and Title III of the ADA, Part 36 and the implementing standards ("2010 ADA Standards") at 28 C.F.R. Part 36, subpart D; California Government Code Section 11135 et seq.; the federal Fair Housing Act of 1968, as amended ("FHA"), 42 U.S.C. §§3601-3620; and its implementing regulations as 24 C.F.R. Parts 100, 103, 108, 110, and 121; implementing regulations and design standards for each of the preceding statutes; and the California Building Code Sections 11A and 11B.

"Accessibility Standards" means the standards adopted under the Accessibility Laws, specifically the following compliance standards:

- a. the new construction requirements of 24 C.F.R. pt. 8, including 24 C.F.R. §§ 8.22 and 8.32 as well as the new construction requirements of UFAS, or their successor standards;
- b. the Alternative Accessibility Standard; or any future accessibility standard and other regulatory requirements applicable to newly constructed facilities in federally-assisted programs that may be adopted in a final rule issued by the U.S. Department of Housing and Urban Development ("HUD") pursuant to notice and comment rulemaking under Section 504 so long as such accessibility standard and regulatory requirements do not provide for less accessibility for persons with disabilities than either (i) or (ii);

- c. the requirements in ANSI A117.1-1986 and the Fair Housing Accessibility Guidelines, March 6, 1991, in conjunction with the Supplement to Notice of Fair Housing Accessibility Guidelines: Questions and Answers About the Guidelines, June 28, 1994;
- d. the accessibility provisions of the California Building Code Chapters 11A and 11B, or any future accessibility standard and other regulatory requirements applicable to newly constructed facilities adopted as part of the California Building Code;
- e. all applicable building codes in effect for the City of Los Angeles Building and Safety Department; and
- f. ADA accessibility provisions relevant to public use areas.

"Alternative Accessibility Standard" means and refers to the alternative accessibility standard for new construction set out in HUD's notice at 79 Fed. Reg. 29,671 - Nondiscrimination on the Basis of Disability in Federally Assisted Programs and Activities (May 23, 2014), when used in conjunction with the new construction requirements of HUD's regulations at 24 C.F.R. pt. 8, 24 C.F.R. § 8.22, and the new construction requirements of 28 C.F.R. pt. 35, including the 2010 Standards for Accessible Design as defined in 28 C.F.R. § 35.104 and as applied to public entities (excluding any elevator exceptions).

"Fair Housing Policy Related to Disability" means the document containing the policy of the City, as amended periodically, that all affordable housing developments monitored by the City be constructed and operated in accordance with all applicable disability and fair housing laws and under which the Borrower is required to create a Property Management Plan ("Property Management Plan" or "PMP") as described in Section 4 that must comply with the requirements and guidance in the Fair Housing Policy Related to Disability. The PMP must be consistent with the City's Property Management Plan template and must be approved by the City along with other requirements, as amended periodically.

"Housing Development" or "Development" means the whole of one or more residential structures and appurtenant structures in the project, including common walkways and parking lots that were or are designed, constructed, altered, operated, administered or financed in whole or in part in connection with the issuance of Bonds.

"Housing Unit" means a single unit of residence in the Housing Development that provides spaces for living, bathing, and sleeping.

"Housing Unit with Hearing/Vision Features" means a Housing Unit that is located on an accessible route and complies with 24 C.F.R. §§ 8.22 and 8.23 and all applicable provisions of UFAS, or the comparable provisions of the Alternative Accessibility Standard including but not limited to § 809 and specifically subsection § 809.5 of the 2010 ADA Standards for Accessible Design, and with the California Building Code Chapters 11A & 11B. Hearing/Vision Features include but are not limited to visual alarms (UFAS §§ 4.34.10, 4.28.3), auxiliary alarms (UFAS §§ 4.34.10, 4.28.4), protections against protruding objects (UFAS § 4.4), stairway requirements (UFAS §§ 4.9, 4.26.4), protections against exposed pipes and surfaces (UFAS §§ 4.19.4, 4.24.6, 4.34.6.5(8)), audible alarms (UFAS § 4.28.2), signage (UFAS § 4.30), consumer information (UFAS § 4.34.4), and range, cooktop, and oven controls (UFAS §§ 4.34.6.6, 4.34.6.7).

"Housing Unit with Mobility Features" means a Housing Unit that is located on an

accessible route and complies with 24 C.F.R. §§ 8.22 and 8.23 and all applicable provisions of UFAS, or the comparable provisions of the Alternative Accessibility Standard including but not limited to § 809 and specifically subsections §§ 809.2 through 809.4 of the 2010 ADA Standards, and with the California Building Code Chapter 11 B. A Housing Unit with Mobility Features can be approached, entered and used by persons with mobility disabilities, including people who use wheelchairs.

"UFAS" means the Uniform Federal Accessibility Standards and refers to a set of scoping requirements and standards for the design and construction of buildings and facilities to ensure that they are readily accessible to and usable by persons with disabilities. See Appendix A to 24 C.F.R. subpart 40 for residential structures and Appendix A to 41 C.F.R. subpart 101-19.6 for general-type buildings (UFAS is also available on-line at http://www.access-board.gov).

**Section 2. Borrower Obligations.** The Borrower represents, warrants, covenants and agrees as follows:

- a. A State of California Certified Access Specialist ("CASp") who is a licensed architect or engineer must be identified as part of the development team. The CASp cannot be the architect of record for the project. The cost of CASp activities and certifications should be included in the application's project budget.
- b. The Housing Development shall be constructed in accordance with the Accessibility Standards in Section 1 above to ensure accessibility for persons with disabilities. Borrower must work with their CASp to ensure that the Housing Development complies with those Accessibility Standards.
- c. An accessibility report by a CASp inspector certifying that the Housing Development is compliant with all applicable Accessibility Standards, as defined in Section 1, above,, must be submitted to and approved by the City at the following phases of the project development:
  - 1. Accessibility Design Review Report and a pdf copy of the plans are due for review by the City when construction documents have been developed, and prior to the submission of plans to Los Angeles Building and Safety Department.
  - 2. The Accessibility Design Review Report must be approved by the City before building permits can be issued.
  - 3. Accessibility Progress Inspection Reports conducted after all rough inspections have been signed off by the Los Angeles Building and Safety Department and prior to closing of walls.
  - 4. The Final Accessibility Report at completion of construction must be approved by the City before any retention payment, certificate of occupancy, and final building permit can be issued by the Los Angeles Building and Safety Department.
  - 5. A list of State Certified Disabled Access Specialists can be found at the following link: https://www.apps.dgs.ca.gov/casp/casp/casp certified list.aspx.

- d. If applicable, applicants/developers/owners must list all applicable standards on title page of plans and include the following note: "This is a publicly funded housing project and must comply with California Building Code Chapters 11A & 11B").
- e. If the Development is to be rehabilitated, accessibility retrofit of the Housing Development shall take place concurrently with any project rehabilitation.
- f. The Accessible Housing Units shall be prioritized for persons with disabilities who have a disability-related need for the accessibility features of the Accessible Housing Unit.
- g. At least ten percent (10%) of the total Housing Units in the Housing Development shall be constructed and maintained by the Borrower as Housing Units with Mobility Features.
- h. At least <u>four percent</u> (4%) of the total Housing Units in the Housing Development shall be constructed and maintained by the Borrower as Housing Units with Hearing/Vision Features
- i. The 4% and 10% calculations shall each be based on the total number of units in the Housing Development. In determining the number of Accessible Housing Units required, any fractions of units shall be rounded up to the next whole number. Required Accessible Housing Units shall, to the maximum extent feasible and subject to reasonable health and safety requirements, be distributed throughout projects and sites, and shall be available in a sufficient range of sizes and amenities so that a qualified individual with a disability has a choice of living arrangements that is, as a whole, comparable to that of other persons eligible for housing assistance under the same program.
- j. The Accessible Housing Units shall be affordable for households with incomes at 30%, 50%, and 80% of area median income.
- k. The project shall comply with the City's Accessibility Regulations Matrix & Overview, Accessible Design/Construction Compliance Requirements, and the Accessibility Report Requirements.
- 1. Borrowers shall adopt and comply with the City's Fair Housing Policy Related to Disability, as amended.
- m. Borrowers and property managers (including resident managers and on-site managers) of the Housing Development shall attend the City's Fair Housing for People with Disabilities workshops.
- n. Following reasonable notice to Borrower, Borrower shall allow the City to conduct periodic onsite inspections of the Housing Development in order to verify compliance with the Accessibility Standards and policies.

- o. The Housing Development as a whole and all Housing Units shall meet the requirements of the Accessibility Standards as defined in Section 1, above, and any requirements of the City.
- p. Borrower shall provide a list to the City of all Accessible Housing Units with unit number, bedroom size and type of Accessible Housing Unit ("Housing Unit with Hearing/Vision Features" or "Housing Unit with Mobility Features").

Section 3. Occupancy of Accessible Housing Units. Borrower shall follow the requirements of the City's Fair Housing Policy Related to Disability to assure that information regarding the availability of Accessible Housing Units reaches eligible individuals with disabilities, and will take reasonable, nondiscriminatory steps to maximize the utilization of such units by eligible individuals whose disability requires the accessibility features of the particular unit. To this end, Borrower will take the following steps when an Accessible Housing Unit becomes vacant:

- a. First, Borrower will offer the unit to a current occupant of the Housing Development who has requested and needs the features of an Accessible Housing Unit;
- b. Second, Borrower will offer the unit to a current occupant of a Housing Development under common control who has requested and needs the features of an Accessible Housing Unit;
- c. Third, Borrower will offer the unit to an eligible, qualified applicant on the waiting list for Accessible Housing Units who needs the features of an Accessible Housing Unit;
- d. Fourth, Borrower will make reasonable efforts to advertise the Accessible Housing Unit to qualified individuals who need the accessible features, including listing it as available to individuals who need the accessible features at http://www/Housing.LACity.org, distributing the information about the accessible vacancy in accord with the Borrower's City approved Property Management Plan, distributing it to the most recent list from the City of organizations that serve people with disabilities, and sending an e-blast to parties on the Housing.LACity.org website Outreach List.

In the event that more than one household has requested an Accessible Housing Unit, Borrowers will offer the Unit to households in order on the appropriate waiting list within each category.

If, after using the process identified above, there are no households who need the features of that Accessible Housing Unit, then Borrower may offer the unit to the next household on the conventional waiting list. Should that household choose not to occupy the Accessible Housing Unit, they will remain at the same position on the conventional waiting list. If the household chooses to occupy the Accessible Housing Unit, the tenant must sign a lease addendum in the form approved by the City. The lease addendum requires the household to move to the next available, comparable, conventional unit, when given legal notice by the Housing Development that there is an eligible applicant or existing resident with a disability who requires the accessibility features of that Unit

For individuals who are required to vacate an Accessible Unit because it is needed by an individual with a disability, Borrowers will pay the costs of transferring to a comparable conventional unit, including new utility deposit(s), if required, and reasonable moving expenses.

**Section 4. Rental Policies.** The Borrower shall adopt rental policies that meet the requirements of the ADA, the Fair Housing Act, FEHA, and other federal and state laws and regulations as applicable, and of the Fair Housing Policy Related to Disability of the City, as amended. Borrower shall develop and utilize a PMP approved by the City which describes affirmative marketing, tenanting, and other procedures to ensure that the Housing Development meets all of the fair housing requirements for individuals with disabilities. Within 90 days of bond issuance, the Housing Development must have an approved PMP.

Rental applications will include a section to be filled out by applicants requesting an accommodation or modification. Applicants will not be required to disclose a disability under any circumstances unless requesting an accommodation or modification and that disclosure shall pertain only to the accommodation or modification being requested. Outreach efforts to the disability community shall include, but not be limited to, notices and other communications describing the availability of such units, specific information regarding the features of accessible units, eligibility criteria, and application procedures. These, and additional procedures, are incorporated into the City's Fair Housing Policy Related to Disability, as amended over time.

**Section 5. Residential Rental Property.** The Borrower hereby represents, covenants, warrants and agrees as follows:

- a. All of the housing units in the Housing Development will be similarly constructed units, and each income restricted unit in the project will contain complete separate and distinct facilities for living, sleeping, eating, cooking and sanitation for a single person or a family, including a sleeping area, bathing and sanitation facilities and cooking facilities equipped with a cooking range and oven, a sink and a refrigerator. Each of Accessible Housing Units comply the shall also with requirements. Notwithstanding the foregoing, a unit shall not fail to be treated as a residential unit merely because such unit is a single room occupancy unit within the meaning of Section 42(i)(3)(B)(iv) of the Code even though such housing may provide eating, cooking and sanitation facilities on a shared basis.
- b. All of the housing units (which shall not include any manager units) will be available for rental on a continuous basis to members of the general public, and the Borrower will not give preference to any particular class or group in renting the housing units in the project, except to the extent that for the following: (1) Accessible Housing Units shall be made available on a priority basis to persons who need the accessible features, as described in Section 3 above; (2) any dwelling units are required to be leased or rented to low income tenants or persons [62] years of age and older, (3) the requirements of any regulatory agreement executed between the Borrower and HUD or between the Borrower and a subordinate lender (including the City), (4) the requirements of any Section 8 Housing Assistance Payments Contract with respect to the project, and (5) any preference Borrower gives to a class of persons permitted to be given preference pursuant to the Code, State law and other applicable federal law.

Section 6. Monitoring Requirements. The City will monitor the initial production and ongoing occupancy of the Accessible Housing Units and the Accessible Housing Development to ensure full compliance with the Accessibility Standards and the policies in Sections 1 - 4, above. In order to determine compliance with the Accessibility Standards, Borrower shall submit and the City shall review and approve a CASp Inspection Report of the housing development that identifies the necessary and required design elements to make the units and site accessible for individuals with disabilities. The City shall inspect the construction and/or rehabilitation to verify that the correct number of Accessible Housing Units have been produced and that the necessary and required design elements have been constructed to make the units and site accessible for individuals with disabilities, in compliance with Sections 1 - 4 above, and supported by an independent CASp consultant's report.

The City will utilize the Housing Development's City approved PMP and Fair Housing Policy Related to Disability, to monitor ongoing occupancy compliance of the Accessible Housing Units and nondiscrimination in regards to individuals with disabilities. Compliance with the policies shall include, but not be limited to, target marketing, establishing and monitoring a waiting list specific to the Accessibility Housing Units, responding to reasonable accommodation and modification requests, implementation of the service animal policies, implementation of the policies for re-leasing empty Accessible Housing Units, and all elements contained in the Fair Housing Policy Related to Disability, as amended over time.

**Section 7. Maintenance of Records.** The Borrower agrees to keep and maintain books, accounts, reports, files, records, and other documents in accordance with the Fair Housing Policy Related to Disability. Borrower shall maintain all records until August 1, 2026 or for 5 years, whichever is later.

**Section 8. Notices, Demands, Payments and Communication**. Formal notices, demands, payments and communications between the City and the Borrower shall by sufficiently given and shall not be given unless dispatched by registered or certified mail, postage prepaid, return receipt requested, or delivered personally to the principal office of the City as follows:

CITY:

City of Los Angeles
Housing and Community Investment Department
P.O. Box #532729
Los Angeles, CA 90053-2729

Attention: Portfolio Management and Accessible Housing Program

BORROWER:		
W/copy to:		

Section 9. Term of the Accessibility Covenants. The covenants contained in the Covenants shall become effective upon the issuance of the Bond and shall terminate no earlier than at the end of the CDLAC Conditions set forth in the California Debt Limit Allocation Committee Resolution for the project, it being expressly agreed and understood that the provisions hereof are intended to survive the retirement of the Bond.

Section 10. Covenant to Run with the Land. The Borrower hereby subjects the project to the covenants, reservations, and restrictions set forth in the Covenants. The City and the Borrower hereby declare their express intent that the covenants, reservations and restrictions set forth herein shall be deemed covenants running with the land and shall pass to and be binding upon the Borrower's successors in title to the project. Each and every contract, deed, or other instrument hereafter executed covering or conveying the project or any portion thereof shall conclusively be held to have been executed, delivered and accepted subject to such covenants, reservations, and restrictions, regardless of whether such covenants, reservations and restrictions are set forth in such contract, deed or other instruments.

Section 11. Default; Enforcement. As part of ensuring compliance with the Accessibility Standards, the City or its agent, will conduct periodic on-site visits inspecting the Housing Development, which inspections may include inspecting the Housing Units and common areas, tenant files, logs and other records. Should the Borrower fail to comply with the Accessibility Standards or any of the provisions of the Covenants, the City will first issue an Order to Comply ("Order") stating the element of the Housing Development that is out of compliance, and providing a date by which the Borrower must comply. The Order shall give the Borrower not more than 30 days to correct the violation, or such additional time as the City may grant if the Borrower is taking steps to correct the violation ("Compliance Date"), and diligently pursues such action until the default is corrected, which extension is in the City's sole discretion. The City shall re-inspect the Housing Development within 10 days of the Compliance Date specified in the Order or any extension, however failure to inspect or re-inspect within the time frame does not remove the obligation of the Borrower to comply with the Order.

If the Order is issued and the violation continues to exist after the Compliance Date, then the City shall declare an "Event of Default" and may take any one or more of the following steps:

a. Inspection Fee for Non-Compliance. In the event the Borrower fails to comply with the Order within the Compliance Date, the Borrower shall be liable for subsequent inspection fees in the amount approved by Council until compliance has been achieved. Failure to pay the assessed inspection fee within 30 days of the date of invoice, will result in a late charge equal to or two times the fees and a collection fee equal to 50 percent of the original fee shall be imposed if any fee imposed is not paid within 30 days of service of notice of the imposition of the fee.

The late fee may be imposed without a hearing but may be appealed to the General Manager of HCID. The appeal shall be made in writing, and shall specify the grounds for the appeal. The appeal shall be filed with HCID within ten calendar days of the

issuance of the imposition of the late fees and costs. The General Manager or his designee shall issue a decision within ten calendar days of the filing of the appeal. A copy of the decision shall be served on the person or entity subject to the Order or fee by first class United States mail, postage prepaid, or in person. The City shall have the right to bring legal action in any court to enforce the Order and collect the amount of outstanding fees and penalties. City may waive the penalty imposed pursuant to this section if City determines that good causes exist for the Borrower's failure to pay in a timely manner.

- b. By mandamus or other suit, action or proceeding at law or in equity, including injunctive relief, require the Borrower to perform its obligations and covenants hereunder or enjoin any acts or things which may be unlawful or in violation of the rights of the City hereunder;
- c. Filing of a complaint or referral to HUD or other appropriate agencies for further enforcement actions;
- d. Have access to and inspect, examine and make copies of all or a portion of the books and records of the Borrower pertaining to the project, in order to ensure compliance with all provisions of the Covenants, including records relating to the accessibility of the Accessible Housing Units; and
- e. Take such other action at law or in equity as may appear necessary or desirable to enforce the obligations, covenants and agreements of the Borrower hereunder.

Section 12. Americans with Disabilities Act. The Borrower hereby certifies that it and its property manager and any agent, contractor and subcontractor will comply with the Accessibility Standards, as defined, and the policies described in Sections 2-5. The Borrower and any contractor and subcontractor will provide reasonable accommodations and modifications to allow qualified individuals with disabilities to have access to and to participate in its programs, services, and activities in accordance with the applicable provisions of the ADA, the ADAAG, Section 504, the UFAS, the FHA, California Government Code 11135 *et seq.*, the California Building Code Chapters 11A and 11B, and all subsequent amendments to those laws. The Borrower and any contractor and subcontractor will not discriminate against persons with disabilities or against persons due to their relationship or association with a person with a disability. Any contract and subcontract entered into by the Borrower, relating to this Regulatory Agreement and the Project, to the extent allowed hereunder, shall be subject to the provisions of this paragraph.

**Section 13. Governing Law**. The Covenants shall be governed by the laws of the State of California.

**Section 14. Parties Bound**. The provisions of the Covenants shall be binding upon and inure to the benefit of the City and Borrower and their respective successors and assigns.

**Section 15. Severability.** Every provision of the Covenants is intended to be severable. If any provision of the Covenants shall be held invalid, illegal, or unenforceable by a court of

competent jurisdiction, the validity, legality, and enforceability of the remaining provisions shall not in any way be affected or impaired.

**Section 16. Waiver.** Any waiver by the City of any obligation in the Covenants shall be in writing. No waiver will be implied from any delay or failure by the City to take action on any breach or default of Borrower or to pursue any remedy allowed under the Covenants or applicable law. Any extension of time granted to Borrower to perform any obligation under the Covenants shall not operate as a waiver or release from any of its obligations under the Covenants. Consent by the City to any act or omission by Borrower shall not be construed to be a consent to any other or subsequent act or omission or to waive the requirement for the City's written consent to future waivers.

**Section 17. Modifications**. There shall be no amendment or modification of the Covenants without the prior written approval of the City. Any amendment or modification of the Covenants shall be by a written instrument executed by the City and the parties to the Covenants and the Regulatory Agreement or their successors in title, and duly recorded in the real property records of the County of Los Angeles, California.

**Section 18. Conflicts.** If the provisions of the Covenants are inconsistent with the provisions of the Regulatory Agreement, the provisions of the Covenants shall be controlling.

**Section 19. Third Party Beneficiary**. The City is intended to be and is a third party beneficiary of the Covenants, and the City shall have the right (but not the obligation) to enforce the provisions of the Covenants and to pursue an action for specific performance of such provisions or other available remedy at law or in equity.

### **EXHIBIT J**

## FORM OF CERTIFICATE OF QUALIFIED PROJECT PERIOD

Project Name:	SP7 Apartments
	nce the award of allocation please note the original project name and request a change in the CDLAC resolution.)
CDLAC Application No.:	18-316
Name of Note Issuer:	City of Los Angeles
Name of Borrower: (If Borrower has changed since Borrower and request a change	the award please note the original Borrower as well as the new
Project meets the general federal Yes No	al rule for a Qualified Project Period.
(a) 10% of the dwelling units in Note were first occupied on	n the project financed in part from the proceeds of the captioned, 20; and
(b) 50% of the dwelling units in Note were first occupied on	n the project financed in part from the proceeds of the captioned, 20
Project meets the special federal Yes No	l rule for a Qualified Project Period. —
	uisition/rehabilitation where more than 90% of the units were not 60 days of the earlier of the project acquisition or the Note Issuance
(a) Note was issued on (b) Date 12 months after the No	
Signature of Officer	
Printed Name of Officer	
Title of Officer	
Phone Number	

### **HCIDLA** Request for Issuance of Bonds for the SP7 Project

## Attachment E

Bond Regulatory Agreement for SP7 (647 S. San Pedro Street) on next page.

RECORDING REQUESTED BY AND WHEN RECORDED RETURN TO:

KUTAK ROCK LLP 1650 FARNAM STREET OMAHA, NE 68102 ATTENTION: J. TOGER SWANSON, ESQ.

# REGULATORY AGREEMENT AND DECLARATION OF RESTRICTIVE COVENANTS

by and among

#### CITY OF LOS ANGELES,

as City

and

### [FISCAL AGENT],

as Fiscal Agent

and

#### SP7 APARTMENTS LP,

as Borrower

relating to

\$[32,007,814] City of Los Angeles Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C

Dated as of [September] 1, 2018

647 S. San Pedro

### TABLE OF CONTENTS

		Page
Section 1.	Definitions and Interpretation	2
Section 2.	Acquisition, Rehabilitation and Equipping of the Project	9
Section 3.	Residential Rental Property	11
Section 4.	Low Income Tenants; Records and Reports	13
Section 5.	Tax-exempt Status of the Note	16
Section 6.	Additional Requirements of the Act	17
Section 7.	Additional Requirements of the City	18
Section 8.	Modification of Covenants	24
Section 9.	Indemnification	25
Section 10.	Consideration	27
Section 11.	Reliance	27
Section 12.	Project in the City	27
Section 13.	Sale or Transfer of the Project; Equity Interests	27
Section 14.	Term	29
Section 15.	Covenants To Run With the Land	30
Section 16.	Burden and Benefit	31
Section 17.	Uniformity; Common Plan	31
Section 18.	Default; Enforcement	31
Section 19.	The Fiscal Agent	32
Section 20.	Recording and Filing.	33
Section 21.	Governing Law	33
Section 22.	Amendments	33
Section 23.	Notices	33
Section 24.	Severability	35
Section 25.	Multiple Counterparts	36
Section 26.	Nondiscrimination and Affirmative Action	36
Section 27.	Business Tax Registration Certificate	36
Section 28.	Financial Obligations Personal to Borrower	36
Section 29.	[Reserved]	37
Section 30.	Child Support Assignment Orders	37

Section 31.	Americans with Disabilities Act	38
Section 32.	Slavery Disclosure Ordinance	38
Section 33.	Requirements of CDLAC	38
EXHIBIT A	DESCRIPTION OF PROJECT SITE	
EXHIBIT B	FORM OF CERTIFICATE OF CONTINUING PROGRAM COMPLIAN	CE
EXHIBIT C	FORM OF INCOME CERTIFICATION	
EXHIBIT D	FORM OF ANNUAL TENANT INCOME RECERTIFICATION	
EXHIBIT E	FORM OF CERTIFICATE OF CDLAC PROGRAM COMPLIANCE	
EXHIBIT F	FORM OF REHABILITATION COMPLETION CERTIFICATE	
EXHIBIT G	CDLAC RESOLUTION	
EXHIBIT H	FORM OF SLAVERY DISCLOSURE ORDINANCE CERTIFICATE	
EXHIBIT I	ACCESSIBILITY COVENANTS	
EXHIBIT J	FORM OF CERTIFICATE OF QUALIFIED PROJECT PERIOD	

# REGULATORY AGREEMENT AND DECLARATION OF RESTRICTIVE COVENANTS

THIS REGULATORY AGREEMENT AND DECLARATION OF RESTRICTIVE

COVENANTS (this "Agreement" or this "Regulatory Agreement") is made and entered into and dated as of [September] 1, 2018 by and among the CITY OF LOS ANGELES, a charter city and municipal corporation in the State of California (together with any successor to its rights, duties and obligations, the "City"), [FISCAL AGENT], a national banking association in its capacity as Fiscal Agent (the "Fiscal Agent") under the Funding Loan Agreement dated as of [September] 1, 2018 (the "Funding Loan Agreement") by and among the City, MUFG Union Bank, N.A., as Funding Lender and the Fiscal Agent, with an office in Los Angeles, California, and SP7 APARTMENTS LP, a California limited partnership (the "Borrower").

#### WITNESSETH:

WHEREAS, pursuant to Section 248 of the City Charter of the City and Article 6.3 of Chapter 1 of Division 11 of the Los Angeles Administrative Code, as amended (collectively, the "Law"), and in accordance with Chapter 7 of Part 5 of Division 31 (commencing with Section 52075) of the Health and Safety Code of the State of California, as amended (the "Act"), the City is empowered to issue notes and other evidence of indebtedness to finance the acquisition, rehabilitation and equipping of multifamily rental housing; and

WHEREAS, on August 25, 2017, the City indicated its intent to provide for the issuance of a revenue bond or note to finance a portion of the acquisition, rehabilitation, construction and equipping of SP7 Apartments, a multifamily residential rental housing project a portion of which is located in the City of Los Angeles at 647 South San Pedro on the site more particularly described in Exhibit A hereto (the "Project") and the City Council of the City subsequently adopted a resolution (the "Resolution") authorizing the issuance of a note for such purpose; and

WHEREAS, in furtherance of the purposes of the Law, the Act and the Resolution, and as a part of the City's program of financing housing, the City has issued \$[32,007,814] maximum aggregate principal amount of its Multifamily Mortgage Revenue Note (SP7 Apartments) Series 2018C (the "Note") the proceeds of which will be used to fund a loan (the "Loan") to the Borrower to finance a portion of the acquisition, rehabilitation and equipping of the Project and the acquisition, construction and equipping of that companion multifamily housing project located at 519 East 7th Street (the "Companion Project"); and

WHEREAS, in order for interest on the Note to be excluded from gross income for federal income tax purposes under the Internal Revenue Code of 1986 (the "Code"), and the below-defined Regulations and rulings with respect to the Code, and in order to comply with the Law, the Act and the policies with respect to the City's housing program, the use and operation of the Project must be restricted in certain respects; and

WHEREAS, the City, the Fiscal Agent and the Borrower have determined to enter into this Regulatory Agreement in order to set forth certain terms and conditions relating to the acquisition, rehabilitation and equipping of the Project and in order to ensure that the Project will be used and

operated in accordance with the Code, the Law, the Act and the additional requirements of the City and the California Debt Limit Allocation Committee ("CDLAC");

NOW, THEREFORE, in consideration of the mutual covenants and undertakings set forth herein, and other good and valuable consideration, the receipt and sufficiency of which hereby are acknowledged, the City, the Fiscal Agent and the Borrower hereby agree as follows:

**Section 1. Definitions and Interpretation**. Terms not otherwise defined herein shall have the meanings assigned thereto in the Funding Loan Agreement or Loan Agreement, as applicable. The following terms shall have the respective meanings assigned to them in this Section 1 unless the context in which they are used clearly requires otherwise:

"Act" means Chapter 7 of Part 5 of Division 31 of the Health and Safety Code of the State of California, as the same may be amended from time to time (but only to the extent any such amendments, by their terms or by appropriate election of the City, apply to the Note outstanding as of the effective date of such amendments).

"Adjusted Income" means the adjusted income of a person (together with the adjusted income of all persons who intend to reside with such person in one residential unit) calculated pursuant to Section 142(d)(2)(B) of the Code.

"Affiliated Party" means a limited or general partner or member of the Borrower, a person whose relationship with the Borrower would result in a disallowance of losses under Section 267 or 707(b) of the Code or a person who, together with the Borrower, is a member of the same controlled group of corporations (as defined in Section 1563(a) of the Code, except that "more than 50 percent" shall be substituted for "at least 80 percent" each place it appears therein).

"Agreement" or "Regulatory Agreement" means this Regulatory Agreement and Declaration of Restrictive Covenants, as it may be amended from time to time.

"Area" means the Los Angeles Primary Metropolitan Statistical Area.

"Authorized Borrower Representative" means any person who, at any time and from time to time, may be designated as the Borrower's authorized representative by written certificate furnished to the City and the Fiscal Agent containing the specimen signature of such person and signed on behalf of the Borrower by or on behalf of any authorized general partner of the Borrower if the Borrower is a general or limited partnership, by any authorized managing member of the Borrower if the Borrower is a limited liability company, or by any authorized officer of the Borrower if the Borrower is a corporation, which certificate may designate an alternate or alternates, or in the event that such term shall refer to successors or assigns of the Borrower, any authorized general partner if the successor or the assignee is a general or limited partnership, any authorized managing member if the successor or assignee is a limited liability company or any authorized officer if the successor or the assignee is a corporation. The Fiscal Agent may conclusively presume that a person designated in a written certificate filed with it as an Authorized Borrower Representative is an Authorized Borrower Representative until such time as the Borrower files with it (with a copy to the City) a written certificate identifying a different person or persons to act in such capacity.

"Bond Counsel" means an attorney at law or firm of attorneys of nationally recognized standing in matters pertaining to the validity of, and the Tax-exempt nature of interest on, obligations issued by states and their political subdivisions, selected by the City and duly admitted to the practice of law before the highest court of any state of the United States of America or the District of Columbia but shall not include counsel for the Borrower or the Fiscal Agent.

"Borrower" means SP7 Apartments LP, a California limited partnership, and its successors and assigns.

"CDLAC" means the California Debt Limit Allocation Committee or its successors.

"CDLAC Conditions" has the meaning given such term in Section 33 hereof.

"CDLAC Resolution" means CDLAC Resolution No. 18-011, adopted on March 21, 2018, attached to this Regulatory Agreement as Exhibit G and related to the Project as such resolution may be modified or amended from time to time.

"Certificate of CDLAC Program Compliance" means the Certification of Compliance II for Qualified Residential Rental Projects to be filed with the City at the times specified in Section 33(a) of this Regulatory Agreement, such report to contain the information set forth in and to be in substantially the form attached hereto as Exhibit E or such other form required or otherwise provided by CDLAC from time to time.

"Certificate of Continuing Program Compliance" means the Certificate of Continuing Program Compliance and Statistical Report to be filed by the Borrower with the City and the Fiscal Agent at the times specified in Sections 4(d) and (f) of this Regulatory Agreement, such report to contain the information set forth in and to be in substantially the form attached hereto as Exhibit B or such other form as may from time to time be prescribed by the City.

"Certificate of Qualified Project Period" means the certificate to be filed by the Borrower upon commencement of the Qualified Project Period in substantially the form attached hereto as Exhibit J.

"City" means the City of Los Angeles, a charter city and municipal corporation of the State of California.

"Closing Date" or "Note Closing Date" means the date upon which the Note is initially funded in an amount equal to at least \$50,001.

"Code" means the Internal Revenue Code of 1986; each reference to the Code shall be deemed to include (a) any successor internal revenue law and (b) the applicable regulations whether final, temporary or proposed under the Code or such successor law.

"Companion Project" means the parcel or parcels of real property located at 519 East 7th Street in the City of Los Angeles, California, and all rights and appurtenances thereunto appertaining, as more particularly described in Exhibit A to the Companion Project Regulatory Agreement.

"Companion Project Regulatory Agreement" means that certain Regulatory Agreement and Declaration of Restrictive Covenants of even date herewith, by and among the City, the Fiscal Agent and the Borrower executed in connection with the issuance of the Note and restricting the real property comprising the Companion Project.

"Completion Date" means the date of the completion of the acquisition, rehabilitation and equipping of the Project, as that date shall be certified as provided in Section 2(i) hereof as specified in the Rehabilitation Completion Certificate.

"Compliance Period" means the period beginning on the first day of the Qualified Project Period and ending on the later of the end of the Qualified Project Period or such later date as set forth in Section 33 of this Regulatory Agreement.

"Costs of Issuance" means costs of issuing the Note as set forth in the Funding Loan Agreement.

"Determination of Taxability" means either (a) refusal by the Borrower to consent to any amendment or supplement hereto or to the Funding Loan Agreement which, in the written opinion of Bond Counsel delivered to the City, the Fiscal Agent, the Funding Lender and the Borrower, is necessary or advisable to maintain the exclusion of interest on the Note from gross income for federal income tax purposes; or (b) any of (i) the enactment of applicable legislation of which the Fiscal Agent has actual knowledge, (ii) a final judgment or order of a court of original or appellate jurisdiction of which the Fiscal Agent has actual knowledge, (iii) a final ruling or decision of the Internal Revenue Service of which the Fiscal Agent has actual knowledge or (iv) the filing with the Fiscal Agent of an opinion of Bond Counsel, in each case to the effect that the interest on the Note (other than interest on the Note for any period during which such Note is held by a "substantial user" of any facility financed with the proceeds of the Note or a "related person," as such terms are used in Section 147(a) of the Code) is includable in the gross incomes of all recipients thereof for federal income tax purposes. With respect to the foregoing, a judgment or order of a court or a ruling or decision of the Internal Revenue Service shall be considered final only if no appeal or action for judicial review has been filed and the time for filing such appeal has expired.

"Fiscal Agent" means [FISCAL AGENT] in its capacity as fiscal agent under the Funding Loan Agreement, together with its successors and assigns.

"Funding Loan Agreement" means the Funding Loan Agreement dated as of [September] 1, 2018 by and among the City, the Funding Lender and the Fiscal Agent relating to the issuance of the Note, as amended, modified, supplemented or restated from time to time.

"Hazardous Materials" means petroleum and petroleum products and compounds containing them, including gasoline, diesel fuel and oil; explosives; flammable materials; radioactive materials; polychlorinated biphenyls ("PCBs") and compounds containing them; lead and lead-based paint; asbestos or asbestos-containing materials in any form that is or could become friable; underground or above-ground storage tanks, whether empty or containing any substance; any substance the presence of which on the Project is prohibited by any federal, state or local authority; any substance that requires special handling and any other material or substance now or

in the future that (i) is defined as a "hazardous substance," "hazardous material," "hazardous waste," "toxic substance," "toxic pollutant," "contaminant," or "pollutant" by or within the meaning of any Hazardous Materials Law, or (ii) is regulated in any way by or within the meaning of any Hazardous Materials Law.

"Hazardous Materials Laws" means all federal, state, and local laws, ordinances and regulations and standards, rules, policies and other governmental requirements, administrative rulings and court judgments and decrees in effect now or in the future and including all amendments, that relate to Hazardous Materials or the protection of human health or the environment and apply to the Borrower or to the Project. Hazardous Materials Laws include, but are not limited to, the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. Section 9601, et seq., the Resource Conservation and Recovery Act of 1976, 42 U.S.C. Section 6901, et seq., the Toxic Substance Control Act, 15 U.S.C. Section 2601, et seq., the Clean Water Act, 33 U.S.C. Section 1251, et seq., and the Hazardous Materials Transportation Act, 49 U.S.C. Section 5101 et seq., and their state analogs.

"Housing Act" means the United States Housing Act of 1937, as amended, or its successor.

"Income Certification" means, initially, an Income Certification in the form attached hereto as Exhibit C or in such other form as may from time to time be provided by the City to the Borrower and, with respect to recertifications, the Annual Tenant Income Recertification attached hereto as Exhibit D or such other form as may, from time to time, be provided by the City to the Borrower.

"Inducement Date" means August 25, 2017.

"Loan" means the loan of the sale proceeds of the Note by the City to the Borrower pursuant to the Loan Agreement for the purpose of providing funds for the acquisition, construction or rehabilitation and equipping of the Project and the Companion Project.

"Loan Agreement" means the [Construction and Permanent Loan Agreement (Multifamily Housing Back to Back Loan Program)] dated as of [\_\_\_\_\_] 1, 2018 by and among the City, MUFG Union Bank, N.A. and the Borrower, as amended or supplemented from time to time.

"Low Income Tenant" means a tenant whose Adjusted Income does not exceed limits determined in a manner consistent with determinations of lower-income families under Section 8 of the Housing Act, except that the percentage of median gross income that qualifies as lower income shall be [60]% of median gross income for the Area with adjustments for family size. Except as otherwise provided herein, the occupants of a unit in the Project shall not be considered to be Low Income Tenants if all the occupants of a unit are students (as defined in Section 152(f)(2) of the Code) and any one of those students is not (1) a single parent living with his/her children; (2) a student receiving assistance under Title IV of the Social Security Act (Temporary Assistance for Needy Families); (3) a student enrolled in a job training program receiving assistance under the Job Training Partnership Act or under other similar federal, state, or local laws; (4) a student who was previously under the care and placement responsibility of a foster care program (under part B or E of Title IV of the Social Security Act); or (5) a student who is married and files a joint return. Single parents described in (1) above may not be dependents of another individual and their children may not be dependents of another individual other than their parents. The determination of a tenant's status as a Low Income Tenant shall be made by the Borrower upon

initial occupancy of a unit in the Project by such Tenant and annually thereafter and at any time the Borrower has knowledge that the number of occupants in that unit has increased, on the basis of an Income Certification executed by the tenant.

"Low Income Units" means the units in the Project required to be rented to, or held available for occupancy by, Low Income Tenants pursuant to Sections 4(a), 4(b) and 6(a) hereof.

"Net Proceeds" means the total proceeds derived from the issuance, sale and delivery of the Note, representing the total purchase price of the Note, including any premium paid as part of the purchase price of the Note, but excluding the accrued interest, if any, on the Note paid by the initial purchaser of the Note.

"Note" means the note authorized, authenticated and delivered under the Funding Loan Agreement.

"Note Documents" means the Funding Loan Agreement, the Loan Agreement, this Regulatory Agreement, the Companion Project Regulatory Agreement, the Tax Certificate and any other document now or hereafter executed by the Borrower, City, Fiscal Agent or Noteholder in connection with the Note.

"Noteholder" or "Owner" or "Holder" means the party identified as the owner of the Note on the registration books maintained by the Fiscal Agent on behalf of the Governmental Lender.

"Project" means the Project Facilities and the Project Site.

"Project Costs" means, to the extent authorized by the Code, the Regulations, the Law and the Act, any and all costs incurred by the Borrower with respect to the acquisition and rehabilitation and equipping, the credit enhancement fees, if any, attributable to the period of the rehabilitation of the Project, whether paid or incurred prior to or after the Inducement Date, including, without limitation, costs for site preparation, the planning of housing, related facilities and improvements, the acquisition of property, the removal or demolition of existing structures, the construction or rehabilitation of housing and related facilities and improvements, and all other work in connection therewith, including Qualified Project Costs, and all costs of financing, including, without limitation, the cost of consultant, accounting and legal services, other expenses necessary or incident to determining the feasibility of the Project, contractors' and developer's overhead and supervisors' fees and costs directly allocable to the Project, administrative and other expenses necessary or incident to the Project and the financing thereof (including reimbursement to any municipality, county or other entity or person for expenditures made, for the Project).

"Project Facilities" means the buildings, structures and other improvements on the Project Site to be acquired, rehabilitated, constructed, equipped or improved by the Borrower, and all fixtures and other property owned by the Borrower and located on, or used in connection with, such buildings, structures and other improvements constituting the Project. Project Facilities do not include retail sales facilities, leased office space, commercial facilities or recreational, fitness, parking or business facilities available to members of the general public.

"Project Site" means the parcel or parcels of real property having the street address of 647 South San Pedro in the City of Los Angeles, California and all rights and appurtenances thereunto appertaining, as more particularly described in Exhibit A hereto.

"Qualified Project Costs" means the Project Costs (excluding issuance costs) incurred not earlier than the date 60 days prior to the Inducement Date which either constitute land or property of a character subject to the allowance for depreciation under Section 167 of the Code, or are chargeable to a capital account with respect to the Project for federal income tax and financial accounting purposes, or would be so chargeable either with a proper election by the Borrower or but for the proper election by the Borrower to deduct those amounts; provided, however, that only such portion of the interest accrued on the Note during the rehabilitation of the Project shall constitute Qualified Project Costs as bear the same ratio to all such interest or fees, as applicable, as the Qualified Project Costs bear to all Project Costs; and provided further that interest accruing on or after the Completion Date shall not be Qualified Project Costs; and provided finally that if any portion of the Project is being rehabilitated or constructed by the Borrower or an Affiliated Party (whether as a general contractor or a subcontractor), "Qualified Project Costs" shall include only (a) the actual out-of-pocket costs incurred by the Borrower or such Affiliated Party in rehabilitating or constructing the Project (or any portion thereof), (b) any reasonable fees for supervisory services actually rendered by the Borrower or such Affiliated Party (but excluding any profit component) and (c) any overhead expenses incurred by the Borrower or such Affiliated Party which are directly attributable to the work performed on the Project, and shall not include, for example, intercompany profits resulting from members of an affiliated group (within the meaning of Section 1504 of the Code) participating in the rehabilitation of the Project or payments received by such Affiliated Party due to early completion of the Project (or any portion thereof). Qualified Project Costs do not include Costs of Issuance. Notwithstanding anything herein to the contrary, no Project Costs relating to the acquisition of the Project or any assets relating thereto (including, without limitation, rights and interests with respect to development of the Project) shall constitute "Qualified Project Costs" unless, at the time Note proceeds are expended to pay such costs, the Borrower and the seller of such assets are not "related parties" as such term is defined in Section 1.150-1(b) of the Regulations.

"Qualified Project Period" means the period beginning on the Closing Date and ending on the latest of (a) the date which is 15 years after the Closing Date, (b) the first date on which no Tax-exempt private activity bond (as that phrase is used in Section 142(d)(2) of the Code) issued with respect to the Project is outstanding or (c) the date on which any assistance provided with respect to the Project under Section 8 of the Housing Act terminates. The CDLAC Conditions apply for a period which, in some cases, exceeds the Qualified Project Period.

"Qualified Rehabilitation Expenditures" means any amount properly chargeable to the Project's capital account which is incurred no earlier than 60 days prior to the Inducement Date by the person acquiring the building or property (or additions or improvements to property) or by the seller of the property under a sales contract between the Borrower and the seller of the Project to the Borrower in connection with the rehabilitation of a building. In the case of an integrated operation contained in a building before its acquisition, such term includes rehabilitating existing equipment in such building or replacing it with equipment having substantially the same function. "Qualified Rehabilitation Expenditures" do not include any amount which is incurred after the date that is two years after the later of the date on which the building was acquired by the Borrower

or the date on which the Note was issued. "Qualified Rehabilitation Expenditures" do not include any expenditure described in Section 47(c)(2)(B) of the Code. All amounts constituting Qualified Rehabilitation Expenditures must be depreciated on a straight-line basis over 27.5 years (unless otherwise provided in the Code).

"Regulations" means the Income Tax Regulations promulgated or proposed (if deemed appropriate in the opinion of Bond Counsel) by the Department of the Treasury pursuant to the Code from time to time.

"Rehabilitation Completion Certificate" means a written certification signed by an Authorized Borrower Representative certifying among other things to the substantial completion of the Project and that the Borrower has incurred Qualified Rehabilitation Expenditures with respect to the Project in an amount equal to or greater than 15% of the portion of the cost, if any, of acquiring the Project (exclusive of any acquisition costs attributable to land) financed with the Net Proceeds of the Note and delivered to the City, the Fiscal Agent and CDLAC not more than 25 months after the Closing Date, in substantially the form of Exhibit F hereto or such other form required or otherwise provided by CDLAC from time to time.

"*Tax Certificate*" means the Tax Certificate as to Arbitrage and the Provisions of Sections 103 and 141-150 of the Internal Revenue Code of 1986 dated the Closing Date, executed and delivered by the City and the Borrower, as amended, modified, supplemented or restated from time to time.

"Tax-exempt" means, with respect to interest on any obligations of a state or local government, including the Note, that such interest is excluded from gross income for federal income tax purposes (other than interest on the Note for any period during which the Note is held by a "substantial user" of any facility financed with the proceeds of the Note or a "related person," as such terms are used in Section 147(a) of the Code); provided, however, that such interest may be includable as an item of tax preference or otherwise includable directly or indirectly for purposes of calculating other tax liabilities, including any alternative minimum tax or environmental tax, under the Code.

Unless the context clearly requires otherwise, as used in this Regulatory Agreement, words of the masculine, feminine or neuter gender shall be construed to include each other gender and words of the singular number shall be construed to include the plural number, and vice versa. This Regulatory Agreement and all the terms and provisions hereof shall be construed to effectuate the purposes set forth herein and to sustain the validity hereof.

The defined terms used in the preamble and recitals of this Regulatory Agreement have been included for convenience of reference only, and the meaning, construction and interpretation of all defined terms shall be determined by reference to this Section I, notwithstanding any contrary definition in the preamble or recitals hereof. The titles and headings of the sections of this Regulatory Agreement have been inserted for convenience of reference only, and are not to be considered a part hereof and shall not in any way modify or restrict any of the terms or provisions hereof or be considered or given any effect in construing this Regulatory Agreement or any provisions hereof or in ascertaining intent, if any question of intent shall arise.

In the event of any conflict between this Regulatory Agreement and the CDLAC Conditions, the most restrictive requirement shall govern.

- **Section 2. Acquisition, Rehabilitation and Equipping of the Project**. The Borrower hereby represents as of the date hereof, covenants and agrees with the City and the Fiscal Agent as follows:
  - (a) The Borrower has incurred, or will incur within six months after the Closing Date, a substantial binding obligation to commence the acquisition and rehabilitation of the Project, pursuant to which the Borrower is or will be obligated to expend at least 5% of the proceeds of the Loan financed from proceeds of the Note.
  - (b) The Borrower's reasonable expectations respecting the total cost of the acquisition, rehabilitation and equipping of the Project are accurately set forth in the Borrower Cost Certificate (the "Borrower Cost Certificate") submitted to the City on the Closing Date.
  - (c) The Borrower has acquired the Project Site and will, within six months following the Note Closing Date, commence the rehabilitation of the Project and will proceed with due diligence to complete the same. Notwithstanding anything herein to the contrary, no Project Costs relating to the acquisition of the Project or any assets relating thereto (including, without limitation, rights and interests with respect to development of the Project) shall constitute "Qualified Project Costs" unless, at the time Note proceeds are expended to pay such costs, the Borrower and the seller of such assets are not "related parties" as such term is defined in Section 1.150-1(b) of the Regulations. The Borrower reasonably expects to complete the acquisition and rehabilitation of the Project and to expend the full amount of the proceeds of the Loan for Project Costs prior to the date which is 24 months after the Closing Date.
  - The Borrower agrees that the full amount of each disbursement of Note (d) proceeds pursuant to the Funding Loan Agreement and the Loan Agreement will be applied to pay or to reimburse the Borrower for the payment of Project Costs as set forth in the Borrower Cost Certificate and that, after taking into account each such disbursement, (i) the aggregate disbursements of Note proceeds will have been applied to pay or to reimburse the Borrower for the payment of Qualified Project Costs in an aggregate amount equal to 97% or more of the aggregate disbursements of the Loan; provided, however, that if the Borrower provides the Fiscal Agent with an opinion of Bond Counsel to the effect that the Tax-exempt status of interest on the Note will not be adversely affected if less than the aforesaid percentage, but not less than 95%, is disbursed for such purpose, then the certificate may refer to such lesser percentage as may be specified by Bond Counsel; and (ii) less than 25% of the proceeds of the Note expended relative to the Project Site will be disbursed to pay or to reimburse the Borrower for the cost of acquiring land or rights with respect to land relative to the Project Site (exclusive of the cost of acquiring improvements on such land).
    - (e) [Reserved].

- (f) No proceeds of the Note will be used to pay or reimburse any cost (i) incurred more than 60 days prior to the Inducement Date, or (ii) incurred more than three years prior to such payment or reimbursement. Any allocation of Note proceeds to the reimbursement of previously incurred costs shall be made not later than 18 months after the later of (i) the date the original expenditure was paid or (ii) the date the Project is placed in service or abandoned. The acquisition, rehabilitation and equipping of the Project by the Borrower commenced less than 60 days prior to the Inducement Date, and as of 60 days prior to the Inducement Date (A) neither the Borrower nor any "related person" (as such phrase is used in Section 147(a)(2) of the Code) has made any expenditure in connection with the acquisition, rehabilitation or equipping of the Project, (B) no on-site work has been commenced by the Borrower or any related person in connection with the rehabilitation of the Project, and (C) no off-site fabrication of any portion of the Project has been commenced by the Borrower or any related person. The Project consists, and shall at all times consist, of property which is land or is subject to the allowance for depreciation provided in Section 167 of the Code.
- (g) The Borrower has or shall, prior to the date which is 24 months (unless extended pursuant to subsection "(j)" below) after the Note Closing Date, expend proceeds of the Note equal to not less than 15% of the amount of Note proceeds expended to acquire the Project (exclusive of any acquisition costs attributable to land) on Qualified Rehabilitation Expenditures which expenditures shall be confirmed in writing through a Rehabilitation Completion Certificate delivered to the City and the Fiscal Agent not later than 25 months (unless extended pursuant to subsection "(j)" below) after the Closing Date.
- (h) The Borrower (and any Affiliated Party) will not take or omit to take, as is applicable, any action if such action or omission would in any way cause the proceeds from the Loan to be applied in a manner contrary to the requirements of this Regulatory Agreement, nor will it take or omit to take any such action if the Borrower (or any Affiliated Party) knows that such action or omission may cause the proceeds from the sale of the Note to be applied in a manner contrary to the Funding Loan Agreement, the Loan Agreement, the Law, the Act or the Code.
- by providing a Rehabilitation Completion Certificate to CDLAC, the Fiscal Agent and the City, signed by the Authorized Borrower Representative, stating the total cost of the Project and identifying the total acquisition cost, if any (which shall specify the costs attributable to land and the costs attributable to buildings) and the total Qualified Project Costs and Qualified Rehabilitation Expenditures and further stating that (A) rehabilitation of the Project has been completed substantially in accordance with the plans, specifications and work orders therefor, and all labor, services, materials and supplies used in rehabilitation or construction have been paid for and (B) all other facilities necessary in connection with the Project have been acquired, rehabilitated, constructed and installed substantially in accordance with the plans, specifications, work write-up and work orders therefor and all costs and expenses incurred in connection therewith have been paid. Notwithstanding the foregoing, such certificate may state that it is given without prejudice to any rights of the Borrower against third parties for the payment of any amount not then due and payable which exist at the date of such certificate or which may subsequently exist. The

Rehabilitation Completion Certificate shall be delivered to the Fiscal Agent no later than the date 25 months from the Closing Date unless the Borrower delivers to the Fiscal Agent a certificate of the City consenting to an extension of such date, accompanied by an opinion of Bond Counsel to the effect that such extension will not result in interest on the Note being included in gross income for federal income tax purposes.

- (j) The Borrower agrees to spend additional moneys for payment of any costs of the Project sufficient to reduce the portion of Note proceeds (A) spent on land by the Borrower relative to the Project Site to an amount that is less than 25% of the amount of Note proceeds spent by the Borrower relative to the Project Site for all purposes and (B) spent on costs of the Project paid or incurred by or on account of the Borrower or any "related person" (as such term is used in Section 147(a)(2) of the Code) on or after the date 60 days prior to the Inducement Date and chargeable to the capital account of the Project (or so chargeable either with a proper election by the Borrower to deduct such amounts, within the meaning of Treasury Regulation Section 1.103-8(a)(1)) so that the amount of Note proceeds expended on such Qualified Project Costs are at least 97% of the amount of Note proceeds spent for all purposes related to the Project, except that, upon receipt by the Borrower, the Fiscal Agent and the City of an approving opinion of Bond Counsel, the percentage of such amounts so used may be 95%.
- (k) No Note proceeds shall be expended to acquire any structures other than buildings within the meaning of Section 147(d) of the Code.
- **Section 3. Residential Rental Property**. The Borrower hereby acknowledges and agrees that the Project is to be owned, managed and operated as a "qualified residential rental project" (within the meaning of Section 142(d) of the Code) for a term equal to the Qualified Project Period. To that end, and for the Qualified Project Period, the Borrower hereby represents, covenants, warrants and agrees as follows:
  - (a) The Project Facilities will be developed for the purpose of providing multifamily residential rental property, and the Borrower will own, manage and operate the Project Facilities as a project to provide multifamily residential rental property comprising a building or structure or several interrelated buildings or structures, together with any functionally related and subordinate facilities, and no other facilities in accordance with Section 142(d) of the Code and Section 1.103-8(b) of the Regulations, the Law and the Act, and in accordance with such requirements as may be imposed thereby on the Project from time to time. For purposes of this Section 3(a), the term "functionally related and subordinate facilities" includes facilities for use by the tenants (for example, swimming pools, other recreational facilities and parking areas) and other facilities which are reasonably required for the Project, for example, heating and cooling equipment, trash disposal equipment and units for resident managers and maintenance personnel. Substantially all of the Project will contain such units and functionally related and subordinate facilities.
  - (b) All of the dwelling units in the Project will be similarly constructed units, and each Low Income Unit in the Project will contain complete separate and distinct facilities for living, sleeping, eating, cooking and sanitation for a single person or a family,

including a sleeping area, bathing and sanitation facilities and cooking facilities equipped with a cooking range and oven, a sink and a refrigerator. Each of the Accessible Housing Units (as defined in Exhibit I hereto) shall also comply with the requirements of Exhibit I. Notwithstanding the foregoing, a unit shall not fail to be treated as a residential unit merely because such unit is a single-room occupancy unit within the meaning of Section 42(i)(3)(B)(iv) of the Code even though such housing may provide eating, cooking and sanitation facilities on a shared basis.

- (c) None of the dwelling units in the Project will at any time be utilized on a transient basis or will ever be used as a hotel, motel, dormitory, fraternity house, sorority house, rooming house, nursing home, hospital, sanitarium, rest home or trailer court or park. Notwithstanding the foregoing, single-room occupancy units provided under Section 42(i)(3)(B)(iv) of the Code shall not be considered to be utilized on a transient basis.
- (d) No part of the Project will at any time be owned by a cooperative housing corporation, nor shall the Borrower take any steps in connection with a conversion to such ownership or uses. Other than filing a condominium map and a final subdivision map on the Project and obtaining a Final Subdivision Public Report from the California Department of Real Estate, the Borrower shall not take any steps in connection with a conversion of the Project to condominium ownership during the Qualified Project Period.
- (e) All of the dwelling units (which shall not include any manager units) in the Project will be available for rental on a continuous basis to members of the general public, and the Borrower will not give preference to any particular class or group in renting the dwelling units in the Project, except to the extent of the following: (1) any dwelling units that are required to be leased or rented to Low Income Tenants, (2) the requirements of any regulatory agreement executed between the Borrower and HUD or between the Borrower and a subordinate lender (including the City), (3) the requirements of any Section 8 Housing Assistance Payments Contract with respect to the Project, (4) any preference Borrower gives to a class of persons permitted to be given preference pursuant to the Code, State law and other applicable federal law and (5) Accessible Housing Units shall be made available to persons with disabilities as provided in Exhibit I.
- (f) The Project Site consists of a parcel or parcels that are contiguous except for the interposition of a road, street or stream, and all of the Project Facilities comprise a single geographically and functionally integrated project for residential rental property, as evidenced by the ownership, management, accounting and operation of the Project.
- (g) No dwelling unit in the Project shall be occupied by the Borrower; provided, however, that if the Project contains five or more dwelling units, this subsection shall not be construed to prohibit occupancy of not more than one dwelling unit by one or more resident managers or maintenance personnel any of whom may be the Borrower.
- (h) The Project shall be maintained in conformity with the habitability and fire codes of the City of Los Angeles.

- (i) The Project shall be managed in a manner consistent with prudent property management standards and in compliance with all state and local laws, ordinances and regulations relating thereto.
- (j) Should involuntary noncompliance with the provisions of Regulations Section 1.103-8(b) be caused by fire, seizure, requisition, foreclosure, transfer of title by deed in lieu of foreclosure, change in a federal law or an action of a federal agency after the Closing Date which prevents the City from enforcing the requirements of the Regulations, or condemnation or similar event, the Borrower covenants that, within a "reasonable period" determined in accordance with the Regulations, and subject to the provisions of the Funding Loan Agreement and the Loan Agreement, it will either prepay the Loan or apply any proceeds received as a result of any of the preceding events to reconstruct the Project to meet the requirements of Section 142(d) of the Code and the Regulations.
- **Section 4. Low Income Tenants; Records and Reports**. Pursuant to the requirements of the Code and the City, the Borrower hereby represents, warrants and covenants as follows:
  - (a) The Project will be developed for the purpose of providing multifamily residential rental property, and the Borrower will own, manage and operate the Project as a project to provide multifamily residential rental property comprising a building or structure or several interrelated buildings or structures, together with any functionally related and subordinate facilities, and no other facilities in accordance with the Law and the Act, and in accordance with such requirements as may be imposed thereby on the Project from time to time. The Borrower hereby certifies that on the Closing Date not less than 50% of the dwelling units in the Project are occupied by tenants providing an Income Certification. Based on the foregoing, the beginning date of the Qualified Project Period is the Closing Date and the earliest ending date of the Qualified Project Period is 15 years after the Closing Date.
  - (b) Commencing on the first day of the Qualified Project Period, Low Income Tenants shall occupy at least [40]% of all completed and occupied units in the Project (excluding units occupied by property managers) before any additional units are occupied by persons who are not Low Income Tenants; and for the Qualified Project Period no less than [40]% of the total number of completed units of the Project (excluding units occupied by property managers) shall at all times be rented to and occupied by Low Income Tenants. For the purposes of this subsection (b), a vacant unit which was most recently occupied by a Low Income Tenant is treated as rented and occupied by a Low Income Tenant until reoccupied, other than for a temporary period of not more than 31 days, at which time the character of such unit shall be redetermined. In determining whether the requirements of this subsection (b) have been met, fractions of units shall be treated as entire units.
  - (c) No tenant qualifying as a Low Income Tenant shall be denied continued occupancy of a unit in the Project because, after admission, such tenant's Adjusted Income increases to exceed the qualifying limit for Low Income Tenants; provided, however, that should a Low Income Tenant's Adjusted Income, as of the most recent determination thereof, exceed 140% of the then applicable income limit for a Low Income Tenant of the

same family size, the next available unit of comparable or smaller size must be rented to (or held vacant and available for immediate occupancy by) a Low Income Tenant; and provided further that, until such next available unit is rented to a tenant who is not a Low Income Tenant, the former Low Income Tenant who has ceased to qualify as such shall be deemed to continue to be a Low Income Tenant for purposes of the [40]% requirement of paragraph (b) of this Section 4 (if applicable). If the Project consists of more than one building, this requirement shall apply on a building-by-building basis.

- The Borrower will obtain, complete and maintain on file Income Certifications from each Low Income Tenant, including (i) an Income Certification dated no later than the day prior to the initial occupancy of such Low Income Tenant in the Project and, in the case of tenants residing in the Project as of the date of issuance of the Note (if applicable), dated no later than the day prior to the disbursement of Note proceeds to fund acquisition and rehabilitation of the Project and (ii) thereafter, annual Income Certifications dated as of the anniversary date of each initial Income Certification. The Borrower will obtain such additional information as may be required in the future by the State of California, by the City and by Section 142(d) of the Code, as the same may be amended from time to time, or in such other form and manner as may be required by applicable rules, rulings, policies, procedures, Regulations or other official statements now or hereafter promulgated, proposed or made by the Department of the Treasury or the Internal Revenue Service with respect to obligations which are Tax-exempt under Section 142(d) of the Code. A copy of the most recent Income Certification for Low Income Tenants commencing or continuing occupation of a Low Income Unit (and not previously filed with the City) shall be attached to the Certificate of Continuing Program Compliance which is to be filed with the City no later than the fifteenth day of each month until such report indicates compliance with Section 4(b) and thereafter on the fifteenth day of each [ ] and [ until the end of the Qualified Project Period. The Borrower shall make a good-faith effort to verify that the income information provided by an applicant in an Income Certification is accurate by obtaining the acceptable forms of verification enumerated in Chapter 3 of the most current, amended edition of HUD Handbook 4350.3, or such instruction by HUD that may supersede this handbook, and any additional documentation that the City shall deem relevant, such as the two most recent years' tax returns or other forms of independent verification satisfactory to the City.
- (e) The Borrower will use its best efforts to maintain complete and accurate records pertaining to the Low Income Units, and will with reasonable notice permit any duly authorized representative of the City, the Fiscal Agent, the Department of the Treasury or the Internal Revenue Service to inspect the books and records of the Borrower pertaining to the Project during regular business hours, including those records pertaining to the occupancy of the Low Income Units.
- (f) The Borrower will prepare and submit to the City and the Fiscal Agent, no later than the fifteenth day of each month following the receipt by the Fiscal Agent of the Rehabilitation Completion Certificate to and including the month in which such report indicates that [40]% of the occupied units (excluding units occupied by property managers) are occupied by Low Income Tenants, and thereafter no later than the fifteenth day of each and [\_\_\_\_\_] until the end of the Qualified Project Period, a Certificate of

Continuing Program Compliance executed by the Borrower stating (i) the percentage of the dwelling units of the Project which were occupied or deemed occupied, pursuant to subsection (b) of this Section 4, by Low Income Tenants during such period; (ii) that either (A) no unremedied default has occurred under this Regulatory Agreement, or (B) a default has occurred, in which event the certificate shall describe the nature of the default in detail and set forth the measures being taken by the Borrower to remedy such default; and (iii) that, to the knowledge of the Borrower, no Determination of Taxability has occurred, or if a Determination of Taxability has occurred, setting forth all material facts relating thereto.

- (g) On or before each February 15 during the Qualified Project Period, the Borrower will submit to the City a draft of the completed Internal Revenue Service Form 8703 or such other annual certification required by the Code to be submitted to the Secretary of the Treasury as to whether the Project continues to meet the requirements of Section 142(d) of the Code. On or before each March 31 during the Qualified Project Period, the Borrower will submit such completed form to the Secretary of the Treasury, regardless of whether or not the City has responded to such draft.
- (h) Subject to the requirements of any Section 8 Housing Assistance Payments Contract with respect to the Project, each lease or rental agreement pertaining to a Low Income Unit shall contain a provision to the effect that the Borrower has relied on the Income Certification and supporting information supplied by the Low Income Tenant in determining qualification for occupancy of the Low Income Unit and that any material misstatement in such certification (whether or not intentional) will be cause for immediate termination of such lease or rental agreement. Each such lease or rental agreement shall also provide that the tenant's income is subject to annual certification in accordance with Section 4(c) hereof and to recertification if the number of occupants in the units changes for any reason (other than the birth of a child to an occupant of such unit) and that if upon any such certification such tenant's Adjusted Income exceeds 140% of the then applicable income limit for a Low Income Tenant of the same family size, such tenant may cease to qualify as a Low Income Tenant, and such tenant's rent is subject to increase. Notwithstanding anything in this Section 4(h) to the contrary, such tenant's rent may be increased only pursuant to Section 7(1) hereof.
- (i) Pursuant to the CDLAC Conditions attached hereto and for the entire term of the Regulatory Agreement, the Project, together with the Companion Project, shall consist of 19 units of which at least 19 qualified residential units shall be rented or held vacant for rental for persons or families whose income is at or below 50% of the area median income as shown in the chart below:

Unit Type	Units at 35% AMI	Units at 50% AMI	Total Number of Units
Studio	12	7	19
Total	12	7	19

- **Section 5. Tax-exempt Status of the Note**. The Borrower and the City make the following representations, warranties and agreements for the benefit of the holder of the Note from time to time:
  - (a) The Borrower and the City will not knowingly take or permit actions within their control, or omit to take or cause to be taken, as is appropriate, any action that would adversely affect the Tax-exempt nature of the interest on the Note and, if either should take or permit, or omit to take or cause to be taken, any such action, it will take all lawful actions necessary to rescind or correct such actions or omissions promptly upon obtaining knowledge thereof, provided that the Borrower shall not have violated these covenants if the interest on the Note becomes taxable to a person solely because such person is a "substantial user" of the Project or a "related person" within the meaning of Section 147(a) of the Code.
  - (b) The Borrower and the City will take such action or actions as may be necessary, in the written opinion of Bond Counsel filed with the City and the Fiscal Agent, with a copy to the Borrower and the Funding Lender, to comply fully with all applicable rules, rulings, policies, procedures, Regulations or other official statements promulgated, proposed or made by the Department of the Treasury or the Internal Revenue Service pertaining to obligations the interest on which is Tax-exempt under Section 142(d) of the Code.
  - (c) The Borrower and the City will file or record such documents and take such other steps as are necessary, in the written opinion of Bond Counsel filed with the City and the Fiscal Agent, with a copy to the Borrower and the Funding Lender, in order to insure that the requirements and restrictions of this Regulatory Agreement will be binding upon all owners of the Project, including, but not limited to, the execution and recordation of this Regulatory Agreement in the real property records of the County of Los Angeles.
  - (d) The Borrower will not knowingly enter into any agreements which would result in the payment of principal or interest on the Note being "federally guaranteed" within the meaning of Section 149(b) of the Code.
  - (e) Subject to Section 14 hereof, the Borrower hereby covenants to include the requirements and restrictions contained in this Regulatory Agreement in any documents transferring any interest in the Project prior to the expiration of the Qualified Project Period to another person to the end that such transferee has notice of, and is bound by, such restrictions, and to obtain the agreement from any transferee to abide by all requirements and restrictions of this Regulatory Agreement; provided, however, that so long as any former Borrower has no remaining interest in the Project, such former Borrower shall have no obligation to monitor such transferee's compliance with such restrictions, and such former Borrower shall incur liability if such transferee fails to comply with such restrictions only in proportion to its then remaining interest.
  - (f) The Borrower and any related party (as defined in Section 1.150-1(b) of the Regulations) thereto shall not acquire the Note in an amount related to the amount of the Loan.

- **Section 6.** Additional Requirements of the Act. In addition to the requirements set forth in Sections 2 through 5, and without limiting any additional requirements in Section 7, during the Qualified Project Period, the Borrower and the City hereby agree to comply with each of the requirements of the Act, and, without limiting the foregoing, the Borrower hereby specifically agrees to comply with each of the requirements set forth in this Section 6, as follows:
  - (a) As provided in Section 52097.5 of the Act, not less than [40]% of the total number of units in the Project (excluding units occupied by property managers) shall be reserved for occupancy by tenants whose adjusted gross income does not exceed [60]% of the median adjusted gross income for the Area, adjusted for family size, as determined pursuant to Section 8 of the Housing Act.
  - (b) The rents paid by the tenant for the units reserved pursuant to paragraph (a) of this Section (excluding any supplemental rental assistance from the State, the federal government, or any other public agency to those occupants or on behalf of those units) shall not exceed the amount derived by multiplying 30% times [60]% of the median gross income for the Area, adjusted for family size, as determined pursuant to Section 8 of the Housing Act.
  - (c) During the Qualified Project Period the Borrower shall file Certificates of Continuing Program Compliance in the form and at the time required by Sections 4(d) and (f) hereof that shall contain sufficient information to allow the City to file any annual report required by the Act or pursuant to California Government Code Section 8855.5, and the Borrower shall provide to the California Debt and Investment Advisory Commission any annual report information required by California Government Code Section 8855(k)(1) until the later of the date the Note is no longer outstanding or the proceeds of the Note have been fully spent.
  - (d) No portion of the Note shall be used to finance the acquisition, construction, rehabilitation, equipping, refinancing or development of commercial property for lease.
  - (e) The Borrower shall not apply selection criteria to certificate holders under Section 8 of the United States Housing Act of 1937, as amended, that are more burdensome than the criteria applied to all other prospective tenants.
  - (f) Following the expiration or termination of the Qualified Project Period with respect to the Project, except in the event of foreclosure and redemption of the Note, deed in lieu of foreclosure, eminent domain or action of a federal agency preventing enforcement, units required to be reserved for occupancy pursuant hereto shall remain available to any eligible household occupying a reserved unit at the date of expiration or termination, at a rent not greater than the amount set forth in (b) above, until the earliest of any of the following occur:
    - (i) The household's income exceeds 140% of the maximum eligible income specified herein;
    - (ii) The household voluntarily moves or is evicted for "good cause." "Good cause," for the purposes of this Section, means the nonpayment of rent or

allegation of facts necessary to prove major, or repeated minor, violations of material provisions of the occupancy agreement which detrimentally affect the health and safety of other persons or the structure, the fiscal integrity of the Project, or the purposes or special programs of the Project;

- (iii) Thirty (30) years after the date of the commencement of the Qualified Project Period relative to the Project; and
- (iv) The Borrower pays the relocation assistance and benefits to tenants as provided in subdivision (b) of Section 7264 of the California Government Code.
- (g) During the three years prior to expiration of the Qualified Project Period, the Borrower shall continue to make available to Low Income Tenants reserved units that have been vacated to the same extent that nonreserved units are made available to tenants other than Low Income Tenants.

Notwithstanding Section 1461 of the California Civil Code, the provisions hereof shall run with the land and may be enforced either in law or in equity by any resident, local agency, entity or any other person adversely affected by the Borrower's failure to comply with this Regulatory Agreement.

- **Section 7. Additional Requirements of the City**. In addition to, and not in derogation of, the requirements set forth in the preceding and following sections of this Regulatory Agreement, each of which is hereby incorporated in this Section as a specific requirement of the City, whether or not required by California or federal law, the Borrower represents, warrants, covenants and agrees as follows:
  - (a) The Borrower shall promptly provide to the City such information with respect to the Project or the Note as the City shall from time to time request. The Borrower shall provide written notice to the City of receipt of a certificate of occupancy or other official authorization to occupy the Project immediately upon receipt.
  - (b) The Low Income Units shall be of comparable quality to all other units in the Project, shall be dispersed throughout the Project, and shall offer a range of size and number of bedrooms comparable to those units which are available to other tenants; and Low Income Tenants shall have access to and enjoyment of all common areas and facilities of the Project on the same basis as tenants of other units.
  - (c) The Borrower agrees that it will not discriminate in the rental of units or in its employment practices against any employee or applicant for employment because of the applicant's race, religion, national origin, ancestry, sex, age, sexual orientation, gender identity/expression, transgender status, disability (except to give priority to persons with disabilities for the occupancy of Accessible Housing Units), marital status, domestic partner status or medical condition. All contracts entered into by the Borrower which relate to the Project shall contain a like provision. The Borrower shall comply with the provisions of Sections 10.8.2 and 10.8.4 of the Administrative Code of the City, the provisions of which are hereby incorporated by reference.

## (d) [Reserved].

- (e) For the Qualified Project Period, the Borrower will comply with the provisions of the Unruh Civil Rights Act, including, without limitation, Sections 51.2 and as applicable, 51.3 of the California Civil Code, as amended, and Sections 45.50 et seq. of the Los Angeles Municipal Code, as amended.
- (f) The lease to be utilized by the Borrower in renting any residential units in the Project to Low Income Tenants shall provide for termination of the lease and consent by such person to immediate eviction, subject to applicable provisions of California law, for any tenant who fails to qualify as a Low Income Tenant and who has made a material misrepresentation on the Income Certification as to such tenant's qualification as a Low Income Tenant. All such leases shall contain clauses, among others, wherein each individual lessee (i) certifies the accuracy of the statements made in the Income Certification and (ii) agrees that the family income, family composition and other eligibility requirements shall be deemed substantial and material obligations of the lessee's tenancy; that the lessee will comply promptly with all requests for information with respect thereto from the Borrower or the City; and that the lessee's failure to provide accurate information in the Income Certification or refusal to comply with a request for information with respect thereto shall be deemed a violation of a substantial obligation of the lessee's tenancy and shall be a default thereunder. Additionally, such lease shall contain provisions informing any tenant of the possibility of rental payment increases in accordance with the terms of this Regulatory Agreement.
- (g) All Income Certifications will be maintained on file at the Project or, with the prior written consent of the City, at the principal place of business of the Borrower or the property manager of the Project, so long as this Regulatory Agreement is in effect and for five years thereafter with respect to each Low Income Tenant who occupied a residential unit in the Project during the Qualified Project Period.
- (h) The Borrower will accept as tenants, on the same basis as all other prospective tenants, persons who are recipients of federal certificates for rent subsidies pursuant to the existing program under Section 8 of the Housing Act, or its successor. The Borrower shall not apply selection criteria to Section 8 certificate or voucher holders that are more burdensome than criteria applied to all other prospective tenants.
- (i) The Borrower shall submit to the City (i) at the times specified in Sections 4(d) and (f) herein, a Certificate of Continuing Program Compliance, which shall include the information called for therein, including occupancy records for all units in the Project, and (ii) within 15 days after receipt of a written request, any other information or completed forms requested by the City, in each case in order to comply with reporting requirements of the Internal Revenue Service or the State of California, including, without limitation, information necessary for the City to file any periodic report, or any other information concerning the Project as the City may reasonably request.
- (j) All workers performing construction or rehabilitation work for the Project employed by the Borrower or by any contractor or subcontractor shall be compensated in

an amount no less than the greatest of (i) the general prevailing rate of per diem wages ("Prevailing Wages") as determined pursuant to Labor Code Sections 1770-1781 and implementing regulations of the Department of Industrial Relations, (ii) the general prevailing rate of per diem wages as determined by the U.S. Labor Department pursuant to the Davis-Bacon Act under 40 U.S.C.S. 3141-3148 and implementing regulations ("Davis-Bacon Wages"), if applicable; and (iii) the "Living Wage" as determined by the policies and procedures of the City of Los Angeles. The Borrower shall comply with all reporting and recordkeeping requirements of the City's prevailing wage policy. The Borrower shall, and shall cause the contractors and subcontractors to, submit data and documents related to Prevailing Wages or Davis-Bacon Wages, if applicable, using the LCP Tracker or comparable HCIDLA-approved program. The fee for the LCP Tracker, or comparable HCIDLA-approved program, will be in the amount equal to 0.03% of the total rehabilitation and construction cost, which fee shall be paid in full to the City within 30 days of execution of this Agreement.

- (k) The City may, at its option and at its expense, at any time appoint an administrator to administer this Regulatory Agreement and to monitor performance by the Borrower of the terms, provisions and requirements hereof. Following any such appointment, the Borrower shall comply with any request by the City to deliver to such administrator, in addition to or instead of the City, any reports, notices or other documents required to be delivered pursuant hereto, and upon reasonable notice to the Borrower to make the Project and the books and records with respect thereto available for inspection during regular business hours by such administrator as an agent of the City.
- (l) If upon the annual certification or recertification required in Section 4(d) a tenant's Adjusted Income exceeds 140% of the then applicable income limit for a Low Income Tenant of the same family size, all rental limits herein previously applicable to the unit occupied for such tenant shall continue to apply until the next available unit is rented to a tenant who is a Low Income Tenant.
- (m) The Borrower shall give written notice to Low Income Tenants at the following four points in time:
  - (i) Upon initial move-in/lease execution, the Borrower shall give written notice to all tenants of Low Income Units, of the duration of the rent restrictions under this Regulatory Agreement. The Borrower must maintain, in its files, a copy of each notice containing each tenant's signed acknowledgment of the notice required hereunder. The notice shall, at the least, contain language that the rent restrictions under this Regulatory Agreement shall be for a term equal to the later of the expiration of: (A) the Qualified Project Period; or (B) the CDLAC Conditions. Upon termination of the rent restriction period under this Regulatory Agreement, rents may be set at market rates unless otherwise restricted by some other legal, regulatory, or contractual requirement.
  - (ii) Twelve months prior to the termination of the rent restriction period under this Regulatory Agreement, the Borrower must give written notice to its tenants of the termination of the restrictions on the Low Income Units before their

rents may be raised to market rent levels. The Borrower must also give written notice, pursuant to California Government Code Section 65863.10, to the Mayor of the City of Los Angeles, the Housing Authority of the City of Los Angeles, the California Department of Housing and Community Development and the Los Angeles Housing and Community Investment Department.

- (iii) Six months prior to the termination of the rent restriction period under this Regulatory Agreement, the Borrower must give written notice to its tenants of the termination of the restrictions on the Low Income Units before their rents may be raised to market rent levels. The Borrower must also give written notice, pursuant to California Government Code Section 65863.10, to the Mayor of the City of Los Angeles, the Housing Authority of the City of Los Angeles, the California Department of Housing and Community Development and the Los Angeles Housing and Community Investment Department.
- (iv) Ninety days prior to the termination of the rent restriction period under this Regulatory Agreement, the Borrower must again give written notice to its tenants of the termination of the restrictions on the Low Income Units before their rents may be raised to market rent levels.

Unless the Borrower meets the requirements of California Government Code 65863.13, pursuant to California Government Code 65863.11, prior or concurrent with the 12-month notice referenced above in (ii), the Borrower must provide notice of the opportunity to offer to purchase the assisted housing development to all qualified entities on the list maintained by the California Department of Housing and Community Development as well as to those qualified entities that contact the Borrower directly. The notice shall conform to the requirements of California Government Code 65863.11(h) and shall be sent to the entities by registered or certified mail, return receipt requested. The Borrower shall also post a copy of the notice in a conspicuous place in the common area of the Project.

The Borrower shall, on the Note Closing Date, pay to the City its initial fee and thereafter pay to the City its ongoing fees with respect to the issuance of the Note as follows. The Borrower shall pay the City an initial fee immediately upon issuance of the Note equal to \$[80,019.54] (.25% of the aggregate maximum principal amount of the Note issuable under the Funding Loan Agreement (\$[32,007,814])). In addition, the Borrower shall, as compensation for the City's monitoring of the provisions of this Regulatory Agreement, pay to the City, semiannually in arrears, prorated for the \_] commencing initial payment, on the first day of each [\_\_\_ ] and [ 1, 201[9] for the period from the date of issuance of the Note through the later of: (i) the end of the Qualified Project Period; or (ii) the termination of the CDLAC Conditions, prorated for the initial and any subsequent partial period, a semiannual amount equal to the greater of: (A) during the period from the Closing Date to the Conversion Date, \$1,250 or one-half of 0.125% of the maximum principal amount of Note issuable under the Funding Loan Agreement (\$[32,007,814]); and (B) from and after the Conversion Date \$1,250 or one half of 0.125% of the outstanding principal amount of the Note Outstanding under the Funding Loan Agreement immediately after the Conversion Date; or in either case, such lesser amount as shall be necessary in the opinion of Bond Counsel to preserve the exemption of interest on the Note from gross income for federal income tax purposes. Throughout the term of this Agreement, the Fiscal Agent, or the City, as applicable, shall provide an invoice to the Borrower at least 30 days prior to the due date of each such payment (and, if applicable, a copy of which shall be provided to the City) and shall collect such payments from the Borrower and immediately remit such funds to the City. In the event of any prepayment of the Note in whole, prior to the later of: (i) the end of the Qualified Project Period; or (ii) the termination date of the CDLAC Conditions, the Borrower, at its election, shall either: (A) pay to the City, on or before such payment, an amount equal to the present value of the remaining City fees payable hereunder, as calculated by the City, using a discount rate equal to the yield on the date of prepayment on the United States treasury security maturing on the date nearest the later of: (1) the end of the Qualified Project Period or (2) the termination date of the CDLAC Conditions, or such lesser amount as shall be necessary in the opinion of Bond Counsel to preserve the exemption of interest on the Note from gross income for federal income tax purposes; or (B) enter into a fiscal agent agreement with a corporate fiscal agent acceptable to the City requiring the fiscal agent appointed thereunder to bill and collect from the Borrower and to pay to the City on an annual basis, in arrears on or before each [ 11, the annual fee described above. The Borrower shall bear the cost of such fiscal agent through the term of this Regulatory Agreement. The Borrower shall not be required to pay the fee described in the preceding sentences if the Note is prepaid in whole under circumstances which permit termination of this Regulatory Agreement pursuant to Section 14 hereof. Provisions identical to the foregoing appear in Section 7(n) of the Companion Project Regulatory Agreement. Such fees are calculated and due in connection with the Project and Companion Projects as a whole. To the extent that the foregoing fees are paid in full pursuant to this Regulatory Agreement, no additional fees shall be due under the Companion Project Regulatory Agreement.

The Borrower shall pay to the City a processing fee equal to: (i) prior to the Conversion Date the greater of \$5,000 or 0.125% of the maximum principal amount of the Note issuable under the Funding Loan Agreement; and (ii) following the Conversion Date the greater of \$5,000 or 0.125% of the principal amount of the Note outstanding under the Funding Loan Agreement immediately after the Conversion Date, plus any expenses incurred by the City, including, without limitation, Bond Counsel, City attorney and financial advisor fees, as a condition to the consideration and receipt of any consent, approval, amendment, transfer or waiver requested of the City with respect to the Project, the Project Site or the Note. The City shall provide an invoice directly to the Borrower for such amounts. Provisions identical to the foregoing appear in Section 7(o) of the Companion Project Regulatory Agreement. Such fees are calculated and due in connection with the Project and Companion Projects as a whole. To the extent that the foregoing fees are paid in full pursuant to this Regulatory Agreement, no additional fees shall be due under the Companion Project Regulatory Agreement.

- (p) The Borrower shall pay the City its then-current fees in connection with any consent, approval, transfer, amendment or waiver requested of the City, together with any expenses incurred by the City in connection therewith.
- (q) The Fiscal Agent shall report to the City in writing semiannually, within 10 days of each June 30 and December 31, the principal amount of the Note outstanding as of such June 30 or December 31, as appropriate.
- (r) The Borrower shall promptly provide the City such information with respect to the Project or the Note as the City shall from time to time request.
- (s) The Borrower shall include the City as an additional insured on all liability insurance policies relating to the Borrower or the Project.
- (t) The Borrower shall not rent any Low Income Unit to: (i) any individual who (A) holds an ownership interest in the Borrower or any general partner or member (or owner of such general partner or member) of the Borrower, (B) is an officer, board member, employee or agent of, or consultant to, the Borrower or any general partner or member thereof or owner of such general partner or member or (C) is a developer of the Project (collectively, an "Owner/Developer"); (ii) any Immediate Family Member of an Owner/Developer ("Immediate Family Members" consists of: (A) spouses, (B) children, (C) parents and grandparents, (D) siblings, (E) in-laws, including brothers/sisters-in-law and mothers/fathers-in-law and sons/daughters-in-law or (F) significant others or domestic partners); or (iii) any elected official or his or her spouse/partner who participated in the deliberative process, vote or consideration of legislative action regarding the issuance of the Note or other loan in support of the Project, unless such person otherwise qualifies for tenancy under this Agreement and such tenancy is approved in writing by the City.
- (u) The Borrower shall include a certification in each tenant application that the applicant is not an Owner/Developer, an elected official who participated in the issuance of the Note or an Immediate Family Member thereof. The Borrower recognizes and agrees that the penalty for violation of the above covenant shall be a fine of \$5,000 per violation/per unit.
- (v) Neither the Borrower nor any general partner thereof shall issue any publicity release or other communication to any print, broadcast or online media, post any sign or in any other way identify the City as the source of the financing provided for the Project, without the prior written approval of the City (provided that nothing herein shall prevent the Borrower or any general partner thereof from identifying the City as the source of such financing to the extent that the Borrower or any general partner thereof is required to do so by disclosure requirements applicable to publicly held companies).

Any of the foregoing requirements of the City may be expressly waived by the City in writing in the City's sole discretion, but (i) no waiver by the City of any requirement of this Section 7 shall, or shall be deemed to, extend to or affect any other provision of this Regulatory

Agreement, including particularly but without limitation the provisions of Sections 2 through 6 hereof, except to the extent the City has received an opinion of Bond Counsel that any such provision is not required by the Act or the Law and may be waived without adversely affecting the exclusion from gross income of interest on the Note for federal income tax purposes; and (ii) any requirement of this Section 7 shall be void and of no force and effect if the City and the Borrower receive a written opinion of Bond Counsel to the effect that compliance with any such requirement would cause interest on the Note to become includable in gross income for federal income tax purposes, if such opinion is accompanied by a copy of a ruling from the Internal Revenue Service to the same effect or to the effect that compliance with such requirement would be in conflict with the Act or the Law.

**Section 8. Modification of Covenants**. The Borrower, the Fiscal Agent and the City hereby agree as follows:

- (a) To the extent any amendments to the Law, the Act, the Regulations or the Code shall, in the written opinion of Bond Counsel filed with the City, the Fiscal Agent and the Borrower (with a copy to the Funding Lender), impose requirements upon the ownership or operation of the Project more restrictive than those imposed by this Regulatory Agreement in order to maintain the Tax-exempt status of interest on the Note, this Regulatory Agreement shall be deemed to be automatically amended, without the consent or approval of any other person, to impose such additional or more restrictive requirements. The parties hereto hereby agree to execute such amendment hereto as shall be necessary to document such automatic amendment hereof.
- (b) To the extent that the Law, the Act, the Regulations or the Code, or any amendments thereto, shall, in the written opinion of Bond Counsel filed with the City, the Fiscal Agent and the Borrower (with a copy to the Funding Lender), impose requirements upon the ownership or operation of the Project less restrictive than imposed by this Regulatory Agreement, this Regulatory Agreement may be amended or modified to provide such less restrictive requirements but only by written amendment signed by the City, the Fiscal Agent and the Borrower and approved by the written opinion of Bond Counsel to the effect that such amendment is permitted by the Law and the Act and will not affect the Tax-exempt status of interest on the Note. The City shall be under no obligation to agree to any such amendment, it being understood that each of the requirements of this Regulatory Agreement is a specific requirement of the City, whether or not required by California or federal law.
- (c) The Borrower, the City and, if applicable, the Fiscal Agent shall execute, deliver and, if applicable, file or record any and all documents and instruments necessary to effectuate the intent of this Section 8, and the City hereby appoints the Fiscal Agent as its true and lawful attorney-in-fact to execute, deliver and, if applicable, file or record on behalf of the City, as is applicable, any such document or instrument (in such form as may be approved in writing by Bond Counsel) if the City defaults in the performance of its obligations under this subsection (c); provided, however, that unless directed in writing by the City, the Fiscal Agent shall take no action under this subsection (c) without first notifying the City and without first providing the City an opportunity to comply with the

requirements of this Section 8. Nothing in this Section 8(c) shall be construed to allow the Fiscal Agent to execute an amendment to this Regulatory Agreement on behalf of the City.

Section 9. Indemnification. The Borrower shall defend, indemnify and hold harmless the City and the Fiscal Agent and the respective officers, members, supervisors, directors, officials and employees, counsel, attorneys and agents, past, present and future of each of them (collectively, the "Indemnified Parties"), against all loss, costs, damages, expenses, suits, judgments, actions and liabilities of whatever nature (including, without limitation, reasonable attorneys' fees, litigation and court costs, amounts paid in settlement, and amounts paid to discharge judgments) directly or indirectly resulting from or arising out of or related to (a) the design, construction, rehabilitation, installation, operation, use, occupancy, maintenance, financing or ownership of the Project (including compliance with laws, ordinances and rules and regulations of public authorities relating thereto), (b) any written statements or representations with respect to the Borrower, the Project or the Note made or given to the City or the Fiscal Agent, or any underwriters or purchaser of the Note, or any tenants or applicants for tenancy in the Project or any other person, by the Borrower, or any Authorized Borrower Representative, including, but not limited to, statements or representations of facts, financial information or limited partnership affairs, (c) the Note or the Tax-exempt status of interest on the Note, (d) the failure or alleged failure of any person or entity (including the Borrower or its contractor or subcontractors) to pay the general prevailing rate of per diem wages as determined pursuant to Labor Code Sections 1770-1781 and implementing regulations of the Department of Industrial Relations in connection with the rehabilitation or construction of the improvements or any other work undertaken or in connection with the Project, or (e) any actual or alleged violation of any Hazardous Materials Law or with respect to the presence of Hazardous Materials on or under the Project or in any of improvements or on or under any property of the Borrower that is adjacent to the Project (whether before or after the date of this Agreement and whether or not the Borrower knew of the same); provided, however, that this provision shall not require the Borrower to indemnify the Indemnified Parties from any claims, costs, fees, expenses or liabilities arising from its active negligence or willful misconduct or, in the case of the Fiscal Agent, its negligence, fraud or willful misconduct. The Borrower also shall pay and discharge and shall indemnify and hold harmless the City and the Fiscal Agent from (i) any lien or charge upon payments by the Borrower to the City and the Fiscal Agent hereunder or under the Note Documents and (ii) any taxes (including, without limitation, all ad valorem taxes and sales taxes), assessments, impositions and other charges in respect of any portion of the Project. If any such claim is asserted, or any such lien or charge upon payments, or any such taxes, assessments, impositions or other charges are sought to be imposed, the City or the Fiscal Agent shall give prompt notice to the Borrower and the Borrower shall, as provided in the following paragraph, have the right to assume the defense thereof, with full power to litigate, compromise or settle the same in its sole discretion, provided that the City and the Fiscal Agent shall have the right to review and approve or disapprove any such compromise or settlement. In addition thereto, the Borrower will pay upon demand all of the reasonable fees and expenses paid or incurred by the Fiscal Agent and/or the City in enforcing the provisions hereof against the Borrower. The Borrower shall also pay the City its standard fees and reimburse the City for its expenses in connection with any consent, approval, amendment, waiver or other action taken at the request or for the benefit of the Borrower in connection with this Regulatory Agreement, the Note or any other document or agreement relating thereto. In the event of any audit or inquiry regarding the Note or the Project from any governmental entity, the Borrower shall, at the election of the City, be responsible for responding to and resolving such audit or inquiry at the expense of the Borrower.

Promptly after receipt by any party entitled to indemnification under this Section 9 of notice of the commencement of any suit, action or proceeding, such Indemnified Party shall, if a claim in respect thereof is to be made against the indemnifying party under this Section 9, notify the indemnifying party in writing of the commencement thereof; but the omission so to notify the indemnifying party shall not relieve the indemnifying party from any liability which it may have to any Indemnified Party otherwise than under this Section 9 or from any liability under this Section 9 unless the failure to provide notice prejudices the defense of such suit, action or proceeding. In case any such action is brought against any Indemnified Party, and it notifies the indemnifying party, the indemnifying party shall be entitled to participate in, and to the extent that it may elect by written notice delivered to the Indemnified Party within five Business Days after receiving the aforesaid notice from such Indemnified Party (but shall not be required) to assume, the defense thereof, with counsel reasonably satisfactory to such Indemnified Party; provided, however, if the defendants in any such action include both the Indemnified Party and the indemnifying party and the Indemnified Party shall have reasonably concluded that there are legal defenses available to it and/or other Indemnified Parties which are different from or additional to those available to the indemnifying party, the Indemnified Party or parties shall have the right to select separate counsel to assert such legal defenses and otherwise to participate in the defense of such action on behalf of such Indemnified Party or parties. Upon the indemnifying party's receipt of notice from the Indemnified Party of such Indemnified Party's election so to assume the defense of such action and selection by the Indemnified Party of counsel, the indemnifying party shall not be liable to such Indemnified Party under this Section 9 for any attorneys' fees or expenses subsequently incurred by such Indemnified Party for the engagement of separate counsel in connection with defense thereof unless (i) the Indemnified Party shall have employed separate counsel in connection with the assertion of legal defenses in accordance with the proviso to the next preceding sentence, (ii) the indemnifying party shall not have employed counsel reasonably satisfactory to the Indemnified Party to represent the Indemnified Party or shall not have employed such counsel within a reasonable time after notice of commencement of the action or (iii) the indemnifying party has authorized the employment of separate counsel to represent the Indemnified Party at the expense of the indemnifying party. Notwithstanding the foregoing, the Fiscal Agent shall not be indemnified for income tax, franchise tax or similar tax liability relating to the Fiscal Agent's own income and operations.

**Section 10. Consideration**. The City has issued the Note to provide funds to finance the acquisition, rehabilitation and equipping of the Project, all for the purpose, among others, of inducing the Borrower to acquire, rehabilitate, construct and equip the Project. In consideration of the issuance of the Note by the City, the Borrower has entered into this Regulatory Agreement and has agreed to restrict the uses to which this Project can be put on the terms and conditions set forth herein.

**Section 11. Reliance**. The City and the Borrower hereby recognize and agree that the representations and covenants set forth herein may be relied upon by all persons interested in the legality and validity of the Note and in the exemption from federal income taxation and California personal income taxation of the interest on the Note. In performing their duties and obligations hereunder, the City and the Fiscal Agent may rely upon statements and certificates of the Low

Income Tenants and upon audits of the books and records of the Borrower pertaining to the Project. In addition, the City and the Fiscal Agent may consult with counsel, and the written opinion of such counsel shall be full and complete authorization and protection in respect of any action taken or suffered by the City or the Fiscal Agent hereunder in good faith and in conformity with such opinion. In determining whether any default or lack of compliance by the Borrower exists under this Regulatory Agreement, the Fiscal Agent may, but shall not be required to, conduct any investigation into or review of the operations or records of the Borrower and may rely solely on any written notice or certificate delivered to the Fiscal Agent by the Borrower or the City with respect to the occurrence or absence of a default unless it knows that the notice or certificate is erroneous or misleading.

**Section 12. Project in the City**. The Borrower hereby represents and warrants that the Project is located entirely within the City.

Section 13. Sale or Transfer of the Project; Equity Interests. The Borrower hereby covenants and agrees not to voluntarily (which term shall not be interpreted to include a foreclosure of any security for the Loan, the granting by the Borrower of a deed in lieu of foreclosure, or any other comparable conversion of the Loan) sell, transfer or otherwise dispose of the Project, or any portion thereof (other than for individual tenant use as contemplated hereunder), equity interests in the Borrower aggregating more than 50% of the equity interest in the Borrower, or any general partner interests in the Borrower, without obtaining the prior written consent of the City, which consent shall not be unreasonably withheld by the City and shall be given by the City if (a) the Borrower is not in default hereunder or under the Loan Agreement; (b) the purchaser or assignee is not in default under any obligations it may have to the City and is not the subject of any legal or enforcement actions by the City, and the purchaser or assignee certifies that the continued operation of the Project will comply with the provisions of this Regulatory Agreement; (c) evidence reasonably satisfactory to the City is presented to establish that the purchaser or assignee is willing to comply and capable of complying with the terms and conditions of this Regulatory Agreement; (d) either (i) evidence satisfactory to the City is presented to establish that the purchaser or assignee has at least three years' experience in the ownership, operation and management of rental housing projects, without any record of material violations of discrimination restrictions or other state or federal laws or regulations applicable to such projects, (ii) the purchaser or assignee agrees to retain a property management firm which the City determines has the experience and record described in subclause (i) above or (iii) the City determines that it has no reason to believe that the purchaser or assignee is incapable, financially or otherwise, of complying with, or may be unwilling to comply with, the terms of all agreements binding on such purchaser or assignee and relating to the Project; (e) the City and the Fiscal Agent shall have received (i) with respect to any transfer of the Project, reasonable evidence satisfactory to the City that the Borrower's purchaser or transferee has assumed in writing and in full the Borrower's duties and obligations under this Regulatory Agreement and the Loan Agreement, (ii) with respect to any transfer of the Project to a new Borrower, an opinion of counsel to the transferee that the transferee has duly assumed the obligations of the Borrower under this Regulatory Agreement and that such obligations and this Regulatory Agreement are binding on the transferee, (iii) unless waived by the City, an opinion of Bond Counsel that such transfer will not adversely affect the Tax-exempt nature of the interest on the Note, (iv) from the Borrower, a Certificate of Continuing Program Compliance (and a "bring-down" certificate, if necessary) current as of the date of transfer and (v) evidence satisfactory to the City that the purchaser or assignee does not have pending against

it, nor does it have a history of, building or fire code violations as identified by the City, the State of California or federal regulatory agencies; (f) the purchaser or assignee complies with the provisions of the Los Angeles Administrative Code Section 10.8.4 Affirmative Action Program Provisions; (g) the Borrower or transferee pays all costs of the transfer of title, including, but not limited to, the cost of meeting the conditions specified in this Section 13; and (h) such other conditions are met as the City may reasonably impose to assure compliance by the Project with the requirements of this Regulatory Agreement. It is hereby expressly stipulated and agreed that, except for any such sale, transfer or disposition agreed to by the City in a separate writing, any sale, transfer or other disposition of the Project in violation of this Section 13 shall be null, void and without effect, shall cause a reversion of title to the Borrower, and shall be ineffective to relieve the Borrower of its obligations under this Regulatory Agreement. Upon any sale or other transfer which complies with this Regulatory Agreement, the Borrower shall be fully released from its obligations hereunder, but only to the extent such obligations have been assumed by the transferee of the Project, without the necessity of further documentation. Any transfer of the Project to any entity, whether or not affiliated with the Borrower, shall be subject to the provisions of this Section 13.

Nothing contained in this Section 13 shall affect any provision of the Deed of Trust, or any of the other Loan Documents to which the Borrower is a party, which requires the Borrower to obtain the consent of the Funding Lender as a precondition to sale, transfer or other disposition of, or any direct or indirect interest in, the Project or of any direct or indirect interest in the Borrower or which otherwise gives the Funding Lender the right to accelerate the maturity of the Loan or any obligations of the Borrower under the Loan Documents, or to take some other similar action with respect to the Loan or any obligations of the Borrower under the Loan Documents, upon the sale, transfer or other disposition of the Project or any such other interest. Notwithstanding anything contained in this Section 13 to the contrary, neither the consent of the City nor the delivery of items (a) through (h) of the preceding paragraph shall be required in the case of, and nothing in this Section 13 shall otherwise affect the right of the Funding Lender, the Fiscal Agent or a designee or third-party purchaser to effect, a foreclosure or deed in lieu of foreclosure, whereby the Funding Lender, the Fiscal Agent or a designee or third-party purchaser forecloses on the Project or accepts a deed in lieu of a foreclosure or effects a comparable conversion of the Loan or the Loan Documents. Consent of the City and delivery of items (a) through (h) of the preceding paragraph shall be required for any transfer of the Project subsequent to the purchase at a foreclosure or transfer pursuant to deed in lieu of foreclosure as described in the preceding sentence.

Notwithstanding the foregoing, if the Fiscal Agent acquires title to the Project by foreclosure or deed in lieu of foreclosure, no consent of the City shall be required to such transfer under this Regulatory Agreement and no other conditions shall be required to be satisfied. However, if the Fiscal Agent acquires title to the Project by foreclosure or deed in lieu of foreclosure and this Regulatory Agreement has not been terminated pursuant to Section 14 below, consent of the City and delivery of items (a) through (h) above shall be required for any transfer of the Project subsequent to the Fiscal Agent's acquisition of the Project by foreclosure or deed in lieu of foreclosure.

Notwithstanding anything to the contrary contained herein, the interest of the Borrower's limited partners shall, with prior written notice to the City, be transferable under this Regulatory

Agreement to any affiliate of the limited partners of the Borrower, without the consent of the City and/or Fiscal Agent but with prior written notice thereto.

The Borrower acknowledges and recognizes that in addition to the above requirements, the consent of CDLAC in the manner and to the extent as may at the time be required by CDLAC, among other parties, may be required in connection with any transfer of the Project.

**Section 14. Term**. This Regulatory Agreement and all and each of the provisions hereof shall become effective upon its execution and delivery and shall remain in full force and effect for the periods provided herein and, except as otherwise provided in this Section 14, shall terminate in its entirety at the end of the Qualified Project Period (or such later date provided in Section 33 hereof pursuant to the CDLAC Resolution, which imposes restrictions for a term of at least 55 years), it being expressly agreed and understood that the provisions hereof are intended to survive the retirement of the Note, discharge of the Loan and termination of the Funding Loan Agreement and the Loan Agreement.

Notwithstanding the foregoing, the provisions of Section 9 hereof shall, in the case of the Fiscal Agent, survive the term of this Regulatory Agreement or the replacement of the Fiscal Agent, but only as to claims arising from events occurring during the term of this Regulatory Agreement or the Fiscal Agent's tenure as Fiscal Agent under the Funding Loan Agreement, and shall, in the case of the City, survive the term of this Regulatory Agreement, but only as to claims arising from events occurring during the term of this Regulatory Agreement.

The terms of this Regulatory Agreement to the contrary notwithstanding, this Regulatory Agreement and all the requirements set forth herein (except Section 9 as aforesaid) shall terminate and be of no further force and effect in the event of (a) involuntary noncompliance with the provisions of this Regulatory Agreement caused by fire, seizure, requisition, change in a federal law or an action of a federal agency after the Closing Date which prevents the City or the Fiscal Agent from enforcing the provisions hereof, or (b) condemnation, foreclosure, delivery of a deed in lieu of foreclosure or a similar event, but only if, within a reasonable period thereafter, either the portion of the Note attributable to the affected portion of the Project is retired or amounts received as a consequence of such event are used to provide a project which meets the requirements of the Code set forth in Sections 2 through 6 of this Regulatory Agreement and provided that, in either case, an opinion of Bond Counsel (unless waived by the City) is delivered to the Fiscal Agent to the effect that the exclusion from gross income for federal income tax purposes of interest on the Note will not be adversely affected thereby. The provisions of the preceding sentence shall cease to apply and the requirements referred to therein shall be reinstated if, at any time during the Qualified Project Period after the termination of such requirements as a result of involuntary noncompliance due to foreclosure, transfer of title by deed in lieu of foreclosure or similar event, the Borrower or any "related party" (within the meaning of Section 1.150-1(b) of the Regulations) or "related person" (defined in Section 1.103-10(e) of the Regulations) obtains an ownership interest in the Project for tax purposes. The Borrower hereby agrees that, following any foreclosure, transfer of title by deed in lieu of foreclosure or similar event, neither the Borrower nor any related party as described above will obtain an ownership interest in the Project for tax purposes.

Upon the termination of this Regulatory Agreement, the parties hereto agree to execute, deliver and record appropriate instruments of release and discharge of the terms hereof; provided, however, that the execution and delivery of such instruments shall not be necessary or a prerequisite to the termination of this Regulatory Agreement in accordance with its terms.

Section 15. Covenants To Run With the Land. The Borrower hereby subjects the Project (including the Project Site) to the covenants, reservations and restrictions set forth in this Regulatory Agreement. The City and the Borrower hereby declare their express intent that the covenants, reservations and restrictions set forth herein shall be deemed covenants running with the land and shall pass to and be binding upon the Borrower's successors in title to the Project; provided, however, that on the termination of this Regulatory Agreement said covenants, reservations and restrictions shall expire. The City and, if necessary, the Fiscal Agent agree to execute a quitclaim deed or other documents required to remove this Regulatory Agreement from title after the covenants, agreements and restrictions herein have expired. Each and every contract, deed or other instrument hereafter executed covering or conveying the Project or any portion thereof shall conclusively be held to have been executed, delivered and accepted subject to such covenants, reservations and restrictions, regardless of whether such covenants, reservations and restrictions are set forth in such contract, deed or other instruments.

No breach of any of the provisions of this Regulatory Agreement shall impair, defeat or render invalid the lien of any security instrument, deed of trust or like encumbrance made in good faith and for value encumbering the Project or any portion thereof.

Section 16. Burden and Benefit. The City and the Borrower hereby declare their understanding and intent that the burden of the covenants set forth herein touch and concern the land in that the Borrower's legal interest in the Project is rendered less valuable thereby. The City and the Borrower hereby further declare their understanding and intent that the benefit of such covenants touch and concern the land by enhancing and increasing the enjoyment and use of the Project by Low Income Tenants, the intended beneficiaries of such covenants, reservations and restrictions, and by furthering the public purposes for which the Note was issued. Notwithstanding the foregoing or any other provision of this Regulatory Agreement, no person, other than the parties hereto, shall have any rights of enforcement of this Regulatory Agreement.

**Section 17. Uniformity; Common Plan**. The covenants, reservations and restrictions hereof shall apply uniformly to the entire Project in order to establish and carry out a common plan for the use, development and improvement of the Project Site.

Section 18. Default; Enforcement. If the Borrower defaults in the performance or observance of any covenant, agreement or obligation of the Borrower set forth in this Regulatory Agreement, and if such default remains uncured for a period of 60 days after notice thereof shall have been given by the City to the Borrower, then the City shall declare an "Event of Default" to have occurred hereunder; provided, however, that if the default stated in the notice is of such a nature that it cannot be corrected within 60 days, such default shall not constitute an Event of Default hereunder so long as (i) the Borrower institutes corrective action within said 60 days and diligently pursues such action until the default is corrected and (ii) in the opinion of Bond Counsel, the failure to cure said default within 60 days will not adversely affect the Tax-exempt status of interest on the Note. The Fiscal Agent hereby consents to any correction of the default by the City

on behalf of the Borrower. The City hereby consents to any correction of a default on the part of the Borrower hereunder made by the Borrower's limited partners on behalf of the Borrower within the time periods provided in this Section. Copies of any notices sent to the Borrower hereunder shall simultaneously be sent to the Borrower's limited partners at the addresses set forth in Section 23.

Following the declaration of an Event of Default hereunder, the Fiscal Agent, as directed by the City and subject to the provisions of the Funding Loan Agreement relative to the Fiscal Agent's duty to exercise remedies generally, or the City, may, at its option, take any one or more of the following steps:

- (a) by mandamus or other suit, action or proceeding at law or in equity, including injunctive relief, require the Borrower to perform its obligations and covenants hereunder or enjoin any acts or things which may be unlawful or in violation of the rights of the City or the Fiscal Agent hereunder;
- (b) have access to and inspect, examine and make copies of all or any portion of the books and records of the Borrower pertaining to the Project; and
- (c) take such other action at law or in equity as may appear necessary or desirable to enforce the obligations, covenants and agreements of the Borrower hereunder.

During the Qualified Project Period, the Borrower hereby grants to the City the option, upon either (a) the expiration of 60 days after the giving of the notice to the Borrower referred to in the first paragraph of this Section 18 of the Borrower's default under this Regulatory Agreement or (b) the vacancy of a Low Income Unit for more than six months (other than a unit while undergoing rehabilitation) and the submission by the City to the Borrower during such six-month or longer period of at least five proposed tenants which meet the qualifications of Low Income Tenants and the qualifications of a reasonable landlord, to lease up to [40]% of the units in the Project for a rental of \$1.00 per unit per year for the sole purpose of subleasing such units to Low Income Tenants for a period of not less than six months, but only to the extent necessary to comply with the provisions of Sections 2 through 7 of this Regulatory Agreement and to insure full occupancy of the Low Income Units. The option granted in the preceding sentence shall be effective only if the Borrower or the Fiscal Agent has not instituted corrective action before the end of such 60-day period referenced in (a) above, or the Borrower has not rented the unit during the six-month or longer period referenced in (b) above, to a qualified Low Income Tenant. The option and any leases to the City under this provision shall terminate with respect to each default upon the achievement, by the Borrower, the Fiscal Agent or the City, of compliance with the requirements of Sections 2 through 7 hereof, and any subleases entered into pursuant to the City's option shall be deemed to be leases from the Borrower. The City shall make diligent effort, but shall not be required, to rent Low Income Units to Low Income Tenants at the highest rents practicable, subject to the limits of Sections 5, 6 and 7 hereof. Any rental paid under any such sublease shall be paid to the Borrower after the City has been reimbursed for any reasonable expenses incurred in connection with such sublease, provided that if the Borrower is in default under the Loan Agreement, such rental shall be paid to the Fiscal Agent for credit against payments due under the Loan Agreement. The Fiscal Agent shall have the right, as directed by the City, in accordance with this Section 18 and the provisions of the Funding Loan Agreement, to exercise

any or all of the rights or remedies of the City hereunder, provided that prior to taking any such action the Fiscal Agent shall give the City written notice of its intended action. All reasonable fees, costs and expenses of the City and the Fiscal Agent incurred in taking any action pursuant to this Section 18 shall be the sole responsibility of the Borrower.

After the Funding Loan Agreement has been discharged, the City may act on its own behalf to declare an "Event of Default" to have occurred and to take any one or more of the steps specified hereinabove to the same extent and with the same effect as if taken by the Fiscal Agent.

The obligations of the Borrower hereunder are not secured by a lien on the Project and the Loan shall not be accelerated as a result of any default hereunder. The Borrower hereby agrees that specific enforcement of the Borrower's agreements contained herein is the only means by which the City may obtain the benefits of such agreements made by the Borrower herein and the Borrower therefore agrees to the imposition of the remedy of specific performance against it in the case of any default by the Borrower hereunder.

The occurrence of a Determination of Taxability shall not, in and of itself, constitute a default hereunder.

**Section 19. The Fiscal Agent**. The Fiscal Agent shall act as specifically provided herein and in the Funding Loan Agreement. The Fiscal Agent is entering into this Regulatory Agreement solely in its capacity as fiscal agent under the Funding Loan Agreement, and the duties, powers, rights and liabilities of the Fiscal Agent in acting hereunder shall be subject to the provisions of the Funding Loan Agreement.

The City shall be responsible for monitoring and verifying compliance by the Borrower with the terms of this Regulatory Agreement. The Fiscal Agent may at all times assume compliance with this Regulatory Agreement unless otherwise notified in writing by the City or unless it has actual knowledge of noncompliance.

After the date on which no principal of the Note remains outstanding as provided in the Funding Loan Agreement, the Fiscal Agent shall no longer have any duties or responsibilities under this Regulatory Agreement and all references to the Fiscal Agent in this Regulatory Agreement shall be deemed references to the City.

**Section 20. Recording and Filing**. The Borrower shall cause this Regulatory Agreement, and all amendments and supplements hereto and thereto, to be recorded and filed in the real property records of the County of Los Angeles and in such other places as the City or the Fiscal Agent may reasonably request. The Borrower shall pay all fees and charges incurred in connection with any such recording. This Regulatory Agreement shall be recorded in the grantor-grantee index to the name of the Borrower as grantor and the City as grantee.

**Section 21. Governing Law**. This Regulatory Agreement shall be governed by the laws of the State of California. The Fiscal Agent's rights, duties and obligations hereunder are governed in their entirety by the terms and provisions of the Funding Loan Agreement.

**Section 22. Amendments**. Except as provided in Section 33(e), this Regulatory Agreement shall be amended only by a written instrument executed by the parties hereto or their

successors in title, and duly recorded in the real property records of the County of Los Angeles, California, and only upon receipt by the City of an opinion from Bond Counsel that such amendment will not adversely affect the Tax-exempt status of interest on the Note and is not contrary to the provisions of the Law or the Act and with the written consent of the Fiscal Agent.

The City, the Fiscal Agent and the Borrower hereby agree to amend this Regulatory Agreement to the extent required, in the opinion of Bond Counsel (subject to the approval of the City Attorney of the City), in order that interest on the Note remains Tax-exempt. The party or parties requesting such amendment shall notify the other parties to this Regulatory Agreement of the proposed amendment, with a copy of such requested amendment to Bond Counsel and the City Attorney of the City and a request that such Bond Counsel render to the City an opinion as to the effect of such proposed amendment upon the Tax-exempt status of interest on the Note.

**Section 23. Notices**. Any notice required to be given hereunder shall be made in writing and shall be given by personal delivery, certified or registered mail, postage prepaid, return receipt requested, at the addresses specified below, or at such other addresses as may be specified in writing by the parties hereto:

If to the City:	City of Los Angeles c/o Housing and Community Investment Department 8th Floor 1200 West 7th Street Los Angeles, CA 90017 Attention: Supervisor, Affordable Housing Bond Program  HIMS# [] Facsimile: (213) 808-8918
	Facsimile: (213) 808-8918
with a copy to:	Los Angeles Housing and Community Investment Department P.O. Box 532729
	Los Angeles, CA 90053-2729
	Attention: Supervisor, Affordable Housing Bond Program HIMS# []
If to the Borrower:	SP7 Apartments LP
	c/o The Skid Row Housing Trust
	1317 E. 7th St.
	Los Angeles, CA 90021
	Attention: []
	Facsimile: []
with a copy to:	[BORROWER'S COUNSEL]
	[]
	[]
	Attention: []
	Facsimile: []

with a copy to:	[INVESTOR LIMITED PARTNER]  c/o []  []  Attention: []  Facsimile: []	
with a copy to:	[INVESTOR LIMITED PARTNER'S COUNSEL]  []  []  Attention: []  Facsimile: []	
with copy to:	MUFG Union Bank, N.A. 200 Pringle Avenue, Suite 355 Walnut Creek, CA 94596 Attention: CDF Division Head Facsimile: (925) 947-2455	
with a copy to:	[LENDER'S COUNSEL] [] [] Attention: []	
If to the Funding Lender:		
	MUFG Union Bank, N.A. Loan Administration Department 3151 East Imperial Highway, 1st Floor Brea, CA 92821 Attention: Manager Telephone (714) 990-7300 Facsimile: (323) 720-2433	
If to the Fiscal Agent:	[FISCAL AGENT]  []  Attention: Global Corporate Trust Services  Ref: LA MF (SP7 Apartments 2018C)  Telephone: (213) []  Facsimile: (213) []	
If to CDLAC:	California Debt Limit Allocation Committee Room 311 915 Capitol Mall Sacramento, CA 95814 Attention: Executive Director	

Notice shall be deemed given three Business Days after the date of mailing.

A duplicate copy of each notice, certificate or other communication given hereunder by any party hereto to another party hereto shall also be given to all of the parties specified above. Failure to provide any such duplicate notice pursuant to the foregoing sentence, or any defect in any such duplicate notice so provided, shall not constitute a default hereunder. All other documents required to be submitted to any of the foregoing parties shall also be submitted to such party at its address set forth above. Any of the foregoing parties may, by notice given hereunder, designate any further or different addresses to which subsequent notices, certificates, documents or other communications shall be sent.

**Section 24. Severability**. If any provision of this Regulatory Agreement or if the applicability of any such provision shall be invalid, illegal or unenforceable, the validity, legality, enforceability or applicability with respect to the validity, legality and enforceability of the remaining portions hereof shall not in any way be affected or impaired thereby.

**Section 25. Multiple Counterparts**. This Regulatory Agreement may be simultaneously executed in multiple counterparts, all of which shall constitute one and the same instrument, and each of which shall be deemed to be an original.

Section 26. Nondiscrimination and Affirmative Action. The Fiscal Agent and the Borrower shall comply with the applicable nondiscrimination and affirmative action provisions of the laws of the United States of America, the State of California, and the City of Los Angeles. The Fiscal Agent and the Borrower shall not discriminate in their employment practices against any employee or applicant for employment; denial of family and medical care leave; denial of pregnancy disability leave or reasonable accommodations against any employee or applicant for employment because of such person's race, ancestry, color, citizenship, national origin, religion, sex, sexual orientation, gender identity/expression, transgender status, age, marital status, familial status, domestic partner status, physical handicap, mental disability, medical condition, political affiliation or belief. The Fiscal Agent and the Borrower shall comply with Executive Order 11246, entitled "Equal Employment Opportunity," as amended by Executive Order 11375, and as supplemented in Department of Labor regulations (41 C.F.R. Part 60). The Fiscal Agent and the Borrower shall comply with the provisions of the Los Angeles Administrative Code Sections 10.8 through 10.13, to the extent applicable hereto. The affirmative action program of the Fiscal Agent and the Borrower shall include the mandatory contract provisions set forth in Los Angeles Administrative Code Section 10.8.4, and said provisions are incorporated herein by this reference. The Borrower and the Fiscal Agent shall also comply with all rules, regulations, and policies of the City of Los Angeles' Board of Public Works, Office of Contract Compliance relating to nondiscrimination and affirmative action, including the filing of all forms required by the City. Any subcontract entered into by the Borrower or Fiscal Agent relating to this Regulatory Agreement, to the extent allowed hereunder, shall be subject to the provisions of this Section. No person shall on the grounds of race, ancestry, color, citizenship, national origin, religion, sex, sexual orientation, gender identity/expression, transgender status, age, marital status, familial status, domestic partner status, physical handicap, mental disability, medical condition, political affiliation or belief be excluded from participation in, be denied the benefit of, or be subjected to discrimination under, this Regulatory Agreement. For purposes of this Section, Title 24 Code of Federal Regulations Part 107 and Section 570.601(b) define specific discriminatory actions that are prohibited and corrective action that shall be taken in a situation as defined therein.

Section 27. Business Tax Registration Certificate. Subject to any exemption available to it, the Fiscal Agent and the Borrower each represent that they will obtain and hold the Business Tax Registration Certificate(s) required by the City's Business Tax Ordinance (Article 1, Chapter 2, Section 21.00 and following, of the Los Angeles Municipal Code). For the term covered by this Regulatory Agreement, the Fiscal Agent and the Borrower shall maintain, or obtain as necessary, all such Business Tax Registration Certificates required of them under said Ordinance and shall not allow any such Business Tax Registration Certificate to be revoked or suspended.

Section 28. Financial Obligations Personal to Borrower. The City acknowledges that the Project shall be encumbered by the Borrower Loan Documents. Notwithstanding any provisions of this Regulatory Agreement to the contrary, all obligations of the Borrower under this Regulatory Agreement for the payment of money and all claims for damages against the Borrower occasioned by breach or alleged breach by the Borrower of its obligations under this Regulatory Agreement, including indemnification obligations, shall not be a lien on the Project and no Person shall have the right to enforce such obligations other than directly against the Borrower as provided in Section 18 of this Regulatory Agreement, except that the City shall have the right at all times to enforce the rights contained in the third paragraph of Section 18 hereof. No subsequent owner of the Project shall be liable or obligated for the breach or default of any obligations of the Borrower under this Regulatory Agreement on the part of any prior Borrower, including, but not limited to, any payment or indemnification obligation. Such obligations are personal to the Person who was the Borrower at the time the default or breach was alleged to have occurred and such Person shall remain liable for any and all damages occasioned thereby even after such Person ceases to be the Borrower. Each Borrower shall comply with and be fully liable for all obligations of an "owner" hereunder during its period of ownership. Notwithstanding the foregoing, neither the Borrower nor its partners shall be personally liable for any indemnification obligation under the Borrower Loan Documents which would result in payment of principal of or interest on the Note.

## Section 29. [Reserved].

Section 30. Child Support Assignment Orders. This Regulatory Agreement is subject to Section 10.10 of the Los Angeles Administrative Code, Child Support Assignment Orders Ordinance. Pursuant to this Ordinance, each of the Borrower and the Fiscal Agent certifies that (a) it will fully comply with all State and federal employment reporting requirements applicable to Child Support Assignment Orders; (b) the principal owner(s) of the Borrower are in compliance with any Wage and Earnings Assignment Orders and Notices of Assignment applicable to them personally; (c) it will fully comply with all lawfully served Wage and Earnings Assignment Orders and Notices of Assignment in accordance with California Family Code Section 5230 et seq.; and (d) it will maintain such compliance throughout the term of this Regulatory Agreement. Pursuant to Section 10.10.b of the Los Angeles Administrative Code, failure of the Borrower or the Fiscal Agent to comply with all applicable reporting requirements or to implement lawfully served Wage and Earnings Assignment Orders and Notices of Assignment or the failure of any principal owner(s) of the Borrower or the Fiscal Agent to comply with any Wage and Earnings Assignment Orders and Notices of Assignment applicable to them personally shall constitute a default by the Borrower or the Fiscal Agent, as appropriate, under the terms of this Regulatory Agreement,

subjecting (i) the Borrower to the remedies provided herein and (ii) the Fiscal Agent to termination under the Funding Loan Agreement where, in either case, such failure shall continue for more than 90 days after notice of such failure to the Borrower or the Fiscal Agent by the City. Any subcontract entered into by the Borrower or the Fiscal Agent relating to this Regulatory Agreement, to the extent allowed hereunder, shall be subject to the provisions of this paragraph and shall incorporate the provisions of the Child Support Assignment Orders Ordinance. Failure of the Borrower or the Fiscal Agent to obtain compliance of its subcontractors shall constitute a default by the Borrower or the Fiscal Agent, as appropriate, under the terms of this Regulatory Agreement, subjecting (A) the Borrower to the remedies provided herein and (B) the Fiscal Agent to termination under the Funding Loan Agreement where such failure shall continue for more than 90 days after notice of such failure to the Borrower or the Fiscal Agent by the City.

The Borrower and the Fiscal Agent shall comply with the Child Support Compliance Act of 1998 of the State of California Employment Development Department. Each of the Borrower and the Fiscal Agent assures that to the best of its knowledge it is fully complying with the earnings assignment orders of all employees and is providing the names of all new employees to the New Hire Registry maintained by the Employment Development Department as set forth in subdivision (b) of the Public Contract Code Section 7110.

Section 31. Americans with Disabilities Act. Each of the Borrower and the Fiscal Agent hereby certifies that it and any contractor and subcontractor will comply with the Accessibility Requirements (as defined in Exhibit I). The Borrower and any contractor and subcontractor will provide reasonable accommodations to allow qualified individuals with disabilities to have access to and to participate in its programs, services, and activities in accordance with the applicable provisions of the ADA, the ADAAG, Section 504, the UFAS, the FHA (each as defined in Exhibit I) and all subsequent amendments. The Borrower, the Fiscal Agent and each and any contractor and subcontractor will not discriminate against persons with disabilities or against persons due to their relationship to or association with a person with a disability. Any contract and subcontract entered into by the Borrower or the Fiscal Agent, relating to this Regulatory Agreement and the Project, to the extent allowed hereunder, shall be subject to the provisions of this paragraph. The Borrower hereby agrees to observe all of the covenants contained in Exhibit I to this Regulatory Agreement as if contained herein.

**Section 32. Slavery Disclosure Ordinance**. This Regulatory Agreement is subject to the Slavery Disclosure Ordinance, Section 10.41 of the Los Angeles Administrative Code, as it may be amended from time to time. The Borrower hereby agrees to execute and deliver a certificate in the form attached hereto as Exhibit H (or such other form as is required by the City) certifying that it has complied with the applicable provisions of this Ordinance. The Borrower acknowledges that failure to fully and accurately complete the affidavit may result in a default under this Regulatory Agreement.

**Section 33. Requirements of CDLAC**. In addition to other requirements set forth herein and to the extent not prohibited by the requirements set forth in Sections 2 through 6 hereof, the Borrower hereby agrees to comply with each of the requirements of CDLAC set forth in this Section 33, as follows:

- (a) The Borrower shall comply with the conditions set forth in Exhibit A to the CDLAC Resolution (the "CDLAC Conditions"), as they may be modified or amended from time to time, which conditions are incorporated herein by reference and made a part hereof and are attached hereto as Exhibit G. Notwithstanding anything to the contrary herein, the provisions of this Section 33 shall remain effective for the period specified in the CDLAC Conditions, unless this Regulatory Agreement shall terminate as otherwise provided in Section 14 hereof.
- (b) The Borrower acknowledges that the City shall monitor the Borrower's compliance with the terms of the CDLAC Conditions. The Borrower will cooperate fully with the City in connection with such monitoring and reporting requirements as provided herein. Compliance with the terms of the CDLAC Conditions not contained within this Regulatory Agreement, but referred to in the CDLAC Conditions, is the responsibility of the Borrower to report to the City.
  - (i) The Borrower shall prepare and deliver a Certificate of CDLAC Program Compliance pursuant to the terms of the CDLAC Conditions. The Borrower acknowledges that the Borrower will prepare and submit to the City, not later than January 15 of each year, and the City will submit to CDLAC, not later than March 1 of each year, until the Borrower has submitted to the City and CDLAC a Rehabilitation Completion Certificate, and on March 1 every three years thereafter (such that the next succeeding year shall be the beginning of each such three-year period) until the end of the term of the CDLAC Conditions, a Certificate of CDLAC Program Compliance, executed by an Authorized Borrower Representative.
  - (ii) The Borrower shall prepare and deliver a Self-Certification Certificate pursuant to the terms of the CDLAC Conditions. The Borrower acknowledges that the Borrower will prepare and submit to the City, not later than January 15 of each year, and the City will submit to CDLAC, not later than March 1 of each year, until the Borrower has submitted to the City and CDLAC a Rehabilitation Completion Certificate, and on March 1 every three years thereafter (such that the next succeeding year shall be the beginning of each such three-year period) until the end of the term of the CDLAC Conditions, a Self-Certification Certificate in the form provided by CDLAC.
  - (iii) Within 30 days following the completion of the Project, the Borrower will prepare and submit to the City, the Fiscal Agent and CDLAC a Rehabilitation Completion Certificate. Following the submission of the Rehabilitation Completion Certificate, the Borrower will prepare and submit to the City, not later than January 15 every three years thereafter until the end of the Compliance Period, a California Tax Credit Allocation Committee Project Status Report or equivalent documentation in substantially the form required or otherwise provided by CDLAC from time to time.
- (c) Except as otherwise provided in Section 14 of this Regulatory Agreement, this Regulatory Agreement shall terminate on the date 55 years after the date on which at

least 50% of the units in the Project are first occupied or such later date as the Qualified Project Period shall begin, as required by the CDLAC Conditions.

- (d) The Borrower shall notify CDLAC in writing of: (i) any change in ownership of the Project, (ii) any change in the issuer of the Note, (iii) any change in the name of the Project or the Project manager; (iv) any default under the Indenture, the Loan Agreement or this Regulatory Agreement; or (v) termination of this Regulatory Agreement.
- (e) Any of the foregoing requirements of CDLAC contained in this Section 33 may be expressly waived by CDLAC, in its sole discretion, in writing, but (i) no waiver by CDLAC of any requirement of this Section 33 shall, or shall be deemed to, extend to or affect any other provision of this Regulatory Agreement except to the extent the City has received an opinion of Bond Counsel that any such provision is not required by the Code, the Act and the Law and may be waived without adversely affecting the exclusion from gross income of interest on the Note for federal income tax purposes; and (ii) any requirement of this Section 33 shall be void and of no force and effect if the City and the Borrower receive a written opinion of Bond Counsel to the effect that compliance with any such requirement would cause interest on the Note to cease to be Tax-exempt or to the effect that compliance with such requirement would be in conflict with the Code, the Act, the Law or any other state or federal law.
- (f) CDLAC is intended to be and is a third-party beneficiary of this Regulatory Agreement, and CDLAC shall have the right (but not the obligation) to enforce, separately or jointly with the City and/or the Fiscal Agent or to cause the City or the Fiscal Agent to enforce, the provisions of Section 33 of this Regulatory Agreement and to pursue an action for specific performance of such provisions or other available remedy at law or in equity in accordance with Section 18 hereof, provided that any such action or remedy shall not materially adversely affect the interests and rights of the Noteholders and shall otherwise be subject to the terms, conditions and limitations applicable to the enforcement of remedies under this Regulatory Agreement.
- CDLAC shall have the right, but not the obligation, to deliver revised CDLAC Conditions to the Borrower after the Closing Date at any time, that are not more restrictive than the original CDLAC Conditions; provided, however, that, with the prior written consent of the Noteholders, which will not be unreasonably withheld: (i) any changes in the terms and conditions of the CDLAC Conditions prior to the recordation against the Project in the real property records of Los Angeles County, California, of a regulatory agreement between the Borrower and the California Tax Credit Allocation Committee ("TCAC Regulatory Agreement") shall be limited to such changes as are necessary to correct any factual errors or to otherwise conform the CDLAC Conditions to any change in facts or circumstances applicable to the Borrower or the Project; and (ii) after recordation of the TCAC Regulatory Agreement, any changes in the terms and conditions of the CDLAC Conditions shall be limited to such changes as are necessary to conform Items 1, 6, 7, 10, 11, 12, 14, 15, 16, 18, 19, 20, 21, 22, 23, 24, 25, 26 or 37 of Exhibit A to the CDLAC Conditions to any change in terms and conditions requested by the Borrower and approved by CDLAC. The City may, in its sole and absolute discretion, require the Borrower to enter into an amendment to this Regulatory Agreement reflecting the revised

CDLAC Conditions, which amendment shall be executed by the parties hereto or their successor in title and duly recorded in the real property records of the County of Los Angeles. The Borrower shall pay any costs and expenses in connection therewith and provide CDLAC with a copy of that recorded amendment reflecting the revised CDLAC Conditions.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the City, the Fiscal Agent and the Borrower have executed this Regulatory Agreement by duly authorized representatives, all as of the date first above written.

CITY OF LOS ANGELES, as City

By Los Angeles Housing and Community Investment Department

	By Sean L. Spear Assistant General Manager	
Approved as to form:		
CITY OF LOS ANGELES MICHAEL N. FEUER, City Attorney		
Deputy/Assistant City Attorney	_	

[Signature Page to *SP7 Apartments* Regulatory Agreement]

By		
Name:		
Title:		

[FISCAL AGENT], as Fiscal Agent

[Signature Page to SP7 Apartments Regulatory Agreement]

SP7 APARTMENTS LP, a California limited partnership

By: SP7 Apartments GP LLC, a California limited liability company, its general partner

> By: The Skid Row Housing Trust, a California nonprofit public benefit corporation, its sole member/manager

By:	
	Ben Rosen,
	Real Estate Development Director

[Signature Page to SP7 Apartments Regulatory Agreement

State of California	)	
County of)		
On	_, before me,	, a Notary Public in and for
said State, personally appeared	ed	
(here insert name(s) of signers), v	who proved to me on t	the basis of satisfactory evidence to be the person(s) whose name(s)
is/are subscribed to the within in	nstrument and acknow	wledged to me that he/she/they executed the same in his/her/their
authorized capacity(ies), and tha	t by his/her/their sign	nature(s) on the instrument the person(s), or the entity upon behalf
of which the person(s) acted, exe	ecuted the instrument	
I certify under PENALTY OF P and correct.	ERJURY under the 1	laws of the State of California that the foregoing paragraph is true
WITNESS my hand and official	seal.	
Signature		[SEAL]

State of California	)		
County of)			
On	_, before me,	, a Notary Public i	n and for
said State, personally appear	red		
(here insert name(s) of signers),	who proved to me on t	ne basis of satisfactory evidence to be the person(s) whose	e name(s)
is/are subscribed to the within i	nstrument and acknow	vledged to me that he/she/they executed the same in his	/her/their
authorized capacity(ies), and that	at by his/her/their sign	ature(s) on the instrument the person(s), or the entity upon	on behalf
of which the person(s) acted, ex	ecuted the instrument.		
I certify under PENALTY OF I and correct.	PERJURY under the la	nws of the State of California that the foregoing paragrap	ph is true
WITNESS my hand and official	l seal.		
Signature		[SEAL]	

State of California )
County of)
On, before me,, a Notary Public in and for
said State, personally appeared
(here insert name(s) of signers), who proved to me on the basis of satisfactory evidence to be the person(s) whose name(s)
is/are subscribed to the within instrument and acknowledged to me that he/she/they executed the same in his/her/their
authorized capacity(ies), and that by his/her/their signature(s) on the instrument the person(s), or the entity upon behalf
of which the person(s) acted, executed the instrument.
I certify under PENALTY OF PERJURY under the laws of the State of California that the foregoing paragraph is true and correct.
WITNESS my hand and official seal.
Signature [SEAL]

State of California	)	
County of	)	
On	, before me,	, a Notary Public in and for
said State, personal	ly appeared	
(here insert name(s) of	signers), who proved to me on the basis of satisfactors	sfactory evidence to be the person(s) whose name(s)
is/are subscribed to the	e within instrument and acknowledged to me	that he/she/they executed the same in his/her/their
authorized capacity(ie	s), and that by his/her/their signature(s) on the	e instrument the person(s), or the entity upon behalf
of which the person(s)	acted, executed the instrument.	
I certify under PENAl and correct.	LTY OF PERJURY under the laws of the Stat	te of California that the foregoing paragraph is true
WITNESS my hand a	nd official seal.	
Signature	[SEA	L]

### **EXHIBIT A**

### **DESCRIPTION OF PROJECT SITE**

### 647 South San Pedro

Real property in the City of Los Angeles, County of Los Angeles, State of California, described as follows:

as 10110 WS.		
	[TO BE PROVIDED]	
Assessor's Parcel Number: [	]	

#### **EXHIBIT B**

#### FORM OF CERTIFICATE OF CONTINUING PROGRAM COMPLIANCE

FOR THE [MONTH/QUARTER] ENDING	
\$[32,007,814]	
City of Los Angeles	
Multifamily Mortgage Revenue Note	
(SP7 Apartments)	
Series 2018C	

The undersigned, being the Authorized Borrower Representative of SP7 Apartments LP, a California limited partnership (the "Borrower"), has read and is thoroughly familiar with the provisions of the various loan documents associated with the Borrower's participation in the multifamily housing program of the City of Los Angeles (the "City"), including, without limitation, the Regulatory Agreement and Declaration of Restrictive Covenants, dated as of [September] 1, 2018 (the "Regulatory Agreement"), among the Borrower, the City and [FISCAL AGENT], as Fiscal Agent relative to the property located at 647 South San Pedro, Los Angeles, California.

As of the date of this Certificate, the following percentages of completed residential units in the Project (as such term is defined in the Regulatory Agreement) (i) are occupied by Low Income Tenants (as such term is defined in the Regulatory Agreement) or (ii) are currently vacant and being held available for such occupancy and have been so held continuously since the date a Low Income Tenant vacated such unit, as indicated:

Occupied by Low Income Tenants:	Unit Nos and size
Held vacant for occupancy continuously since last occupied by Low Income Tenant:	Unit Nos and size
Vacant Units:	
Low Income Tenants who commenced Occupancy of units during the Preceding [month/quarter]:	Unit Nos.

Attached is a separate sheet (the "Occupancy Summary") listing, among other items, the following information for each unit in the Project: the number of each unit, the occupants of each unit and the size, in square feet, of each unit. It also indicates which units are occupied by Low Income Tenants and which units became Low Income Units during the preceding [month/quarter]. The information contained thereon is true and accurate.

The undersigned hereby certifies that (1) a review of the activities of the Borrower during such [month/quarter] and of the Borrower's performance under the Loan Agreement has been made under the supervision of the undersigned; (2) to the best of the knowledge of the undersigned, based on the review described in clause (1) hereof, the Borrower is not in default under any of the terms and provisions of the above documents [OR DESCRIBE THE NATURE OF ANY DEFAULT IN DETAIL AND SET FORTH THE MEASURES BEING TAKEN TO REMEDY SUCH DEFAULT]; and (3) to the knowledge of the Borrower, no Determination of Taxability (as such term is defined in the Regulatory Agreement) has occurred [OR, IF A DETERMINATION OF TAXABILITY HAS OCCURRED, SET FORTH ALL MATERIAL FACTS RELATING THERETO].

SP7 APARTMENTS LP, a California limited partnership

By: SP7 Apartments GP LLC, a California limited liability company, its general partner

> By: The Skid Row Housing Trust, a California nonprofit public benefit corporation, its sole member/manager

By:	
	Ben Rosen,
	Real Estate Development Director

[Signature Page to SP7 Apartments Certificate of Program Compliance]

#### EXHIBIT C

#### FORM OF INCOME CERTIFICATION

NOTE TO APARTMENT OWNER: This form is designed to assist you in computing this form is at all times up to date with the HUD Regulations.

Annual Income in accordance with the method set forth in the Department of Housing and Urban Development ("HUD") Regulations (24 C.F.R. Part 5 Subpart F). You should make certain that SP7 Apartments, 647 South San Pedro, Los Angeles, California Re: The undersigned hereby (certify) (certifies) that: 1. This Income Certification is being delivered in connection with the undersigned's application for occupancy of Apartment #\_\_\_\_ in the SP7 Apartments located at 647 South San Pedro, in Los Angeles, California. List all the occupants of the apartment, the relationship (if any) of the various occupants, their ages, and indicate whether they are students (for this purpose, a student is any individual who has been, or will be, a full-time student at an educational institution during five months (whether consecutive or not) of the year in which this application is submitted, other than a correspondence school, with regular facilities and students). **Social Security** Student Occupant Relationship (Yes or No) Number Age (e) \_\_\_\_\_ (f) If all of the occupants are students, answer the following questions for each 3. occupant Is any student listed in paragraph 2 above married and files a joint return for federal income tax purposes? List any such students. Yes Name(s) No Not Applicable

(b) Is any student listed in paragraph 2 above (i) a single parent living with his/her children, (ii) not a dependent of another individual and (iii) whose children are not dependents of an individual other than their parents? List any such students.

Yes Name(s) No Not Applicable

(c) Is any student listed in paragraph 2 above a student receiving assistance under Title IV of the Social Security Act (Temporary Assistance for Needy Families)? List any such students.

Yes Name(s) No Not Applicable

(d) Is any student listed in paragraph 2 above a student enrolled in a job-training program receiving assistance under the Job Training Partnership Act or under other similar federal, State or local laws? List any such students.

Yes Name(s) No Not Applicable

(e) Is any student listed in paragraph 2 a student who was previously under the care and placement responsibility of a foster care program (under Part B or E of Title IV of the Social Security Act)? List any such students.

Yes Name(s) No Not Applicable

4. The total anticipated income for each person listed in paragraph 2 above during the 12-month period commencing with the date occupancy will begin, including:

full amount, before any payroll deductions, of wages, salaries, overtime, commissions, fees, tips, and bonuses; net income from the operation of a business or profession or from the rental of real or personal property (without deducting expenditures for business expansion or amortization of capital indebtedness or any allowance for depreciation of capital assets); interest and dividends (including income from assets excluded below); the full amount of periodic payments from social security, annuities, insurance policies, retirement funds, pensions, disability or death benefits and other similar types of periodic payments, including any lump-sum payment for the delayed start of a periodic payment; payments in lieu of earnings, such as unemployment and disability compensation, workers' compensation and severance pay; all public assistance income; periodic and determinable allowances such as alimony and child support payments, and regular contributions or gifts received from persons not residing in the dwelling; all regular and special pay and allowances of members of the Armed Forces (whether or not living in the dwelling) who are the head of the family or spouse; and any earned income tax credit to the extent that it exceeds income tax liability;

#### but excluding:

income from employment of children (including foster children) under the age of 18 years; payments received for the care of foster children or foster adults (usually individuals with disabilities, unrelated to the tenant family, who are unable to live alone); lump-sum additions to family assets, such as inheritances, insurance payments (including payments

under health and accident insurance and workers' compensation), capital gains and settlement for personal or property losses; amounts which are specifically for reimbursement of medical expenses; amounts of educational scholarships paid directly to the student or the educational institution, and amounts paid to a veteran for use in meeting the costs of tuition, fees, books and equipment, but in either case only to the extent used for such purposes; special pay to a serviceman head of a family who is away from home and exposed to hostile fire; amounts received under training programs funded by HUD; amounts received under Plan to Attain Self-Sufficiency; amounts for out-of-pocket expenditures incurred in connection with other public assistance programs; resident service stipend (not in excess of \$200 per month); amounts from state or local employment training programs; temporary, nonrecurring or sporadic income (including gifts); reparation payments paid by a foreign government to persons who were persecuted during the Nazi era; earnings in excess of \$480 for each full-time student 18 years old or older (excluding head of family and spouse); adoption assistance payments in excess of \$480 per adopted child; deferred periodic payments of supplemental social security income and benefits received in a lump sum; refunds or rebates of property taxes paid on the unit; payments from state agency to allow developmentally disabled family member to stay home; relocation payments under Title II of the Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970; foster child care payments; the value of coupon allotments for the purchase of food pursuant to the Food Stamp Act of 1964 which is in excess of the amount actually charged for the allotments; and payments to volunteers under the Domestic Volunteer Service Act of 1973; is as follows:

(	Occupant	Anticipated Annual Income	Source of Income or Employer
(a)			
(b)			
		<b>C</b>	
		•	
		_	
		\$	
TOTAL		\$	

5.(a) Do the persons whose income or contributions are included in paragraph 4 above have savings, stocks, bonds, equity in real property or other form of capital investment (excluding the values of necessary items of personal property such as furniture and automobiles and interest in Indian trust land)?

				-	Y	es _	No	
(b) disposed of a less than fair	ny assets	s (other						aragraph 4 above last two years at
				-	Y	es _	No	
(c) assets owned								value of all such
				-	Y	es _	No	
(d)	If the a	ınswer	to (c) above i	s yes,				
	\$		insert the t	otal value	of all	such as	ssets owned	or disposed of
		(ii)	state:					
	12-mo	nth per	riod beginnin					such assets in the the unit that you
	\$							
	above:		the amount	of such inco	ome, if a	ıny, that	t was include	ed in paragraph 4
	\$							
	NT - 141		16		4 - C 41			:

- 6. Neither myself nor any other occupant of the unit I/we propose to rent is the owner of the rental housing project in which the unit is located (hereinafter, the "Borrower"), has any family relationship to the Borrower or owns directly or indirectly any interest in the Borrower. For purposes of this paragraph, indirect ownership by an individual shall mean ownership by a family member, ownership by a corporation, partnership, estate or trust in proportion to the ownership or beneficial interest in such corporation, partnership, estate or trust held by the individual or a family member and ownership, direct or indirect, by a partner of the individual.
- 7. This Income Certification is made with the knowledge that it will be relied upon by the Borrower to determine maximum income for eligibility to occupy the unit, and I/we declare that all information set forth herein is true, correct and complete and based upon information I/we deem reliable and that the statement of total anticipated income contained in paragraph 4 is reasonable and based upon such investigation as the undersigned deemed necessary.
- 8. I/we will assist the Borrower in obtaining any information or documents required to verify the statements made therein, including either an income verification from my/our present employer(s) or copies of federal tax returns for the immediately preceding calendar year.

- 9. I/we acknowledge that I/we have been advised that the making of any misrepresentation or misstatement (whether or not intentional) in this Income Certification will constitute a material breach of my/our agreement with the Borrower to lease the unit and will entitle the Borrower to prevent my/our occupancy of the unit and will be cause for immediate termination of such lease.
- 10. The undersigned hereby acknowledge and agree that on or before January 1 (or upon Lease renewal) of each year the undersigned and any other current residents of such apartment will complete and deliver a new Income Certification, in the form then in use, to the Borrower and that the undersigned's rent is subject to increase 30 days after written notice is given to the undersigned stating that the undersigned no longer qualifies as a Lower Income Tenant under the Tax Regulatory Agreement.
- 11. RESIDENT(S) STATEMENT: I/We certify that the statements are true and complete to the best of my/our knowledge and belief and are given under penalty of perjury. In the event this Income Certification is executed more than five days prior to the date I/we intend to occupy the unit, I/we hereby agree to update and recertify the accuracy of the information herein provided as of the date I/we first occupy the unit:

(a)	Date:	
(b)	Date:	
(c)	Date:	
(d)	Date:	
(e)	Date:	
(f)	Date:	
[Th required]	e signatures of all persons over the age of 18 years listed in po	aragraph 2 above
12.	Calculation of Eligible Income:	
(a)	Enter the amount entered for entire household in 4 above:	\$
(b)	Enter income derived from assets (line 5(d)(2)(A)):	\$
(c)	Subtract (b) from (a)	\$
(d)	Multiply the amount entered in $5(d)(1)$ by the current passbook savings rate to determine the total annual earnings on assets $[5(d)(1)]$ if invested in passbook savings.	
	Passbook rate	

are

(e)	Enter the greater of (b) or (d)	\$
(f)	TOTAL ELIGIBLE INCOME (Line (e) + (c))	\$
13.	The amount entered in 12(f):	
(a)	Qualifies the applicant(s) as a Lower Income	Γenant(s).
(b)	Does not qualify the applicant(s) as Lower Inc	come Tenant(s).
14.	Number of apartment unit assigned:	
	Bedroom size: Rent: \$	

	Tenant-paid		Utilities:				
	Water Trash		Gas	Electric _			
			Other (list type)				
		e aggre	gate anticipa	ted annual in	ccupied for a period of 31 consecutive days by come as certified in the above manner upon their d them as Lower Income Tenants?		
				_	Yes No		
	16.	Metho	od used to ver	rify applicant(	(s) income:		
			Employer in	ncome verific	ation		
			Social Secu	rity Administ	ration verification		
			Department	t of Social Ser	vices verification		
			Copies of ta	ax returns			
			Other (				
	17.	Metho	ethod used to verify responses, if any, in paragraph 3 of this Income Certification:				
			Copies of ta	ax returns			
			Evidence of	f participation	in an enumerated program		
named	in par	cumen agraph	tation submit 2 of this In	ted pursuant come Certific	Based on the representations herein and upon the to paragraph 8 hereof, the family or individual(s) cation is/are eligible under the provisions of the rictive Covenants to live in a unit in the Project.		
Date _					Signature of Authorized Borrower Representative:		
					ByName		
					Title		
	EXEC	UTION	OF PARAC	GRAPHS 19 A	AND 20		
		IS	IS NOT N	IECESSARY			
	Initials	z.					

19. If this Income Ce	ertification was executed by me/us more than five days prior to
my/our occupancy of the unit, I/	we hereby update and recertify the accuracy of the information
herein provided as of	, 20 and state:
•	
(a) No	additional information is required to be provided to make this
Income Certification true	and correct on the date of this certification.
	e following information is provided to update the information
previously provided in the	e Income Certification:
[Rema	ninder of page intentionally left blank
Livenia	inder of page intentionally fert blank

	(a)		Date:	
	(b)		Date:	
	(c)		Date:	
	(d)		Date:	
	(e)		Date:	
	(f)		Date:	
		ne Certificati	on have, pursuar	ENT: The family or individual(s) named in paragraph 2 at to paragraph 19 hereof, updated and recertified the cally set forth in paragraph 19 hereof.
Date _				Signature of Authorized Borrower Representative
				Ву
				Name
				Title

[Remainder of page intentionally left blank]

### **INCOME VERIFICATION**

(for employed persons)

The undersigned employee has applied for a rental unit located in a project financed by the issuance of a note by the City of Los Angeles for persons of low or moderate income. Every income statement of a prospective tenant must be stringently verified. Please indicate below the employee's current annual income from wages, overtime, bonuses, commissions or any other form of compensation received on a regular basis.

	Annual Wag	ges							
	Overtime								
	Bonuses								
	Commission	ns							
	Total Curren	nt Income							
knowl	-	ertify that the	statements	above are	e true and	l complete	to the	best	of my
Date									
				Ву					
				Name_					
				Title					

I hereby grant you permission to disclose my income to
in order that they may determine my income eligibility for rental of an apartment located in their project which has been financed by an issuance of a note by the City of Los Angeles.
Date Signature
Please send form to:

[Income verification signature page]

### **INCOME VERIFICATION**

(for self-employed persons)

I hereby attach copies of my individual federal and state (if applicable) income tax return	S
for the immediately preceding calendar year and certify that the information shown in such incom	e
tax returns is true and complete to the best of my knowledge.	

Date		
	Signature	

#### **EXHIBIT D**

#### FORM OF ANNUAL TENANT INCOME RECERTIFICATION

# CITY OF LOS ANGELES ANNUAL TENANT INCOME RECERTIFICATION

Projec	ct name				
Apart	ment # Date of Orig	ginal Certification_			
Resid	ent name				
This f which follow	HE RESIDENT: form is a continuation of the City was previously discussed with y ving information each year when y 86 and the City to maintain this in	you. In order to ke you renew your leas	ep you on the se. The Borre	ne qualifying list, you ower is required by the	will need to update the
1) 2)	ehold Composition:  Please list all of those individu  Please list the anticipated annu usemaker, or unemployed, etc.—  If college or technical school s	nal income of all occlease list as such).	cupants of yo		18 years of age or older
1)	NAME	SS#	AGE	ANTICIPATED ANNUAL INCOME*	OCCUPATION/STUDEN
2)					
3) 4)					
5)					
6) 7)			+		_
*SEE	INCOME DEFINITION ATTAC OU OWN OR HAVE YOU ACC HE PAST YEAR?	QUIRED OR HAV	E YOU DIS		
	If so please describe and list a	amount and annual i	income expe	cted to be derived from	n such assets

If all persons residing in your apartment are full-time students, please indicate for each such person whether they are: (1) a single parent living with his/her children; (2) a student receiving assistance under Title IV of the Social Security Act (Temporary Assistance for Needy Families); (3) a student enrolled in a job-training program receiving assistance under the Job Training Partnership Act or under other similar federal, state or local laws; (4) a student who was previously under the care and placement responsibility of a foster care program (under Part B or E of Title IV of the Social Security Act); or (5) a student who is married and files a joint return. Single parents described in (1) above may not be dependents of another individual and their children may not be dependents of another individual other than their parents.

#### Please have all occupants over the age of 18 sign this certification.

I/we acknowledge that I/we have been advised that the making of any misrepresentation or misstatement in this declaration will constitute a material breach of my/our agreement with the Borrower to lease the unit and will entitle the Borrower to prevent or terminate my/our occupancy of the unit by institution of an action for ejection or other appropriate proceedings.

	I/we declare under penalty of perjury that the foregoing is true and correct.		
	SIGNATURES:	DATE:	
1)			
2)			
3)			
4)			
MA	NAGER'S SIGNATURE:		

#### **DEFINITION OF INCOME**

The full amount, before any payroll deductions, of wages, salaries, overtime, commissions, fees, tips, and bonuses; net income from the operation of a business or profession or from the rental of real or personal property (without deducting expenditures for business expansion or amortization of capital indebtedness or any allowance for depreciation of capital assets); interest and dividends (including income from assets excluded below); the full amount of periodic payments from social security, annuities, insurance policies, retirement funds, pensions, disability or death benefits and other similar types of periodic payments, including any lump-sum payment for the delayed start of a periodic payment; payments in lieu of earnings, such as unemployment and disability compensation, workers' compensation and severance pay; all public assistance income; periodic and determinable allowances such as alimony and child support payments, and regular contributions or gifts received from persons not residing in the dwelling; all regular and special pay and allowances of members of the Armed Forces (whether or not living in the dwelling) who are the head of the family or spouse; and any earned income tax credit to the extent that it exceeds income tax liability;

#### but excluding:

income from employment of children (including foster children) under the age of 18 years; payments received for the care of foster children or foster adults (usually individuals with disabilities, unrelated to the tenant family, who are unable to live alone); lump-sum additions to family assets, such as inheritances, insurance payments (including payments under health and accident insurance and workers' compensation), capital gains and settlement for personal or property losses; amounts which are specifically for reimbursement of medical expenses; amounts

of educational scholarships paid directly to the student or the educational institution, and amounts paid to a veteran for use in meeting the costs of tuition, fees, books and equipment, but in either case only to the extent used for such purposes; special pay to a serviceman head of a family who is away from home and exposed to hostile fire; amounts received under training programs funded by HUD; amounts received under Plan to Attain Self-Sufficiency; amounts for out-of-pocket expenditures incurred in connection with other public assistance programs; resident service stipend (not in excess of \$200 per month); amounts from state or local employment training programs; temporary, nonrecurring or sporadic income (including gifts); reparation payments paid by a foreign government to persons who were persecuted during the Nazi era; earnings in excess of \$480 for each full-time student 18 years old or older (excluding head of family and spouse); adoption assistance payments in excess of \$480 per adopted child; deferred periodic payments of supplemental social security income and benefits received in a lump sum; refunds or rebates of property taxes paid on the unit; payments from state agency to allow developmentally disabled family member to stay home; relocation payments under Title II of the Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970; foster child care payments; the value of coupon allotments for the purchase of food pursuant to the Food Stamp Act of 1964 which is in excess of the amount actually charged for the allotments; and payments to volunteers under the Domestic Volunteer Service Act of 1973.

### **EXHIBIT E**

### FORM OF CERTIFICATE OF CDLAC PROGRAM COMPLIANCE

Project Name Change:	No	Yes
(If project name has change as well as the new project na		llocation please note the original project name
If yes provide old and new I	Project Name:	
CDLAC Application No.:	18-317	
Bond Issuer Change:	No	Yes
(If Bond Issuer has changed please note the original Issuer		esult of refinance or refunding of an allocation ssuer.)
If yes provide the Name of e	xisting and New Issue	r
Contact Information		
Change in Borrower		
(If Borrower has changed sin Borrower as well as the new		the CDLAC resolution please note the original
If yes provide the Name of t	he existing and New B	orrower
Contact Information		
Change in Management Cor	npany No	Yes
If yes provide the Name of t	he New Management (	Company
Has the Qualified Project Pe	riod commenced?	No Yes
No Yes_	Already Subr	mitted Certification
If yes please submit the Reh	abilitation Completion	Certificate (one time only)
Has the project been comple	ted and placed in servi	ce?
No Yes_	Already Subr	nitted Certification
If yes please submit Comple	etion Certification (one	time only)

Have any of the following event limited to: defaults associated w Bond Default?		
No Yes	_	
If so, please describe and explain	n?	
Has a termination of the Regular year? Has proper noticing occur		termination planned in the next
No Yes	_	
If so, please describe and explain	n?	
Federally Bond Restricted Units (Reflected in in PSR)	Other Restrictions (reflected in PSR)	Total (Reported in CDLAC Resolution)
at 50% AMI	at 50% AMI	at 50% AMI
at 60% AMI	at 60% AMI	at 60% AMI
Total	Total	Total
Please attach a copy of the documentation.	project's TCAC Project Statu	s Report (PSR) or equivalent
Please indicate the distribution o	f the CDLAC-restricted 10% of	the 50% AMI units.
Bedroom Type	# of Units in PSR	# of Units in CDLAC Resolution
1 bedroom		
2 bedroom		
3 bedroom		

If the Project has committed to and is currently providing the service amenities for a term as specified in the CDLAC resolution, please verify the services are being provided: on a regular and ongoing basis, which are provided free of charge, and all hour requirements are being met:
After-school Programs
Educational, health and wellness, or skill building classes
Health and Wellness services and programs (not group classes)
Licensed Childcare provided for a minimum of 20 hours per week (Monday-Friday)
Bona Fide Service Coordinator/ Social Worker
Is the service being offered on an ongoing basis and provided free of charge (childcare excepted)?
No Yes If no please explain.
Are all hour requirements being met?
·
No Yes If no please explain.
Attach evidence demonstrating that the above listed services are being provided and have met the requirements in the CDLAC Resolution. Including but not limited to MOUs and or contracts associated with the services rendered, a 12-month schedule (current reporting year) of the services offered, flyers, sign-up sheets, etc.
Pursuant to Section 13 of Resolution No. 18-011 (the "Resolution"), adopted by the California Debt Limit Allocation Committee (the "Committee") on March 21, 2018, I, [], an Officer of the Borrower, hereby certify under penalty of perjury that, as of the date of this Certification, the above-mentioned Project is in compliance with the terms and conditions set forth in the Resolution. I further certify that I have read and understand the Resolution, which specifies that once the Note is issued, the terms and conditions set forth in the Resolution Exhibit A shall be enforceable by the Committee through an action for specific performance, negative points, withholding future allocations or any other available remedy.
Signature of Officer Date

Printed Name of Officer	
Title of Officer	

### **EXHIBIT F**

### FORM OF REHABILITATION COMPLETION CERTIFICATE

1) Project Name:	SP7 Apartments
as well as the new project name	ince the award of allocation please note the original project name e and request a change in the CDLAC resolution.)
2) CDLAC Application No.:	18-317
3) Name of Bond Issuer:	City of Los Angeles
new Borrower and request a ch	SP7 Apartments LP e since the award please note the original Borrower as well as the nange in the CDLAC resolution.)
5) The undersigned hereby cert, 20	ifies that all work on the Project was substantially completed as of
The undersigned hereby further	certifies that:
(a) the aggregate an	nount disbursed on the Loan to date is \$;
reimburse the undersign Regulatory Agreement)	bursed from proceeds of the Note have been applied to pay or ned for the payment of Project Costs (as that term is used in the and none of the amounts disbursed from the proceeds of the Note y or reimburse any party for the payment of costs or expenses other
applied to pay or reimb that term is used in the F from the proceeds of the	the amounts disbursed from the proceeds of the Note have been urse the Borrower for the payment of Qualified Project Costs (as Regulatory Agreement) and less than 25% of the amounts disbursed Note, exclusive of amounts applied to pay the costs of issuing the I to pay or reimburse the Borrower for the cost of acquiring land.
6) The undersigned hereby ce Project Period.	rtifies the Project meets the general federal rule for a Qualified
No Yes	
	lling units in the Project financed in part from the proceeds of the d on, 20 and
	lling units in the Project financed in part from the proceeds of the d on

7) If no to Qualified Pr	), the undersigned hereby certifies the Project meets the special federal rule ject Period.	for a
No_	Yes	
` .	ifies if it is an acquisition/rehabilitation where no more than 90% of the units for occupancy within 60 days of the earlier of the Project acquisition or the .)	
(a)	Note was issued on	
(b)	Property was acquired on	
(c) of th	The date 10% of the units were available to occupy (within 60 days of the e acquisition or bond issuance) is, 20	arlier
Signature of	Officer Date	
Printed Nan	of Officer	
Title of Offi	<del></del> er	
Phone Num	 er	

### **EXHIBIT G**

## **CDLAC RESOLUTION**

### **EXHIBIT H**

### FORM OF SLAVERY DISCLOSURE ORDINANCE CERTIFICATE

#### **EXHIBIT I**

#### **ACCESSIBILITY COVENANTS**

The Accessibility Covenants (the "Covenants") herein are attached to the Regulatory Agreement as an exhibit as an exhibit and the Borrower hereby agrees to comply with each of the requirements of the City set forth as follows:

**Section 1. Definitions.** Terms not otherwise defined herein shall have the meanings assigned thereto in the Regulatory Agreement as applicable. The following terms shall have the respective meanings assigned to them in this Section 1 unless the context in which they are used clearly requires otherwise:

"Accessible," when used with respect to a Housing Unit or a Housing Development, means and refers to full compliance with the requirements of the Accessibility Standards.

"Accessible Housing Development" means a Housing Development that is Accessible, including Accessible public and common use areas, as well as the number and type of Accessible Housing Units that are required to be Accessible by this Agreement.

"Accessible Housing Units" or "Accessible Unit" refers collectively to Housing Units with Mobility Features and Housing Units with Hearing/Vision Features that are Accessible, on an accessible route, and in an Accessible Housing Development.

"Accessibility Laws" means Section 504 of the Rehabilitation Act of 1973, 29 U.S.C. §794 et seq, and the implementing regulations at 24 C.F.R. Part 8, as well as the requirements of the Uniform Federal Accessibility Standards ("UFAS"); the Americans with Disabilities Act, ("ADA"), 42 U.S.C. §12101, et seq. including (1) Title II of the ADA and the implementing standards ("2010 ADA Standards") at 28 C.F.R. Part 35 and the 2004 ADA Accessibility Guidelines ("ADAAG") and Title III of the ADA, Part 36 and the implementing standards ("2010 ADA Standards") at 28 C.F.R. Part 36, subpart D; California Government Code Section 11135 et seq.; the federal Fair Housing Act of 1968, as amended ("FHA"), 42 U.S.C. §§3601-3620; and its implementing regulations as 24 C.F.R. Parts 100, 103, 108, 110, and 121; implementing regulations and design standards for each of the preceding statutes; and the California Building Code Sections 11A and 11B.

"Accessibility Standards" means the standards adopted under the Accessibility Laws, specifically the following compliance standards:

- a. the new construction requirements of 24 C.F.R. pt. 8, including 24 C.F.R. §§ 8.22 and 8.32 as well as the new construction requirements of UFAS, or their successor standards;
- b. the Alternative Accessibility Standard; or any future accessibility standard and other regulatory requirements applicable to newly constructed facilities in federally-assisted programs that may be adopted in a final rule issued by the U.S. Department of Housing and Urban Development ("HUD") pursuant to notice and comment rulemaking under Section 504 so long as such accessibility standard and regulatory requirements do not provide for less accessibility for persons with disabilities than either (i) or (ii);

- c. the requirements in ANSI A117.1-1986 and the Fair Housing Accessibility Guidelines, March 6, 1991, in conjunction with the Supplement to Notice of Fair Housing Accessibility Guidelines: Questions and Answers About the Guidelines, June 28, 1994;
- d. the accessibility provisions of the California Building Code Chapters 11A and 11B, or any future accessibility standard and other regulatory requirements applicable to newly constructed facilities adopted as part of the California Building Code;
- e. all applicable building codes in effect for the City of Los Angeles Building and Safety Department; and
- f. ADA accessibility provisions relevant to public use areas.

"Alternative Accessibility Standard" means and refers to the alternative accessibility standard for new construction set out in HUD's notice at 79 Fed. Reg. 29,671 - Nondiscrimination on the Basis of Disability in Federally Assisted Programs and Activities (May 23, 2014), when used in conjunction with the new construction requirements of HUD's regulations at 24 C.F.R. pt. 8, 24 C.F.R. § 8.22, and the new construction requirements of 28 C.F.R. pt. 35, including the 2010 Standards for Accessible Design as defined in 28 C.F.R. § 35.104 and as applied to public entities (excluding any elevator exceptions).

"Fair Housing Policy Related to Disability" means the document containing the policy of the City, as amended periodically, that all affordable housing developments monitored by the City be constructed and operated in accordance with all applicable disability and fair housing laws and under which the Borrower is required to create a Property Management Plan ("Property Management Plan" or "PMP") as described in Section 4 that must comply with the requirements and guidance in the Fair Housing Policy Related to Disability. The PMP must be consistent with the City's Property Management Plan template and must be approved by the City along with other requirements, as amended periodically.

"Housing Development" or "Development" means the whole of one or more residential structures and appurtenant structures in the project, including common walkways and parking lots that were or are designed, constructed, altered, operated, administered or financed in whole or in part in connection with the issuance of Bonds.

"Housing Unit" means a single unit of residence in the Housing Development that provides spaces for living, bathing, and sleeping.

"Housing Unit with Hearing/Vision Features" means a Housing Unit that is located on an accessible route and complies with 24 C.F.R. §§ 8.22 and 8.23 and all applicable provisions of UFAS, or the comparable provisions of the Alternative Accessibility Standard including but not limited to § 809 and specifically subsection § 809.5 of the 2010 ADA Standards for Accessible Design, and with the California Building Code Chapters 11A & 11B. Hearing/Vision Features include but are not limited to visual alarms (UFAS §§ 4.34.10, 4.28.3), auxiliary alarms (UFAS §§ 4.34.10, 4.28.4), protections against protruding objects (UFAS § 4.4), stairway requirements (UFAS §§ 4.9, 4.26.4), protections against exposed pipes and surfaces (UFAS §§ 4.19.4, 4.24.6, 4.34.6.5(8)), audible alarms (UFAS § 4.28.2), signage (UFAS § 4.30), consumer information (UFAS § 4.34.4), and range, cooktop, and oven controls (UFAS §§ 4.34.6.6, 4.34.6.7).

"Housing Unit with Mobility Features" means a Housing Unit that is located on an

accessible route and complies with 24 C.F.R. §§ 8.22 and 8.23 and all applicable provisions of UFAS, or the comparable provisions of the Alternative Accessibility Standard including but not limited to § 809 and specifically subsections §§ 809.2 through 809.4 of the 2010 ADA Standards, and with the California Building Code Chapter 11 B. A Housing Unit with Mobility Features can be approached, entered and used by persons with mobility disabilities, including people who use wheelchairs.

"UFAS" means the Uniform Federal Accessibility Standards and refers to a set of scoping requirements and standards for the design and construction of buildings and facilities to ensure that they are readily accessible to and usable by persons with disabilities. See Appendix A to 24 C.F.R. subpart 40 for residential structures and Appendix A to 41 C.F.R. subpart 101-19.6 for general-type buildings (UFAS is also available on-line at http://www.access-board.gov).

**Section 2. Borrower Obligations.** The Borrower represents, warrants, covenants and agrees as follows:

- a. A State of California Certified Access Specialist ("CASp") who is a licensed architect or engineer must be identified as part of the development team. The CASp cannot be the architect of record for the project. The cost of CASp activities and certifications should be included in the application's project budget.
- b. The Housing Development shall be constructed in accordance with the Accessibility Standards in Section 1 above to ensure accessibility for persons with disabilities. Borrower must work with their CASp to ensure that the Housing Development complies with those Accessibility Standards.
- c. An accessibility report by a CASp inspector certifying that the Housing Development is compliant with all applicable Accessibility Standards, as defined in Section 1, above,, must be submitted to and approved by the City at the following phases of the project development:
  - 1. Accessibility Design Review Report and a pdf copy of the plans are due for review by the City when construction documents have been developed, and prior to the submission of plans to Los Angeles Building and Safety Department.
  - 2. The Accessibility Design Review Report must be approved by the City before building permits can be issued.
  - Accessibility Progress Inspection Reports conducted after all rough inspections have been signed off by the Los Angeles Building and Safety Department and prior to closing of walls.
  - 4. The Final Accessibility Report at completion of construction must be approved by the City before any retention payment, certificate of occupancy, and final building permit can be issued by the Los Angeles Building and Safety Department.
  - 5. A list of State Certified Disabled Access Specialists can be found at the following link: https://www.apps.dgs.ca.gov/casp/casp/casp certified list.aspx.

- d. If applicable, applicants/developers/owners must list all applicable standards on title page of plans and include the following note: "This is a publicly funded housing project and must comply with California Building Code Chapters 11A & 11B").
- e. If the Development is to be rehabilitated, accessibility retrofit of the Housing Development shall take place concurrently with any project rehabilitation.
- f. The Accessible Housing Units shall be prioritized for persons with disabilities who have a disability-related need for the accessibility features of the Accessible Housing Unit.
- g. At least ten percent (10%) of the total Housing Units in the Housing Development shall be constructed and maintained by the Borrower as Housing Units with Mobility Features.
- h. At least <u>four percent</u> (4%) of the total Housing Units in the Housing Development shall be constructed and maintained by the Borrower as Housing Units with Hearing/Vision Features.
- i. The 4% and 10% calculations shall each be based on the total number of units in the Housing Development. In determining the number of Accessible Housing Units required, any fractions of units shall be rounded up to the next whole number. Required Accessible Housing Units shall, to the maximum extent feasible and subject to reasonable health and safety requirements, be distributed throughout projects and sites, and shall be available in a sufficient range of sizes and amenities so that a qualified individual with a disability has a choice of living arrangements that is, as a whole, comparable to that of other persons eligible for housing assistance under the same program.
- j. The Accessible Housing Units shall be affordable for households with incomes at 30%, 50%, and 80% of area median income.
- k. The project shall comply with the City's Accessibility Regulations Matrix & Overview, Accessible Design/Construction Compliance Requirements, and the Accessibility Report Requirements.
- 1. Borrowers shall adopt and comply with the City's Fair Housing Policy Related to Disability, as amended.
- m. Borrowers and property managers (including resident managers and on-site managers) of the Housing Development shall attend the City's Fair Housing for People with Disabilities workshops.
- n. Following reasonable notice to Borrower, Borrower shall allow the City to conduct periodic onsite inspections of the Housing Development in order to verify compliance with the Accessibility Standards and policies.

- o. The Housing Development as a whole and all Housing Units shall meet the requirements of the Accessibility Standards as defined in Section 1, above, and any requirements of the City.
- p. Borrower shall provide a list to the City of all Accessible Housing Units with unit number, bedroom size and type of Accessible Housing Unit ("Housing Unit with Hearing/Vision Features" or "Housing Unit with Mobility Features").

Section 3. Occupancy of Accessible Housing Units. Borrower shall follow the requirements of the City's Fair Housing Policy Related to Disability to assure that information regarding the availability of Accessible Housing Units reaches eligible individuals with disabilities, and will take reasonable, nondiscriminatory steps to maximize the utilization of such units by eligible individuals whose disability requires the accessibility features of the particular unit. To this end, Borrower will take the following steps when an Accessible Housing Unit becomes vacant:

- a. First, Borrower will offer the unit to a current occupant of the Housing Development who has requested and needs the features of an Accessible Housing Unit;
- b. Second, Borrower will offer the unit to a current occupant of a Housing Development under common control who has requested and needs the features of an Accessible Housing Unit;
- c. Third, Borrower will offer the unit to an eligible, qualified applicant on the waiting list for Accessible Housing Units who needs the features of an Accessible Housing Unit;
- d. Fourth, Borrower will make reasonable efforts to advertise the Accessible Housing Unit to qualified individuals who need the accessible features, including listing it as available to individuals who need the accessible features at http://www/Housing.LACity.org, distributing the information about the accessible vacancy in accord with the Borrower's City approved Property Management Plan, distributing it to the most recent list from the City of organizations that serve people with disabilities, and sending an e-blast to parties on the Housing.LACity.org website Outreach List.

In the event that more than one household has requested an Accessible Housing Unit, Borrowers will offer the Unit to households in order on the appropriate waiting list within each category.

If, after using the process identified above, there are no households who need the features of that Accessible Housing Unit, then Borrower may offer the unit to the next household on the conventional waiting list. Should that household choose not to occupy the Accessible Housing Unit, they will remain at the same position on the conventional waiting list. If the household chooses to occupy the Accessible Housing Unit, the tenant must sign a lease addendum in the form approved by the City. The lease addendum requires the household to move to the next available, comparable, conventional unit, when given legal notice by the Housing Development that there is an eligible applicant or existing resident with a disability who requires the accessibility features of that Unit

For individuals who are required to vacate an Accessible Unit because it is needed by an individual with a disability, Borrowers will pay the costs of transferring to a comparable conventional unit, including new utility deposit(s), if required, and reasonable moving expenses.

**Section 4. Rental Policies.** The Borrower shall adopt rental policies that meet the requirements of the ADA, the Fair Housing Act, FEHA, and other federal and state laws and regulations as applicable, and of the Fair Housing Policy Related to Disability of the City, as amended. Borrower shall develop and utilize a PMP approved by the City which describes affirmative marketing, tenanting, and other procedures to ensure that the Housing Development meets all of the fair housing requirements for individuals with disabilities. Within 90 days of bond issuance, the Housing Development must have an approved PMP.

Rental applications will include a section to be filled out by applicants requesting an accommodation or modification. Applicants will not be required to disclose a disability under any circumstances unless requesting an accommodation or modification and that disclosure shall pertain only to the accommodation or modification being requested. Outreach efforts to the disability community shall include, but not be limited to, notices and other communications describing the availability of such units, specific information regarding the features of accessible units, eligibility criteria, and application procedures. These, and additional procedures, are incorporated into the City's Fair Housing Policy Related to Disability, as amended over time.

**Section 5. Residential Rental Property.** The Borrower hereby represents, covenants, warrants and agrees as follows:

- a. All of the housing units in the Housing Development will be similarly constructed units, and each income restricted unit in the project will contain complete separate and distinct facilities for living, sleeping, eating, cooking and sanitation for a single person or a family, including a sleeping area, bathing and sanitation facilities and cooking facilities equipped with a cooking range and oven, a sink and a refrigerator. Each of Accessible Housing Units comply the shall also with requirements. Notwithstanding the foregoing, a unit shall not fail to be treated as a residential unit merely because such unit is a single room occupancy unit within the meaning of Section 42(i)(3)(B)(iv) of the Code even though such housing may provide eating, cooking and sanitation facilities on a shared basis.
- b. All of the housing units (which shall not include any manager units) will be available for rental on a continuous basis to members of the general public, and the Borrower will not give preference to any particular class or group in renting the housing units in the project, except to the extent that for the following: (1) Accessible Housing Units shall be made available on a priority basis to persons who need the accessible features, as described in Section 3 above; (2) any dwelling units are required to be leased or rented to low income tenants or persons [62] years of age and older, (3) the requirements of any regulatory agreement executed between the Borrower and HUD or between the Borrower and a subordinate lender (including the City), (4) the requirements of any Section 8 Housing Assistance Payments Contract with respect to the project, and (5) any preference Borrower gives to a class of persons permitted to be given preference pursuant to the Code, State law and other applicable federal law.

**Section 6. Monitoring Requirements.** The City will monitor the initial production and ongoing occupancy of the Accessible Housing Units and the Accessible Housing Development to ensure full compliance with the Accessibility Standards and the policies in Sections 1 - 4, above. In order to determine compliance with the Accessibility Standards, Borrower shall submit and the City shall review and approve a CASp Inspection Report of the housing development that identifies the necessary and required design elements to make the units and site accessible for individuals with disabilities. The City shall inspect the construction and/or rehabilitation to verify that the correct number of Accessible Housing Units have been produced and that the necessary and required design elements have been constructed to make the units and site accessible for individuals with disabilities, in compliance with Sections 1 - 4 above, and supported by an independent CASp consultant's report.

The City will utilize the Housing Development's City approved PMP and Fair Housing Policy Related to Disability, to monitor ongoing occupancy compliance of the Accessible Housing Units and nondiscrimination in regards to individuals with disabilities. Compliance with the policies shall include, but not be limited to, target marketing, establishing and monitoring a waiting list specific to the Accessibility Housing Units, responding to reasonable accommodation and modification requests, implementation of the service animal policies, implementation of the policies for re-leasing empty Accessible Housing Units, and all elements contained in the Fair Housing Policy Related to Disability, as amended over time.

**Section 7. Maintenance of Records.** The Borrower agrees to keep and maintain books, accounts, reports, files, records, and other documents in accordance with the Fair Housing Policy Related to Disability. Borrower shall maintain all records until August 1, 2026 or for 5 years, whichever is later.

**Section 8. Notices, Demands, Payments and Communication**. Formal notices, demands, payments and communications between the City and the Borrower shall by sufficiently given and shall not be given unless dispatched by registered or certified mail, postage prepaid, return receipt requested, or delivered personally to the principal office of the City as follows:

City of Los Angeles Housing and Community Investment Department P.O. Box #532729

Los Angeles, CA 90053-2729

CITY:

Attention: Portfolio Management and Accessible Housing Program

BORROWER:		
W/copy to:		

Section 9. Term of the Accessibility Covenants. The covenants contained in the Covenants shall become effective upon the issuance of the Bond and shall terminate no earlier than at the end of the CDLAC Conditions set forth in the California Debt Limit Allocation Committee Resolution for the project, it being expressly agreed and understood that the provisions hereof are intended to survive the retirement of the Bond.

Section 10. Covenant to Run with the Land. The Borrower hereby subjects the project to the covenants, reservations, and restrictions set forth in the Covenants. The City and the Borrower hereby declare their express intent that the covenants, reservations and restrictions set forth herein shall be deemed covenants running with the land and shall pass to and be binding upon the Borrower's successors in title to the project. Each and every contract, deed, or other instrument hereafter executed covering or conveying the project or any portion thereof shall conclusively be held to have been executed, delivered and accepted subject to such covenants, reservations, and restrictions, regardless of whether such covenants, reservations and restrictions are set forth in such contract, deed or other instruments.

Section 11. Default; Enforcement. As part of ensuring compliance with the Accessibility Standards, the City or its agent, will conduct periodic on-site visits inspecting the Housing Development, which inspections may include inspecting the Housing Units and common areas, tenant files, logs and other records. Should the Borrower fail to comply with the Accessibility Standards or any of the provisions of the Covenants, the City will first issue an Order to Comply ("Order") stating the element of the Housing Development that is out of compliance, and providing a date by which the Borrower must comply. The Order shall give the Borrower not more than 30 days to correct the violation, or such additional time as the City may grant if the Borrower is taking steps to correct the violation ("Compliance Date"), and diligently pursues such action until the default is corrected, which extension is in the City's sole discretion. The City shall re-inspect the Housing Development within 10 days of the Compliance Date specified in the Order or any extension, however failure to inspect or re-inspect within the time frame does not remove the obligation of the Borrower to comply with the Order.

If the Order is issued and the violation continues to exist after the Compliance Date, then the City shall declare an "Event of Default" and may take any one or more of the following steps:

a. Inspection Fee for Non-Compliance. In the event the Borrower fails to comply with the Order within the Compliance Date, the Borrower shall be liable for subsequent inspection fees in the amount approved by Council until compliance has been achieved. Failure to pay the assessed inspection fee within 30 days of the date of invoice, will result in a late charge equal to or two times the fees and a collection fee equal to 50 percent of the original fee shall be imposed if any fee imposed is not paid within 30 days of service of notice of the imposition of the fee.

The late fee may be imposed without a hearing but may be appealed to the General Manager of HCID. The appeal shall be made in writing, and shall specify the grounds for the appeal. The appeal shall be filed with HCID within ten calendar days of the

issuance of the imposition of the late fees and costs. The General Manager or his designee shall issue a decision within ten calendar days of the filing of the appeal. A copy of the decision shall be served on the person or entity subject to the Order or fee by first class United States mail, postage prepaid, or in person. The City shall have the right to bring legal action in any court to enforce the Order and collect the amount of outstanding fees and penalties. City may waive the penalty imposed pursuant to this section if City determines that good causes exist for the Borrower's failure to pay in a timely manner.

- b. By mandamus or other suit, action or proceeding at law or in equity, including injunctive relief, require the Borrower to perform its obligations and covenants hereunder or enjoin any acts or things which may be unlawful or in violation of the rights of the City hereunder;
- c. Filing of a complaint or referral to HUD or other appropriate agencies for further enforcement actions;
- d. Have access to and inspect, examine and make copies of all or a portion of the books and records of the Borrower pertaining to the project, in order to ensure compliance with all provisions of the Covenants, including records relating to the accessibility of the Accessible Housing Units; and
- e. Take such other action at law or in equity as may appear necessary or desirable to enforce the obligations, covenants and agreements of the Borrower hereunder.

Section 12. Americans with Disabilities Act. The Borrower hereby certifies that it and its property manager and any agent, contractor and subcontractor will comply with the Accessibility Standards, as defined, and the policies described in Sections 2-5. The Borrower and any contractor and subcontractor will provide reasonable accommodations and modifications to allow qualified individuals with disabilities to have access to and to participate in its programs, services, and activities in accordance with the applicable provisions of the ADA, the ADAAG, Section 504, the UFAS, the FHA, California Government Code 11135 *et seq.*, the California Building Code Chapters 11A and 11B, and all subsequent amendments to those laws. The Borrower and any contractor and subcontractor will not discriminate against persons with disabilities or against persons due to their relationship or association with a person with a disability. Any contract and subcontract entered into by the Borrower, relating to this Regulatory Agreement and the Project, to the extent allowed hereunder, shall be subject to the provisions of this paragraph.

- **Section 13. Governing Law**. The Covenants shall be governed by the laws of the State of California.
- **Section 14. Parties Bound**. The provisions of the Covenants shall be binding upon and inure to the benefit of the City and Borrower and their respective successors and assigns.
- **Section 15. Severability.** Every provision of the Covenants is intended to be severable. If any provision of the Covenants shall be held invalid, illegal, or unenforceable by a court of

competent jurisdiction, the validity, legality, and enforceability of the remaining provisions shall not in any way be affected or impaired.

**Section 16. Waiver.** Any waiver by the City of any obligation in the Covenants shall be in writing. No waiver will be implied from any delay or failure by the City to take action on any breach or default of Borrower or to pursue any remedy allowed under the Covenants or applicable law. Any extension of time granted to Borrower to perform any obligation under the Covenants shall not operate as a waiver or release from any of its obligations under the Covenants. Consent by the City to any act or omission by Borrower shall not be construed to be a consent to any other or subsequent act or omission or to waive the requirement for the City's written consent to future waivers.

**Section 17. Modifications**. There shall be no amendment or modification of the Covenants without the prior written approval of the City. Any amendment or modification of the Covenants shall be by a written instrument executed by the City and the parties to the Covenants and the Regulatory Agreement or their successors in title, and duly recorded in the real property records of the County of Los Angeles, California.

**Section 18.** Conflicts. If the provisions of the Covenants are inconsistent with the provisions of the Regulatory Agreement, the provisions of the Covenants shall be controlling.

**Section 19. Third Party Beneficiary**. The City is intended to be and is a third party beneficiary of the Covenants, and the City shall have the right (but not the obligation) to enforce the provisions of the Covenants and to pursue an action for specific performance of such provisions or other available remedy at law or in equity.

### **EXHIBIT J**

# FORM OF CERTIFICATE OF QUALIFIED PROJECT PERIOD

Project Name:	SP7 Apartmen	ts		
(If project name has changed s as well as the new project name	since the award o	allocation please		
as well as the new project name	e ana requesi a ci	unge in the CDLA	ic resolution.)	,
CDLAC Application No.:	18-317			
Name of Note Issuer:	City of Los Ar	geles		
Name of Borrower: (If Borrower has changed sinc Borrower and request a change	e the award plea	se note the origina	al Borrower as	s well as the new
Project meets the general feder Yes No		fied Project Period	1.	
(a) 10% of the dwelling units  Note were first occupied on			the proceeds	of the captioned
(b) 50% of the dwelling units Note were first occupied on		nnced in part from	the proceeds	of the captioned
Project meets the special federa Yes No		fied Project Period	l.	
(Project qualifies if it is an acq available for occupancy within Date.)	•			
(a) Note was issued on (b) Date 12 months after the No	, 20 ote Issuance date		_	
Signature of Officer				
Printed Name of Officer				
Title of Officer				
Phone Number				